Soft and Biological Matter

Giuseppe Spoto Roberto Corradini *Editors*

Detection of Non-Amplified Genomic DNA



Detection of Non-Amplified Genomic DNA

SOFT AND BIOLOGICAL MATTER

"Soft and Biological Matter" is a series of authoritative books covering established and emergent areas in the realm of soft matter science, including biological systems spanning all relevant length scales from the molecular to the mesoscale. It aims to serve a broad interdisciplinary community of physicists, chemists, biophysicists and materials scientists.

Pure research monographs in the series, as well as those of more pedagogical nature, will emphasize topics in fundamental physics, synthesis and design, characterization and new prospective applications of soft and biological matter systems. The series will encompass experimental, theoretical and computational approaches.

Volumes published in this series are disseminated both in print and in electronic formats, the electronic archive being available at SpringerLink.com

Series Editors

Wenbing Hu, Nanjing University, China Roland Netz, Technical University Berlin, Germany Roberto Piazza, Politecnico di Milano, Italy Peter Schall, University of Amsterdam, the Netherlands Gerard C.L. Wong, University of California, Los Angeles, USA

Proposals should be sent to one of the Series Editors, or directly to the managing editor at Springer:

Dr. Maria Bellantone Springer-SBM van Godewwijckstraat 30 3311 GX Dordrecht The Netherlands maria.bellantone@springer.com

For further volumes: http://www.springer.com/series/10783 Giuseppe Spoto • Roberto Corradini Editors

Detection of Non-Amplified Genomic DNA



Editors Prof. Giuseppe Spoto Department of Chemical Science University of Catania Catania Italy

Prof. Roberto Corradini Department of Organic and Industrial Chemistry University of Parma Parma Italy

ISSN 2213-1736 ISSN 2213-1744 (electronic) ISBN 978-94-007-1225-6 ISBN 978-94-007-1226-3 (eBook) DOI 10.1007/978-94-007-1226-3 Springer Dordrecht Heidelberg New York London

Library of Congress Control Number: 2012941159

© Springer Science+Business Media Dordrecht 2012

This work is subject to copyright. All rights are reserved by the Publisher, whether the whole or part of the material is concerned, specifically the rights of translation, reprinting, reuse of illustrations, recitation, broadcasting, reproduction on microfilms or in any other physical way, and transmission or information storage and retrieval, electronic adaptation, computer software, or by similar or dissimilar methodology now known or hereafter developed. Exempted from this legal reservation are brief excerpts in connection with reviews or scholarly analysis or material supplied specifically for the purpose of being entered and executed on a computer system, for exclusive use by the purchaser of the work. Duplication of this publication or parts thereof is permitted only under the provisions of the Copyright Law of the Publisher's location, in its current version, and permission for use must always be obtained from Springer. Permissions for use may be obtained through RightsLink at the Copyright Clearance Center. Violations are liable to prosecution under the respective Copyright Law.

The use of general descriptive names, registered names, trademarks, service marks, etc. in this publication does not imply, even in the absence of a specific statement, that such names are exempt from the relevant protective laws and regulations and therefore free for general use.

While the advice and information in this book are believed to be true and accurate at the date of publication, neither the authors nor the editors nor the publisher can accept any legal responsibility for any errors or omissions that may be made. The publisher makes no warranty, express or implied, with respect to the material contained herein.

Printed on acid-free paper

Springer is part of Springer Science+Business Media (www.springer.com)

Preface

The extraordinary effort in the sequencing of living organism genomes, including the human genome project, is continuously providing new targets for the use of genetic information as specific markers for analysis of samples of biological origin; this process has prompted towards the development of new methodologies for an even more simpler, faster and cheaper detection of nucleic acids to be used for both applicative and research needs in different areas, including clinical diagnosis, environmental monitoring and food control.

Most of the currently available methodologies to detect the DNA base sequences exploit the recognition capabilities of nucleic acids and are based on the hybridization reaction between a probe oligonucleotide carrying a known base sequence and the counterpart present in the unknown DNA target. Most of the above methods require the amplification of the target species as a consequence of the very small amount of DNA material often available and detect the duplex formation by using indicators (labels) and transducers able to generate a detectable signal as a consequence of the specific hybridization event.

The polymerase chain reaction (PCR) is the most widely adopted technique used to amplify a target DNA. It amplifies a single piece of DNA across several orders of magnitude, duplicating millions or more copies of a particular DNA sequence, in order to detect the genetic material more easily. Real-time PCR is presently the golden standard for quantitative DNA analysis.

The PCR amplification of the target sequences is used since it enhances both the specificity and the sensitivity of the assay by increasing the target concentration.

Such enhancement is balanced by the need to use complex reaction mixtures and by the potential contamination of the genetic sample both resulting in ambiguities and errors in the final DNA detection process. Moreover, PCR amplification methods suffer from drawbacks such as the difficulty in amplifying large target DNA molecules (more than about 10000 bases) and the possible amplification of DNA sequences different from the target resulting in non-specific signals.

The concept of amplification has also been extended to other types of schemes, in particular those working under isothermal conditions.

Sequence-specificity is one of the major issues in modern DNA-based diagnostics. Single point mutations are very important markers of alterations linked to genetic diseases and tumour insurgence. Furthermore, the identification of single nucleotide polymorphisms (SNPs), variation of a single nucleotide in the genome sequence, is of great importance in the classification of biological material, allowing to discriminate between individuals, varieties and species. Identification of SNP is one of the major outcome of the modern massive sequencing technologies. Detection of SNP and point mutations is not easily performed by PCR platforms, and specialized techniques are continuously being developed to allow better singlebase level recognition ability.

DNA detection is expected to be significantly improved by using simple and economic detection methods which require minimal DNA modifications and provide enhanced signal amplification. In this perspective, the direct analysis of non-amplified genomic DNA is an excellent cost-effective alternative that can be achieved by using ultrasensitive DNA detection protocols, since extra labour and costs from the amplification procedure are reduced.

Scientific literature published over the last 10 years has shown that non-amplified DNA detection can be achieved by adopting different techniques. Such techniques have allowed the commercialization of innovative platforms for DNA detection that are expected to break into the DNA diagnostic market that is expected to be valued at \$20.1 billion in 2013 with an annual growth rate of 13.7% between 2008 and 2013. In particular, new diagnostic assays for the PCR-free detection of DNA are expected to introduce a new segment that will compete with the \$9.5 billion in 2013 PCR-based diagnostic assay market.

The enhanced sensitivity required when non-amplified genomic DNA is going to be detected has prompted efforts aimed at identify new strategies useful for the ultrasensitive DNA detection. It is clear today that ultrasensitivity can be obtained by combining specific materials with specific detection tools.

Advanced materials play different roles in the ultrasensitive detection. For instance, nanostructures such as metallic nanoparticles have been widely investigated and the specific optical properties of gold nanoparticles have prompted the development of new methods in DNA sensing exploiting the solutions colour change generated by a controlled aggregation of the colloidal nanoparticles or the specific detected optical signal enhancement generated by metallic surface plasmon interferences. Cationic conjugated polymers have been deeply investigated for their ability to produce amplified signal upon interaction with dsDNA.

Probe selection, surface composition and architecture play a fundamental role in the enhanced sensitivity detection of non-amplified DNA. Some of the physical processes which can be used to generate a signal in sensors are strongly dependent on the surface architecture and are greatly improved by the possibility to control distances between the captured DNA analyte and the surface with nanometer precision. This enables to distinguish between short- and long-range effects, which is particularly important in the case of optical detection based on plasmonics. For the generation of strong signals, the advantages offered by specifically designed probes such as peptide nucleic acids (PNA), locked nucleic acids (LNA), or morpholino oligonucleotides (MO) can be exploited. In particular, synthetic probes offer improved selectivity and specificity in targeting complementary nucleic acid sequences that can be tuned by properly designing their structure. Pushing the analysis down to very low detection limits can be achieved only if the probe-DNA interaction is sufficiently strong to prevent dissociation at very low concentrations; thus enhancing the performances of probes is one of the issues to be controlled in order to design new ultrasensitive methods. PNA and LNA are particularly suited for ultrasensitive detection, due to their high DNA affinity. Cationic polythiophene polymers has been used for the ultrasensitive detection of non-amplified DNA since their intrinsic fluorescence quenches as a result of the planar, highly conjugated conformation adopted by the polymers when complexed with a single-strand DNA probe but increases again after hybridization with the perfectly matched complementary strand.

Optical and electrochemical detection tools are among the most widely investigated for the analysis of non-amplified nucleic acids. Both the methods have been applied to the direct detection of genomic DNA and non-amplified RNAs from bacteria. Optical detection of genomic DNA has gained a new impulse in recent years after the advent of surface plasmon resonance (SPR). The SPR sensitivity in the direct detection of unamplified genomic DNA has been evaluated by using genomic DNA extracted from different living organisms. The limitation suffered by SPR in the parallel detection of different probe/target interactions are overcome by SPR imaging while the ultrasensitivity is obtained by adopting specific metallic nanoparticle signal enhancement protocols.

Biosensors based on piezoelectric crystals have been also used to detect unamplified genomic DNA. The immobilization of oligonucleotide probes on the piezoelectric crystal surface allows the combination of the high sensitivity to mass changes of the piezoelectric transducer with the high specificity of the hybridization reaction between the probe and the nucleic acid target and provides a tool for the label free detection of unamplified DNA.

Finally, next generation sequencing technologies have been greatly improved, thanks to the 1000\$ genome challenge, and modern platforms are very close to this goal. Thus redundant information on the genome of specific specimens is also a possibility becoming more and more feasible at the point of care level and certainly the possibility to have the complete screen of genomic information at the nucleotide sequence level is going not only to offer new services, but also to substantially change some of our societies' fundamental issues, such as health and food management.

The brief description of the topic above makes it possible to understand just how heterogeneous the interests and the professional background of the people involved are. In fact, chemists, biologists, biotechnologists and physicists have all devoted their attention to the field of nucleic acid detection using their own cultural background. Moreover, the wide spectrum of information provided has established a multi-disciplinary approach but, at the same time, has expanded the information sources which are to be considered in the field. For all these reasons a book like this, where topics in the field of detection of non-amplified DNA are reviewed, is the result of multi-disciplinary contributors. The across-the-board relevance of the topics discussed in the book together with the even greater scientific and economic importance of new diagnostic tools for genomic analyses are two fundamental reasons to expect a high level of interest in the readers. The interest shown in the subject by academic departments, colleges and institutes involved in the field would appear to show that the book would also have educational potential.

The evolution of the topic discussed is itself a factor to be considered when evaluating the interest it may generate.

The book starts with two chapter dedicated to the most important fields of application of the DNA-detection technologies, that is health and food, which account for the most important economic issues presumably enjoying the major advantages from the availability of new simplified, fast and reliable methods for massive detection of DNA sequences. **Chapter 1** deals with the most important issues which are presently at stake in medical sciences, from newborn safety arising from pre-natal diagnostics to tumour progression evaluation and early diagnosis, and list the expectations in this field which can be foreseen for the next years.

Chapter 2 reports the enormous progress made possible by DNA analysis in the field of food sciences, with application to food safety, traceability and authenticity. This field, being of huge social and economic relevance, offers the possibility to apply new DNA detection schemes to the development of new and unprecedented services, which will be made easier to perform by methods detecting unamplified genomic DNA without the use of complex apparata.

Of course these desiderata could not be satisfied if new generations of materials and detection schemes would not have been developed by chemists and material scientists. Thus, the next section deals with the materials and detection schemes needed for obtaining ultrasensitive diagnostic techniques, which are a pre-requisite for unamplified DNA detection. Chapter 3 deals with the use of nanostructured materials such as nanoparticles, nanowires and nanogaps, whose availability paved the way for unamplified DNA detection. Chapter 4 deals with the molecular probes needed to obtain high fidelity of the DNA capture in sensory systems, and introduces the field of chemically engineered oligonucleotide analogs which allow to reach higher levels of sensitivity and selectivity, such as locked nucleic acids (LNA), morpholino oligonucleotides (MO) and peptide nucleic acids (PNA), together with an explanation of how these molecules can be linked to the sensor and to other reporting groups for ultrasensitive detection. Molecular probes used to directly visualize the genomic DNA by optical methods are also described. Chapter 5 describes cationic conjugated polymers (CCP) which have demonstrated great potential as DNA-sensing materials and how their optical properties can be changed in the hybridization states. The application of these materials to direct genomic DNA detection is also discussed.

The technical advancements made in signal collection are also responsible for the crossing of the threshold needed to directly detect unamplified DNA. Thus in Part III, a series of contribution dealing with the analytical techniques is reported.

In Chap. 6, the advances in DNA detection by optical techniques, including colorimetric, fluorescence, luminescence, surface plasmon resonance (SPR) and Raman scattering assays are reported, with up-to-date description of strategies and technologies for ultrasensitive and unamplified DNA analysis. Electrochemical techniques are another key point in modern DNA genosenors, and in Chap. 7 the most recent advances in this field are discussed, with a description of how new fabrication techniques and nanomaterials have improved the ability to provide rapid, multiplexed electrochemical DNA detection essential for point of care clinical diagnostics. The third large class of mechanical sensors is described in Chap. 8 dealing with piezoelectric sensing. In this chapter, the detection principle and the approaches used in DNA-based sensing for microsatellite and target sequence DNA detection will be presented and discussed, together with a useful discussion on sample treatment and probe chemistry. Surface plasmon resonance and related techniques are among the most useful tools for label-free bioanalysis, and this has allowed to more easily achieve PCR-free detection of DNA. In Chap. 9, the principles of these techniques and their most recent applications to genomic DNA analysis are discussed.

Modern technologies and surface architectures allow to perform DNA detection in a multimodal approach. **Chapter 10** shows an instructive example of how these elements can be combined for the ultrasensitive detection of DNA samples, using electrochemical and surface plasmon enhanced fluorescence.

For most scientists interested in DNA sensing, the most important issues are the incredible advances in sequencing technologies. In this field also, non-amplificative techniques have become more and more competitive. **Chapter 11** gives excellent and updated information about these next-generation sequencing techniques, and guides the reader into this complicated field, which represents some of the most advanced frontiers of human technologies so far reached.

We are really grateful to all the qualified scientists, who are listed in the list of authors, for their precious contribution to this book, bringing their experiences and know-how and sharing these with our readers. We are also indebted to Dr. Maria Bellantone, Senior Publishing Editor, Springer, for her encouragement, support and patience.

Catania Parma Giuseppe Spoto Roberto Corradini

Contents

Part I DNA Analysis: Current Issues

1	Genetic Analyses in Health Laboratories: Current Status and Expectations	3
2	DNA Analyses in Food Safety and Quality: Current Status and Expectations Rosangela Marchelli, Tullia Tedeschi, and Alessandro Tonelli	25
Par	t II Advanced Materials for Ultrasensitive Analysis	
3	Engineered Nanostructures for the Ultrasensitive DNA Detection Dong-Kwon Lim, Amit Kumar, and Jwa-Min Nam	67
4	Advanced Molecular Probes for Sequence-Specific DNA Recognition Alessandro Bertucci, Alex Manicardi, and Roberto Corradini	89
5	Charged Conjugated Polymers Fengting Lv and Shu Wang	125
Par	t III Detection Strategies – Sensors	
6	Optical Detection of Non-amplified Genomic DNA Di Li and Chunhai Fan	153
7	Electrochemical Detection of DNA Using Nanomaterials Based Sensors Sergio Marín and Arben Merkoçi	185
8	Piezoelectric Sensing for Sensitive Detection of DNA Maria Minunni	203

9	Surface Plasmon Resonance-Based Methods Roberta D'Agata and Giuseppe Spoto	235		
10	Parallel Optical and Electrochemical DNA Detection			
Part IV Detection Strategies – Sequencing				
11	Recent Advances in Sequencing Technology John F. Thompson, Fatih Ozsolak, and Patrice M. Milos	281		
Inde	ex	309		

Contributors

Alessandro Bertucci Dipartimento di Chimica Organica e Industriale, Università di Parma, Parma, Italy

Monica Borgatti Laboratory for the Development of Gene and Pharmacogenomic Therapy of Thalassemia, Biotechnology Center, Ferrara University, Ferrara, Italy

Giulia Breveglieri Laboratory for the Development of Gene and Pharmacogenomic Therapy of Thalassemia, Biotechnology Center, Ferrara University, Ferrara, Italy

Roberto Corradini Dipartimento di Chimica Organica e Industriale, Università di Parma, Parma, Italy

Roberta D'Agata Dipartimento di Scienze Chimiche, Università di Catania, Catania, Italy

Chunhai Fan Laboratory of Physcial Biology, Shanghai Institute of Applied Physics, Chinese Academy of Sciences, Shanghai, China

Alessia Finotti Laboratory for the Development of Gene and Pharmacogenomic Therapy of Thalassemia, Biotechnology Center, Ferrara University, Ferrara, Italy

Roberto Gambari Laboratory for the Development of Gene and Pharmacogenomic Therapy of Thalassemia, Biotechnology Center, Ferrara University, Ferrara, Italy

Wolfgang Knoll AIT Austrian Institute of Technology, Vienna, Austria

Amit Kumar Department of Chemistry, Seoul National University, Seoul, South Korea

Di Li Laboratory of Physcial Biology, Shanghai Institute of Applied Physics, Chinese Academy of Sciences, Shanghai, China

Dong-Kwon Lim Department of Chemistry, Seoul National University, Seoul, South Korea

Jianyun Liu Max Planck Institute for Polymer Research, Mainz, Germany

Fengting Lv Beijing National Laboratory for Molecular Science, Key Laboratory of Organic Solids, Institute of Chemistry, Chinese Academy of Sciences, Beijing, China

Alex Manicardi Dipartimento di Chimica Organica e Industriale, Università di Parma, Parma, Italy

Rosangela Marchelli Dipartimento di Chimica Organica ed Industriale, Università di Parma, Parma, Italy

Sergio Marín Nanobioelectronics & Biosensors Group, CIN2 (ICN-CSIC) Institut Català de Nanotecnologia, Barcelona, Spain

Arben Merkoçi Nanobioelectronics & Biosensors Group, CIN2 (ICN-CSIC) Institut Català de Nanotecnologia, Barcelona, Spain

Patrice M. Milos Pfizer Center for Therapeutic Innovation, Boston, MA, USA

Maria Minunni Dipartimento di Chimica "Ugo Schiff", Università di Firenze, Sesto Fiorentino, FI, Italy

Jwa-Min Nam Department of Chemistry, Seoul National University, Seoul, South Korea

Peter Eigil Nielsen Department of Biochemistry B, The Panum Institute, IMBG, Copenhagen, Denmark

Lifang Niu Max Planck Institute for Polymer Research, Mainz, Germany

Fatih Ozsolak Helicos BioSciences Corporation, Cambridge, MA, USA

Giuseppe Spoto Dipartimento di Scienze Chimiche, Università di Catania and Istituto Nazionale di Biostrutture e Biosistemi, Catania, Italy

Tullia Tedeschi Dipartimento di Chimica Organica ed Industriale, Università di Parma, Parma, Italy

John F. Thompson NABsys Inc, Providence, RI, USA

Louis Tiefenauer Department of Chemistry, Life Sciences Department, Paul Scherrer Institute, Villingen, PSI, Switzerland

Alessandro Tonelli Dipartimento di Chimica Organica ed Industriale, Università di Parma, Parma, Italy

Shu Wang Beijing National Laboratory for Molecular Science, Key Laboratory of Organic Solids, Institute of Chemistry, Chinese Academy of Sciences, Beijing, China

Part I DNA Analysis: Current Issues

Chapter 1 Genetic Analyses in Health Laboratories: Current Status and Expectations

Alessia Finotti, Giulia Breveglieri, Monica Borgatti, and Roberto Gambari

Abstract Genetic analyses performed in health laboratories involve adult patients, newborns, embryos/fetuses, pre-implanted pre-embryos, pre-fertilized oocytes and should meet the major medical needs of hospitals and pharmaceutical companies. Recent data support the concept that, in addition to diagnosis and prognosis, genetic analyses might lead to development of personalized therapy. Novel frontiers in genetic testing involve the development of single cell analyses and non-invasive assays, including those able to predict outcome of cancer pathologies by looking at circulating tumor cells, DNA, mRNA and microRNAs. In this respect, PCR-free diagnostics appears to be one of the most interesting and appealing approaches.

Abbreviations

ASO	allele-specific oligonucleotide
EPO	erythropoietin
ErPC	erythroid precursor cells
HbA	adult hemoglobin
HbF	fetal hemoglobin
HPLC	high performance liquid chromatography
IVF	in vitro fertilization
miRNA	microRNA
ODN	oligodeoxyribonucleotide
PCR	polymerase-chain reaction

A. Finotti • G. Breveglieri • M. Borgatti • R. Gambari (🖂)

Laboratory for the Development of Gene and Pharmacogenomic Therapy of Thalassemia, Biotechnology Center, Ferrara University, Via Fossato di Mortara 64/b, 44121 Ferrara, Italy e-mail: gam@unife.it

G. Spoto and R. Corradini (eds.), *Detection of Non-Amplified Genomic DNA*, Soft and Biological Matter, DOI 10.1007/978-94-007-1226-3_1, © Springer Science+Business Media Dordrecht 2012

RT-PCR	reverse transcription PCR
SCA	sickle-cell anemia
TOP	termination of pregnancy

1.1 Introduction

Genetic analyses in health laboratories should be designed and developed to meet the major medical needs of hospitals and pharmaceutical companies. Basically, genetic analyses involve adult patients [1-5], newborns [6-8], embryos/fetuses [9-12], pre-implanted pre-embryos [13-16], pre-fertilized oocytes [17-21].

In the case of genetic analyses involving adult patients the major issue is the molecular characterization of the disease, eventually linking diagnosis to prognosis and diagnosis to personalized therapy [5]. In this case, genetic testing is performed usually after the onset of the pathology and evidences for clinical symptoms. On the contrary, screening newborns for inherited disorders provides the possibility for presymptomatic identification and early intervention to prevent or mitigate morbidity and mortality [6–8]. At present newborn screening can be applied to more than 50 disorders. One example is sickle-cell anemia, an inherited disorder that occurs throughout the world with its highest incidence in areas of Africa where malaria is endemic. There are a number of potentially serious complications associated with the condition, and it is suggested that early treatment (before symptoms develop) can improve both morbidity and mortality. Screening for the condition in the neonatal period would enable early diagnosis and therefore early treatment [7]. Genetic testing on cells from embryos/fetuses allows prenatal diagnosis to determine the genotype of embryos and identify genetic diseases before birth [9-12]. This issue is relevant in several pathologies for which a cure is not currently available. In this case screening tests should be offered to all pregnant women to assess their risk of having a baby with a birth defect or genetic disorder [10]. In this respect, it should be underlined that intrauterine therapy can be offered in some cases in which severe fetal abnormalities are prenatally diagnosed, for which natural history anticipates a fatal outcome or the development of severe disability despite optimal postnatal care [11]. Fetal therapy has therefore developed as an alternative to fatalist expectant prenatal management as well as to termination of pregnancy (TOP). The technological progress in genetic testing at present allows the study of pre-implanted embryos (in this case the diagnosis is expected to help the decision of implanting the embryos) [13–16]. The couples seeking in vitro fertilization, who may screen their embryos for aneuploidy, and the couples at risk for a monogenic disorder but averse to abortion of the affected fetuses after prenatal diagnosis, are likely to be the best candidates to undergo this procedure. Finally, genetic testing can be performed on pre-fertilized oocytes (the objective in this case is to identify the oocyte free of the genetic mutation under analysis) [17-21].



Fig. 1.1 Approaches in diagnosis/prognosis with examples of isolation of biological materials, requirement of PCR steps and biomedical applications. (**a**): genetic testing on adult patients and newborns; (**b**): prenatal diagnosis on embryos/fetuses; (**c**): genetic testing on pre-implanted pre-embryos; (**d**): molecular analyses on pre-fertilized oocytes

Figure 1.1 summarizes genetic testing approaches, pointing out the dependence of these analyses from polymerase-chain reaction (PCR) steps. PCR is required in most of the techniques starting from single cells or low numbers of cells. For instance, PCR and fluorescence in situ hybridization (FISH) are the two common techniques employed on a single or two cells obtained via embryo biopsy during pre-implantation diagnosis [14]. On the contrary, when large numbers of cells are available, several technologies (not requiring a PCR step) can be employed, such as Southern blotting (for genomic analyses) and Northern blotting (for studies on RNA expression) using allele-specific oligonucleotide (ASO) probes. In any case, the major parameters of the molecular techniques supporting these issues (given the fact that the technology should be absolutely reproducible and without the risk of errors) are (a) speed, (b) sensitivity, (c) low-cost, (d) portability. This chapter will give examples of the expectations of the health laboratories in respect to the genetic analyses performed to meet the already mentioned medical needs.

As far as the applications, Fig. 1.1 points out essentially the usefulness of genetic testing for diagnosis and/or prognosis. However, recent studies point out the need of genetic information for the design of personalized therapeutic approaches.

1.2 Genetic Analyses and Personalized Therapies

Personalized therapy is one of the major issues in molecular medicine [22–27]. Examples of personalized medicine involve the choice of the therapeutic drug after considering the genotype (in terms of analysis of the basic genetic mutations as well as of the wide-genome polymorphism scanning) [24] and the phenotype (in terms of transcriptomic and proteomic pattern) [26]. In addition the knowledge of the molecular basis of the disease is expected to help in the development of diseasespecific in vivo models as well as disease-specific therapeutic approaches. This issue is very complex in consideration of the fact that even monogenic diseases can be caused by a large variety of genetic mutations. One example is β -thalassemia, a group of hereditary hematological diseases caused by more than 200 mutations of the human β-globin gene, leading to low or absent production of adult β-globin and excess of α -globin content in erythroid cells, causing ineffective erythropoiesis and low or absent production of adult hemoglobin (HbA) [28-35]. Together with sickle-cell anemia (SCA), thalassemia syndromes are the most important problem in developing countries, in which the lack of genetic counseling and prenatal diagnosis have contributed to the maintenance of the high frequency of these genetic diseases in the population [34]. This contributes significantly to driving changes in the distribution of carriers and affected people, in relation to the migration of populations from endemic areas to countries where their prevalence in indigenous populations had been extremely low (USA, Canada, Australia, South America, the United Kingdom, France, Germany, Belgium, the Netherlands and, more recently, Scandinavia). These changes have challenged health professionals and policymakers in providing equitable access to quality services for the prevention and treatment of Hb disorders in these countries. On the other hand, considering limitations and side effects of the therapeutic approaches and management of the thalassemic patients, novel alternative options for therapy are urgently needed, based on the genetic background of the patient.

1.2.1 Polymorphisms Associated with Fetal Hemoglobin Production

This issue has been the object of several studies in the field of β -thalassemia. For instance, Thein and Menzel [35] reported the progress in the understanding of the persistence of HbF in adults. Three major loci (XmnI-HBG2 single nucleotide polymorphism, HBS1L-MYB intergenic region on chromosome 6q, and BCL11A) contribute to high HbF production. While other numerous loci are expected to be present [36–39], they are supposed to play modest effects. It should be pointed out that the identification of the three major loci has not yet been translated into new therapeutic approaches for HbF reactivation but, in consideration of the fact that they might identify putative binding sites for transcription repressors, an immediate

application is expected to be the prediction of the ability of erythroid cells carrying such genetic variations to produce HbF, which in turn, may improve prediction of disease severity and possible personalized therapeutic strategies [35].

1.2.2 Molecular Approaches for a Personalized Therapy of β-Thalassemia: Mutations Affecting Splicing

A very exciting possibility linking diagnostics to therapeutic choice is the study by Svasti et al., who described a very interesting approach finalized to the repair of β globin pre-mRNA rendered defective by a thalassemia-causing splicing mutation, IVSII-654, in intron 2 of the human β -globin gene [40]. This intervention was performed using a mouse model of IVSII-654 thalassemia, and based on the use of the delivery of a splice-switching oligonucleotide (SSO), a morpholino oligomer conjugated to an arginine-rich peptide. The SSO blocked the aberrant splice site in the targeted pre-mRNA and forced the splicing machinery to reselect existing correct splice sites. Repaired β -globin mRNA restored significant amounts of hemoglobin in the peripheral blood of the IVSII-654 mouse, improving the number and quality of erythroid cells. This approach is expected to be used in all the several splicing defects of β -thalassemia which produce large amount of uncorrectly spliced RNA molecules deeply interfering with RNA trafficking and translation [41–43]. These studies represent new hopes for specific classes of β -thalassemia patients carrying splicing mutations.

1.2.3 Molecular Approaches for a Personalized Therapy of β-Thalassemia: Stop-Codon Mutations

Another example of the need of genomic characterization of β -thalassemia patients is related to the recently proposed read-through approaches for inducing HbA production in erythroid precursors from patients affected by β^0 39-thalassemia, where the CAG (glutamine) codon is mutated to the UAG stop codon, leading to premature translation termination and to mRNA destabilization through the well-described NMD (nonsense-mediated mRNA decay) [44]. Relevant to this issue, Salvatori et al., after FACS (fluorescence-activated cell sorting) and HPLC (high performance liquid chromatography) analyses, demonstrated that erythroid precursor cells from β^0 39-thalassemia patients are able to produce β -globin and adult hemoglobin after treatment with G418 [45]. This study strongly suggests that ribosomal read-through should be considered a strategy for developing experimental strategies for treatment of β^0 -thalassemia caused by stop codon mutations, and might be combined with DNA-based strategies to reactivate HbF [45, 46]. Accordingly, the identification of patients carrying stop-codon mutations might be relevant to design for them a therapeutic intervention based on the use of read-through molecules. This need does apply also for other pathologies caused by stop-codon mutations, including cystic fibrosis [47] and Duchenne muscular dystrophy [48].

1.3 From Tissues to Single Cell Analysis

Standard genetic testing using the commonly available molecular tools and devices is based on the analysis of large numbers of cells; in this case, however, the results obtained should be considered as an average product and hardly reveal the features and biological parameters of single individual cells. On the other hand, in several fields of applied medicine (for instance in molecular oncology) analysis of the genome and the transcriptome of single cells appear to be of great value. In this case single cells can be isolated from a heterogeneous pool, the genome and the RNA can be extracted and analyzed individually; this "single-cell analysis" is becoming one of the milestones for genetic research in several fields of investigation, such as cancer diagnosis [49–52].

In the field of single cell analysis, movement, isolation and characterization of single cells and cell pairs partnering in heterotypic interactions will contribute to functional studies with unprecedented resolution [53, 54], particularly when either partner belongs to a heterogeneous biological population and/or a scarcely represented cell lineage [55]. Unfortunately, conventional approaches for single-cell handling such as microdissection, micromanipulation and flow cytometry suffer from several limitations. For instance, microdissection is analytical, but does not preserve cell viability and function [56, 57]. Micromanipulation of live cells in suspension may be coupled with microscopical inspection and immunophenotypic tagging, but involves the repetitive, slow handling of one object at a time, and exclusively delivers visual readouts. As for flow cytometry, the ability to detect individual events relies on the serial manipulation of large number of cells (and distinct populations thereof). However, a minimum number of cells (few thousands) in suspension are required [58].

1.3.1 Dielectrophoresis (DEP)

DEP is the movement of particles in non-uniform electric fields [59–61]. Charges in the particle itself are not necessary for the effect to occur. This is due to the fact that when an electric field is applied to a system consisting of particles suspended in a liquid, a dipole moment is induced, due to the electrical polarizations at the interface between the particle and the suspending liquid. If the field is non-uniform, the particles experience a translation force (DEP force), of magnitude and polarity depending not only on the electrical properties of the particles and the medium, but also on the magnitude and frequency of the applied electric field. This means that for a given particle type and suspending medium, the particles can experience, at a certain frequency of the electrode applied voltages, a translation force directed towards regions of high electric field strength (this phenomenon is called pDEP). Or, by simply changing the frequency, they may experience a force that will direct them away from high electric field strength regions (this phenomenon is called nDEP).

1.3.2 Lab-on-a-Chip Devices and Cell Isolation

The development of advanced analytical and bioseparation methodologies based on micro/macro arrays and biosensors is one of the strategic objectives of the so-called post-genomics. In respect to isolation of pure cell populations, approaches leading to the development of entire laboratories on a single substrate, the so-called "Labon-a-Chip", are of great relevance [62–65]. Elsewhere published results (reviewed in [62]) indeed firmly demonstrate that human cell populations of different histotype and differentiation stage could be isolated.

1.3.3 DEP-Based Lab-on-a-Chip Devices

We have recently reported a Printed Circuit Board (PCB)-based chip generating DEP-based cylinder-shaped cages that can entrap cells and move them along the device (Fig. 1.2a, b). This does not require a fluidic system and allows levitation and movement of large numbers of eukaryotic cells [65]. Furthermore, we have reported the development of a DEP-array, constituted of 320×320 electrodes and generating over 10,000 spherical-shaped cages able to accommodate single cells [66]. These can be controlled in parallel and two or more cages can be forced to share the same location, making possible in principle to bring in contact, for instance, different cells (Fig. 1.2c-g). These are the most important features of the proposed Lab-on-a-Chip devices: (1) the movement could be programmed at a singlecell level; (2) the chip carries out functions of both actuating and sensing type; (3) single cells can be forced to contact antibody-exposing microspheres. These features allow proposing Lab-on-a-Chip devices for diagnostic applications, cells separation and characterization in the field of non-invasive prenatal diagnosis [29– 31]. The development of a dielectrophoresis (DEP)-based Lab-on-a-Chip approach is important to optimize enrichment, identification and genetic characterization of fetal cells in blood of pregnant women for non-invasive prenatal diagnosis. The optimization of prenatal diagnosis for chromosomal and monogenic disorders will be greatly facilitated if obtaining fetal material avoids the risks associated with amniocentesis, chorionic villus sampling and fetal blood sampling.



Fig. 1.2 Lab-on-a-Chip based manipulation of small number of cells (a, b) or single cells and single microparticles (a, c-g). The DEP-based SMART Slide and the DEP-array have been described in Altomare et al. [65] and Borgatti et al. [66]. The schemes of panel (c) are modified from Gambari et al. [63]. In panel (c) the DEP-based separation of single cells is shown; in panels (d-g) an example is shown demonstrating the possibility to target a single cell (see panel d) with one (e), two (f) or three (g) microspheres (Modified from Borgatti et al. [66]). This example gives clear evidence of the possibility to isolate/manipulate single cells with the use of DEP-based Labon-a-Chip platforms

1.4 Non-invasive Diagnostic Strategies in Prenatal Diagnosis of Genetic Diseases

Current methods for the diagnosis of aneuploidies and monogenic disorders require invasive testing by amniocentesis, chorionic villus biopsy or fetal blood sampling. These diagnostic techniques increase the frequency of fetal loss by about 0.5% [67]. One alternative way for obtaining information on the gestating fetus involves recovery of fetal cells from maternal blood.

In 1969, Walknowska et al. [68] described Y chromatin in blood cells from women carrying male fetuses. Ten years later, recovery of fetal leukocytes from maternal blood by fluorescence-activated cell sorting (FACS) with antibodies against paternally derived HLA (human leukocyte antigen) antigens was reported [69]. Schmorl [70] described deportation of fetal trophoblasts to the lungs of pregnant

women suffering from eclampsia. Trophoblast cells have not found widespread application in diagnostic studies because they are rapidly cleared by the maternal pulmonary circulation and are likely to exhibit confined chromosomal mosaicism [71]. Nucleated red blood cells (NRBC) are the most common cells in fetal blood during early pregnancy. Because they have a relatively short half-life, and because they express hematopoietic plasma membrane antigens, such as the transferrin receptor (CD71), the glycophorin A cell surface molecule and intra-cellular markers (epsilon and gamma globin chains), fetal NRBC have become the targets of choice. In 1990, Bianchi et al. [72] recovered fetal NRBC in maternal blood by using FACS. Fetal origin of the separated cells was confirmed by the presence of Y-chromosome DNA, corresponding to the sex of the fetus. Although the most represented fetal cells, NRBC are rare respect to maternal cells. In maternal blood samples they must be enriched by using antibodies against specific fetal antigens or by physical methods as gradients or electrophoretic mobility-based separations. Antibodies have been used with FACS [73] or magnetic-activated cell sorting (MACS) [74]. These methods exploit antigenic differences between cells. FACS is able to enrich cells with high purity so that slides with sorted cells can be readily scanned manually. It also allows multiparameter sorting and can be adapted for use with intracytoplasmic antigens. MACS, on the other hand, by using magnetic beads coated with specific antibodies, is a faster bench-top technique better suited to process larger cell numbers. Both negative and positive selections can be performed on the same population of cells. Detection levels of fetal cells with these two methods is compounded by small numbers of circulating fetal cells and loss of fetal cells during enrichment procedures. Bischoff et al. [75] reported a simple and rapid density-based progenitor cell enrichment approach. The samples were labeled with a RosetteSepTM progenitor antibody cocktail to remove unwanted maternal white cells (mature T-cells, B-cells, granulocytes, natural killer, neutrophils and myelomonocytic cells).

Cellular fraction collected was analyzed by either fluorescent in situ hybridization (FISH) or real-time PCR for the presence of intact fetal cells and to quantify Y-chromosome-specific DYS1 sequences, respectively. Correct detection rates of the progenitor enrichment approach were 53-89% and fetal sequences were detected in the range from 0.067 to 1.167 genome equivalents per milliliter of blood. This method targets progenitor cells that are not necessarily of the erythroid lineage and may also allow expansion in culture and characterization of the fetal cell types that circulate in maternal blood. Some lymphocytes are long lived and by this approach there is concern that enriched progenitors may be the vestiges of previous pregnancies and do not represent the true fetal genetic status of the current pregnancy [76]. Charge flow separation (CFS) is an antibody independent selection method of fetal cells that relies on the behavior of cells in an electric field and a buffer counterflow gradient [77]. It is an automated rapid method that purifies NRBC from maternal blood and approximately 30% of these cells are fetal. For non-invasive prenatal diagnosis NRBC from fetus must be precisely identified. They could be distinguished from maternal cells by identifying paternal DNA sequences [78]. This method can be extended using micromanipulation of candidate fetal cells and amplification of chromosome-specific short tandem repeats [79]. It is actually under investigation if recovery of fetal cells with CFS is higher than those obtained with other separation methods. The most important molecular techniques that have allowed genetic analysis of enriched fetal cells are PCR and FISH. The ability of PCR to amplify minute quantities of DNA (even single copies) over a billion fold has been exploited for the prenatal diagnosis of monogenic disorders from maternal blood [80–82]. Chromosomal FISH allows the detection of aneuploidies and chromosomal rearrangements in interfase nuclei. It has been used to detect most of the major fetal aneuploidies within fetal cells isolated from maternal blood [73, 83–85]. Fetal traits so far identified among the separated fetal cells include blood group antigen, the common trisomies, triploidy, polymorphic DNA repeats and some single gene disorders [80, 85–88].

All of these methods result in the enrichment of fetal cells among larger populations of maternal cells, but they do not enable recovery of pure populations of fetal cells. Experimental approaches that combine fetal cell identification with molecular genetic diagnosis in an in situ technique circumvent these limitations and are especially suited for automation [89]. Unfortunately the prenatal diagnosis from maternal blood is not practicable for approximately 30% of pregnant women, because the NRBC are not recovered. It is then important to optimize enrichment identification and diagnostic protocols (Fig. 1.3).

1.5 Non-invasive Diagnostic/Prognostic Strategies in Oncology

Diagnosis and prognosis in oncology is of course a primary goal and in this specific field of investigation, non-invasive approaches avoiding biopsies are of great interest. In most tumors, it is known since a long time that tumor cells are present within the circulatory system of cancer patients as "circulating tumor cells" (CTCs) [90-92]. For this reason not only they represent unique diagnostic targets, but they should be considered important for prognosis. In fact, CTCs are released from primary tumors into blood circulation and then they can initiate new cancer growth at novel tissues/organs. Recent advancements in detection and characterization of CTCs can have a significant impact on early cancer detection. For instance, the quantitation of CTCs can be used for prognosis of the outcomes of metastatic diseases [92]. In another example, the identification of cancer stem cells, a subpopulation of CTCs believed to be tumorigenic, can be used for identifying metastasis and relapse [93]. These assays are based on the use of monoclonal antibodies targeting tumor-associated antigens (TAA). Recently, the efficiency and selectivity of this approach was validated using protocols based on the detection of free nucleic acids in the blood circulation of cancer patients.



Fig. 1.3 Example of prenatal diagnosis using maternal peripheral blood, isolation of fetal enriched genomic DNA and DNA sequencing (Breveglieri et al., manuscript in preparation). In this example, the sequencing of the mother DNA is shown in (a) and (b), demonstrating a $\beta^+ 110/\beta^0$ 39 genotype. The sequencing of fetal enriched DNA demonstrates an increase of the $\beta^+ 110$ fraction and a decrease of the β^0 39 fraction (panels c and d), suggesting a $\beta^+ 110/\beta^N$ genotype, that was confirmed after birth, as shown in panels (e) and (f)

1.5.1 Free Tumor DNA as Molecular Target

First of all, it should be underlined that tumor cells from each cancer patient can be isolated and characterized with respect to the genetic changes associated with cancer onset and, more importantly, metastatic progression [94–98]. Regardless of whether these genetic alterations are a direct cause of malignant transformation or whether they are a secondary molecular feature of tumor cells, several reports demonstrated that their analysis can provide useful information for clinical management of the cancer patients. Several molecular genetic alterations including somatic mutations in oncogenes and tumor suppressor genes have been described in melanomas, breast carcinomas, colon carcinomas. Some of these alterations, which are identical to those found in primary tumor DNA, have been observed in extracellular circulating DNA in blood samples of cancer patients. In an attempt to evaluate whether circulating DNA with characteristics of tumor DNA in cancer patients has prognostic value, several studies have reported a direct correlation with poor survival in pancreatic carcinoma, esophageal cancer, non-small cell lung cancer (NSCLC), melanoma, head and neck carcinomas, and B-cell neoplasias. Cancer patients with circulating tumor DNA may have a different outcome than those patients who do not have this molecular change.

1.5.2 Free Tumor RNA as Molecular Target

In addition to circulating DNA, several studies have demonstrated that tumor cellderived RNA is present in the plasma of cancer patients [99–104]. This is quite surprising and deserves further validation, since RNA is far less stable even than shared genomic DNA; however free circulating RNA has been found in several cancer patients, including for instance patients carrying breast cancer. One relevant example is the expression of erbB2 mRNA in the plasma analyzed by nested RT-PCR strategy. The available information suggests that cancer patients with erbB2 mRNA in the plasma may have a high malignancy or an aggressive phenotype, and frequently detecting plasma erbB2 mRNA may provide a novel approach for monitoring breast cancer progression or response to treatment.

1.5.3 MicroRNA as Molecular Target

MicroRNAs (miRNAs, miRs) are a family of small non coding RNAs that regulate gene expression by targeting mRNAs in a sequence specific manner, inducing translational repression or mRNA degradation [105-108]. The genes encoding miRNAs are longer than the processed miRNA molecules: miRNAs are first transcribed as primary transcripts or pri-miRNA and then processed to short, approximately 70-nucleotide stem-loop structures known as pre-miRNA in the nucleus. This process is carried out by a protein complex known as the Microprocessor complex, including the nuclease Drosha and the double-stranded RNA binding protein DGCR8 [105]. These pre-miRNAs are then processed to mature miRNAs in the cytoplasm following interaction with the endonuclease Dicer assisted by TRBP (human immunodeficiency virus (HIV-1) transactivating response (TAR) RNA-binding protein), which also initiates the formation of the RNA-induced silencing complex (RISC) [108]. This complex is responsible for the gene silencing. MicroRNAs have been so far described to be involved in all the differentiation steps leading to production of highly-specialized cells. For instance, miRNAs have been described to be involved in cell cycle, apoptosis, inflammation, terminal differentiation [105–108]. Recently, it has been demonstrated that miRNAs circulate in a highly stable, cell-free form in the blood (i.e., they can be detected in plasma and serum). Tumor cells have been shown to release miRNAs into the circulation and profiles of miRNAs in plasma and serum have been found to be altered in cancer, suggesting broad opportunities for development of circulating miRNAs as bloodbased markers for molecular diagnostics [96, 100, 109–120]. For instance miR-21 [116], let-7a [113], miR-16 [113] levels in plasma might be employed for early detection of cancer as well as sensitivity to anticancer-based chemotherapy.

1.6 Molecular Diagnosis on Human Embryos

Two important parameters deserve great consideration in procedures related to in vitro fertilization and embryo transfer (IVF-ET), in the case molecular diagnosis of the embryo is required: (a) the choice of the best embryo to be transferred and (b) the fast, reliable and non-invasive diagnosis of genetic diseases. As far as point (a), several investigations on human in vitro fertilization procedures have recently associated the production of soluble HLA-G (sHLAG) molecules by 48-72 h early embryos with an increased implantation rate [121-123]. Previous studies demonstrated that HLA-G antigen modulation creates, by direct and indirect mechanisms, a tolerogenic microenvironment at the feto-maternal interface that is essential for a positive pregnancy outcome. In fact, HLA-G molecules, in both membrane bound and soluble isoforms, inhibit the innate Natural Killer response against cytotrophoblast cells that lack HLA class I and II expression [124]. Moreover, HLA-G antigens affect the adaptative cellular response by inducing the apoptosis of cytotoxic CD8+ T lymphocytes [125], impairing CD4+ T cell functions and preventing dendritic cell maturation [126]. Of course, despite the fact that this and similar approaches allow to select the best pre-embryos to be transferred, no information following this approach can be obtained on the genetic pattern of the analyzed embryo, i.e. whether it carries a gene defect at the homozygous state (point (b)). This is very important for embryo selection, in avoiding implantation of pre-embryos carrying life-threating genetic mutations (such as those causing β -thalassemia). In this case the medical need is a protocol allowing genetic testing on a single cell isolated from 8-cell embryos. The cells in this case are totipotent and, therefore, the pre-embryo from which a single cell has been isolated and genetically characterized can originate, after implantation and fetal development, a fully normal individual. This approach is safe for the diagnostic point of view, and carries low risk for the pre-embryos, as reported in several reviews [13–15].

1.7 Molecular Diagnosis on Pre-fertilized Oocytes

Currently, several oocytes are fertilized during IVF-ET procedures and two or more embryos could be transferred to the uterus in order to increase the chance of a positive pregnancy. However, this approach demonstrates a high risk of multiple pregnancies and perinatal mortality and morbidity. The necessity to overcome these problems has stimulated several studies to obtain pregnancy by a single embryo transfer and for ethical reasons to select the best gametes with a higher competency in order to avoid the production of surplus embryos. In this context the identification of validated markers to select oocytes represents a fundamental step in IVF-ET [127]. Nowadays the oocyte selection is mainly performed by intra and extracytoplasmic morphological characteristics, but no definitive association between the morphology and the implantation outcome were obtained [128]. Several studies suggested that the embryo developmental potential and protein production at 4-8 cell stage depend mainly on the maternal mRNA [129]. As the oocyte is the keeper of maternal mRNA, the identification of oocvtes with a higher production of molecules with a pivotal role in implantation could represent a marker of embryo functionality [130]. In order to understand if the oocytes could be implicated in sHLA-G production during the maturation process, Rizzo et al. have analyzed the culture supernatants of "in vitro matured" oocytes for sHLA-G presence. In order to analyze the sHLA-G production these authors have established a detection system that could reveal low levels of sHLA-G. The results obtained evidenced, for the first time, the ability of the oocyte-cumulus complex to produce sHLA-G antigens and proposed these molecules as a marker for oocyte maturation and consequently with a higher probability to develop in a viable embryo [131]. As far as detection of genetic mutations present in the pre-fertilized oocvtes, the molecular analysis concerns the isolation of the first "polar body" (PB), which should be considered to have a complementary genetic pattern in respect to the relative oocyte. Therefore, polar bodies can be used for genetic testing. By cytogenetic analysis of the PB using fluorescent in-situ hybridization (FISH) or chromosome painting, partial or full chromosomal status in the oocyte can be predicted. For example, if the oocyte is derived from a heterozygous β -thalassemia patient and the polar body carries a normal β -globin allele, then the oocyte carries a mutated β -globin allele, with the risk to generate a homozygous β -thalassemia subject if fertilized by a spermatozoon carrying a mutated β -globin allele. Conversely, if the polar body carries a mutated β -globin allele, then the oocyte carries a normal β -globin allele, without any risk to generate a homozygous β-thalassemia subject. This technology has been applied to several genetic diseases and exhibits low risk for oocyte damage and subsequent fertilization rates or embryonic growth to the blastocyst stage [18–21].

1.8 From Complex Protocols and Instrumentation to Low-Tech Technologies

The medical needs in developed countries can be reached employing complex instruments governed by highly-trained researchers. However, even in developing countries, psychosocial and financial factors, inadequate insurance coverage, and the inability to pay for health care services are some of the known barriers to health-care [132]. When prenatal diagnosis is taken as an example, from an economical standpoint it has been demonstrated that these procedures have the potential of saving millions of dollars to healthcare systems. However, several patients do not

have the resources to access prenatal diagnosis in several countries. Developing countries, on the contrary, have completely different needs, most of which concern the identification of the primary mutation of the pathology, the possibility to perform genetic counseling and, possibly, to consider prenatal diagnosis in the case of the life-threating pathologies. However, in developing countries we have to face several problems, which are (a) low-tech facilities; (b) lack of advanced instrumentation in most hospitals; (c) lack of trained technicians; (d) ethical and social issues, making difficult the performance of genetic counseling and prenatal diagnosis. Despite the fact that social issues are difficult to be overcome, the need for advanced diagnostic protocols and tools employing low-tech approaches could be of great interest. Protocols which do not need complex, enzyme-based methodologies are appealing. An example is the deep interest in PCR-free diagnostic approaches [133–139]. Finally, a key point in this context is the portability of the instrument/technology [140] to reach, for example, villages located in geographical positions difficult to be reached, instead of recruiting patients and relatives to hospitals located distally.

Acknowledgments R.G. is granted by Fondazione Cariparo (Cassa di Risparmio di Padova e Rovigo), CIB, by UE ITHANET Project (Infrastructure for the Thalassaemia Research Network), by Telethon (contract GGP07257) and by COFIN-2007. This research is also supported by Associazione Veneta per la Lotta alla Talassemia (AVLT), Rovigo, and by FIRB (The impact of biotechnological innovations on fundamental rights: an interdisciplinary and comparative survey).

References

- Allingham-Hawkins, D.: Successful genetic tests are predicated on clinical utility. Genet. Eng. Biotechnol. News 28(14), 6–9 (2008)
- Hunter, D.J., Khoury, M.J., Drazen, J.M.: Letting the genome out of the bottle. N. Engl. J. Med. 358(2), 105–107 (2008)
- Gollust, S.E., Hull, S.C., Wilfond, B.S.: Limitations of direct-to-consumer advertising for clinical genetic testing. JAMA 288(14), 1762–1767 (2002)
- Marić, P., Ozretić, P., Levanat, S., Oresković, S., Antunac, K., Beketić-Oresković, L.: Tumor markers in breast cancer–evaluation of their clinical usefulness. Coll. Antropol. 35(1), 241–247 (2011)
- Eddy, J.A., Sung, J., Geman, D., Price, N.D.: Relative expression analysis for molecular cancer diagnosis and prognosis. Technol. Cancer Res. Treat. 9(2), 149–159 (2010)
- 6. Sahai, I., Marsden, D.: Newborn screening. Crit. Rev. Clin. Lab. Sci. 46(2), 55-82 (2009)
- Lees, C.M., Davies, S., Dezateux, C.: Neonatal screening for sickle cell disease. Cochrane Database Syst. Rev. (2), CD001913 (2000)
- Zhang, Y.H., McCabe, L.L., Wilborn, M., Therrell Jr., B.L., McCabe, E.R.: Application of molecular genetics in public health: improved follow-up in a neonatal hemoglobinopathy screening program. Biochem. Med. Metab. Biol. 52(1), 27–35 (1994)
- Devaney, S.A., Palomaki, G.E., Scott, J.A., Bianchi, D.W.: Noninvasive fetal sex determination using cell-free fetal DNA: a systematic review and meta-analysis. JAMA 306(6), 627–636 (2011)
- Izquierdo, L.A., Berkshire, S.: Access, quality and costs of prenatal diagnosis. Bol. Asoc. Med. P. R. 102(4), 25–29 (2010)
- 11. Ville, Y.: Fetal therapy: practical ethical considerations. Prenat. Diagn. 31(7), 621–627 (2011)

- Hui, L., Bianchi, D.W.: Prenatal pharmacotherapy for fetal anomalies: a 2011 update. Prenat. Diagn. 31(7), 735–743 (2011)
- Checa, M.A., Alonso-Coello, P., Solà, I., Robles, A., Carreras, R., Balasch, J.: IVF/ICSI with or without preimplantation genetic screening for aneuploidy in couples without genetic disorders: a systematic review and meta-analysis. J. Assist. Reprod. Genet. 26(5), 273–283 (2009)
- Adiga, S.K., Kalthur, G., Kumar, P., Girisha, K.M.: Preimplantation diagnosis of genetic diseases. J. Postgrad. Med. 56(4), 317–320 (2010)
- Geraedts, J.P., De Wert, G.M.: Preimplantation genetic diagnosis. Clin. Genet. 76(4), 315–325 (2009)
- Pauli, S.A., Berga, S.L., Shang, W., Session, D.R.: Current status of the approach to assisted reproduction. Pediatr. Clin. North Am. 56(3), 467–488 (2009)
- 17. Bing, Y., Ouellette, R.J.: Fertilization in vitro. Methods Mol. Biol. 550, 251–266 (2009)
- Gitlin, S.A., Gibbons, W.E., Gosden, R.G.: Oocyte biology and genetics revelations from polar bodies. Reprod. Biomed. Online 6(4), 403–409 (2003)
- 19. Rienzi, L., Vajta, G., Ubaldi, F.: Predictive value of oocyte morphology in human IVF: a systematic review of the literature. Hum. Reprod. Update **17**(1), 34–45 (2011)
- Basille, C., Frydman, R., El Aly, A., Hesters, L., Fanchin, R., Tachdjian, G., Steffann, J., LeLorc'h, M., Achour-Frydman, N.: Preimplantation genetic diagnosis: state of the art. Eur. J. Obstet. Gynecol. Reprod. Biol. 145(1), 9–13 (2009)
- Dawson, A., Griesinger, G., Diedrich, K.: Screening oocytes by polar body biopsy. Reprod. Biomed. Online 13(1), 104–109 (2006)
- Kamat, A.M., Mathew, P.: Bladder cancer: imperatives for personalized medicine. Oncology (Williston Park) 25(10), 951–958, 960 (2011)
- Picker, A., Jackson, D.B.: Genetic determinants of anticancer drug activity: towards a global approach to personalized cancer medicine. Expert Rev. Mol. Diagn. 11(6), 567–577 (2011)
- Mestroni, L., Taylor, M.R.: Pharmacogenomics, personalized medicine, and heart failure. Discov. Med. 11(61), 551–561 (2011)
- Offit, K.: Personalized medicine: new genomics, old lessons. Hum. Genet. 130(1), 3–14 (2011)
- Gonem, S., Desai, D., Siddiqui, S., Brightling, C.C.: Evidence for phenotype-driven treatment in asthmatic patients. Curr. Opin. Allergy Clin. Immunol. 11(4), 381–385 (2011)
- Mehta, R., Jain, R.K., Badve, S.: Personalized medicine: the road ahead. Clin. Breast Cancer 11(1), 20–26 (2011)
- Patrinos, G.P., Kollia, P., Papadakis, M.N.: Molecular diagnosis of inherited disorders: lessons from hemoglobinopathies. Hum. Mutat. 26, 399–412 (2005)
- Old, J.M.: Screening and genetic diagnosis of haemoglobin disorders. Blood Rev. 17(1), 43–53 (2003)
- 30. Galanello, R., Origa, R.: Beta-thalassemia. Orphanet J. Rare Dis. 5, 11 (2010)
- Colah, R., Gorakshakar, A., Nadkarni, A.: Global burden, distribution and prevention of βthalassemias and hemoglobin E disorders. Expert Rev. Hematol. 3(1), 103–117 (2010)
- 32. Quek, L., Thein, S.L.: Molecular therapies in beta-thalassaemia. Br. J. Haematol. 136(3), 353–365 (2007)
- 33. Lederer, C.W., Basak, A.N., Aydinok, Y., et al.: An electronic infrastructure for research and treatment of the thalassemias and other hemoglobinopathies: the Euro-mediterranean ITHANET project. Hemoglobin 33, 163–176 (2009)
- Gambari, R., Fibach, E.: Medicinal chemistry of fetal hemoglobin inducers for treatment of beta-thalassemia. Curr. Med. Chem. 14, 199–212 (2007)
- Thein, S.L., Menzel, S.: Discovering the genetics underlying foetal haemoglobin production in adults. Br. J. Haematol. 145(4), 455–467 (2009)
- 36. Sankaran, V.G., Menne, T.F., Xu, J., Akie, T.E., Lettre, G., Van Handel, B., Mikkola, H.K., Hirschhorn, J.N., Cantor, A.B., Orkin, S.H.: Human fetal hemoglobin expression is regulated by the developmental stage-specific repressor BCL11A. Science **322**(5909), 1839–1842 (2008)

- Sankaran, V.G., Xu, J., Orkin, S.H.: Transcriptional silencing of fetal hemoglobin by BCL11A. Ann. N. Y. Acad. Sci. 1202, 64–68 (2010)
- 38. Nguyen, T.K., Joly, P., Bardel, C., Moulsma, M., Bonello-Palot, N., Francina, A.: The XmnI (G)gamma polymorphism influences hemoglobin F synthesis contrary to BCL11A and HBS1L-MYB SNPs in a cohort of 57 beta-thalassemia intermedia patients. Blood Cells Mol. Dis. 45(2), 124–127 (2010)
- Jiang, J., Best, S., Menzel, S., Silver, N., Lai, M.I., Surdulescu, G.L., Spector, T.D., Thein, S.L.: cMYB is involved in the regulation of fetal hemoglobin production in adults. Blood 108(3), 1077–1083 (2006)
- Svasti, S., Suwanmanee, T., Fucharoen, S., Moulton, H.M., Nelson, M.H., Maeda, N., Smithies, O., Kole, R.: RNA repair restores hemoglobin expression in IVS2-654 thalassemic mice. Proc. Natl. Acad. Sci. USA 106(4), 1205–1210 (2009)
- Suwanmanee, T., Sierakowska, H., Fucharoen, S., Kole, R.: Repair of a splicing defect in erythroid cells from patients with beta-thalassemia/HbE disorder. Mol. Ther. 6(6), 718–726 (2002)
- Zeng, Y., Gu, X., Chen, Y., Gong, L., Ren, Z., Huang, S.: Reversal of aberrant splicing of beta-thalassemia allele by antisense RNA in vitro and in vivo. Chin. Med. J. (Engl). 112(2), 107–111 (1999)
- 43. El-Andaloussi, S., Johansson, H.J., Lundberg, P., Langel, U.: Induction of splice correction by cell-penetrating peptide nucleic acids. J. Gene Med. **8**(10), 1262–1273 (2006)
- 44. Neu-Yilik, G., Amthor, B., Gehring, N.H., Bahri, S., Paidassi, H., Hentze, M.W., Kulozik, A.E.: Mechanism of escape from nonsense-mediated mRNA decay of human beta-globin transcripts with nonsense mutations in the first exon. RNA 17(5), 843–854 (2011)
- Salvatori, F., Breveglieri, G., Zuccato, C., Finotti, A., Bianchi, N., Borgatti, M., Feriotto, G., Destro, F., Canella, A., Brognara, E., Lampronti, I., Breda, L., Rivella, S., Gambari, R.: Production of beta-globin and adult hemoglobin following G418 treatment of erythroid precursor cells from homozygous beta(0)39 thalassemia patients. Am. J. Hematol. 84(11), 720–728 (2009)
- 46. Salvatori, F., Cantale, V., Breveglieri, G., Zuccato, C., Finotti, A., Bianchi, N., Borgatti, M., Feriotto, G., Destro, F., Canella, A., Breda, L., Rivella, S., Gambari, R.: Development of K562 cell clones expressing beta-globin mRNA carrying the beta039 thalassaemia mutation for the screening of correctors of stop-codon mutations. Biotechnol. Appl. Biochem. 54(1), 41–52 (2009)
- Linde, L., Kerem, B.: Nonsense-mediated mRNA decay and cystic fibrosis. Methods Mol. Biol. 741, 137–154 (2011)
- Malik, V., Rodino-Klapac, L.R., Viollet, L., Mendell, J.R.: Aminoglycoside-induced mutation suppression (stop codon readthrough) as a therapeutic strategy for Duchenne muscular dystrophy. Ther. Adv. Neurol. Disord. 3(6), 379–389 (2010)
- 49. Hannemann, J., Meyer-Staeckling, S., Kemming, D., Alpers, I., Joosse, S.A., Pospisil, H., Kurtz, S., Görndt, J., Püschel, K., Riethdorf, S., Pantel, K., Brandt, B.: Quantitative highresolution genomic analysis of single cancer cells. PLoS One 6(11), e26362 (2011)
- Ståhlberg, A., Kubista, M., Aman, P.: Single-cell gene-expression profiling and its potential diagnostic applications. Expert Rev. Mol. Diagn. 11(7), 735–740 (2011)
- 51. Hennig, G., Gehrmann, M., Stropp, U., Brauch, H., Fritz, P., Eichelbaum, M., Schwab, M., Schroth, W.: Automated extraction of DNA and RNA from a single formalin-fixed paraffinembedded tissue section for analysis of both single-nucleotide polymorphisms and mRNA expression. Clin. Chem. 56(12), 1845–1853 (2010)
- Sun, J., Masterman-Smith, M.D., Graham, N.A., Jiao, J., et al.: A microfluidic platform for systems pathology: multiparameter single-cell signaling measurements of clinical brain tumor specimens. Cancer Res. **70**(15), 6128–6138 (2010)
- Kim, J.B., Stein, R., O'Hare, M.J.: Three-dimensional in vitro tissue culture models of breast cancer – a review. Breast Cancer Res. Treat. 85(3), 281–291 (2004)
- Knudsen, K.A., Soler, A.P.: Cadherin-mediated cell-cell interactions. Methods Mol. Biol. 137, 409–440 (2000)

- 55. Bischoff, F.Z., Sinacori, M.K., Dang, D.D., Marquez-Do, D., Horne, C., Lewis, D.E., Simpson, J.L.: Cell-free fetal DNA and intact fetal cells in maternal blood circulation: implications for first and second trimester non-invasive prenatal diagnosis. Hum. Reprod. Update 8(6), 493–500 (2002)
- Kamme, F., Zhu, J., Luo, L., Yu, J., Tran, D.T., Meurers, B., Bittner, A., Westlund, K., Carlton, S., Wan, J.: Single-cell laser-capture microdissection and RNA amplification. Methods Mol. Med. 99, 215–223 (2004)
- Burgemeister, R.: New aspects of laser microdissection in research and routine. J. Histochem. Cytochem. 53(3), 409–412 (2005)
- Kamihira, M., Kumar, A.: Development of separation technique for stem cells. Adv. Biochem. Eng. Biotechnol. 106, 173–193 (2007)
- 59. Pohl, H.A., Crane, J.S.: Dielectrophoretic force. J. Theor. Biol. 37(1), 1–13 (1972)
- Crane, J.S., Pohl, H.A.: Theoretical models of cellular dielectrophoresis. J. Theor. Biol. 37(1), 15–41 (1972)
- Gascoyne, P.R., Vykoukal, J.: Particle separation by dielectrophoresis. Electrophoresis 23(13), 1973–1983 (2002)
- 62. Gambari, R., Borgatti, M., Altomare, L., Manaresi, N., Medoro, G., Romani, A., Tartagni, M., Guerrieri, R.: Applications to cancer research of "lab-on-a-chip" devices based on dielectrophoresis (DEP). Technol. Cancer Res. Treat. 2(1), 31–40 (2003)
- 63. Gambari, R., Borgatti, M., Fabbri, E., Gavioli, R., Fortini, C., Nastruzzi, C., Altomare, L., Abbonnec, M., Manaresi, N., Medoro, G., Romani, A., Tartagni, M., Guerrieri, R.: "Lab-on-achip" devices for cellular arrays based on dielectrophoresis. In: Appasani, K. (ed.) Bioarrays – From Basics to Diagnostics, pp. 231–243. Humana Press Inc, Totowa (2007)
- Medoro, G., Guerrieri, R., Manaresi, N., Nastruzzi, C., Gambari, R.: Lab-on-a-chip for live cell manipulation. Des. Test Comput. IEEE 24(1), 26–36 (2007)
- Altomare, L., Borgatti, M., Medoro, G., Manaresi, N., Tartagni, M., Guerrieri, R., Gambari, R.: Levitation and movement of human tumor cells using a printed circuit board device based on software-controlled dielectrophoresis. Biotechnol. Bioeng. 82, 474–479 (2003)
- 66. Borgatti, M., Altomare, L., Abonnenc, M., Fabbri, E., Manaresi, N., Medoro, G., Romani, A., Tartagni, M., Nastruzzi, C., Di Croce, S., Tosi, A., Mancini, I., Guerrieri, R., Gambari, R.: Dielectrophoresis (DEP) based 'lab-on-a-chip' devices for efficient and programmable binding of microspheres to target cells. Int. J. Oncol. 27, 1559–1566 (2005)
- Jackson, L., Wapner, R.J.: Chorionic villus sampling. In: Simpson, J.L., Elias, S. (eds.) Essentials of Prenatal Diagnosis, pp. 45–61. Churchill Livingstone, New York (1993)
- Walknowska, J., Conte, F.A., Grumbach, M.M.: Practical and theoretical implication of fetal: maternal lymphocyte transfer. Lancet 1, 1119–1122 (1969)
- Herzenberg, L.A., Bianchi, D.W., Schroder, J., Cann, H.M., Iverson, G.M.: Fetal cells in the blood of pregnant women: detection and enrichment by fluorescence-activated cell sorting. Proc. Natl. Acad. Sci. USA 76, 1453–1455 (1979)
- Schmorl, G.: Pathologisch-Anatomische Untersuchungen uber Puerperal Eklampsie. Vogel, Leipzig (1893)
- Hahnemann, J.M., Vejerslev, L.O.: Accuracy of cytogenetic findings on chorionic villus sampling (CVS)-diagnostic consequences of CVS mosaicism and non-mosaic discrepancy in centres contributing to EUCROMIC 1986-1992. Prenat. Diagn. 17, 801–820 (1997)
- Bianchi, D.W., Flint, A.F., Pizzimenti, M.F., Knoll, J.H.M., Latt, S.A.: Isolation of fetal DNA from nucleated erythrocytes in maternal blood. Proc. Natl. Acad. Sci. USA 87, 3279–3283 (1990)
- 73. Price, J.O., Elias, S., Wachtel, S.S., Klinger, K., Dockter, M., Tharapel, A., Shulman, L.P., Phillips, O.P., Meyers, C.M., Shook, D., et al.: Prenatal diagnosis with fetal cells isolated prenatal diagnosis with fetal cells isolated from maternal blood by multiparameter flow cytometry. Am. J. Obstet. Gynecol. 165, 1731–1737 (1991)
- 74. Ganshirt, D., Smeets, F.W., Dohr, A., Walde, C., Steen, I., Lapucci, C., Falcinelli, C., Sant, R., Velasco, M., Garritsen, H.S., Holzgreve, W.: Enrichment of fetal nucleated red blood cells from the maternal circulation for prenatal diagnosis: experiences with triple density gradient and MACS based on more than 600 cases. Fetal Diagn. Ther. **13**, 276–286 (1998)

- Bischoff, F.Z., Marquez-Do, D.A., Martinez, D.I., Dang, D., Horne, C., Lewis, D., Simpson, J.L.: Intact fetal cell isolation from maternal blood: improved isolation using a simple whole blood progenitor cell enrichment approach (RosetteSep). Clin. Genet. 63, 483–489 (2003)
- Ho, S.S., O'Donoghue, K., Choolani, M.: Fetal cells in maternal blood: state of the art for non-invasive prenatal diagnosis. Ann. Acad. Med. Singapore 32, 597–603 (2003)
- Wachtel, S.S., Sammons, D., Manley, M., Wachtel, G., Twitty, G., Utermohlen, J., Phillips, O.P., Shulman, L.P., Taron, D.J., Muller, U.R., Koeppen, P., Ruffalo, T.M., Addis, K., Porreco, R., Murata-Collins, J., Parker, N.B., McGavran, L.: Fetal cells in maternal blood: recovery by charge flow separation. Hum. Genet. 98, 162–166 (1996)
- von Eggeling, F., Michel, S., Gunther, M., Schimmel, B., Claussen, U.: Determination of the origin of single nucleated cells in maternal circulation by means of random PCR and a set of length polymorphisms. Hum. Genet. **99**, 266–270 (1997)
- 79. Samura, O., Sohda, S., Johnson, K.L., Pertl, B., Ralston, S., Delli-Bovi, L.C., Bianchi, D.W.: Diagnosis of trisomy 21 in fetal nucleated erythrocytes from maternal blood by use of short tandem repeat sequences. Clin. Chem. 47(9), 1622–1626 (2001)
- Cheung, M.C., Goldberg, J.D., Kan, Y.W.: Prenatal diagnosis of sickle cell anaemia and thalassaemia by analysis of fetal cells in maternal blood. Nat. Genet. 14(3), 264–268 (1996)
- Suzumori, K., Adachi, R., Okada, S., Narukawa, T., Yagami, Y., Sonta, S.: Fetal cells in the maternal circulation: detection of Y-sequence by gene amplification. Obstet. Gynecol. 80(1), 150–154 (1992)
- Sekizawa, A., Kimura, T., Sasaki, M., Nakamura, S., Kobayashi, R., Sato, T.: Prenatal diagnosis of Duchenne muscular dystrophy using a single fetal nucleated erythrocyte in maternal blood. Neurology 46(5), 1350–1353 (1996)
- Ganshirt-Ahlert, D., Borjesson-Stoll, R., Burschyk, M., Dohr, A., Garritsen, H.S., Helmer, E., Miny, P., Velasco, M., Walde, C., Patterson, D., et al.: Detection of fetal trisomies 21 and 18 from maternal blood using triple gradient and magnetic cell sorting. Am. J. Reprod. Immunol. 30(2–3), 194–201 (1993)
- Cacheux, V., Milesi-Fluet, C., Tachdjian, G., Druart, L., Bruch, J.F., Hsi, B.L., et al.: Detection of 47,XYY trophoblast fetal cells in maternal blood by fluorescence in situ hybridization after using immunomagnetic lymphocyte depletion and flow cytometry sorting. Fetal Diagn. Ther. 7, 190–194 (1992)
- Pezzolo, A., Santi, F., Pistoia, V., De Biasio, P.: Prenatal diagnosis of triploidy using fetal cells in the maternal circulation. Prenat. Diagn. 17(4), 389 (1997)
- Geifman-Holtzman, O., Bernstein, I.M., Berry, S.M., Holtzman, E.J., Vadnais, T.J., DeMaria, M.A., Bianchi, D.W.: Fetal Rh genotyping in fetal cells flow sorted from maternal blood. Am. J. Obstet. Gynecol. **174**(3), 818–822 (1996)
- Oosterwijk, J.C., Mesker, W.E., Ouwerkerk-Van Velzen, M.C., Knepflé, C.F., Wiesmeijer, K.C., Beverstock, G.C., Van Ommen, G.J., Tanke, H.J., Kanhai, H.H.: Prenatal diagnosis of trisomy 13 on fetal cells obtained from maternal blood after minor enrichment. Prenat. Diagn. 18(10), 1082–1085 (1998)
- de Graaf, I.M., van Bezouw, S.M., Jakobs, M.E., Leschot, N.J., Zondervan, H.A., Bilardo, C.M., Hoovers, J.M.: First trimester non-invasive prenatal diagnosis of triploidy. Prenat. Diagn. 19(2), 175–177 (1999)
- Choolani, M., O'Donnell, H., Campagnoli, C., Kumar, S., Roberts, I., Bennett, P.R., Fisk, N.M.: Simultaneous fetal cell identification and diagnosis by epsilon globin chain immunophenotyping and chromosomal fluorescence in situ hybridization. Blood 98(3), 554–557 (2001)
- Yu, J.Q., Cristofanilli, M.: Circulating tumor cells and PET. J. Nucl. Med. 52(10), 1501–1504 (2011)
- Lianidou, E.S., Markou, A.: Circulating tumor cells in breast cancer: detection systems, molecular characterization, and future challenges. Clin. Chem. 57(9), 1242–1255 (2011)
- Paterlini-Bréchot, P.: Organ-specific markers in circulating tumor cell screening: an early indicator of metastasis-capable malignancy. Future Oncol. 7(7), 849–871 (2011)

- Schwarz-Cruz-y-Celis, A., Meléndez-Zajgla, J.: Cancer stem cells. Rev. Invest. Clin. 63(2), 179–186 (2011)
- Schwarzenbach, H., Hoon, D.S., Pantel, K.: Cell-free nucleic acids as biomarkers in cancer patients. Nat. Rev. Cancer 11(6), 426–437 (2011)
- Jung, K., Fleischhacker, M., Rabien, A.: Cell-free DNA in the blood as a solid tumor biomarker – a critical appraisal of the literature. Clin. Chim. Acta 411(21–22), 1611–1624 (2010)
- Vlassov, V.V., Laktionov, P.P., Rykova, E.Y.: Circulating nucleic acids as a potential source for cancer biomarkers. Curr. Mol. Med. 10(2), 142–165 (2010)
- Ellinger, J., Müller, S.C., Stadler, T.C., Jung, A., von Ruecker, A., Bastian, P.J.: The role of cell-free circulating DNA in the diagnosis and prognosis of prostate cancer. Urol. Oncol. 29(2), 124–129 (2011)
- Gahan, P.B.: Circulating nucleic acids in plasma and serum: roles in diagnosis and prognosis in diabetes and cancer. Infect. Disord. Drug Targets 8(2), 100–108 (2008)
- 99. Qiu, M.Z., Li, Z.H., Zhou, Z.W., Li, Y.H., Wang, Z.Q., Wang, F.H., Huang, P., Aziz, F., Wang, D.Y., Xu, R.H.: Detection of carcinoembryonic antigen messenger RNA in blood using quantitative real-time reverse transcriptase-polymerase chain reaction to predict recurrence of gastric adenocarcinoma. J. Transl. Med. 8, 107 (2010)
- 100. Casciano, I., Vinci, A.D., Banelli, B., Brigati, C., Forlani, A., Allemanni, G., Romani, M.: Circulating tumor nucleic acids: perspective in breast cancer. Breast Care (Basel) 5(2), 75–80 (2010)
- 101. Deligezer, U., Yaman, F., Darendeliler, E., Dizdar, Y., Holdenrieder, S., Kovancilar, M., Dalay, N.: Post-treatment circulating plasma BMP6 mRNA and H3K27 methylation levels discriminate metastatic prostate cancer from localized disease. Clin. Chim. Acta **411**(19–20), 1452–1456 (2010)
- 102. Garcia, V., Garcia, J.M., Silva, J., Martin, P., Peña, C., Dominguez, G., Diaz, R., Herrera, M., Maximiano, C., Sabin, P., Rueda, A., Cruz, M.A., Rodriguez, J., Canales, M.A., Bonilla, F., Provencio, M.: Extracellular tumor-related mRNA in plasma of lymphoma patients and survival implications. PLoS One 4(12), e8173 (2009)
- 103. Silva, J., García, V., García, J.M., Peña, C., Domínguez, G., Díaz, R., Lorenzo, Y., Hurtado, A., Sánchez, A., Bonilla, F.: Circulating Bmi-1 mRNA as a possible prognostic factor for advanced breast cancer patients. Breast Cancer Res. 9(4), R55 (2007)
- 104. Apostolaki, S., Perraki, M., Pallis, A., Bozionelou, V., Agelaki, S., Kanellou, P., Kotsakis, A., Politaki, E., Kalbakis, K., Kalykaki, A., Vamvakas, L., Georgoulias, V., Mavroudis, D.: Circulating HER2 mRNA-positive cells in the peripheral blood of patients with stage I and II breast cancer after the administration of adjuvant chemotherapy: evaluation of their clinical relevance. Ann. Oncol. 18(5), 851–858 (2007)
- 105. He, L., Hannon, G.J.: MicroRNAs: small RNAs with a big role in gene regulation. Nat. Rev. Genet. 5, 522–531 (2010)
- 106. Krol, J., Loedige, I., Filipowicz, W.: The widespread regulation of microRNA biogenesis, function and decay. Nat. Rev. Genet. **11**, 597–610 (2010)
- 107. Sontheimer, E.J., Carthew, R.W.: Silence from within: endogenous siRNAs and miRNAs. Cell 122, 9–12 (2005)
- Alvarez-Garcia, I., Miska, E.A.: MicroRNA functions in animal development and human disease. Development 132, 4653–4662 (2005)
- 109. Kroh, E.M., Parkin, R.K., Mitchell, P.S., Tewari, M.: Analysis of circulating microRNA biomarkers in plasma and serum using quantitative reverse transcription-PCR (qRT-PCR). Methods 50(4), 298–301 (2010)
- Heneghan, H.M., Miller, N., Lowery, A.J., Sweeney, K.J., Newell, J., Kerin, M.J.: Circulating microRNAs as novel minimally invasive biomarkers for breast cancer. Ann. Surg. 251(3), 499–505 (2010)
- 111. Brase, J.C., Johannes, M., Schlomm, T., Fälth, M., Haese, A., Steuber, T., Beissbarth, T., Kuner, R., Sültmann, H.: Circulating miRNAs are correlated with tumor progression in prostate cancer. Int. J. Cancer **128**(3), 608–616 (2011)
- 112. Silva, J., García, V., Zaballos, Á., Provencio, M., Lombardía, L., Almonacid, L., García, J.M., Domínguez, G., Peña, C., Diaz, R., Herrera, M., Varela, A., Bonilla, F.: Vesicle-related microRNAs in plasma of nonsmall cell lung cancer patients and correlation with survival. Eur. Respir. J. **37**(3), 617–623 (2011)
- 113. Zuo, Z., Calin, G.A., de Paula, H.M., Medeiros, L.J., Fernandez, M.H., Shimizu, M., Garcia-Manero, G., Bueso-Ramos, C.E.: Circulating microRNAs let-7a and miR-16 predict progression-free survival and overall survival in patients with myelodysplastic syndrome. Blood **118**(2), 13–15 (2011)
- 114. Mostert, B., Sieuwerts, A.M., Martens, J.W., Sleijfer, S.: Diagnostic applications of cell-free and circulating tumor cell-associated miRNAs in cancer patients. Expert Rev. Mol. Diagn. 11(3), 259–275 (2011)
- 115. Duttagupta, R., Jiang, R., Gollub, J., Getts, R.C., Jones, K.W.: Impact of cellular miRNAs on circulating miRNA biomarker signatures. PLoS One 6(6), e20769 (2011)
- 116. Wei, J., Gao, W., Zhu, C.J., Liu, Y.Q., Mei, Z., Cheng, T., Shu, Y.Q.: Identification of plasma microRNA-21 as a biomarker for early detection and chemosensitivity of non-small cell lung cancer. Chin. J. Cancer **30**(6), 407–414 (2011)
- 117. Xie, L., Chen, X., Wang, L., Qian, X., Wang, T., Wei, J., Yu, L., Ding, Y., Zhang, C., Liu, B.: Cell-free miRNAs may indicate diagnosis and docetaxel sensitivity of tumor cells in malignant effusions. BMC Cancer 10, 591 (2010)
- 118. Criscitiello, C., Sotiriou, C., Ignatiadis, M.: Circulating tumor cells and emerging blood biomarkers in breast cancer. Curr. Opin. Oncol. 22(6), 552–558 (2010)
- 119. Weigel, M.T., Dowsett, M.: Current and emerging biomarkers in breast cancer: prognosis and prediction. Endocr. Relat. Cancer **17**(4), R245–R262 (2010)
- 120. Gilad, S., Meiri, E., Yogev, Y., Benjamin, S., Lebanony, D., Yerushalmi, N., Benjamin, H., Kushnir, M., Cholakh, H., Melamed, N., Bentwich, Z., Hod, M., Goren, Y., Chajut, A.: Serum microRNAs are promising novel biomarkers. PLoS One 3(9), e3148 (2008)
- 121. Fuzzi, B., Rizzo, R., Criscuoli, L., Noci, I., Melchiorri, L., Scarselli, B., Bencini, E., Menicucci, A., Baricordi, O.R.: HLA-G expression in early embryos is a fundamental prerequisite for the obtainment of pregnancy. Eur. J. Immunol. 32(2), 311–315 (2002)
- 122. Sher, G., Keskintepe, L., Nouriani, M., Roussev, R., Batzofin, J.: Expression of sHLA-G in supernatants of individually cultured 46-h embryos: a potentially valuable indicator of 'embryo competency' and IVF outcome. Reprod. Biomed. Online 9(1), 74–78 (2004)
- 123. Sher, G., Keskintepe, L., Batzofin, J., Fisch, J., Acacio, B., Ahlering, P., Ginsburg, M.: Influence of early ICSI-derived embryo sHLA-G expression on pregnancy and implantation rates: a prospective study. Hum. Reprod. 20(5), 1359–1363 (2005)
- 124. Pazmany, L., Mandelboim, O., Valés-Gómez, M., Davis, D.M., Reyburn, H.T., Strominger, J.L.: Protection from natural killer cell-mediated lysis by HLA-G expression on target cells. Science 274(5288), 792–795 (1996)
- 125. Fournel, S., Aguerre-Girr, M., Huc, X., Lenfant, F., Alam, A., Toubert, A., Bensussan, A., Le Bouteiller, P.: Cutting edge: soluble HLA-G1 triggers CD95/CD95 ligand-mediated apoptosis in activated CD8+ cells by interacting with CD8. J. Immunol. **164**(12), 6100–6104 (2000)
- 126. Bainbridge, D.R., Ellis, S.A., Sargent, I.L.: HLA-G suppresses proliferation of CD4(+) T-lymphocytes. J. Reprod. Immunol. **48**, 17–26 (2000)
- 127. Gasca, S., Pellestor, F., Assou, S., Loup, V., Anahory, T., Dechaud, H., De Vos, J., Hamamah, S.: Identifying new human oocyte marker genes: a microarray approach. Reprod. Biomed. Online 14(2), 175–183 (2007)
- Balaban, B., Urman, B.: Effect of oocyte morphology on embryo development and implantation. Reprod. Biomed. Online 12(5), 608–615 (2006)
- 129. Tesarík, J., Kopecný, V., Plachot, M., Mandelbaum, J., Da Lage, C., Fléchon, J.E.: Nucleologenesis in the human embryo developing in vitro: ultrastructural and autoradiographic analysis. Dev. Biol. 115(1), 193–203 (1986)
- von Otte, S.: In vitro maturation is an attractive new approach to human assisted reproduction. Expert Rev. Obstet. Gynecol. 2(4), 407–408 (2007)

- 131. Rizzo, R., Dal Canto, M.B., Stignani, M., Fadini, R., Fumagalli, D., Renzini, M.M., Borgatti, M., Gambari, R., Baricordi, O.R.: Production of sHLA-G molecules by in vitro matured cumulus-oocyte complex. Int. J. Mol. Med. 24(4), 523–530 (2009)
- 132. Hatemi, P.K., Dawes, C.T., Frost-Keller, A., Settle, J.E., Verhulst, B.: Integrating social science and genetics: news from the political front. Biodemography Soc. Biol. 57(1), 67–87 (2011)
- 133. D'Agata, R., Breveglieri, G., Zanoli, L.M., Borgatti, M., Spoto, G., Gambari, R.: Direct detection of point mutations in nonamplified human genomic DNA. Anal. Chem. 83(22), 8711–8717 (2011)
- 134. Girigoswami, A., Jung, C., Mun, H.Y., Park, H.G.: PCR-free mutation detection of BRCA1 on a zip-code microarray using ligase chain reaction. J. Biochem. Biophys. Methods **70**(6), 897–902 (2008)
- Dubus, S., Gravel, J.F., Le Drogoff, B., Nobert, P., Veres, T., Boudreau, D.: PCR-free DNA detection using a magnetic bead-supported polymeric transducer and microelectromagnetic traps. Anal. Chem. 78(13), 4457–4464 (2006)
- 136. Ho, H.A., Doré, K., Boissinot, M., Bergeron, M.G., Tanguay, R.M., Boudreau, D., Leclerc, M.: Direct molecular detection of nucleic acids by fluorescence signal amplification. J. Am. Chem. Soc. **127**(36), 12673–12676 (2005); Gao, Z., Tansil, N.C.: An ultrasensitive photoelectrochemical nucleic acid biosensor. Nucleic Acids Res. **33**(13), e123 (2005)
- 137. Minunni, M., Tombelli, S., Fonti, J., Spiriti, M.M., Mascini, M., Bogani, P., Buiatti, M.: Detection of fragmented genomic DNA by PCR-free piezoelectric sensing using a denaturation approach. J. Am. Chem. Soc. **127**(22), 7966–7967 (2005)
- 138. Uslu, F., Ingebrandt, S., Mayer, D., Böcker-Meffert, S., Odenthal, M., Offenhäusser, A.: Labelfree fully electronic nucleic acid detection system based on a field-effect transistor device. Biosens. Bioelectron. 19(12), 1723–1731 (2004)
- Wang, J., Liu, G., Jan, M.R.: Ultrasensitive electrical biosensing of proteins and DNA: carbonnanotube derived amplification of the recognition and transduction events. J. Am. Chem. Soc. 126(10), 3010–3011 (2004)
- 140. Senapati, S., Basuray, S., Slouka, Z., Cheng, L.J., Chang, H.C.: A nanomembrane-based nucleic acid sensing platform for portable diagnostics. Top. Curr. Chem. 304, 153–169 (2011)

Chapter 2 DNA Analyses in Food Safety and Quality: Current Status and Expectations

Rosangela Marchelli, Tullia Tedeschi, and Alessandro Tonelli

Abstract Food safety and quality are very important issues receiving a lot of attention in most countries by producers, consumers and regulatory and control authorities. In particular, DNA analysis in food is becoming popular not only in relation to genetically modified products (GMOs), in which DNA modification is the "clue" of the novelty, but also in other fields like microbiology and pathogen detection, which require long times for the cultivation and specially in cases in which the microorganisms are not cultivable like some viruses, as well as for authenticity and allergen detection. A new topic concerning "nutrigenetics and nutrigenomics" has also been mentioned, very important but still in its infancy, which could lead in the future to a personalized diet. In this chapter we have described the main areas of food research and fields of application where DNA analysis is being performed and the relative methods of detection, which are generally based on PCR. The possibility/opportunity to detect DNA without previous amplification (PCR-free) will be discussed. We have examined the following areas: (1) genetically modified foods (GMOs); (2) food allergens; (3) microbiological contaminations; (4) food authenticity; (5) nutrigenetics/nutrigenomics.

2.1 Introduction

Methods for the evaluation of food safety and quality have been greatly improved in the last years and received a beneficial input, in particular, from the availability of technologies deriving from molecular biology, which are producing new results and open new perspectives in food science, even giving rise to a new global approach called "Foodomics". "Foodomics" has recently been defined as a new discipline

R. Marchelli (🖂) • T. Tedeschi • A. Tonelli

Dipartimento di Chimica Organica ed Industriale, Università di Parma, Parco Area delle Scienze 17/A, 43124 Parma, Italy e-mail: rosangela.marchelli@unipr.it

G. Spoto and R. Corradini (eds.), *Detection of Non-Amplified Genomic DNA*, Soft and Biological Matter, DOI 10.1007/978-94-007-1226-3_2, © Springer Science+Business Media Dordrecht 2012



Fig. 2.1 Foodomics: covered areas, tools, and goals [3]

that studies food and nutrition domains through the application of advanced "omic" technologies including the genomic, transcriptomic, proteomic, and metabolomic studies of foods for profiling, authenticity, biomarker-detection related to food quality and safety, development of new transgenic foods, food contaminants, food effects on human health, etc. [1, 2]. Thus, Foodomics is becoming a global discipline that includes all the emerging working areas in which food, advanced analytical techniques (mainly "omic" tools) and bioinformatics are combined. A representation of the areas covered by Foodomics and its main goals can be seen in Fig. 2.1 [3].

This global interest in food also coincides with a clear shift in medicine and biosciences toward prevention of future diseases through adequate food intakes, and the development of the so-called functional foods.

Foodomics involves the use of multiple tools to deal with the different applications included in this field. In particular, research of new and selective methods for fast, reliable and sensitive detection of specific DNA sequences has received a lot of attention in connection with specific topics, such as GMO detection [4], microbial pathogen determination [5], investigation on the presence of undeclared allergenic ingredients [6], authenticity assessments [7].

This chapter does not intend to provide an exhaustive report of the many works published so far on food analysis concerning DNA detection but to provide an overview of the current status of application, of the methods utilized and of the eventual perspectives.

The detection of DNA in foods always starts with its extraction, which must be carefully optimized, since it is strongly dependent on the nature and complexity of the food matrices and in some cases, such as oils, on the low abundance of DNA [8]. Indeed, extraction and purification procedures of genomic DNA from food strongly depends not only on the matrix, but also on the molecular transformations of food components induced by the different technologies applied and on the eventual additives present in food. High temperatures can give rise to xenobiotics which can interfere with the amplification process, DNA may be degraded or insolubilised [9]. Thus, although several kits are available on the market, no methodology can be considered universal so far [10].

Once the purity of the target DNA extracted has been verified, amplification is usually performed by Polymerase Chain Reaction (PCR) using specific primers in order to amplify selectively the region of interest of the DNA [11]. For routine purposes, the presence of an amplified DNA product of the expected size, normally assessed by gel electrophoresis, is usually considered a sufficient proof of the presence of the sequence of interest. In many cases the identity of the amplified DNA must be confirmed. This is usually performed by "nested PCR", which consists in a further amplification of a shorter sequence within the former strand. Alternatively, recognition of the presence of the target sequence can be performed by hybridisation with specific probes (mostly oligonucleotides) via Watson-Crick base pairing. In place of DNA oligonucleotides, other nucleotide analogues, in particular locked nucleic acids (LNA) [12], morpholino oligonucleotides (MO) [13] and peptide nucleic acids (PNA) [14] and their analogs [15] can be used, since these compounds have shown to improve DNA binding affinity and selectivity. Peptide nucleic acids are particularly suited, since they are robust molecule from the chemical and biological point of view and have shown a high sequence selectivity, thus allowing to detect single nucleotide polymorphisms (SNP) very efficiently [16]. Recognition should be as specific as possible, and must be followed by a change in some measurable properties either in solution or on a sensing surface. This can be achieved by subsequent reactions (mostly enzymatic) or by changes in some physico-chemical properties [17].

End point PCR may be performed at the qualitative level also on devices based on the Lab-on-chip technology, in which sample extraction, amplification and detection occur on a surface functionalised with oligonucleotide probes [18].

Real time PCR is considered the golden method for DNA quantification, it requires more expensive laboratory equipment, but it is extremely accurate and less labour-intensive than other DNA quantification methods. Real-time PCR allows product detection 'in real time'. Amplification occurs in the presence of a targetspecific oligonucleotide probe carrying a reporter dye and a quencher group: due to the proximity of the quencher to the reporter, fluorescence is suppressed. The probe anneals to the single-stranded amplified segment within the region flanked by the two oligonucleotides priming the enzyme reaction and, during amplification, the 5'-exonuclease activity of the polymerase cleaves the probe and separates the fluorophore from the quencher, thus switching on fluorescence of the free reporter producing a measurable signal. The increase in fluorescence is proportional to the amount of the specific PCR product. The possibility to use authentic DNA standards is highly desirable for a proper quantification, though not often available. It is possible to perform multiplex PCR, which requires two or more couples of primers, which must be compatible, i.e. not complementary to each other, not giving rise to duplexes or secondary structures, which may prevent or limit annealing.

Potentials and limitations of molecular diagnostic methods based on PCR in food safety have been reported [19] and will be discussed in connection with the specific topics reported below.

Detection of DNA without previous amplification has been recently achieved by adopting different techniques, described in this book. These techniques require an enhanced sensitivity in order to detect non amplified genomic DNA, which can be obtained with different strategies by combining specific materials with specific detection tools. The possibility/opportunity to detect DNA in food without previous amplification will be discussed in relation to the different subjects reported in this chapter.

In this chapter we'll describe the main areas of food research where DNA analysis is being performed and the relative applications with favorable results or eventual drawbacks which need to be evidenced and overcome. We will examine the following areas: (1) genetically modified foods (GMOs); (2) food allergens; (3) microbiological contaminations; (4) food authenticity. (5) nutrigenetics/nutrigenomics.

2.2 Genetically Modified Organisms (GMOs)

The development and diffusion of genetically modified plants and organisms and their increasing diversification aimed at introducing new characteristics and properties in food and feed creates new challenges for their identification, so that this very complex matter requires not only new methodological and technical approaches but also a particular attention to the legal aspects and general information. According to the latest International Service for the Acquisition of Agribiotech Applications (ISAAA) report [20], 144 GM events representing 24 crops, have received international regulatory approval so far and it is believed that this number is going to increase. The approval and insertion on the market are regulated in different countries by the respective legal frameworks. While in the European Union the legislation is quite severe and requires stringent approval, labelling and traceability of foods and ingredients [21-23], in the USA labelling is only voluntary [24], other countries (Canada, Japan, Australia and New Zealand) have adopted mixed systems of the previous ones [24], others do not have any regulation at all [25]. Thus, a food which is allowed in a country is not necessarily approved in another, causing serious problems for trade.

Real Time PCR is the standard method for the detection of genetically modified organisms (GMOs), since the structure modification of DNA with the insertion of exogenous sequences is the clue to discriminate the GM modified from the wild type genome. Information about the target sequences of GM crops may be obtained through several website: CERA (2011) [26], BATS (2003) [27], GMDD (2011)



Fig. 2.2 Multiplex PCR for GMO. A mix of the DNA extracted from the lectin and zein wild type and five trangenes containing 20% of each transgene was amplified using, from *lines* 1 to 7, one, two, three, four, five, six, and seven primer pairs, following their relative amplicon length (Reproduced from Germini et al. [33] with permission)

[28], GMO-Compass (2009) [29], CRL (2009) [30], and more general database such as NCBI [31]. The most utilised targets are the gene expression cassettes for herbicide and insect resistance. Therefore, methods to detect genetic traits coding for these characters are mostly utilised together with the promoter DNA (such as the 35S cauliflower mosaic virus promoter, CaMV), terminators (such as the inducible nopaline synthetase terminator iNOS), and genetic markers such as antibiotic resistance. The detection of the 35S CaMV promoter is used for the general screening of the presence of a GM material, regardless of the identity of the crop. Indeed, for the quantitative determination of DNA, specific reference materials are required, which are not often available.

In the last years GMOs carrying a combination of genetically modified traits (i.e. herbicide and insect resistance) have become quite popular so that the detection of a single gene expression cassette is not sufficient to characterise the genetic material, but rather a combination of several targets must be addressed. Thus, new methods are required for the multiple detection of DNA. Several targets can be tracked simultaneously by means of multiple PCR [32]. The various PCR products are distinguished one from another on the basis of their differential migration through agarose gels. A typical procedure for multiplex PCR has been outlined by our group [33], who showed the simultaneous amplification of seven sequences of maize (endogenous maize, Bt11, Bt176, Mon810 and GM25) (Fig. 2.2). This method is qualitative and useful for screening purposes, preliminary to qPCR for the certification procedure.

Confirmatory analysis are often required specially for legal purposes. A general method to address this problem is represented by nested PCR, in which one or more internal traits are amplified via a careful choice of the primers[34]. PCR clamping with PNA probes flanking the sequence targeted by primers have been used in order to assess the identity of the PCR products and the semi-quantitative determination of the GMO content without the need of other instrumentation [35].

Another strategy implies hybridisation of the amplified DNA with specific probes in order to switch on a signal through a sensing platform. Several probes have been reported, mostly oligonucleotides complementary to the target sequence [16]. Chromatographic methods have also been reported for the assessment of the PCR product, based on the different migration time of the PCR product and of the probe-PCR amplicon complex [36]. Fluorescent switching probes such as molecular beacons are very useful for improving sensitivity of the method [37].

A real-time PCR-based approach is the GMO screening platform CoSYPS, which is based on the SYBR[®] Green real-time PCR method [38] and combines the detection of the presence of major commodity crops (such as soybean, maize, oilseed rape, rice and cotton) with the detection of common genetic recombinant elements (such as the p35S/tNOS elements) and GM-specific elements (such as herbicide-resistance genes and insect-resistance genes).

Another approach for the detection of multiple GM events in a single experiment is the "realtime PCR based ready-to-use multitarget analytical system" recently developed by Querci et al. [39], which is based on the TaqMan[®] real-time PCR technology and consists of 96-well plates containing lyophilized primers and probes for the individual detection and the simultaneous identification of 39 GM events in seven plant species.

Other technologies have been proposed as a tool for the simultaneous detection of different targets. PNA microarrays [40, 41] have also been shown to be very specific tools, allowing to analyse simultaneously endogenous soy and Roundup Ready soy, and five sequences of maize (endogenous maize, Bt11, Bt176, Mon810 and GM25) with sound results.

Hamels et al. [42] have also developed a multiplex PCR-microarray method for the screening of GMOs.

Currently, only a few publications have reported the use of alternative methods to PCR-based methods for detecting GMOs. One of these is the loop-mediated isothermal amplification (LAMP) method, a technique based on the use of a set of four specially designed primers that recognize a total of six distinct sequences on the sense and antisense strands of the target DNA [43]. LAMP is a highly sensitive isothermal method without need for thermal cycling or any heat denaturation of the template DNA, so that it can be used without a dedicated equipment. Real-time LAMP has been used in the field of GMO detection to target the promoter of the cauliflower mosaic virus 35S, one of the most common genetic elements used in transgenic constructs. The method was tested on a range of different Round-Up Ready soy contents and was shown to be satisfactory for semi-quantitative GMO content evaluation [44]. However, the complex design of primer sets required for each target element may pose very serious problems for the use of LAMP in a multiplex design.

Another technique, which can be used with or without PCR is the Ligase Detection Reaction (LDR), which utilises a thermostable DNA ligase and a pair of probes, which bind to one strand of the DNA target sequence, leaving a nick between them which acts as a template for highly specific ligation by the DNA ligase.

Successive cycles of denaturation-annealing and ligation lead to amplification of the desired fragment [45, 46]. In the PCR/LDR assay, a preliminary PCR step amplifies the target sequence, and subsequently the target signal is detected by LDR. If the ligation primers are labelled with a fluorescent donor and an acceptor group, a ligation event will enable fluorescence resonance energy transfer (FRET) to occur, thus enabling real-time detection. The use of PCR/LDR in combination with universal array capture has been utilized for GMO detection in food, demonstrating high sensitivity and specificity of the method [32, 47, 48]. The simultaneous detection of endogenous genes of soybean and maize (lectin and zein, respectively) and five DNA constructs of genetically modified crops: Roundup Ready soybean and Bt11, Bt176, GA21, and MON810 lines of maize was performed. However, the multiplex PCR/LDR method allowed for qualitative (and semiquantitative) analysis of the GMO content, since the PCR step was unable to amplify the targets proportionally to the amount of the target DNA present.

The Nucleic Acid Sequence-Based Amplification (NASBA) [49] is an isothermal amplification technique that mimics retroviral replication. So it is applied to RNA amplification. In GMO detection, amplification of RNA is very limited due to the low RNA stability in food samples. Thus, NASBA retains a potential for GMO analysis only if adapted to a DNA amplification method. A big advantage in comparison with the PCR technology could be a fast multiplex GMO detection and quantification method if associated with microarray hybridization. This was introduced in GMO screening by Morisset et al. [50, 51], who have developed a novel multiplex quantitative amplification method to be used in combination with a microarray detection named NASBA Implemented Microarray Analysis (NAIMA). In the first step, tailed primers allow for the multiplex synthesis of template DNA. In the second step the transcription-based amplification using universal primers is performed, leading to formation of the complementary RNA product, which is then directly ligated to 3 DNA dendrimers labelled with fluorescent dyes, allowing for signal amplification, and hybridised on an oligonucleotide-based microarray for multiplex detection. Two triplex systems have been applied to test maize samples containing several transgenic lines. The great advantage and further potential of the NAIMA method is its possibility for quantification in a multiplex platform [52].

Several attempts are being studied in order to design platforms for the screening of all authorized and not-authorised GMO in Europe. For this purpose, Waiblinger et al. [53] have established an Excel spread-sheet describing the application of five different screening real-time PCR methods: P35S [54, 55]; T-nos [54]; CTP2-EPSPS [56]; bar [57]; P35S-pat [54] which cover, with the exemption of only three events (maize, LY038, soybean 305423 and cotton 281-24-236 × 3006-210-23), all authorised and non-authorised GMPs in the EU. This is important not only for the GM plants already authorised for placing on the market, but also for upcoming GM crops that might be commercialized in the near future [58]. The screening table will be updated regularly by a network of German enforcement laboratories [55].

The latest progress made in GMO analysis, taking examples from the most recently developed strategies and tools, is reported in a review by Querci et al. [4].

However, in spite of the fact that PCR technology is the method of choice for GMO detection so far, its application in the detection of several targets is limited by the poor multiplexing capabilities of PCR.

Moreover, due to the increasing number of events related to crops and of different crops subjected to genetic transformations, it is expected that in the near future the picture will be even more complex. Traditional approaches, based on the detection of one target at a time, or on a limited multiplexing, no longer will meet the testing requirements. Along with new analytical technologies, new approaches for the detection of GMOs are needed.

The future will be in avoiding PCR and performing the direct detection of GM materials using ultrasensitive PCR-free techniques? Some of these techniques have already been developed, though not yet validated for official purposes. Quartz crystal microbalance (QCM), using oligonucleotide "openers" have been reported in the first "PCR-free" detection scheme [59]. Surface plasmon resonance imaging of DNA targets was also used for ultra-sensitive detection of unamplified DNA according to a sandwich-like test with gold nanoparticles [60]. Using this procedure on genomic DNA extracted from certified standard soy flour, it was possible to detect directly Roundup Ready transgenic soy extracted from certified standard soy flour without any enzymatic amplification [61]. The method allowed to detect 0.1% of transgenic soy, well below the EU legal limit of 0.9%.

Thus, ultra-sensitive methods to detect even few copies of DNA are available, though at the moment quite sophisticated and expensive. Certainly, the advantages to avoid PCR would be tremendous (less steps, less reagents, less possibility of contamination, more accurate quantitative determinations, etc.), but other problems are still to be challenged in order to make these methods suitable for official screening, such as the multiple detection. In any case, the accessibility to correct DNA sequence information, validated methods and reference materials are fundamental prerequisites for setting up an effective strategy.

2.3 Food Allergens

Food allergies might be defined as adverse, immune-mediated (IgE mediated) reactions to foods that can occur in sensitised individuals. The prevalence of IgE-mediated food allergies is estimated to be around 4% of the general population [62] and even higher in infants and children reaching 8% [63]. The severity of the reactions varies from mild symptoms to potentially lethal anaphylactic shocks.

Recently, in order to protect the consumers, the EU introduced several Directives listing the common allergenic ingredients to be declared on the label when present in a given food: Directives 2000/13/CE [64], 2003/89/EC [65], 2006/142/EC [66] and 2007/68/EC [67]. The list, reported in Annex IIIa of the 2003/89/EC [65], and modified by European Directive 2006/142/EC, includes cow's milk, molluscs, crustaceans, hen's egg, fish, tree nuts, peanuts, soy, cereals, celery, mustard, sesame and lupin. Allergies to fruits and vegetables (peach, apple, cherry, kiwi, tomato, spinach, carrot, etc.), less common, are not to be declared so far.

Indeed, of particular concern are the so called "hidden allergens", i.e. allergenic ingredients accidentally present in food, and thus not declared on the label, which may trigger severe allergic reactions if inadvertently consumed by susceptible subjects.

Allergens are proteins, thus the detection of allergenic ingredients in food is usually performed by methods of protein analysis combined with immunochemical protocols such as radio-allergosorbent test (RAST), enzyme allergosorbent test (EAST), rocket immuno-electrophoresis (RIE), immunoblotting and enzyme-linked immunosorbent assay (ELISA). While RIE e immunoblotting provide only qualitative or semiquantitative results, RAST, EAST e ELISA may be utilized as quantitative methods [68, 69].

At the moment ELISA is the technique mostly utilized for the routine analysis of allergens in foods by Control Institutions as well as by Industry. With this method allergens are detected by complexation with a specific antibody and by measuring the concentration on the basis of a color reaction. Two different approaches are available: competitive and sandwich. Several kits are commercially available mainly using the sandwich approach [70]. ELISA methods offer several advantages, being easy to handle, fast and quite sensitive, though suffering from different drawbacks, since they are very depending on the antibody lot and manufacturer, the food matrix effects, the extraction procedures, cross reactivity and the technological treatments of food, which may alter the structure of the native protein, leading to false positive or false negative [71–75]. Moreover, the availability of certified standard materials is fundamental to perform quantitative determinations [76]. Unfortunately, not many certified allergenic proteins are available on the market; some are produced by IRMM 481 Peanut Test Material Kit, NIST RM 8445 for egg, NIST RM 1549 and NIST RM 8435, respectively for screamed milk powder and whole milk, and NIST RM 2387 for peanut [77]. Limits of detection (LOD) and of quantification (LOQ) are generally quite low, though sometime not defined properly.

Mass spectrometry techniques for detecting proteins are becoming popular, although not yet suitable for quantitative determinations and for routine analysis [78]. In most cases identification of the protein is made through enzymatic digestion and identification of peptides by LC/MS/MS [79–82].

Whenever direct methods fail if applied to complex food mixtures or to severely processed foods, in which proteins may be heavily modified and, therefore, not detectable by antibodies directed to recognize their native forms, indirect methods, based on the detection of DNA sequences specific for a given allergenic ingredient may be used. Indeed, DNA detection is more feasible in these cases, DNA being more resistant to drastic thermal treatments, so that, in general, methods for DNA detection are considered more robust.

Amplification of DNA for allergen detection is usually performed in the two canonical ways: End point PCR (or qualitative PCR), which allows to detect the presence of a specific DNA sequence, but is not quantitative, and Real time PCR (quantitative PCR or qPCR), which allows the detection in real time of amplified DNA and also its quantitative determination, by utilising an adequate reference material. Although methods for DNA detection for all foods listed in Annex IIIa

of the 2003/89/EC [65], and modified by the European Directive 2006/142/EC and following exemptions have been reported in the literature [6], the methods of DNA detection most intensively studied in the last years are for wheat, peanut and hazelnut, followed by sesame and soy. Not always the DNA sequences studied are located on the genes expressing the allergen. In these cases the analysis will not establish the presence of the allergenic protein but rather the fact that the food contains (or has been in contact with) the allergenic ingredient (i.e. the hazelnut). This message is already important for the allergic patient.

In some cases a multiplex PCR may be used, for the simultaneous detection of different DNA sequences of the same organism or of different organisms. These systems are very convenient allowing to save time and resources, but they must be carefully designed and validated. Not many are the "multiplex" protocols reported in the literature for the detection of allergens, in particular those based on multiplex Real time PCR [83]. Amplification of different genes codifying for different allergenic proteins of the same organism (i.e., Cor a 1, Cor a 8, Cor a 14 for hazelnut) requires a careful experimental design and a critical selection of the target sequences for obtaining equivalent amplification yields [84]. In general, the amplification of a specific single sequence, both in End point PCR and in Real time PCR, is considered more reliable on account of the higher specificity and sensitivity [23].

In order to favor identification of the DNA of an organism codifying for an allergen, it can be useful to increase the yield by using target sequences which are repeated in the genome [85]. Alternatively, in the case of low amount of DNA present in food (for instance in oils) a very specific confirmatory analysis is usually mandatory. Peptide Nucleic Acids (PNA), thanks to their very specific DNA binding properties, their chemical and enzymatic stability and the possibility to be used in connection with several detection methods, are ideal candidates for the task [16]. The possibility to use PNAs as confirmatory probes post-PCR analysis has been first demonstrated in combination with HPLC/FLD for hazelnut [86]. In this case a PNA probe complementary to a sequence of Cor a 1 was injected together with an amplified fluorescently labelled DNA strand in HPLC: the formation of a PNA-DNA duplex was visualized by a shift of the retention times of the complexed DNA. Successively, the post-PCR confirmation was tackled with a different technique, i.e. by using PNA microarrays, demonstrating that multiple allergen detection is feasible by this platform [87]. In this case PNA array platforms were devised with two different PNA probes complementary to the gene regions coding for Cor a 1 (the major hazelnut allergen) and Ara h 2 (the major peanut allergen). Then, a duplex PCR was developed and the PCR products were deposited on the plate, giving rise to fluorescent signals whenever complementarity was present.

A mixed technique "PCR-ELISA" has been also proposed, by combining the two standard techniques, allowing to obtain semi-quantitative results [88].

Although methods for DNA detection are considered specific and robust and specially useful for the identification of "hidden allergenes" in food, however, in some cases, PCR may be misleading: DNA may be degraded at acidic pH, at high temperatures (i.e. roasting of hazelnuts and peanuts) leading to under-evaluation

of the real allergen content [89]. A comparative analysis of PCR/ELISA methods was performed and the impact of heat treatment on peanut kernels as well as the impact on a peanut-containing food matrix was investigated. It was shown that heat treatments had a detrimental effect on the detection of peanut with either type of method and that both types of methods were affected in a similar manner. Moreover, the DNA methods are not related to the specific gene expressing the specific protein of a food which triggers a particularly severe allergic reaction, but to the general genomic asset of the organism. Commercial kits concerning most foods reported in the European Regulation 142/2006 EC are available on the market (cereals, crustaceans, molluscs, eggs, fish, peanuts, soy, cow's milk, almonds, hazelnuts, walnuts, pistachios, celery, mustard, sesame and lupin). Other protocols for the detection of DNA of allergenic foods, not listed in Annex IIIa of the 2003/89/EC [65], and modified by the European Directive 2006/142/EC, such as peach, tomatoes, carrots, etc, have been reported in the literature. Fast SYBR Green DNA melting curve temperature analyses and duplex PCR assays with internal control have been developed for detection of these allergenic vegetables and have been tested on commercial foods. Spiking food experiments were also performed, assessing that limits of detection (LOD) of 1 mg/kg for carrot and tomato DNA and 10 mg/kg for celery DNA have been reached [90].

The quantitative analysis of DNA by PCR is strongly dependant on the availability of reference materials and on the knowledge of the genomic sequences. Only a few standard materials are commercially available at the moment (NIST, IRMM), although not yet certified. The limits of detection (LOD) and of quantification (LOQ) are often not reported in the publications and on the commercial kits. However, since in most cases the correlation between the amount of DNA and of the allergenic protein present in the food has not been determined, the quantity of DNA cannot be utilised in risk evaluation so far. Thus, the results obtained by PCR and by ELISA are not immediately comparable in most cases.

The European Committee for Standardization (CEN) has recently issued new standards for the detection of food allergens in order to uniform the allergen analysis in foodstuff across Europe: the EN 15634-1: 2009 "Foodstuffs – Detection of food allergens by molecular biological methods" reports the requirements for the specific amplification of target nucleic acid sequences and for the confirmation of the identity.

More recently, the European Standard EN 15842 (2010) has provided general considerations and validation of the methods (immunoassays, nucleic based and chromatographic) to be utilized for the detection of allergens in food.

As far as the applicability of PCR-free methods to the detection on food allergens, the situation is more critical and complex than for GMOs. In fact, allergens are proteins and the preferential methods of detection are primarily related to proteomics, especially Mass Spectrometry [3]. However, DNA evaluation is complementary or an alternative to protein determination whenever the protein is modified, following technological treatments, or difficult to extract or present in a very low amount. The presence of DNA of a particular species/commodity testifies unequivocally that the food is contaminated or has been in contact with or contains

the ingredient which may cause adverse reactions to susceptible individuals. In these cases the availability of ultra-sensitive PCR-free methods may be very useful, being able to detect even small traces of genomic DNA. An example has already been provided in the previous chapter on GMO, where it has been shown that it was possible to detect DNA from a certified sample of flour containing Roundup Ready soybean down to 41 zeptomolar concentrations, corresponding to 11 copies of DNA bases [61]. The finding is relevant in this case, since soy is an allergenic product to be declared on the food label.

Finally, the PCR-free methods available at the moment cannot be used on a routine basis to prevent exposure to potentially harmful ingredients, however they can already be very useful in particular cases when other methods fail or are equivocal, for confirmatory purposes, for instance for legal controversies.

2.4 Microbial Contamination

Foodborne illness represents a major public health issue. The Centers for Disease Control and Prevention (CDC) has recently estimated that the total effect of contaminated food consumption in the United States amounts to 47.8 million episodes of illness, 127,839 hospitalizations, and 3,037 deaths. Among these cases, CDC has calculated that each year 31 major pathogens in the United States (US) caused 9.4 million episodes of food-borne illness, 55,961 hospitalizations and 1,351 deaths. In the US, most (58%) illnesses were caused by Norovirus, followed by non-typhoidal *Salmonella* spp. (11%), *Clostridium perfringens* (10%), and *Campylobacter* spp. (9%), whereas leading causes of hospitalisation were non-typhoidal *Salmonella* spp. (35%), Norovirus (26%), *Campylobacter* spp. (15%), and *Toxoplasma gondii* (8%). Again, the leading cause of death were non-typhoidal *Salmonella* spp. (28%) followed by *T. gondii* (24%), *Listeria monocytogenes* (19%), and Norovirus (11%) [91, 92].

According to the most recent report produced by the European Food Safety Authority (EFSA) and the European Centre for Disease Prevention and Control (ECDC), a total of approximately 339,000 confirmed cases of zoonoses were reported in 2008. Campylobacteriosis remained by far the most frequently reported zoonotic disease in humans (190,566 confirmed cases) followed by salmonellosis (131,468 confirmed cases). Moreover, 5,332 food-borne outbreaks were reported by the 25 reporting Member States during the same year. Overall, 45,622 human cases, 6,230 hospitalisations and 32 deaths (case fatalities) were related to the reported outbreaks.

In the EU *Salmonella* remained the most commonly known causative agent of food-borne outbreaks being responsible for 35.4% of all reported outbreaks. As in 2007, food-borne viruses were the second most common causative agent of food-borne illness, accounting for the 13.1% of all reported. *Campylobacter* caused 9.2% of all reported outbreaks.

Considering that the source of illness often comes from public settings such as restaurants, cafes, schools and kindergarten and taking into account that for several foodstuff the contamination event may occurs at numerous stages along the food chain, it is clear that the availability of diagnostic tools specifically developed for food safety is crucial to control and prevent food-borne illnesses and out-brakes, protecting the consumers and reducing the loss of money related to medical care and elimination of commodities.

Microbiological testing plays a relevant role in food production and processing and represents a growing field at retail and consumer level. The increasing awareness about food safety and the requirements of guidelines and laws at national and international level are strongly boosting the technical innovation in food microbiology. Culture-independent techniques, molecular tools for genomics and proteomics and sensor based approaches may play a pivotal role in diagnostics, community profiling and gene expression studies in food and food-related ecosystems in the next future. Even now, some of them, such as PCR-based techniques, are fundamental tools for food safety investigations.

A wide range of culture-based methods is currently available to detect common food-borne pathogens. These analyses are generally based on several sequential operations: a cultural enrichment (often composed of pre-enrichment and selective enrichment sub-steps), a selective and differential plating, a confirmation phase and a strain identification process. Usually, the whole process can take from 4 days to a couple of weeks. Although culture-base methods are quite reliable, they have often turned out to be time consuming and strongly labour intensive. The increasing need to reduce the time for detection led to the replacement of the selective and differential plating step with molecular-based assays such as PCR, which has currently become a gold standard for the detection of food-borne pathogens.

PCR-based methods potentially represent appropriate tools to study microbial communities in food-associated matrices and processes [93]. However, nowadays, these techniques are mainly employed for the detection, the identification and the quantification of microbial pathogens or for monitoring specific beneficial population like fermenting microorganisms.

Methods focused on nucleic acid amplification like PCR are often coupled with some degree of cultural enrichment as pre-treatment step and they offer some fundamental advantages over culture-based microbial analysis: (i) they are faster and less labour intensive. PCR takes from 5 to 24 h depending on the method used and the reasonable level of automation available reduce the needs for manual operations, thus simplifying experimental protocols; (ii) they are more sensitive and more specific. Being based on selective amplification of small amounts of DNA fragments, PCR is often more reliable and informative than culture-based approaches; (iii) the detection of sub-populations is possible even in the absence of a selective enrichment medium. Targeting specific genome sequences allows to detect only desired microbial strains even in the presence of other dominant populations, which otherwise equally grow in not selective media. Moreover, it is possible to set up quantitative assays. In order to be compliant with food safety requirements, accurate quantification is crucial for food microbiology applications, specially in food pathogens diagnostics. A wide range of real-time quantitative PCR (qPCR) assays are available for this purpose. Compared with end-point PCR, these approaches do not require post-amplification processing and offer template quantification over a wide dynamic range (7–8 log).

It is possible to selectively detect living cells vs dead cells. Since genomic DNA is present in dead cells for long time, conventional PCR does not allow to discriminate between viable and not-viable microorganisms. To overcome this limitation, quantitative reverse transcriptase PCR (RT-qPCR) assays were developed. These kind of assays use RNA as primary target instead of genomic DNA, thus allowing to detect only the actively transcribed genes. With PCR it is possible to detect viable but non-cultivable (VBNC) microorganisms. In certain conditions PCR provides good results even without pre-enrichment steps and in some cases microorganisms, that cannot grow in standard growth media, can be detected and identified via PCR. Considering that numerous microorganisms, including pathogens, are able to enter in VBNC state, these molecular tools are particularly useful to obtain a more accurate analysis of microbial composition or microbial processes, specially in foods.

Several PCR-based assays allow to monitor the amplification of more than one target in the same reaction tube (multiplexing capability). This approach leads to reduced costs and time-to-results and gives the possibility to introduce internal amplification controls (IACs) and normalization strategies during gene expression analyses.

Several commercial kits and detection methods are currently available. The most commonly employed reporter systems in food microbiology are two: one is based on aspecific DNA binding dyes acting as fluorophores, such as SYBR[®] Green, and another focused on the hydrolysis of peculiar probes, like TaqMan[®], due to the 5'-nuclease activity of certain DNA polymerases. The first one is relatively inexpensive, flexible though relying exclusively on primer specificity; moreover, the only specificity evaluation consists of the indirect measurement of post-amplification melting curves. The latter is more expensive but it ensures monitoring of the specific amplicon and allows multiplex analysis.

Since PCR is a mature technology, several commercial kits, well-defined protocols and numerous scientific papers are available for food safety applications (see Table 2.1). Moreover, it must be pointed out that ISO standards have also been defined for detection of food-borne pathogens by means of PCR, providing guidelines and standardized approaches (ISO/TS 20836:2005; ISO 22174:2005; ISO 20837:2006; ISO 20838:2006; ISO 22118:2011; ISO 22119:2011).

It is possible to couple PCR techniques with other technologies and methods such as Surface Plasmon Resonance (SPR), Quartz Crystal Microbalance (QCM), Enzyme-linked ImmunoSorbent Assay (ELISA), Fluorescent In-Situ Hybridization (FISH).

In order to simplify the operative protocols or to increase sensitivity and specificity, several PCR-free amplification strategies for nucleic acids detection have been proposed and developed so far. Although PCR-based methods are currently recognised as gold standard in nucleic acid detection, isothermal amplification

39

technologies such as Nucleic Acid Sequence-Based Amplification (NASBA) [106], Loop-Mediated Isothermal Amplification (LAMP) [107, 108] recombinase polymerase amplification (RPA) [109], Helicase-dependent isothermal DNA amplification (HDA) [110, 111], or Rolling Circle Amplification (RCA) [112] are increasingly gaining popularity as effective alternatives for specific diagnostic purposes. These diagnostic tools are generally based on the employment of enzymes and primers that allow the amplification of DNA or RNA in isothermal conditions, eliminating the needs for complex thermal cycling instrumentation. Some of these methods are extensively investigated in clinical, environmental and food diagnostics. For example, NASBA has been used for food safety purposes to detect Norovirus [113–115], Campylobacter jejuni [116, 117], Lysteria monocitogenes [118]; Hepatitis A [119] and LAMP assays were developed to detect Salmonella spp. [120–126], Listeria monocytogenes [127, 128], Legionella spp. [129], Escherichia coli [130], Clostridium perfringens [131], Vibrio vulnificus [132, 133], human Norovirus [134], Fusarium graminearum [135], and Campylobacter coli [136]. Although quite recent, isothermal amplification methods are rapidly gaining ground in the diagnostics area because they present some advantages over the PCR-based approaches: the majority of the isothermal methods provide shorter time-to-result thanks to massive amplification processes and no needs for complex post-amplification operations. In some cases, such as in LAMP applications, the whole analysis takes around half an hour to be completed [107].

The robust amplification platform, such as in some NASBA applications, brings up to 10¹⁴-fold amplification in less than 90 min (a couple of order of magnitude more than PCR), thus helping to increase the sensitivity of the whole detection method [137]. Some applications of these methods are usually scarcely labour intensive and do not require highly trained operators to perform the analysis. Another extremely important aspect consists on the ability of some isothermal systems to directly target RNA molecules instead of double stranded DNA, without any additional step in the procedure such as in NASBA. This helps to identify microorganisms that have an active transcription machinery (often corresponding to live vs death condition) even in a highly concentrated DNA background. Moreover, targeting RNA can help to increase sensitivity when the targeted gene is intensively expressed and enable the easy detection of viruses with a RNA-based genome, like Norovirus. Compared to PCR-based strategies, also the need for dedicated instrumentation is less demanding and in some cases completely absent.

All these reasons make the isothermal technologies more suitable for integration with miniaturized and integrated systems with "lab-on-a-chip" formats or other microfluidic devices [138–146] and with "rapid diagnostic methods" like lateral flow devices [147–155]. These emerging diagnostic approaches are at their initial development stage but they are undergoing to a rapid evolution in the field and some of these technologies should provide the next generation of rapid, low cost and easy-to-use diagnostic kits for laboratory, *in field* or *point-of-care* applications.

However, despite the advancements in techniques based on nucleic acid amplification, there are still some strong limitations specially in the field of food safety applications. Since food industry includes a wide spectra of production and

larget	Technology	Matrices	Time/enrichment	Limit of detection
Salmonella [94]	Cultural enrichment method vs real time PCR	lce cream	PCR ca. 3 h cultural enrichment method 5 to 7 days	10 ³ CFU/ml
Salmonella [95]	Multiplex and quantitative real-time PCR	Cheddar cheese, raw and turkey	48 h of selective enrichment + analysis	$6.1 \times 10^1 \text{ CFU/ml}$
Salmonella [96]	TaqMan quantitative real-time PCR	Raw milk, chicken meat, salmon	16 h enrichment + analysis	5 CFU/25 ml (raw milk), 2.5-5.0 CFU/25 g (chicken and salmon)
Salmonella [97]	TaqMan quantitative real-time PCR	Chili powder and shrimp	48 h (24 for pre-enrichment and 24 for selective enrichment)	0.04 CFU/g
Salmonella [98]	TaqMan quantitative real-time RT-PCR (qRT-PCR)	Spinach, tomatoes, jalapeno peppers and serrano peppers	24 h pre-enrichment	40 copies of invA mRNA per reaction
Campylobacter jejuni and Campylobacter coli [99]	TaqMan multiplex real-time quantitative PCR	Chicken	3 h	1.18–5.30 log CFU/ml
Campylobacter, Salmonella spp. [100]	Multiplex real-time quantitative PCR	Chicken skin rinses	1	$3.0 \times 10^3 \text{ CFU/ml}$
Escherichia coli 0157:H7 [101]	RT-qPCR (Scorpion probe)	Artificially contaminated milk natural samples of raw milk, pasteurized milk, ice cream, kulfi (frozen dessert), paneer (soft cheese), infant foods	~	10 ³ CFU/ml

onditic
al c
experiment
and
matrices
food
n different
pathogens i
Detection of 1
e 2.1

Escherichia coli 0157:H7	Real-time quantitative	Milk and milk products	With and without enrichment	1 and 3 log CFU (with
and Listeria	PCR – Molecular		(e h)	and without
monocytogenes [102]	Beacons)			enrichment)
Listeria monocytogenes	Real-time quantitative	Collard green, cabbage,	48 h total time of analysis	1–5 CFU/50 g
[103]	PCR – (SYBR Green)	lettuce, mixed parsley		
		and spring onion		
		bunches, Chinese		
		cabbage, arugula,		
		chicory, wild chicory,		
		spinach and watercress		
Listeria monocytogenes	TaqMan quantitative	Artificially contaminated	24 h	7.5 CFU/25 ml milk,
[104]	real-time PCR	raw milk, salmon, pate,		9 CFU/15 g
		green-veined cheese		salmon,
		naturally contaminated		1 CFU/15 g pate
		fish, meat, meat		and cheese
		products, and dairy		
		products		
Hepatitis A virus and	Multiplex real-time PCR	Different food surfaces	Concentration using	10 ¹ viral particles for
norovirus GI and GII		(lettuce, strawberry,	electrostatic binding	HAV and norovirus
[105]		raspberries and green		GII, and 10^2 viral
		onions)		particles for
				norovirus GI

processing methods and food contaminations can often occur at every stage before food consumption, the needs for suitable diagnostic tests all along the food chain are far to be fulfilled. In several processes, food might be tested for contaminations at the level of raw materials, during processing or stocking, after packaging or at retail and sometimes consumer level. Obviously, all these potential levels of control require different diagnostic approaches: in some an accurate quantitative test represents the best choice, such as in the case of foods that have to pass mandatory requirements. However, these kind of tests are often expensive, take long time and need dedicated laboratories to be accomplished. Sometimes short time-to-results is the key factor to be preferred in order to quickly move big commodity stocks or introduce an *in line* check point along a food process. On the other hand, at retail and consumer level, the demand for easy to perform, rapid and cheap diagnostic tests is rapidly increasing. The nucleic acid amplification methods developed in the last decades are focused on quite complex laboratory based procedures and appear to be not flexible enough to fit the vast majority of the above mentioned settings and conditions.

Although less cumbersome and time-consuming compared to culture-based methods, PCR methods still require well trained operators, are relatively expensive and rely on complex equipments for thermal cycling. Moreover, they need a dedicated area in the laboratory in order to limit accidental contaminations. Some of these drawbacks might be minimised using the proper "new generation" isothermal technology. For example LAMP allows to greatly reduce the instrumentation requirements since the amplification occurs without thermal cycling steps, at constant temperature (65°C) and a wide range of solutions for the detection step have already been developed, moving from fluorescent instrument-based detection to low cost approaches based on naked eye visualization or real time monitoring using an inexpensive turbidimeter. LAMP, however, is based on the interaction of six specially designed primers and sometimes the difficulties in primers definition may discourage from using this technique [108].

As previously mentioned, these methods may open a new era in nucleic acid diagnostics however there is currently no single method that meets all the requirements of microbial diagnostics devoted to food safety. For a comprehensive description and comparison of the most employed isothermal methods see Gill and Ghaemi [156].

Other factors that limit the employment of nucleic acid-based amplification methods common to PCR, as well as the isothermal approaches, are intrinsic to the employment of enzymes during the amplification steps. The activity of these enzymes is dramatically affected by the presence of inhibitory compounds commonly present in foods that might be co-extracted and concentrated from the usual procedures of extraction and concentration of nucleic acids from complex food matrices.

Beside the development of affordable nucleic acid detection methods, it must be pointed out that, concerning food diagnostics, one of the most critical phase involves all the operations performed upstream the analysis, during sample preparation. Thus separation and concentration procedures need to be optimised more carefully to obtain rapid, direct, and quantitative methods [157].

When real food samples have to be analysed, all the methods targeting nucleic acids depend on some degree of cultural enrichment in order to be reasonably affordable. Food samples often consist of heterogeneous and complex matrices, containing numerous components that are incompatible with the majority of the analytical methods, such as fats and complex carbohydrates. A second relevant issue comes from the volume usually obtained from food manipulations that is often much bigger than the volume that can be employed in a nucleic acid-based detection method. In this conditions, the targeted nucleic acid is often under the detection or the quantification limit of the chosen method.

In order to circumvent these critical issues, numerous approaches have been developed with the aims to purify and concentrate the microbial targets combined with the removal of matrix-associated components that act as inhibitors for down-stream processes. These operations have been collectively defined pre-analytical processing, or "upstream" sample processing [5].

Beside the culture-based and the nucleic acids-based amplification methods, numerous biosensors have been developed for microbial and food safety applications. Numerous techniques have been exploited such as SPR, QCM, Surface Enhanced Raman Scattering (SERS), Evanescent Wave and a wide spectra of other optical and electrochemical applications [158, 159], but only a few of them have been used to target nucleic acids. However, tremendous limitations to the use of these technologies arise from the lack of effective pre-analytical processing methods designed to be integrated *ad hoc* in sensors for food safety [160].

Thus, according to Dwivedi and Jaykus, the pre-analytical sample processing still represent the "crucial unresolved step for successful application of biosensor-based technology in food microbiology" [5]. The aspects described above fit in the same way once transposed to the development of amplification-free methods for nucleic acids detection, that currently have no examples in food-borne pathogen diagnostics.

Despite the wide range of diagnostic methods, further efforts must be made in order to fullfil all the requirements of food analysis, specially for microbial pathogen diagnostics. Among the most innovative strategies, the direct amplification-free detection of nucleic acids appears to be one of the most promising approach. However, there are some critical unresolved issues that need to be faced independently from the detection method.

2.5 Authenticity

"Food traceability", under the EU legislation, is the possibility to track any food, feed, food-producing animal or substance that will be used for consumption, through all stages of production, processing and distribution [161].

Requirements for traceability have been introduced in national regulations as an effective mean to contribute to product safety and consumer confidence. In recent years, food traceability has become a matter of great importance in relation not only to safety and quality issues, but even more significant for fresh and typical

food products. In the latter case, in particular, the problems of authenticity and of adulteration recognition have become of paramount importance so that it is requested to verify the identity of products, under compositional and geographic aspects, in order to protect both producers and consumers from frauds.

The possibility to confirm the authenticity of traded foods is vital to guarantee the integrity of the food supply chain. Using a variety of advanced chemical, biochemical and molecular biological techniques, many laboratories are now able to assess the origin of food and feed [7].

Although proteins and other chemicals have been used to identify food products, DNA markers are the best-suited for identification of biological samples. This is because of the remarkable durability of DNA, even in hostile environments which are encountered during many processing steps. The use of DNA markers as diagnostic tools for food authenticity, origin, and traceability of variety/type composition of complex food matrices has been investigated in an increasing number of projects worldwide. However, some processed food contains highly degraded DNA and/or polymerase chain reaction (PCR) inhibitors, both of which may affect the subsequent PCRs used for the amplification of diagnostic DNA sequences. These effects may be overcome by modifying the DNA extraction process and the PCR design and conditions. It is sometimes possible to overcome the inhibitory effects by extensive dilution of the DNA extract; however, this may not be an option when the amount of DNA in the sample is limited. In these cases, a sensitive method for the detection of small amounts of highly degraded DNA is necessary. Therefore, as first step, it is important to be able to optimize the DNA extraction method, which should be adapted to the nature of the food matrices.

Here we report some examples concerning the most frequently used foods, which need traceability inspections.

2.5.1 Meat

Rapid progress in farm animal breeding has been made in the last few decades. Advanced technologies for genomic analysis in molecular genetics have led to the identification of genes or markers, in particular genome sequencing and SNPs maps, which can be useful to trace farm animals back to their source breed, to detect eventual frauds and to validate typical productions.

In the studies reported in the literature, DNA tracts from meat belong to mitochondrial DNA extracted from the muscle tissue. Mitochondrial DNA (mtDNA) is practically the only reliable genomic target for PCR in processed meat products, and its haploid nature and strict maternal inheritance greatly facilitate genetic analysis. The use of mtDNA as a target for PCR enabled to obtain results in highly processed products from which nuclear DNA is often unavailable for reliable and reproducible analysis [162].

The PCR reaction can be associated to a multiplex single base extension (SBE) assay, which makes use of diagnostic single nucleotide polymorphisms (SNPs)

45

for the identification of a particular type of meat. For example, this methodology was successfully applied to the discrimination between Asian and European *Sus scrofa* lineages [163]. The test was robust, sensitive and accurate in a wide range of meat foodstuffs (fresh or frozen or smoked, dry-cured and complex mixtures such as patè and sausages) allowing accurate detection of pig genetic material and identification of maternal ancestry. As it comes out from these studies, the success of the amplification methodologies with cooked-meat is strictly dependent on the design of short PCR amplicons.

Recently, gold nanoparticles have been used to detect swine-specific sequences and nucleotide mismatches in PCR-amplified and non-amplified mitochondrial DNA mixtures in order to authenticate the species [164]. The entire assay (hybridization plus visual detection) was performed in less than 10 min. The LOD (for genomic DNA) of the assay was $6 \ \mu g \ ml^{-1}$ of swine DNA in mixed meat samples. This is a really very promising study, which could be applied in the future for species assignment in food analysis, mismatch detection in genetic screening and homology studies between closely related species. It is also a rare example of first attempts to detect non-amplified DNA for food authenticity.

2.5.2 Fish

Also in the case of fish which has undergone different treatments such as canned fish, it is not possible to amplify PCR products of a large size, because the thermal treatment generates DNA fragmentation. However, it is important to have at one's disposal a method that can be used routinely, allowing the easy amplification of DNA and providing a reliable species identification. The best situation is the amplification of a small fragment which contains enough information to enable the differentiation of all the species studied.

The presence of additives used in seafood industry as spices or sauces may attenuate or inhibit DNA amplification. Moreover, the different kinds of sauces added produce differences in the quantity and quality of extracted DNA as this molecule is very sensitive to acid and alkaline agents. To this purpose, it is worth highlighting the pickled products, in which the low pH produces high DNA degradation.

In a recent study on the amplification of a fragment of mitochondrial DNA by means of PCR, combined to the Forensically Informative Nucleotide Sequencing (FINS), a method which allowed the genetic identification of the most important commercialized species of horse mackerel and related species [165] was developed. The main advantage of this method is that, besides including a high number of species (20 species), it is based on the use of a DNA fragment with a size below 250 bp and can be applied to all kinds of processed products, including those that have undergone intensive transformation processes. The possible applications of this method are the following: normative control of raw and processed products, particularly aimed at verifying the authenticity of imported species; traceability of

different fish batches along the commercial chain; correct labelling; protection of the consumer's rights; fair competence among fish operators; and control of fisheries.

Moreover, PCR-RFLP (Restriction Fragment Length Polymorphism) amplification methodology has been successfully applied for the characterization of fish species in processed food products labelled "anchovy" [166]. This methodology combines PCR amplification to the digestion of DNA fragments which show variations in homologous sequences.

The same amplification method combined to the use of lab-on-a-chip capillary electrophoresis for endpoint analysis enabled accurate sizing of DNA fragments and identification of white fish species at a level of 5% (w/w) in a fish mixture [167]. The method was applied to a range of products and subjected to an inter-laboratory study carried out by five U.K. food control laboratories. One hundred percent correct identification of the single species samples and ca. 70% identification of nine mixture samples was achieved by all laboratories. The results indicated that fish species identification could be carried out using a database of PCR-RFLPprofiles without the need for reference materials.

The technique is also proposed as a qualitative test for screening raw, frozen, and lightly processed samples for the presence or absence of fish species. As such, this method is extremely useful as it has the potential to detect a wide range of fish species following the development of suitable PCR-RFLP profiles or can be applied as a rapid mean of comparing samples.

2.5.3 Milk and Dairy Products

The dairy sector has taken "the viewpoint that traceability/product tracing as a tool is not only relevant in the context of food inspection and certification systems, but may also be applied for industry-driven business reasons and is mainly concerned with food safety aspects" (Bulletin of the International Dairy Federation 412/2007; IDF Guiding Principles for Traceability/Product Tracing).

In particular, molecular markers are becoming useful tools to trace the whole chain: they provide a diagnostic approach for food authenticity that can benefit the farmers, the dairy industries and the traders. Species-specific PCR has shown to be a suitable method to control food authenticity because a specific target sequence can be detected even in matrices containing a pool of heterogeneous genomic DNA, such as milk or other dairy products.

Molecular markers can be found in DNA of different grass species and forages commonly fed to animals, or DNA coming from the microbial environment.

In a recent study [168], using a PCR approach, the rbcL marker, a chloroplastbased gene, was selected to amplify plant DNA fragments in raw cow milk samples, collected from stock farms or bought on the Italian market. rbcL-specific DNA fragments could be found in total milk, as well as in the skimmed and the cream fractions. Upon sequencing of the PCR amplified fragments, a nucleotide composition of the chromatogram was obtained which reflected the multiple contents of the polyphytic diet. The unambiguous characterisation and correlation of the amplified DNA present in milk with that present in the commercial fodder or in a specific pasture area can allow traceability and/or authenticity assessment specific for the food chain of dairy products. These new tools open the possibility that, in the future, many type of cheeses requiring the Protected Designation of Origin (PDO) certification could be validated by searching the plastidial DNA fragments present in the milk used for their production.

By a different approach, analytical support for providing authentication of the geographic origin of alpine milk productions [169] was obtained by studying milk bacterial composition, by using intergenic transcribed spacer–PCR fingerprinting. This work showed for the first time that the composition of the bacterial communities is related to the altitude of pastures strongly enough to allow the distinction of milk origin in terms of altitude, by means of the adopted ITS-PCR (Internal Transcribed Spacer) fingerprinting protocol, followed by multivariate statistical analysis of the resulting band patterns. Although further research is required to better elucidate the relationship between bacteria and environment (type of flora, climate, altitude), this new analytical approach and its findings can be of significant interest for the manufacturing, protection, traceability and promotion of the PDO Fontina cheese and other alpine dairy productions.

The same methodology has been also applied to distinguish italian Protected Designation of Origin (PDO) water buffalo Mozzarella cheese from different producers on a molecular basis in relation to the place of manufacturing within the production district [170]. Microbial DNA was isolated from the governing liquid of the cheese, the whole microflora ribosomal 16S–23S internal transcribed spacers ITS-PCR fingerprinting was amplified by means of an original primer pair. Although further analyses are required to better investigate the capabilities and limitations of the method, and in particular, the possibility to univocally identify samples and their places of production, different PDO water buffalo Mozzarella cheeses were identified by using the described protocol. Moreover, the bacterial populations present in this cheese was shown to be substantially stable over time, as the electrophoretic profiles obtained did not show major changes over 1 year.

Recently it was demonstrated for the first time that cow's milk contains large amounts of microRNAs (miRNAs) and that the unique expression profile of milkspecific miRNAs can serve as a novel indicator and possible new standard for the quality control of raw milk and milk-related commercial products, such as fluid milk and powdered formula milk [171]. A total of 245 miRNAs were identified in raw milk. By using TaqMan probe-based miRNA quantitative RT-PCR, seven miRNAs that have a relatively consistent expression throughout the lactation process were identified. The expression profile of these milk-specific miRNAs can serve as an ideal biomarker for discriminating poor-quality or "manipulated" milk from pure raw milk, as well as for the quality control of commercial milk products, such as fluid milk and powdered formula milk. Very interestingly, these findings provide a basis also for understanding the physiological role of milk miRNAs.

2.5.4 Coffee

Coffee is one of the most important world food commodities, the commercial trade consisting almost entirely of Arabica and Robusta varieties. The former is considered to be of superior quality and thus attracts a premium price. Methods to discriminate between the two coffee cultivars could be useful for unveiling either deliberate or accidental adulteration. Spaniolas et al. have described a molecular genetic approach to distinguish Arabica from Robusta coffee beans [172]. A PCR-RFLP (Restriction Fragment Length Polymorphism) was used to monitor a single nucleotide polymorphism within the chloroplastic genome. Samples were analyzed with a Lab-on-a-Chip capillary electrophoresis system. Coffee powder mixtures were analysed with this technique, showing a 5% limit of detection. The plastid copy number was found to be relatively constant across a wide range of bean samples, suggesting that this methodology can also be employed for the quantification of any adulteration of Arabica with Robusta beans.

This methodology, analogously to the previous studies on fish, is very easy to use, and the analysis performed by a Lab-on-a-Chip technique, is straightforward and relatively fast, approximately 60 min per 12 samples after the restriction digest step. This would mean that this methodology may be suitable for the routine analysis of coffee samples. A similar approach could be employed for the analysis of roasted beans and even ground coffee, providing that DNA suitable for PCR-RFLP could be extracted from these commodities. There is a strong possibility that during the process, either through roasting or other treatments such as decaffeination, degradation of DNA can occur. However, a previous study [173] has shown that DNA suitable for PCR analysis can be extracted from beans subjected to various levels of roasting and from commercial coffee powder, although with low yields and reproducibility.

2.5.5 Olive Oil

Olive (*Olea europaea* L.) is one of the oldest agricultural tree crops spread worldwide and is an important source of oil with beneficial properties for human health, due to its content in polyphenols. In spite of its economic and nutritional importance, few data are available in the literature related to the genetics of olive in comparison with other fruit crops. Available molecular data are especially related to the application of molecular markers to the analysis of genetic variability in *Olea europaea* with the aim of developing efficient molecular tools for defining the olive oil quality and traceability [174], as required for PDO extra-virgin olive oil. At the moment, DNA analysis is more than a promising approach to distinguish the different cultivars from which the oil is produced, since it is less influenced by environmental and processing conditions in respect to other methods (i.e., metabolites). However, although methods of extraction of DNA from olive oil have been developed by several authors, all studies confirmed that DNA extraction from extra-virgin olive oil is a critical point, since DNA is present in very low amount.

DNA extracted by olive oil was studied by means of different amplification and molecular markers techniques. Pafundo et al. [175] traced the cultivar composition of monocultivar olive oils by Amplified Fragment Length Polymorphism (AFLP) markers, suggesting that DNA extraction is the most critical step affecting the procedure. The same group [176], starting from AFLP amplified in olive oil, developed some Sequenced Characterised Amplified Region Markers (SCAR) that amplified successfully on DNA extracted from olive oil.

By using Simple Sequence Repeats (SSR, also referred to as microsatellites) analysis, it was also demonstrated that microsatellites are useful for checking the presence of a specific cultivar in a PDO oil, thus verifying the identity of the product [177]. However, the marker profile of the main cultivar in the oil was only obtained: no signals were detected for the minor cultivars eventually present.

Consolandi et al. [178] reported the development of a semi-automated SNP genotyping assay to verify the authenticity of extra-virgin olive oils. The authors developed a Ligation Detection Reaction (LDR)/Universal Array (UA) platform by using several olive SNPs. They found that 13 accurately chosen SNPs were sufficient to unequivocally discriminate a panel of 49 different cultivars.

Finally, in a recent study [179], the effect of storage on the degradation of DNA purified from oil was investigated. Interestingly, a negative correlation between storage time and quality–quantity of recovered DNA was observed. In fact, 1 month after the production of the oil the degradation increased, making traceability efforts rather hard.

Considering all these investigations, it is possible to note that DNA-based olive oil traceability is a topic of great interest, but a cautionary note on the use of DNA markers for origin testing is mandatory, as reported also by Doveri et al. [180]. Their observations were based on a non-concordance between the genetic profiles of olive oil and of the fruit. The authors suggested that this could be due to the contribution of pollen donors in DNA extracted from the paste obtained by crushing whole fruits. Moreover, standardization of the genetic profile for both trees and oil should be made for each territory to account for eventual variations due the climate. Thus, care must be taken in the interpretation of DNA profiles obtained from DNA extracted from olive oil for assessing the origin and authenticity of the products.

An interesting application of DNA analysis is the possibility to detect adulteration of olive oil with other chipper oils. In our group, a method based on PNA-microarrays allowed to verify adulteration of olive oil with hazelnut oil by detecting the presence of DNA from hazelnut [16].

Although many efforts have been made in the last years, genome studies in *Olea europaea* L. are currently behind those of other crops. Several groups have started to work on the olive genome sequencing (i.e., OLEAGEN genomics project, Fundacion Genoma, Spain, www.chirimoyo.ac.uma.es/oleagen/) and, thanks to the rapid development of the new sequencing technologies, soon the complete sequence of olive genome will be available. This will allow to identify new sequence polymorphisms, which will be very useful for the development of new cultivar-specific molecular markers (e.g., SNPs) and for the implementation of more efficient protocols for tracking and protect olive oil quality.

2.6 Nutrigenetics/Nutrigenomics

Sequencing of human genome and of other organisms has determined a definitive revolution in the biomedical sciences and has provided a strong impulse to the development of the so called "omic sciences" (genomics, transcriptomics, proteomics, metabolomics, etc.). Integration of these sciences with informatics provides the basis for functional genomics, which aims at clarifying the role of single genes and how they interact with each other, with reciprocal influence. This complex network of genetic information applied to nutrition sciences leads to nutrigenetics and nutrigenomics, with the primary objective to investigate the interaction mechanisms between nutrients and genes and their role in the development of diseases for the general population and also for the single individual, opening the way to personalised diet. The development of concepts and research on genetic variations and dietary response has been named nutrigenetics [181] (e.g., individuals responding differently to the same diet by attaining different levels of serum cholesterol and blood pressure because of genetic variations), while studies on the role of nutrients in gene expression are defined as nutrigenomics (e.g., polyunsaturated fatty acids suppressing fatty acid synthase (mRNA) gene expression).

Relevant research projects have started in order to define the mechanisms by which genes influence nutrient absorption, metabolism and excretion, taste perception, and degree of satiation and the mechanisms by which nutrients influence gene expression [182]. Moreover, some important food industries have started programs to develop novel foods based on nutrigenetic/nutrigenomic concepts.

It is well known that the response of blood cholesterol concentration to diet is genotype-dependent [183–185] and therefore general recommendations such as to increase the uptake of polyunsaturated fatty acids in order to reduce the plasma level of cholesterol and consequently the risk of developing cardiovascular disease are not equally relevant for all individuals [186].

Despite the high complexity of these researches, it has been possible, at list in some cases, to identify correlations between several genetic variants, diet and development of cancer diseases. A number of epidemiologic studies and animal experiments suggest that omega-6 fatty acids increase the risk of certain cancers, whereas omega-3 fatty acids decrease this risk. But the correlation between the consumption of omega-6 and development of breast cancer has only been seen in subjects carrying a polymorphism of the protein activating the 5-lipoxygenase (ALOX5AP-4900 A > G) [187]. Another example is given by studies on the protective role of omega-3 fatty acids against the risk of developing prostate cancer only in correlation with specific genetic variants [188, 189]. Other examples about dietary factors and genetic interactions are reported in a recent review by Simopoulus [190].

As far as the effects of nutrients on gene expression (nutrigenomics), particularly relevant are the effects of dietary cholesterol and fatty acids. Dietary cholesterol exerts a strong inhibitory effect on the transcription of the gene for β -hydroxy- β -methyl-glutaryl (HMG)-CoA reductase [191]. Polyunsaturated fatty acids (PUFAs)

suppress the hepatic mRNA production of fatty acid synthase for lipoproteinemia in adult, according to the degree of fatty acid unsaturation. Thus, eicosapentaenoic acid (EPA), and docosahexaenoic acid (DHA) contained in fish oils are more effective than arachidonic acid (AA) (which has 4 double bonds) [192, 193].

Different types of carbohydrates (a rye pasta diet characterised by a low postprandial insulin response and an oat-wheat-potato diet characterised by a high postprandial insulin response) were shown to have a different effect on gene expression in subcutaneous adipose tissue in persons with metabolic syndrome [194]. The insulinogenic index improved after the rye-pasta diet and not after the oat-wheat-potato diet, indicating that different carbohydrates modulate gene expression and metabolism in patients with the metabolic syndrome.

The studies on the interactions gene-food components are in their infancy and performed with the methods and technologies derived from the "omic" sciences, as already partially described in this chapter.

A recent paper by Masotti et al. [195] reports how the microarray technology can be a promising tool in nutrigenomics, for the global evaluation of gene expression profiles in tissues and for understanding the factors controlling the regulation of gene transcription. These Authors show that oligonucleotide and cDNA microarrays represent powerful tools for a rapid and high-throughput gene expression evaluation, since they allow a quantitative assessment of the relative cell or tissue concentration of the specific messenger RNA (mRNA) that is directly related to the level of expression of that particular gene. The mRNA transcript present in the cells or tissues is extracted and is used to create a complementary labeled strand of DNA (cDNA), which can be hybridised with a complementary probe, previously spotted on a glass slide or nylon substrate containing a known set of gene sequences (genome). The intensity of the color, quantified by a fluorescent scanner after the hybridisation process, is directly related to the amount of target mRNA and reflects the expression level of that particular gene. In this way, it is possible to determine which gene is up-regulated or down-regulated as compared to the control sample.

This technology, however, has certain drawbacks. In fact, in order to obtain reliable and reproducible results, several parameters of the experimental protocol (array production, RNA extraction, cDNA labeling and hybridisation, and data analysis techniques) should be optimised.

The MicroArray Quality Control Consortium (MAQC) has recently evaluated the performance and reproducibility of the most common platforms and generated a set of data that justifies their use for gene expression profiling, both in basic and applied research leading to clinical diagnostic tools [196].

In conclusion, Nutrigenomics analyzes the physical and chemical characteristics of food components, as well as the structures and polymorphisms within the genome, with the aim to identify a pool of biomarkers that can predict the beneficial or adverse effects of dietary nutrients or components on the general population or on individuals, thus being used as predictive or screening tools to promote health and prevent disease. However, it is only by combining information from the different sciences and appropriate bioinformatics, that it will be possible to understand all aspects and implications of nutrition on health and its adverse toxicological effects.

2.7 General Conclusions and Perspectives

The development of affordable and robust quantitative assays for DNA in food represents a fundamental task in food safety and quality. The definition of reliable and harmonised strategies for nucleic acids quantification represent a big challenge both for scientists and international organisms involved in food safety regulation. One of the major limitations in this process is represented by the employment of techniques for the detection of nucleic acids that often are organism- and matrix-dependent. These are among the reasons that explain why the definition of common norms and guidelines takes very long times requiring a huge amount of experimental work. After several years, international organisms have defined some minimal requirements for PCR but the debate on which are the best quantification practices for PCR applications is still going on at international level. Moreover, a similar process needs to be developed for the new generation of isothermal methods and for emerging pathogens like foodborne viruses.

The development of direct, amplification-free approaches for the detection of nucleic acids represents an opportunity to improve the area of quantitative diagnostics for food analysis. The possibility to remove enzymes from the process, exclusively relying on direct nucleic acids binding should minimize the effects of the matrix on the final results. On the other hand, this approach may present other issues to be faced: one of the most relevant should be the definition of reference materials for calibration curves. Absolute quantification for Real Time PCR is usually performed by means of plasmid DNA, RT-PCR products or synthetic DNA/RNA oligonucleotides. These products behave similarly to the targeted genes in terms of amplification efficiency because they are designed to be similar (sometimes identical) in length and in sequence composition. Whenever amplification-free nucleic acid methods are supposed to be designed to detect direct signals from a genomic DNA portion or a transcribed RNA, this aspect should be taken into account: since genomic DNAs and cellular RNAs are often bigger and highly structurated molecules compared to the oligonucleotides usually employed for absolute quantification, it is possible to obtain very different signals. However, if this issue will be successfully minimised, quantitative direct nucleic acids detection will open a completely new era in food diagnostics, nutrigenetics and nutrigenomics.

Another point concerns the concomitant needs for effective pre-analytical methods to be easily integrated with the detection methods, which seem to grow slowly and often not considered for an integrated development. This issue becomes more and more relevant for the direct nucleic acids detection methods that are theoretically able to move from sample preparation directly to the detection step, without any additional process. Since the pre-analytical treatments represent the bottle-neck of numerous analytical processes because are still highly time-consuming and labour intensive, the pre-analytical processing must be contextually investigated and developed with the methods focused on amplification-free detection of nucleic acids. Otherwise, even the most efficient method will not fully accomplish the needs in the food analysis field, reducing the impact of these innovative techniques on real diagnostics.

Moreover, the possibility to integrate all the concepts previously described might lead to the development of automated, miniaturised and portable devices that will help rapid detection of foodborne pathogens and process monitoring all along the food-chain, strengthening the chance to develop a "food diagnostics from farm-tofork". Methods that rely on direct detection of nucleic acids appear to be particularly suitable for this purpose.

Since foods have complex and heterogeneous matrices, the next-generation methods should be able to detect targeted sequences in a robust, accurate matrixindependent way. This means nucleic acids coming from the same sample should be directly detected giving comparable results, regardless of the methods employed during the concentration/purification steps.

Finally, the development of customised methods for food diagnostics, covering all the critical steps in food safety and quality, will enable to face the actual limitations in contexts like *in-line* processes or *on-site* settings. Optimising the time and costs required for a single analysis, along with easy-to-perform procedures will open interesting opportunities in food diagnostics.

References

- 1. Cifuentes, A.: Food analysis and foodomics. J. Chromatogr. A 1216, 7109 (2009)
- Herrero, M., Garcıa-Canas, V., Simo, C., Cifuentes, A.: Recent advances in the application of CE methods for food analysis and foodomics. Electrophoresis 31, 205–228 (2010)
- Herrero, M., Simó, C., Garcia-Canas, V., Ibanez, E., Cifuentes, A.: Foodomics: MS-based strategies in modern food science and nutrition. Mass Spectrom. Rev. 31, 49–69 (2012)
- Querci, M., Van den Bulcke, M., Žel, J., Van den Eede, G., Broll, H.: New approaches in GMO detection. Anal. Bioanal. Chem. 396, 1991–2002 (2010)
- Dwivedi, H.P., Jaykus, L.A.: Microbial detection of pathogens in foods: the current state-ofthe-art and future directions. Crit. Rev. Microbiol. 37, 40–63 (2011)
- Monaci, L., Visconti, A.: Immunochemical and DNA-based methods in food allergen analysis and quality assurance perspectives. Trends Food Sci. Technol. 21, 272–283 (2010)
- Mafra, I., Ferreira, I.M.P.L.V.O., Oliveira, M.B.P.P.: Food authentication by PCR-based methods. Eur. Food Res. Technol. 227, 649–665 (2008)
- Di Pinto, A., Forte, V.T., Guastadisegni, M.C., Martino, C., Schena, F.P., Tantillo, G.A.: Comparison of DNA extraction methods for food analysis. Food Control 18, 76–80 (2007)
- Stephan, O., Vieths, S.: Development of a real-time PCR and a sandwich ELISA for detection of potentially allergenic trace amounts of peanut (Arachis hypogaea) in processed foods. J. Agric. Food Chem. 52, 3754–3760 (2004)
- Di Bernardo, G., Del Gaudio, S., Galderisi, U., Cascino, A., Cipollaro, M.: Comparative evaluation of different DNA extraction procedures from food samples. Biotechnol. Prog. 23, 297–301 (2007)

- Mullis, K., Faloona, F., Scharf, S., Saiki, R., Horn, G., Erlich, H.: Specific enzymatic amplification of DNA in vitro: the polymerase chain reaction. Biotechnology 24, 17–27 (1986)
- 12. Vester, B., Wengel, J.: LNA (Locked Nucleic Acid): high-affinity targeting of complementary RNA and DNA. Biochemistry **43**, 13233–13241 (2004)
- Summerton, J., Weller, D.: Morpholino antisense oligomers: design, preparation, and properties. Antisense Nucleic Acid Drug Dev. 7, 187–195 (1997)
- Nielsen, P.E., Egholm, M. (eds.): Peptide Nucleic Acids: Protocols and Applications, 2nd edn. Horizon Press, Wymondham (2004)
- Corradini, R., Sforza, S., Tedeschi, T., Totsingan, F., Manicardi, A., Marchelli, R.: Peptide nucleic acids with a structurally biased backbone. Updated review and emerging challenges. Curr. Top. Med. Chem. 11, 1535–1554 (2011)
- Sforza, S., Corradini, R., Tedeschi, T., Marchelli, R.: Food analysis and food authentication by peptide nucleic acid (PNA)-based technologies. Chem. Soc. Rev. 40, 221–232 (2011)
- 17. Weighardt, F.: GMO quantification in processed food and feed. Nat. Biotechnol. 25, 1213–1214 (2007)
- Coisson, J.D., Cereti, E., Garino, C., D'Andrea, M., Recupero, M., Restani, P., Arlorio, M.: Microchip capillary electrophoresis (Lab-on-chip (R)) improves detection of celery (Apium graveolens L.) and sesame (Sesamum indicum L.) in foods. Food Res. Int. 43, 1237–1243 (2010)
- 19. Lauri, A., Mariani, P.O.: Potentials and limitations of molecular diagnostic methods in food safety. Genes Nutr. 4, 1–12 (2009)
- 20. James, C.: ISAAA Brief No. 39. ISAAA, Ithaca (2008)
- 21. European Commission Off.: J. Eur. Communities L106, 1-38 (2001)
- Regulation (EC) No 1829/2003: The European Parliament and the Council of the European Union on genetically modified food and feed. Off. J. Eur. Union L268, 1–23 (2003)
- 23. Regulation (EC) No 1830/2003: The European Parliament and the Council of the European Union concerning the traceability and labelling of genetically modified organisms and the traceability of food and feed products produced from genetically modified organisms and amending Directive 2001/18/EC. Off. J. Eur. Union L268, 24–28 (2003)
- 24. Carter, C.A., Gruere, G.P.: International approval and labeling regulations of genetically modified food in major trading countries. In: Just, R., Alston, J.M., Zilberman, D. (eds.) Regulating Agricultural Biotechnology. Economics and Policies, pp. 459–480. Springer, New York (2006)
- Ramessar, K., Capell, T., Twyman, R.M., Quemada, H., Christou, P.: Trace and traceability-a call for regulatory harmony. Nat. Biotechnol. 26, 975–978 (2008)
- 26. CERA (Center for Environmental Risk Assessment): GM database. Home page. http://www. cera-gmc.org/?action=gm_crop_database (2009)
- BATS (Centre for Biosafety and Sustainability): Genetically modified (GM) crops: molecular and regulatory details. Home page. http://www.bats.ch/gmo-watch/ (2003)
- GMDD (Genetically Modified Organism Detection Method Database): Home page. http:// gmdd.shgmo.org/ (2011)
- 29. GMO-Compass: Home page. http://www.gmo-compass.org/eng/ (2010)
- 30. CRL (Community Reference Laboratory): Home page. http://gmo-crl.jrc.ec.europa.eu/ (2011)
- NCBI (National Centre for Biotechnology Information): Home page. http://blast.ncbi.nlm. nih.gov/ (2011)
- Peano, C., Bordoni, R., Gulli, M., Mezzelani, A., Samson, M.C., DeBellis, G., Marmiroli, N.: Multiplex polymerase chain reaction and ligation detection reaction/universal array technology for the traceability of genetically modified organisms in foods. Anal. Biochem. 346, 90–100 (2005)
- 33. Germini, A., Zanetti, A., Salati, C., Rossi, S., Forré, C., Schmid, S., Marchelli, R.: Development of a seven-target multiplex PCR for the simultaneous detection of transgenic soybean and maize in feeds and foods. J. Agric. Food Chem. 52, 3275–3280 (2004)

- 34. Ao, J.X., Li, Q.Z., Gao, X.J., Yu, Y.B., Li, L., Zhang, M.H.: A multiplex nested PCR assay for the simultaneous detection of genetically modified soybean, maize and rice in highly processed products. Food Control 22, 1617–1623 (2011)
- Peano, C., Lesignoli, F., Gulli, M., Corradini, R., Samson, M.C., Marchelli, R., Marmiroli, N.: Development of a peptide nucleic acid polymerase chain reaction clamping assay for semiquantitative evaluation of genetically modified organism content in food. Anal. Biochem. 344, 174–182 (2005)
- 36. Rossi, S., Lesignoli, F., Germini, A., Faccini, A., Sforza, S., Corradini, R., Marchelli, R.: Identification of PCR-amplified genetically modified organisms (GMOs) DNA by peptide nucleic acid (PNA) probes in anion-exchange chromatographic analysis. J. Agric. Food Chem. 55, 2509–2516 (2007)
- 37. Totsingan, F., Rossi, S., Corradini, R., Tedeschi, T., Sforza, S., Juris, A., Scaravelli, E., Marchelli, R.: Label-free selective DNA detection with high mismatch recognition by PNA beacons and ion exchange HPLC. Org. Biomol. Chem. 6, 1232–1237 (2008)
- 38. Van den Bulcke, M., Lievens, A., Barbeau-Piednoir, E., MbongoloMbella, G., Roosens, N., Sneyers, M., Casi, A.L.: A theoretical introduction to "Combinatory SYBR[®]Green qPCR Screening", a matrix-based approach for the detection of materials derived from genetically modified plants. Anal. Bioanal. Chem. **396**, 2113–2123 (2010)
- Querci, M., Foti, N., Bogni, A., Kluga, L., Broll, H., Van den Eede, G.: Real-time PCRbased ready-to-use multi-target analytical system for GMO detection food. Anal. Methods 2, 325–336 (2009)
- Germini, A., Mezzelani, A., Lesignoli, F., Corradini, R., Marchelli, R., Bordoni, R., Consolandi, C., De Bellis, G.: Detection of genetically modified soybean using peptide nucleic acids (PNAs) and microarray technology. J. Agric. Food Chem. 52, 4535–4540 (2004)
- Germini, A., Rossi, S., Zanetti, A., Corradini, R., Fogher, C., Marchelli, R.: Development of a PNA array platform for the detection of genetically modified organisms in food. J. Agric. Food Chem. 53, 3958–3962 (2005)
- Hamels, S., Glouden, T., Gillard, K., Mazzara, M., Debode, F., Foti, N., Sneyers, M., Esteve Nuez, T., Pla, M., Berben, G., Moens, W., Bertheau, Y., Audéon, C., Van den Eede, G., Remacle, J.: A PCR-microarray method for the screening of genetically modified organisms. Eur. Food Res. Technol. 228, 531–541 (2009)
- Notomi, T., Okayama, H., Masubuchi, H., Yonekawa, T., Watanabe, K., Amino, N., Hase, T.: Loop-mediated isothermal amplification of DNA. Nucleic Acids Res. 28, E63 (2000)
- 44. Fukuta, S., Mizukami, Y., Ishida, A., Ueda, J., Hasegawa, M., Hayashi, I., Hashimoto, M., Kanbe, M.: Real-time loop-mediated isothermal amplification for the CaMV-35S promoter as a screening method for genetically modified organisms. Eur. Food Res. Technol. 218, 496–500 (2004)
- Barany, F.: Genetic disease detection and DNA amplification using cloned thermostable ligase. Proc. Natl. Acad. Sci. USA 88, 189–193 (1991)
- Gerry, N.P., Witowski, N.E., Day, J.P., Hammer, R.P., Barany, G., Barany, F.: Universal DNA microarray method for multiplex detection of low abundance point mutations. J. Mol. Biol. 292, 251–262 (1999)
- Bordoni, R., Germini, A., Mezzelani, A., Marchelli, R., De Bellis, G.: A microarray platform for parallel detection of five transgenic events in foods: a combined polymerase chain reaction-ligation detection reaction-universal array method. J. Agric. Food Chem. 53, 912–918 (2005)
- Bordoni, R., Mezzelani, A., Consolandi, C., Frosini, A., Rizzi, E., Castiglioni, B., Salati, C., Marmiroli, N., Marchelli, R., Bernardi, L.R., Battaglia, C., De Bellis, G.: Detection and quantitation of genetically modified maize (Bt-176 transgenic maize) by applying ligation detection reaction and universal array technology. J. Agric. Food Chem. **52**, 1049–1054 (2004)
- Kievits, T., van Gemen, B., van Strijp, D., Schukkink, R., Dircks, M., Adriaanse, H.M.A., Malek, L., Sooknanan, R., Lens, P.: NASBA isothermal enzymatic in vitro nucleic acid amplification optimized for the diagnosis of HIV-1 infection. J. Virol. Methods 35, 273–286 (1991)

- Morisset, D., Stebih, D., Cankar, K., Žel, J., Gruden, K.: Alternative DNA amplification methods to PCR and their application in GMO detection: a review. Eur. Food Res. Technol. 227, 1287–1297 (2008)
- Morisset, D., Dobnik, D., Hamels, S., Žel, J., Gruden, K.: NAIMA: target amplification strategy allowing quantitative on-chip detection of GMOs. Nucleic Acids Res. 36, e118 (2008)
- 52. Dobnik, D., Morisset, D., Gruden, K.: NAIMA as a solution for future GMO diagnostics challenges. Anal. Bioanal. Chem. **396**, 2229–2233 (2010)
- 53. Waiblinger, H.U., Boernsen, B., Pietsch, K.: Praktische Anwendung für die Routineanalytik Screening-Tabelle für den Nachweis zugelassener und nicht zugelassener gentechnisch veränderter Pflanzen. Dtsch Lebensm Rundsch 104, 261–264 (2008)
- 54. Waiblinger, H.U., Ernst, B., Anderson, A., Pietsch, K.: Validation and collaborative study of a P35S and T-nos duplex real-time PCR screening method to detect genetically modified organisms in food products. Eur. Food Res. Technol. 226, 1221–1228 (2007)
- Waiblinger, H.U., Grohmann, L., Mankertz, J., Engelbert, D., Pietsch, K.: A practical approach to screen for authorized and unauthorised genetically modified plants. Anal. Bioanal. Chem. **396**, 2065–2072 (2010)
- 56. Grohmann, L., Brünen-Nieweler, C., Nemeth, A., Waiblinger, H.U.: Collaborative trial validation studies of real-time PCR-based GMO screening methods for detection of the bar gene and the ctp2-cp4epsps construct. J. Agric. Food Chem. 57, 8913–8920 (2009)
- Reiting, R., Broll, H., Waiblinger, H.U., Grohmann, L.: Collaborative study of a T-nos realtime PCR method for screening of genetically modified organisms in food products. J. Verbr. Lebensm. 2, 116–121 (2007)
- 58. Clive James International Service for the Acquisition of Agri-biotech Applications (ISAAA): Executive Summary brief 42. Global status of Commercialized biotech/GM Crops. http:// www.isaaa.org/resources/publications/briefs/42/executivesummary/pdf/Brief%2042%20-%20Executive%20Summary%20-%20English.pdf (2010)
- Minunni, M., Tombelli, S., Fonti, J., Spiriti, M.M., Mascini, M., Bogani, P., Buiatti, M.: Detection of fragmented genomic DNA by PCR-free piezoelectric sensing using a denaturation approach. J. Am. Chem. Soc. **127**, 7966–7967 (2005)
- D'Agata, R., Corradini, R., Grasso, G., Marchelli, R., Spoto, G.: Ultrasensitive detection of DNA by PNA and nanoparticle-enhanced surface plasmon resonance imaging. Chembiochem 9, 2067–2070 (2008)
- 61. D'Agata, R., Corradini, R., Ferretti, C., Zanoli, L., Gatti, M., Marchelli, R., Spoto, G.: Ultrasensitive detection of non-amplified genomic DNA by nanoparticle-enhanced surface plasmon resonance imaging. Biosens. Bioelectron. 25, 2095–2100 (2010)
- 62. Boyce, J.A., Assa'ad, A., Burks, A.W., Jones, S.M., Sampson, H.A., Wood, R.A., Plaut, M., Cooper, S.F., Fenton, M.J., Arshad, S.H., Bahna, S.L., Beck, L.A., Byrd-Bredbenner, C., Camargo Jr., C.A., Eichenfield, L., Furuta, G.T., Hanifin, J.M., Jones, C., Kraft, M., Levy, B.D., Lieberman, P., Luccioli, S., McCall, K.M., Schneider, L.C., Simon, R.A., Simons, F.E., Teach, S.J., Yawn, B.P., Schwaninger, J.M.: NIAID-sponsored expert panel: guidelines for the diagnosis and management of food allergy in the United States: summary of the NAID sponsored expert panel. J. Allergy Clin. Immunol. **126**, 1105–1118 (2010)
- 63. Sicherer, S.H.: Epidemiology of food allergy. J. Allergy Clin. Immunol. 127, 594-602 (2011)
- 64. Commission Directive 2000/13/EC: Off. J. Eur. Union L109, 29-42 (2000)
- 65. Commission Directive 2003/89/EC: Off. J. Eur. Union L308, 15-18 (2003)
- 66. Commission Directive 2006/142/EC: Off. J. Eur. Union L368, 110-111 (2006)
- 67. Commission Directive 2007/68/EC: Off. J. Eur. Union L310, 11-14 (2007)
- Poms, R.E., Klein, C.L., Anklam, E.: Methods for allergen analysis in food: a review. Food Addit. Contam. 21, 1–31 (2004)
- van Hengel, A.J.: Food allergen detection methods and the challenge to protect food-allergic consumers. Anal. Bioanal. Chem. 389, 111–118 (2007)

- Schubert-Ullrich, P., Rudolf, J., Ansari, P., Galler, B., Führer, M., Molinelli, A., Baumgartner, S.: Commercialized rapid immunoanalytical tests for determination of allergenic food proteins: an overview. Anal. Bioanal. Chem. **395**, 69–81 (2009)
- Westphal, C.D., Pereira, M.R., Raybourne, R.B., Williams, K.M.: Evaluation of extraction buffers using the current approach of detecting multiple allergenic and nonallergenic proteins in food. J. AOAC Int. 87, 1458–1465 (2004)
- 72. Poms, R.E., Capelletti, C., Anklam, E.: Effect of roasting history and buffer composition on peanut protein extraction efficiency. Mol. Nutr. Food Res. **48**, 459–464 (2004)
- Taylor, S.L., Nordlee, J.A., Niemann, L.M., Lambrecht, D.M.: Allergen immunoassays considerations for use of naturally incurred standards. Anal. Bioanal. Chem. **395**, 83–92 (2009)
- 74. Matsuda, R., Yoshioka, Y., Akiyama, H., Aburatani, K., Watanabe, Y., Matsumoto, T., Morishita, N., Sato, H., Mishima, T., Gamo, R., Kihira, Y., Maitani, T.: Interlaboratory evaluation of two enzyme-linked immunosorbent assay kits for the detection of egg, milk, wheat, buckwheat, and peanut in foods. J. AOAC Int. 89, 1600–1608 (2006)
- 75. Hefle, S., Yeung, J., Helm, R.: Antibodies. In: Koppermann, J., Hefle, L. (eds.) Detecting Allergens in Food, pp. 65–78. CRC Press, Boca Raton (2006)
- 76. Poms, R., Emons, H., Anklam, E.: Reference materials and method validation in allergen detection. In: Koppermann, J., Hefle, L. (eds.) Detecting Allergens in Food, pp. 348–356. CRC Press, Boca Raton (2006)
- Abbott, M., Hayward, S., Ross, W., Godefroy, S.B., Ulberth, F., Van Hengel, A.J., Roberts, J., Akiyama, H., Popping, B., Yeung, J.M., Wehling, P., Taylor, S.L., Poms, R.E., Delahaut, P.: Validation procedures for quantitative food allergen ELISA methods: community guidance and best practices. J. AOAC Int. 93, 442–450 (2010)
- Monaci, L., van Hengel, A.J.: Development of a method for the quantification of whey allergen traces in mixed-fruit juices based on liquid chromatography with mass spectrometric detection. J. Chromatogr. A **1192**, 113–120 (2008)
- Schneider, N., Becker, C.M., Pischetsrieder, M.: Analysis of lysozyme in cheese by immunocapture mass spectrometry. J. Chromatogr. B Analyt. Technol. Biomed. Life Sci. 878, 201–206 (2010)
- Shefcheck, K.J., Callahan, J.H., Musser, S.M.: Confirmation of peanut protein using peptide markers in dark chocolate using liquid chromatography-tandem mass spectrometry (LC-MS/MS). J. Agric. Food Chem. 54, 7953–7959 (2006)
- Careri, M., Elviri, L., Lagos, J.B., Mangia, A., Speroni, F., Terenghi, M.: Selective and rapid immunomagnetic bead-based sample treatment for the liquid chromatography-electrospray ion-trap mass spectrometry detection of Ara h3/4 peanut protein in foods. J. Chromatogr. A 1206, 89–94 (2008)
- Cavatorta, V., Sforza, S., Mastrobuoni, G., Pieraccini, G., Francese, S., Moneti, G., Dossena, A., Pastorello, E.A., Marchelli, R.: Unambiguous characterization and tissue localization of Pru P 3 peach allergen by electrospray mass spectrometry and MALDI imaging. J. Mass Spectrom. 44, 891–897 (2009)
- Schoringhumer, K., Redl, G., Cichna-Markl, M.: Development and validation of a duplex real-time PCR method to simultaneously detect potentially allergenic sesame and hazelnut in food. J. Agric. Food Chem. 57, 2126–2134 (2009)
- 84. D'Andrea, M., Coïsson, J.D., Locatelli, M., Garino, C., Cereti, E., Arlorio, M.: Validating allergen coding genes (Cor a 1, Cor a 8, Cor a 14) as target sequences for hazelnut detection via Real-Time PCR. Food Chem. **124**, 1164–1171 (2011)
- Dovicovicova, L., Olexova, L., Pangallo, D., Siekel, P., Kuchta, T.: Polymerase chain reaction (PCR) for the detection of celery (Apium graveolens) in food. Eur. Food Res. Technol. 218, 493–495 (2004)
- 86. Germini, A., Scaravelli, E., Lesignoli, F., Sforza, S., Corradini, R., Marchelli, R.: Polymerase chain reaction coupled with peptide nucleic acid high-performance liquid chromatography for the sensitive detection of traces of potentially allergenic hazelnut in foodstuffs. Eur. Food Res. Technol. 220, 619–624 (2005)

- Rossi, S., Scaravelli, E., Germini, A., Corradini, R., Fogher, C., Marchelli, R.: A PNA-array platform for the detection of hidden allergens in foodstuffs. Eur. Food Res. Technol. 223, 1–6 (2006)
- Holzhauser, T., Stephan, O., Vieths, S.: Detection of potentially allergenic hazelnut (Corylus avellana) residues in food: a comparative study with DNA PCR-ELISA and protein sandwich-ELISA. J. Agric. Food Chem. 50, 5808–5815 (2002)
- Scaravelli, E., Brohee, M., Marchelli, R., van Hengel, A.J.: Development of three real-time PCR assays to detect peanut allergen residue in processed food products. Eur. Food Res. Technol. 227, 857–869 (2008)
- Pafundo, S., Gullì, M., Marmiroli, N.: Comparison of DNA extraction methods and development of duplex PCR and real-time PCR to detect tomato, carrot, and celery in food. J. Agric. Food Chem. 12, 10414–10424 (2011)
- Scallan, E., Griffin, P.M., Angulo, F.J., Tauxe, R.V., Hoekstra, R.M.: Foodborne illness acquired in the United States—unspecified agents. Emerg. Infect. Dis. 17, 16–22 (2011)
- Scallan, E., Hoekstra, R.M., Angulo, F.J., et al.: Foodborne illness acquired in the United States—major pathogens. Emerg. Infect. Dis. 17, 7–15 (2011)
- Juste, A., Thomma, B.P., Lievens, B.: Recent advances in molecular techniques to study microbial communities in food-associated matrices and processes. Food Microbiol. 25, 745e761 (2008)
- 94. Seo, K.H., Valentin-Bon, I.E., Brackett, R.E.: Detection and enumeration of Salmonella Enteritidis in homemade ice cream associated with an outbreak, comparison of conventional and real-time PCR methods. J. Food Prot. 69, 639–643 (2006)
- McCarthy, N., Reen, F.J., Buckley, J.F., Frye, J.G., Boyd, E.F., Gilroy, D.: Sensitive and rapid molecular detection assays for Salmonella enterica serovars Typhimurium and Heidelberg. J. Food Prot. 72, 2350–2357 (2009)
- Hein, I., Flekna, G., Krassnig, M., Wagner, M.: Real-time PCR for the detection of Salmonella spp in food. An alternative approach to a conventional PCR system suggested by the FOOD-PCR project. J. Microbiol. Methods 66, 538–547 (2006)
- Cheng, C.M., Lin, W., Van, K.T., Phan, L., Tran, N.N., Farmer, D.: Rapid detection of Salmonella in foods using real-time PCR. J. Food Prot. 71, 2436–2441 (2008)
- Gonzalez-Escalona, N., Hammack, T.S., Russell, M., Jacobson, A.P., De Jesus, A.J., Brown, E.W., Lampel, K.A.: Detection of live Salmonella sp cells in produce by a TaqMan-based quantitative reverse transcriptase real-time PCR targeting invA mRNA. Appl. Environ. Microbiol. 75, 3714–3720 (2009)
- 99. Hong, J., Jung, W.K., Kim, J.M., Kim, S.H., Koo, H.C., Ser, J., Park, Y.H.: Quantification and differentiation of Campylobacter jejuni and Campylobacter coli in raw chicken meats using a real-time PCR method. J. Food Prot. **70**, 2015–2022 (2007)
- 100. Wolffs, P.F., Glencross, K., Norling, B., Griffiths, M.W.: Simultaneous quantification of pathogenic Campylobacter and Salmonella in chicken rinse fluid by a flotation and real-time multiplex PCR procedure. Int. J. Food Microbiol. **117**, 50–54 (2007)
- Singh, J., Batish, V.K., Grover, S.: A scorpion probe-based real-time PCR assay for detection of E. coli O157:H7 in dairy products. Foodborne Pathog. Dis. 6, 395–400 (2009)
- 102. Singh, J., Batish, V.K., Grover, S.: A molecular beacon-based duplex real-time polymerase chain reaction assay for simultaneous detection of Escherichia coli O157:H7 and Listeria monocytogenesin milk and milk products. Foodborne Pathog. Dis. 6, 1195–1201 (2009)
- 103. Aparecida de Oliveira, M., Abeid Ribeiro, E.G., Morato Bergamini, A.M., Pereira De Martinis, E.C.: Quantification of Listeria monocytogenes in minimally processed leafy vegetables using a combined method based on enrichment and 16 S rRNA real-time PCR. Food Microbiol. 27, 19–23 (2010)
- 104. Rossmanith, P., Krassnig, M., Wagner, M., Hein, I.: Detection of Listeria monocytogenesin food using a combined enrichment/real-time PCR method targeting the prfA gene. Res. Microbiol. 157, 763–771 (2006)
- 105. Morales-Rayas, R., Wolffs, P.F., Griffiths, M.W.: Simultaneous separation and detection of hepatitis A virus and norovirus in produce. Int. J. Food Microbiol. 139, 48–55 (2010)
- 106. Deiman, B., van Aarle, P., Sillekens, P.: Characteristics and applications of nucleic acid sequence-based amplification (NASBA). Mol. Biotechnol. 20, 163–179 (2002)
- 107. Parida, M., Sannarangaiah, S., Dash, P.K., Rao, P.V., Morita, K.: Loop mediated isothermal amplification (LAMP): a new generation of innovative gene amplification technique; perspectives in clinical diagnosis of infectious diseases. Rev. Med. Virol. 18, 407–421 (2008)
- Mori, Y., Notomi, T.: Loop-mediated isothermal amplification (LAMP): a rapid, accurate, and cost-effective diagnostic method for infectious diseases. J. Infect. Chemother. 15, 62–69 (2009)
- Piepenburg, O., Williams, C.H., Stemple, D.L., Armes, N.A.: DNA detection using recombination proteins. PLoS Biol. 4, e204 (2006)
- Vincent, M., Xu, Y., Kong, H.: Helicase-dependent isothermal DNA amplification. EMBO Rep. 5, 795–800 (2004)
- 111. Jeong, Y.J., Park, K., Kim, D.E.: Isothermal DNA amplification in vitro: the helicasedependent amplification system. Cell. Mol. Life Sci. 66, 3325–3336 (2009)
- 112. Zhao, W., Ali, M.M., Brook, M.A., Li, Y.: Rolling circle amplification: applications in nanotechnology and biodetection with functional nucleic acids. Angew. Chem. Int. Ed. Engl 47, 6330–6337 (2008)
- 113. Lamhoujeb, S., Fliss, I., Ngazoa, S.E., Jean, J.: Evaluation of the persistence of infectious human noroviruses on food surfaces by using real-time nucleic acid sequence-based amplification. Appl. Environ. Microbiol. 74, 3349–3355 (2008)
- 114. Kou, X., Wu, Q., Zhang, J., Fan, H.: Rapid detection of noroviruses in fecal samples and shellfish by nucleic acid sequence-based amplification. J. Microbiol. 44, 403–408 (2006)
- 115. Jean, J., D'Souza, D.H., Jaykus, L.A.: Multiplex nucleic acid sequence-based amplification for simultaneous detection of several enteric viruses in model ready-to-eat foods. Appl. Environ. Microbiol. **70**, 6603–6610 (2004)
- 116. Churruca, E., Girbau, C., Martínez, I., Mateo, E., Alonso, R., Fernández-Astorga, A.: Detection of Campylobacter jejuni and Campylobacter coli in chicken meat samples by real-time nucleic acid sequence-based amplification with molecular beacons. Int. J. Food Microbiol. 117, 85–90 (2007)
- 117. Cools, I., Uyttendaele, M., D'Haese, E., Nelis, H.J., Debevere, J.: Development of a realtime NASBA assay for the detection of Campylobacter jejuni cells. J. Microbiol. Methods 66, 313–320 (2006)
- 118. Nadal, A., Coll, A., Cook, N., Pla, M.: A molecular beacon-based real time NASBA assay for detection of Listeria monocytogenes in food products: role of target mRNA secondary structure on NASBA design. J. Microbiol. Methods 68, 623–632 (2007)
- 119. Abd el-Galil, K.H., el-Sokkary, M.A., Kheira, S.M., Salazar, A.M., Yates, M.V., Chen, W., Mulchandani, A.: Real-time nucleic acid sequence-based amplification assay for detection of hepatitis A virus. Appl. Environ. Microbiol. **71**, 7113–7116 (2005)
- 120. Zhang, G., Brown, E.W., González-Escalona, N.: Comparison of real-time PCR, reverse transcriptase real-time PCR, loop-mediated isothermal amplification, and the FDA conventional microbiological method for the detection of Salmonella spp in produce. Appl. Environ. Microbiol. 77, 6495–6501 (2011)
- 121. Chen, S., Wang, F., Beaulieu, J.C., Stein, R.E., Ge, B.: Rapid detection of viable salmonellae in produce by coupling propidium monoazide with loop-mediated isothermal amplification. Appl. Environ. Microbiol. **77**, 4008–4016 (2011)
- 122. Techathuvanan, C., Draughon, F.A., D'Souza, D.H.: Comparison of reverse transcriptase PCR, reverse transcriptase loop-mediated isothermal amplification, and culture-based assays for Salmonella detection from pork processing environments. J. Food Prot. 74, 294–301 (2011)
- 123. Yang, J.L., Ma, G.P., Yang, R., Yang, S.Q., Fu, L.Z., Cheng, A.C., Wang, M.S., Zhang, S.H., Shen, K.F., Jia, R.Y., Deng, S.X., Xu, Z.Y.: Simple and rapid detection of Salmonella serovar Entertitidis under field conditions by loop-mediated isothermal amplification. J. Appl. Microbiol. 109, 1715–1723 (2010)

- 124. Techathuvanan, C., Draughon, F.A., D'Souza, D.H.: Loop-mediated isothermal amplification (LAMP) for the rapid and sensitive detection of Salmonella Typhimurium from pork. J. Food Sci. 75, M165–M172 (2010)
- 125. Ueda, S., Kuwabara, Y.: The rapid detection of Salmonella from food samples by loopmediated isothermal amplification (LAMP). Biocontrol Sci. **14**, 73–76 (2009)
- 126. Hara-Kudo, Y., Yoshino, M., Kojima, T., Ikedo, M.: Loop-mediated isothermal amplification for the rapid detection of Salmonella. FEMS Microbiol. Lett. **253**, 155–161 (2005)
- 127. Tang, M.J., Zhou, S., Zhang, X.Y., Pu, J.H., Ge, Q.L., Tang, X.J., Gao, Y.S.: Rapid and sensitive detection of Listeria monocytogenes by loop-mediated isothermal amplification. Curr. Microbiol. 63, 511–516 (2011)
- 128. Wang, L., Li, Y., Chu, J., Xu, Z., Zhong, Q.: Development and application of a simple loop-mediated isothermal amplification method on rapid detection of Listeria monocytogenes strains. Mol. Biol. Rep. **39**, 445–449 (2012)
- 129. Lu, X., Mo, Z.Y., Zhao, H.B., Yan, H., Shi, L.: LAMP-based method for a rapid identification of Legionella spp. and Legionella pneumophila. Appl. Microbiol. Biotechnol. 92, 179–187 (2011)
- Wang, F., Jiang, L., Ge, B.: Loop-mediated isothermal amplification assays for detecting shiga toxin-producing Escherichia coliin ground beef and human stools. J. Clin. Microbiol. 50, 91–97 (2012)
- 131. Kaneko, I., Miyamoto, K., Mimura, K., Yumine, N., Utsunomiya, H., Akimoto, S., McClane, B.A.: Detection of enterotoxigenic Clostridium perfringens in meat samples by using molecular methods. Appl. Environ. Microbiol. **77**, 7526–7532 (2011)
- 132. Surasilp, T., Longyant, S., Rukpratanporn, S., Sridulyakul, P., Sithigorngul, P., Chaivisuthangkura, P.: Rapid and sensitive detection of Vibrio vulnificus by loop-mediated isothermal amplification combined with lateral flow dipstick targeted to rpoS gene. Mol. Cell. Probes 25, 158–163 (2011)
- 133. Han, F., Wang, F., Ge, B.: Detecting potentially virulent Vibrio vulnificus strains in raw oysters by quantitative loop-mediated isothermal amplification. Appl. Environ. Microbiol. 77, 2589–2595 (2011)
- 134. Yoda, T., Suzuki, Y., Yamazaki, K., Sakon, N., Kanki, M., Kase, T., Takahashi, K., Inoue, K.: Application of a modified loop-mediated isothermal amplification kit for detecting Norovirus genogroups I and II. J. Med. Virol. 81, 2072–2078 (2009)
- 135. Niessen, L., Vogel, R.F.: Detection of Fusarium graminearum DNA using a loop-mediated isothermal amplification (LAMP) assay. Int. J. Food Microbiol. **140**, 183–191 (2010)
- 136. Yamazaki, W., Taguchi, M., Ishibashi, M., Kitazato, M., Nukina, M., Misawa, N., Inoue, K.: Development and evaluation of a loop-mediated isothermal amplification assay for rapid and simple detection of Campylobacter jejuni and Campylobacter coli. J. Med. Microbiol. 57, 444–451 (2008)
- Perlin, D.S., Zhao, Y.: Molecular diagnostic platforms for detecting Aspergillus. Med. Mycol. 47(Suppl 1), S223–S232 (2009)
- 138. Wang, C.H., Lien, K.Y., Wu, J.J., Lee, G.B.: A magnetic bead-based assay for the rapid detection of methicillin-resistant Staphylococcus aureus by using a microfluidic system with integrated loop-mediated isothermal amplification. Lab Chip **11**, 1521–1531 (2011)
- 139. Dimov, I.K., Garcia-Cordero, J.L., O'Grady, J., Poulsen, C.R., Viguier, C., Kent, L., Daly, P., Lincoln, B., Maher, M., O'Kennedy, R., Smith, T.J., Ricco, A.J., Lee, L.P.: Integrated microfluidic tmRNA purification and real-time NASBA device for molecular diagnostics. Lab Chip 8, 2071–2078 (2008)
- 140. Liu, C., Geva, E., Mauk, M., Qiu, X., Abrams, W.R., Malamud, D., Curtis, K., Owen, S.M., Bau, H.H.: An isothermal amplification reactor with an integrated isolation membrane for point-of-care detection of infectious diseases. Analyst 136, 2069–2076 (2011)
- 141. Fang, X., Liu, Y., Kong, J., Jiang, X.: Loop-mediated isothermal amplification integrated on microfluidic chips for point-of-care quantitative detection of pathogens. Anal. Chem. 82, 3002–3006 (2010)

- 142. Won, J.Y., Min, J., Park, J.H.: Bacteria adsorption on hydrophilic surfaces for the sensitive detection of pathogenic bacteria using a single tube chamber system. Biosens. Bioelectron. 26, 1763–1767 (2010)
- 143. Lutz, S., Weber, P., Focke, M., Faltin, B., Hoffmann, J., Müller, C., Mark, D., Roth, G., Munday, P., Armes, N., Piepenburg, O., Zengerle, R., von Stetten, F.: Microfluidic lab-ona-foil for nucleic acid analysis based on isothermal recombinase polymerase amplification (RPA). Lab Chip 10, 887–893 (2010)
- 144. Mahalanabis, M., Do, J., ALMuayad, H., Zhang, J.Y., Klapperich, C.M.: An integrated disposable device for DNA extraction and helicase dependent amplification. Biomed. Microdevices 12, 353–359 (2010)
- 145. Liu, C., Mauk, M.G., Hart, R., Qiu, X., Bau, H.H.: A self-heating cartridge for molecular diagnostics. Lab Chip 11, 2686–2692 (2011)
- 146. Asiello, P.J., Baeumner, A.J.: Miniaturized isothermal nucleic acid amplification, a review. Lab Chip 11, 1420–1430 (2011)
- 147. Prompamorn, P., Sithigorngul, P., Rukpratanporn, S., Longyant, S., Sridulyakul, P., Chaivisuthangkura, P.: The development of loop-mediated isothermal amplification combined with lateral flow dipstick for detection of Vibrio parahaemolyticus. Lett. Appl. Microbiol. 52, 344–351 (2011)
- 148. Tomlinson, J.A., Dickinson, M.J., Boonham, N.: Rapid detection of Phytophthora ramorum and P. kernoviae by two-minute DNA extraction followed by isothermal amplification and amplicon detection by generic lateral flow device. Phytopathology **100**, 143–149 (2010)
- 149. Arunrut, N., Prombun, P., Saksmerprome, V., Flegel, T.W., Kiatpathomchai, W.: Rapid and sensitive detection of infectious hypodermal and hematopoietic necrosis virus by loopmediated isothermal amplification combined with a lateral flow dipstick. J. Virol. Methods 171, 21–25 (2011)
- 150. Nimitphak, T., Meemetta, W., Arunrut, N., Senapin, S., Kiatpathomchai, W.: Rapid and sensitive detection of Penaeus monodon nucleopolyhedrovirus (PemoNPV) by loop-mediated isothermal amplification combined with a lateral-flow dipstick. Mol. Cell. Probes 24, 1–5 (2010)
- Soliman, H., El-Matbouli, M.: Loop mediated isothermal amplification combined with nucleic acid lateral flow strip for diagnosis of cyprinid herpes virus-3. Mol. Cell. Probes 24, 38–43 (2010)
- 152. Puthawibool, T., Senapin, S., Kiatpathomchai, W., Flegel, T.W.: Detection of shrimp infectious myonecrosis virus by reverse transcription loop-mediated isothermal amplification combined with a lateral flow dipstick. J. Virol. Methods **156**, 27–31 (2009)
- 153. Jaroenram, W., Kiatpathomchai, W., Flegel, T.W.: Rapid and sensitive detection of white spot syndrome virus by loop-mediated isothermal amplification combined with a lateral flow dipstick. Mol. Cell. Probes **23**, 65–70 (2009)
- 154. Arunrut, N., Seetang-Nun, Y., Phromjai, J., Panphut, W., Kiatpathomchai, W.: Rapid and sensitive detection of Laem-Singh virus by reverse transcription loop-mediated isothermal amplification combined with a lateral flow dipstick. J. Virol. Methods **177**, 71–74 (2011)
- 155. Nimitphak, T., Kiatpathomchai, W., Flegel, T.W.: Shrimp hepatopancreatic parvovirus detection by combining loop-mediated isothermal amplification with a lateral flow dipstick. J. Virol. Methods 154, 56–60 (2008)
- 156. Gill, P., Ghaemi, A.: Nucleic acid isothermal amplification technologies: a review. Nucleosides Nucleotides Nucleic Acids 27, 224–243 (2008)
- Brehm-Stecher, B., Young, C., Jaykus, L.A., Tortorello, M.L.: Sample preparation: the forgotten beginning. J. Food Prot. 72, 1774–1789 (2009)
- Lazcka, O., Del Campo, F.J., Muñoz, F.X.: Pathogen detection: a perspective of traditional methods and biosensors. Biosens. Bioelectron. 22, 1205–1217 (2007)
- Velusamy, V., Arshak, K., Korostynska, O., Oliwa, K., Adley, C.: An overview of foodborne pathogen detection: in the perspective of biosensors. Biotechnol. Adv. 28, 232–254 (2010)
- 160. Nugen, S.R., Baeumner, A.J.: Trends and opportunities in food pathogen detection. Anal. Bioanal. Chem. 391, 451–454 (2008)

- 161. Coff, C., Korthals, M., Barling, D.: Chapter 1: Ethical traceability and informed food choice. In: Coff, C., Barling, D., Korthals, M., Nielsen, T. (eds.) Ethical Traceability and Communicating Food, vol. 15, pp. 1–18. Springer, Dordrecht (2008)
- 162. Aslan, Ö., Hamill, R.M., Sweeney, T., Reardon, W., Mullen, A.M.: Integrity of nuclear genomic deoxyribonucleic acid in cooked meat: implications for food traceability. J. Anim. Sci. 87, 57–61 (2009)
- 163. Van Asch, B., Silva Santos, L., Carneiro, J., Pereira, F., Amorim, A.: Identification of mtDNA lineages of Sus scrofa by multiplex single base extension for the authentication of processed food products. J. Agric. Food Chem. 59, 6920–6926 (2011)
- 164. Ali, M.E., Hashim, U., Mustafa, S., Che Man, Y.B., Yusop, M.H.M., Bari, M.F., Islam, Kh N., Hasan, M.F.: Nanoparticle sensor for label free detection of swine DNA in mixed biological samples. Nanotechnology 22, 195503 (2011)
- 165. Lago, F.C., Herrero, B., Vieites, J.M., Espineira, M.: Genetic identification of horse mackerel and related species in seafood products by means of forensically informative nucleotide sequencing methodology. J. Agric. Food Chem. 59, 2223–2228 (2011)
- 166. Jerôme, M., Martinsohn, J.T., Ortega, D., Carreau, P., Verrez-Bagnis, V., Mouchel, O.: Toward fish and seafood traceability: anchovy species determination in fish products by molecular markers and support through a public domain database. J. Agric. Food Chem. 56, 3460–3469 (2008)
- 167. Dooley, J.J., Sage, H.D., Clarke, M.A.L., Brown, H.M., Garrett, S.D.: Fish species identification using PCR-RFLP analysis and lab-on-a-chip capillary electrophoresis: application to detect white fish species in food products and an interlaboratory study. J. Agric. Food Chem. 53, 3348–3357 (2005)
- 168. Ponzoni, E., Mastromauro, F., Gianì, S., Breviario, D.: Traceability of plant diet contents in raw cow milk samples. Nutrients 1, 251–262 (2009)
- 169. Bonizzi, I., Buffoni, J.N., Feligini, M., Enne, G.: Investigating the relationship between raw milk bacterial composition, as described by intergenic transcribed spacer–PCR fingerprinting, and pasture altitude. J. Appl. Microbiol. **107**, 1319–1329 (2009)
- Bonizzi, I., Feligini, M., Aleandri, R., Enne, G.: Genetic traceability of the geographical origin of typical Italian water buffalo Mozzarella cheese: a preliminary approach. J. Appl. Microbiol. 102, 667–673 (2006)
- 171. Chen, X., Gao, C., Li, H., Huang, L., Sun, Q., Dong, Y., Tian, C., Gao, S., Dong, H., Guan, D., Hu, X., Zhao, S., Li, L., Zhu, L., Yan, Q., Zhang, J., Zen, K., Zhang, C.Y.: Identification and characterization of microRNAs in raw milk during different periods of lactation, commercial fluid, and powdered milk products. Cell Res. **20**, 1128–1137 (2010)
- 172. Spaniolas, S., May, S.T., Bennett, M.J., Tucker, G.A.: Authentication of coffee by means of PCR-RFLP analysis and lab-on-a-chip capillary electrophoresis. J. Agric. Food Chem. 54, 7466–7470 (2006)
- 173. Martellossi, C., Taylor, E.J., Lee, D., Graziosi, G., Donini, P.: DNA extraction and analysis from processed coffee beans. J. Agric. Food Chem. **53**, 8432–8436 (2005)
- 174. Bracci, T., Busconi, M., Fogher, C., Sebastiani, L.: Molecular studies in olive (Olea europaea L.): overview on DNA markers applications and recent advances in genome analysis. Plant Cell Rep. 30, 449–462 (2011)
- 175. Pafundo, S., Agrimonti, C., Marmiroli, N.: Traceability of plant contribution in olive oil by amplified fragment length polymorphisms. J. Agric. Food Chem. 53, 6995–7002 (2005)
- 176. Pafundo, S., Agrimonti, C., Maestri, E., Marmiroli, N.: Applicability of SCAR markers to food genomics: olive oil traceability. J. Agric. Food Chem. 55, 6052–6059 (2007)
- 177. Pasqualone, A., Montemurro, C., Summo, C., Sabetta, W., Caponio, F., Blanco, A.: Effectiveness of microsatellite DNA markers in checking the identity of protected designation of origin extra virgin olive oil. J. Agric. Food Chem. 55, 3857–3862 (2007)
- 178. Consolandi, C., Palmieri, L., Severgnini, M., Maestri, E., Marmiroli, N., Agrimonti, C., Baldoni, L., Donini, P., Bellis, G., Castiglioni, B.: A procedure for olive oil traceability and authenticity: DNA extraction, multiplex PCR and LDR-universal array analysis. Eur. Food Res. Technol. 227, 1429–1438 (2008)

- 179. Pafundo, S., Busconi, M., Agrimonti, C., Fogher, C., Marmiroli, M.: Storage-time effects on olive oil DNA assessed by amplified fragments length polymorphisms. Food Chem. **123**, 787–793 (2010)
- Doveri, S., O'Sullivan, D.M., Lee, D.: Non-concordance between genetic profiles of olive oil and fruit: a cautionary note to the use of DNA markers for provenance testing. J. Agric. Food Chem. 54, 9221–9226 (2006)
- Brennan, R.O.: Nutrigenetics: New Concepts for Relieving Hypoglycemia. Signet Books, New York (1977)
- Ordovas, J.M.: Genetic influences on blood lipids and cardiovascular disease risk: tools for primaryprevention. Am. J. Clin. Nutr. 89, S1509–S1517 (2009)
- 183. Lopez-Miranda, J., Ordovas, J.M., Mata, P., Lichtenstein, A.H., Clevidence, B., et al.: Effect of apolipopro tein E phenotype on diet-induced lowering of plasma low density lipoprotein cholesterol. J. Lipid Res. 35, 1965–1975 (1994)
- 184. Miettinen, T.A., Gylling, H., Vanhanen, H.: Serum cholesterol response to dietary cholesterol and apolipoprotein E phenotype. Lancet **2**, 1261 (1988)
- 185. Mahley, R.W., Weisgraber, K.H., Innerarity, T.L., Rall, S.C.: Genetic defects in lipoprotein metabolism: elevation of atherogenic lipoproteins caused by impaired catabolism. JAMA 265, 78–83 (1991)
- 186. Cobb, M.M., Teitlebaum, H., Risch, N., Jekel, J., Ostfeld, A.: Influence of dietary fat, apolipoprotein E phenotype, and sex on plasma lipoprotein levels. Circulation 86, 849–857 (1992)
- 187. Wang, J., John, E.M., Ingles, S.A.: 5-lipoxygenase and 5-lipoxygenase-activating protein gene polymorphisms, dietary linoleic acid, and risk for breast cancer. Cancer. Epidemiol. Biomarkers Prev. 17, 2748–2754 (2008)
- 188. Fradet, V., Cheng, I., Casey, G., Witte, J.S.: Dietary omega-3 fatty acids, cyclooxgenase-2 genetic variation, and aggressive prostate cancer risk. Clin. Cancer Res. 15, 2559–2566 (2009)
- Hedelin, M., Chang, E.T., Wiklund, F., Bellocco, R., Klint, A., et al.: Association of frequent consumption of fatty fish with prostate cancer risk is modified by COX-2 polymorphism. Int. J. Cancer 120, 398–405 (2007)
- 190. Simopoulus, A.P.: Nutrigenetics/nutrigenomics. Annu. Rev. Public Health 31, 53-68 (2010)
- Osborn, T.F., Goldstein, J.L., Brown, M.S.: 5-End of HMG-CoA reductase gene contains sequences responsible for cholesterol mediated inhibition of transcription. Cell 42, 203–212 (1985)
- Clarke, S.D., Jump, D.B.: Polyunsaturated fatty acid regulation of hepatic gene transcription. Lipids 31(Suppl.), 7–11 (1996)
- Clarke, S.D., Romsos, D.R., Leveille, G.A.: Differential effects of dietary methyl esters of long chain polyunsaturated fatty acids on rat liver and adipose tissue lipogenesis. J. Nutr. 107, 1170–1180 (1977)
- 194. Kallio, P., Kolehmainen, M., Laaksonen, D.E., Kekäläinen, J., Salopuro, T., Sivenius, K., Pulkkinen, L., Mykkänen, H.M., Niskanen, L., Uusitupa, M., Poutanen, K.S.: Dietary carbohydrate modification induces alterations in gene expression in abdominal subcutaneous adipose tissue in persons with the metabolic syndrome: the FUNGENUT study. Am. J. Clin. Nutr. 85, 1417–1427 (2007)
- 195. Masotti, A., Da Sacco, L., Bottazzo, G.F., Alisi, A.: Microarray technology: a promising tool in nutrigenomics. Crit. Rev. Food Sci. Nutr. **50**, 693–698 (2010)
- 196. D'Ambrosio, C., Gatta, L., Bonini, S.: The future of microarray technology: networking the genome search. Allergy 60, 1219–1226 (2005)

Part II Advanced Materials for Ultrasensitive Analysis

Chapter 3 Engineered Nanostructures for the Ultrasensitive DNA Detection

Dong-Kwon Lim, Amit Kumar, and Jwa-Min Nam

Abstract Coupled with nanotechnologies, a wide variety of DNA sensing methods have been developed to achieve ultrahigh sensitivity and selectivity without the aid of enzymatic amplification procedures or complicated assay procedures. Structurally engineered nanomaterials have several useful aspects including their unique optical properties depending on size, shape, composition and structural details and electrical properties, which have been translated into various signal transduction modes. However, the most important challenge in DNA detection assay to compete with or complement the polymerase chain reaction (PCR) is matching the sensitivity of PCR, which can detect 10–100 copies in whole sample via various non-enzymatic amplification strategies. Here, we introduce recent advances in engineered nanostructure-based DNA detection methods that show potential for PCR-like sensitivity and can address the existing issues of conventional DNA detection assays. The basic principles, advantages, and limitations of engineered nanostructure-amplified DNA detection methods will be introduced and discussed.

3.1 Introduction

Immediately after invention, the polymerase chain reaction (PCR) was readily recognized as a powerful tool for amplifying a trace amount of DNA of interest to detectable amount and this method completely changed the landscape of DNA research and detection. The versatile PCR methods are now routinely used in medical and biochemical research; particularly, in cloning for DNA sequencing, DNA-based phylogeny, functional analysis of genes, the identification of genetic fingerprints (used in forensic sciences and paternity testing) and diagnosis of

© Springer Science+Business Media Dordrecht 2012

D.-K. Lim • A. Kumar • J.-M. Nam (🖂)

Department of Chemistry, Seoul National University, Seoul 151-747, South Korea e-mail: limdk@mit.edu; aromamit@gmail.com; jmnam@snu.ac.kr

G. Spoto and R. Corradini (eds.), *Detection of Non-Amplified Genomic DNA*, Soft and Biological Matter, DOI 10.1007/978-94-007-1226-3_3,

hereditary and infectious diseases. However, PCR-based methods require numerous enzymatic target amplification steps and complex instrumentation, resulting in limited multiplexing capability and high susceptibility towards contaminations [1, 2]. Recently, many sophisticated metal nanostructures with strong electromagnetic response, have been used for enzyme-free optical DNA-biosensing up to single molecule detection capability, hence, presenting a better alternative of PCR [2]. This chapter provides an overview of the state-of-the-art DNA-sensing nanostructures, and these nanostructures are categorized as (1) nanoparticle probes for solution-state analysis, (2) nanostructures for DNA detection on solid surface, and (3) nanochannel/nanopores for DNA sequencing. In these methods, detection signals have been obtained and analyzed in various ways, including optical (e.g., fluorescence, Raman scattering, white light, surface plasmon resonance, vibrational spectroscopy, etc.), electrical, mechanical, and magnetic methods [2-4]. Here, we show how these engineered nanostructures, coupled with powerful signal amplification strategies, were designed, synthesized and used for highly sensitive DNA detection that can potentially replace or complement conventional DNA detection platforms such as PCR.

3.2 Nanoprobes for Solution-Based DNA Analysis

A variety of engineered nanoprobes have been developed and investigated for DNA analysis in solution to obtain amplified signals with various signal transduction modes (e.g., optical, electrical, mechanical, and magnetic signals) [3, 4]. Mainly, it is desirable to gain higher sensitivity, selectivity, multiplexing capability, and wide dynamic range using such inorganic/organic functional nanomaterials [1, 3, 4]. Currently, a wide range of nano-sized materials with different physicochemical properties have been used to achieve these purposes. The microscopic images of plasmonic nanosphere, nanorod, nanostar, bipyramids, core-shell, and nanowires composed of gold or silver, quantum dots (QDs), silicon nanowire, single-walled carbon nanotubes (SWCNTs), and graphene are shown in Fig. 3.1. Plasmonic nanoparticles not only show strong scattering signals when illuminated with white light, but also are sensitive to changes in size, shape and interparticle distance. Surface plasmon resonance (SPR) wavelength could be tuned from the visible range to the NIR range by simply changing their shape from spherical to rod, star, bipyramids or nano-shell (Fig. 3.1a-c) [5–7]. Due to the narrow emission spectrum, excellent photo-stability, and small size of quantum dots, it has been extensively studied as promising labeling materials for highly sensitive and stable probes that can replace conventional organic fluorophores. These nanostructuredependent plasmonic and fluorescent properties could be used as basic sensing principles and signal amplification strategies. Finally, field effect transistor devices enabled new functional materials such as silicon nanowire, carbon nanotube, and graphene to be used and extensively studied for DNA detection (Fig. 3.1d-e) [8, 9].

In this chapter we describe how these nanostructures can be combined in different architectures and methods for the ultra-sensitive detection of DNA.



Fig. 3.1 Microscopic images of gold nanoparticles carbon nanotubes and graphene sheets with different shapes. (**a**) gold nanorods, (**b**) gold nanostars, (**c**) gold bipyramids, (**d**) STEM images of the PEG_{1500N}-¹³ C-SWNT sample, (**e**) TEM image of a partially folded water-soluble graphene sheet [5–9] (Reprinted with permission from the American Chemical Society: (**a**) Ref. [5], Copyright (2008); (**b**) Ref. [6], Copyright (2008); (**c**) Ref. [7], Copyright (2011); (**d**) Ref. [8], Copyright (2005); (**e**) Ref. [9], Copyright (2008))

In 1996, Mirkin and coworkers reported interesting colorimetric response of DNA-functionalized gold nanoparticles upon hybridization with complementary oligonucleotide sequence. In this so called cross-linking method (Fig. 3.2a), there was a significant red shift in plasmonic band of gold nanoparticles [10-12], and the solution color changed from red-wine to blue, depending on the amount of target DNA (above \sim nM amount) under certain salt conditions (Fig. 3.2b, c) [11, 13, 14]. The aggregated gold nanoparticles were dehybridized upon increasing the temperature above the melting temperature (Tm) of DNA or under reduced salt concentrations [14]. Mismatch in oligonucleotide sequence triggered distinct change in a sharp melting point curve due to the cooperative binding of densely modified DNA on each particle, which enabled to detect single nucleotide polymorphism [13]. In addition, the simple color changes could be permanently recorded by spotting assay solution on the C_{18} -coated thin-layer chromatography plate, and the sensitivity was greatly increased from ~nM range to ~fM range when illuminating the spots with white-light [15, 16], coupled with silver enhancing method [17], further combined with Raman scattering [18] or electrical sensing method [19].

The surface based assay will be discussed in more details in the next section. Silver nanoparticles also have great potential to be used as a DNA probe. Due to



Fig. 3.2 Plasmonic color change in oligonucleotide-modified gold nanoparticle solution via target DNA detection. (a) Schematic description of a colorimetric assay with DNA-modified gold nanoparticles. (b) *Color* change with or without target DNA. (c) UV-Vis spectral change with or without target DNA [10] (Reprinted with permission from the American Chemical Society: Ref. [10], Copyright (2005))

the larger extinction coefficient of silver nanoparticles, a limit of detection (LOD) in the order of 40 pM has been reported, which is roughly 50-fold better than gold nanoparticles [20]. However, unstable surface chemistry of silver nanoparticles often easily induces nonspecific aggregation in solution state analysis [21]. To address this issue, DNA-modified silver-gold core-shell (Au@Ag) nanoparticles [22], DNA-embedded Au-Ag core-shell (Ag@Au) nanoparticles [23], and multithiol anchored oligonucleotide-modified silver nanoparticles [21] were synthesized and utilized in solution state DNA analyses. Other than conventional crosslinking methods [13, 14], assays based on electrostatic interactions with unmodified metal nanoparticles [24, 25] have been successfully applied for DNA detection in solution state, as shown in Fig. 3.3. One of the methods by Rothberg and Li [24] is based on different behaviors of single stranded DNA and double stranded DNA towards the stabilization of gold colloidal solution under high ionic strength. In other example [25], conjugated polyelectrolyte was used with unmodified gold nanoparticles to detect target DNA. This method is much simpler and sensitive (down to $\sim fM$) with wider application scope as compared to conventional crosslinking methods, however, their limited multiplexing capability is further needed to be improved.



Fig. 3.3 Colorimetric change of gold nanoparticles with target DNA (noncrosslinking). (a) Schematic description of noncrosslinking-based colorimetric assay. (b) *Color* change with or without target DNA. (c) UV-Vis spectral change with or without target DNA. (d) Change in absorbance while increasing the concentration of target DNA [25] (Reprinted with permission from the National Academy of Sciences of the United States of America: Ref. [25]. Copyright (2010))

Mirkin and coworkers introduced ultrasensitive detection methods for DNA and proteins based on bio-barcode DNA-functionalized gold nanoparticles and magnetic separation scheme [26–31]. A typical bio-barcode assay utilizes two different particles that are each designed to bind a target molecule. One of the probes is magnetic particles with target capture elements, and the other is gold nanoparticle probe with target capture elements and a number of barcode DNA strands. The functionalized gold nanoparticle (barcode probe) carried with it hundreds to thousands of barcode DNA molecules which served as surrogates for the target. After target binding, the released barcode oligonucleotides from gold nanoparticles were measured with various signal transduction methods including light scattering with *in-situ* silver enhancement (Fig. 3.4a–c) [26], PCR for additional amplification (30 aM) [27], on-chip [28], scanometric analysis [29], fluorescence intensity measurement (if fluorophore molecules are linked to barcode probes) [30] and colorimetric detection [31].



Fig. 3.4 Ultrasensitive DNA detection methods in solution. Bio-barcode assay with silver staining and UV-Vis spectrophotometer (\mathbf{a} - \mathbf{c}). (\mathbf{d}) Solution-based SERS detection using dye-coded silver nanoparticles [26, 32] (Reprinted with permission from the American Chemical Society: (\mathbf{a} - \mathbf{c}) Ref. [26]. Copyright (2007). Reprinted with permission from Nature Publishing groups: (\mathbf{d}) Ref. [32]. Copyright (2008))

Surface enhanced Raman scattering (SERS) methods also showed great potential for DNA analysis in solution which depends on size, shape, assembly, and interparticle distance of plasmonic nanoparticles [32–40]. Controlling hot-spots between noble metal nanoparticle junctions and placing reporter molecules in the hot-spots were major parameters for generating greatly enhanced Raman scattering in solution and solid surface. Raman tags are promising sensing labels in detecting multiple targets simultaneously because they offer differentiable fingerprint signals from various Raman-active molecules. However, generating target structures with a high accuracy and in a large scale and signal reproducibility have been critical issues in this research area [33–36]. Relatively higher SERS intensity was observed in solid state surface analysis comparing to solution state, owing to the reduced molecular movements, higher density of reporter molecules per unit illuminated area at a time on solid surface, and availability of ultrahigh sensitive detection scheme in surface analysis. However, solution-dispersed SERS particles have potential as cell and in vivo imaging probes as well as drug carriers, which are not possible with surfacebased SERS detection formats. Many research efforts have been made to increase sensitivity and demonstrate multiplexing capability in solution state analysis. In 2008, Graham and coworkers reported dye-coded silver nanoparticles modified with probe DNA to detect target DNA-specific SERS signals in solution (Fig. 3.4d) [32]. The target DNA acted as linker to assemble the Ag nanoparticles into an aggregate plasmonic nanostructure, exhibiting SERS with signal in a range of $\sim 2.5-7.5$ nM. In a different report, similar strategy was applied with different Raman reporters (e.g., malachite green), showing a LOD in the order of 25 nM [37]. Most recently, Johnson and coworkers reported $\sim pM$ level of sensitivity for oligonucleotide sequence derived from the West Nile Virus, using magnetic nanoprobes with Raman reporter molecule-adsorbed gold nanoparticles [38]. Quenching the SERS signal ("off") upon detecting the target DNA with a probe DNA hairpin is another SERS-based detection method [39]. Recent study showed that SERS-based DNA detection method can be integrated on a microfluidic system, which opens the way to future "lab on a chip" device system [40].

Apart from utilizing surface plasmon resonance (SPR) properties of noble metal nanoparticles, super-paramagnetic nanoparticles (e.g., iron oxide) have been investigated and demonstrated as sensitive probes for solution state DNA analysis [41]. The increase of hydrodynamic radius could greatly affect the magnetic susceptibility of nanoparticles in solution. However, relatively lower sensitivity and expensive instrumentation are the major limitations of this method. In an interesting solution-based method, magnetic nanoparticles acted as magnetic relaxation switch (MRS) to detect DNA-DNA interactions with high sensitivity and selectivity [42]. The oligonucleotide-functionalized iron oxide nanoparticles were aggregated in the presence of target DNA and the T_2 relaxation time of the surrounding water molecules decreased. The observed change in T_2 relaxation time was proportional to the concentration of target strands, and the detection limit was in a sub-fM range. Since the magnetic changes are also detectable in turbid media and in whole cell lysates, the MRS could be useful for *ex-vivo* DNA analysis.

The distance-dependent fluorescence quenching effect of dye molecules presents a potential basis for sensitive DNA detection using nanoparticles. The methods involving fluorescence resonance energy transfer (FRET) or metal enhanced fluorescence (MEF) have been reported to show high sensitivity and dynamic detection ranges (from ~nM to sub pM). When CdSe/ZnS core/shell nanocrystal-DNA conjugates were used, limited dynamic range (10–50 nM) and detection sensitivity (2 nM) were observed [43]. Gold nanoparticle-DNA and carboxytetramethylrhodamine-DNA system [44] and gold nanoparticle and DNA-fluorescein system [45] also showed similar dynamic ranges and sensitivities. More complex but sophisticated nanostructures involving Fe/Au core/shell nanoparticle-DNA and methylene blue in solution, enabled to further improve the sensitivity (~pM), selectivity, and dynamic range [46]. Here, the decreased amount of intercalated methylene blue by the formation of double-stranded DNA on the Fe/Au shell after target detection was responsible for the decline in fluorescence intensity. MEF-based DNA detection was demonstrated by the Lakowicz group. They reported a 12-fold fluorescence signal enhancement when DNA-conjugated fluoresceins were attached to Ag nanoparticles rather than being dispersed freely in a solution without Ag nanoparticles [47]. Gunnarsson and coworkers reported a single-molecule readout scheme using total internal reflection fluorescence microscopy (TIRFM) technique and demonstrated 10 fM sensitivity for oligonucleotides [48]. In this strategy, they used rhodaminelabeled lipid vesicles (these 100-nm vesicles were also modified with cholesterol-DNA-duplex structures). As lipid vesicles were immobilized to silicon dioxide surface via target DNA hybridization, the number of surface-bound vesicles was counted as a function of time using the TIRFM.

Solution state analysis is well suited for electrochemical signal transduction mode. Therefore, various signal amplification methods have been developed utilizing distinct electrochemical characters (e.g., electrocatalytic property) of nanomaterials and stripping voltammetry in solution state. Large surface area of nanomaterials and well-established bioconjugation methods enable the use of nanomaterials as efficient carriers of nucleotides, antibodies and smaller nanoparticles and as signal amplification substrates for electroanalysis. Stripping voltammetry is an electrical method for the quantitative determination of specific metallic ion species and other electrochemical analytes. Due to the preconcentration step of the desired analytes on the surface of the electrode by electrodeposition or adsorption, more than 2 orders of magnitude amplified signals can be detected in a stripping voltammetry. Another sensitive electrochemical assay utilized electrocatalytic reduction of hydrogen peroxide by avidin-hydrazine [49]. This assay employed a conducting polymer-linked dendrimer and hydrazine. Gold nanoparticle was used as a substrate that carries numerous polymer-linked dendrimers, and this high loading facilitates the signal amplification. After the human IgG and target DNA were captured by antibodies and DNA probes, respectively, analytes were detected by adding hydrogen peroxide to the electrolytic solution. The detection limits for human IgG and DNA were 1 fM and 50 aM, respectively.

In summary, the engineered nanostructured materials have been utilized as useful and sensitive DNA sensing platforms with several types of signal transduction modes such as naked eye, fluorescence, white-light, SERS, and electrochemical signal detections. A variety of principles have been applied to each sensing method such as inter-particle distance dependent plasmonic shift, fluorescence quenching or enhancement, light-scattering properties, electromagnetic enhancement in SERS, catalytic properties for electrochemical sensing, etc. However, some efforts are still needed in order to further enhance the sensitivity, high throughput capability, and multiplexing capability, and signal reproducibility in real samples without the need of expensive and sophisticated instruments. Therefore, new classes of nanomaterials that can address these issues are always in need in this research field for practical applications.

In Chap. 6, other types of nanostructures (e.g., carbon nanotubes and graphene) and their use in optical detection of DNA are described.

3.3 Engineered Nanostructures for DNA Analysis on Solid Surfaces

Conventionally, DNA analysis on solid surface is performed with glass surface arrayed with DNA, which is allowed to hybridize with complementary sequence, and then imaged with fluorescent molecules used as labeling materials [43]. Bio-functionalized inorganic nanomaterials also have shown great potential as new labeling materials for *in-vitro*, *ex-vivo*, *and in-vivo* detection technologies. With the



Fig. 3.5 Various DNA analytical strategies with nanostructures on surface. (a) Silver-enhanced scanometric, (b) SERS, and (c) electrical signal transduction mode, (d) particle-on-wire and (e) nanodisk-on-wire (Reprinted with permission from the American Chemical Society, (**a**–c) Ref. [10]. Copyright (2005), (d) Ref. [34]. Copyright (2010), (e) Ref. [54]. Copyright (2007))

development of theoretical studies and surface patterning technologies, engineered nanostructures fabricated on 1, 2-dimensional solid surfaces generated sensitive signal changes such as localized surface plasmon resonance, SERS, white-light scattering, electronic signal transduction, quenched or enhanced fluorescence, etc., after the target DNA binding event in heterogeneous state. In spite of non-specific binding and complex washing procedures which is essential in solid surface based analysis, the main advantages of DNA analysis on solid surfaces are high throughput, high sensitivity, stability, feasibility of mass production, and cost-effectiveness.

Biotinylated DNA and streptavidin-modified gold nanoparticles have been widely investigated to identify hybridized target DNA on DNA microarray surface. Gold nanoparticles could be detected by scanning with a flat-bed scanner, and the scattering signals could further be amplified by silver enhancing method [17] (Fig. 3.5a).

Also, gold nanoparticles could be illuminated with white-light scattering [15], which is now a commercialized platform [50]. In 2002, Mirkin and coworkers reported a sensitive (20 fM), multiplexed SERS-based nucleic acid detection method, combining silver enhancing method with Raman spectroscopy [18] (Fig. 3.5b). In this assay, multiplexing was achieved using Raman dye-encoded particles with different probe sequences corresponding to different target DNAs. In spite of a limited multiplexing capability, electrical signals have great potential to detect biological molecules including DNA in a rapid, portable, operationally simple, and cost-effective manner. In an array-based nanoparticle platform, target complementary capture DNAs were immobilized between electronic gap and subsequently hybridized with DNA-AuNPs via target DNA recognition. The immobilized AuNPs were then treated with silver staining, giving a sensitive (~fM) and selective

electrical "on" signal [19] (Fig. 3.5c). Fan and coworkers reported a highly sensitive electrochemical method for the detection of DNA, based on a sandwich detection strategy involving capture probe DNA immobilized on gold electrodes and reporter probe DNA labeled with gold nanoparticles [51]. Another method is involved a real-time DNA detection and utilized single-stranded DNA-modified nanoparticles and micropatterned chemoresponsive diffraction gratings, interrogated simultaneously at multiple laser wavelengths [52]. This chemoresponsive diffraction grating signal transduction scheme permits experimentally simple, real-time observation of DNA hybridization on surfaces with \sim 1 pM sensitivity. Fluorophore-dye doped silica nanoparticles also have been explored as alternative labeling material, generating increased fluorescence signals during DNA detection with high sensitivity [53].

After early stages of conceptual discovery of DNA analysis on surface as shown in Fig. 3.5a–c, more sophisticated nano-patterned structures have been explored to obtain improved sensitivity and reproducibility using optical (Raman, fluorescence, etc.) or electrical transduction modes. In one study, a sandwich type of nanoparticle-linked nanowire complexes were assembled by target DNA hybridization with Raman reporter molecules placed in the junction [34] (Fig. 3.5d). These multiplexable assemblies exhibited greatly amplified Raman signals. Mirkin and coworkers used dispersible arrays of nanodisks, prepared by on-wire lithography and functionalized with Raman-active chromophores, for DNA detection in a multiplexed format with high sensitivity (~100 fM), as shown in Fig. 3.5e [54].

Recently, Nam and coworkers reported SERS-active gold nanodumbbells linked with DNA, synthesizable in a high yield, and these nanogaps structures exhibited single-molecule detection capability repeatedly [55] (Fig. 3.6). In this method, they were able to control the nanogap between gold nanoparticles by varying the thickness of silver nanoshell in order to finely tune the SERS signal. They found that controlling inter-particle nanogap is extremely important for highly sensitive and reproducible single-molecule SERS detection. More recently, they also synthesized nanobridged nanogap Au particles with 1-nm interior gap. With the aid of DNA on Au particle surface and specific reaction conditions, highly uniform 1-nm gap was generated inside a particle, and this interior nanogap was responsible for strong, uniform and quantitative SERS signals [56]. When these NNPs were analyzed on the surface, ~ 90 % of analyzed particles displayed detectable Raman signals. Another simple method to fabricate dimeric Ag nanostructures in solution state is shown in Fig. 3.7a-d [35]. In this method, by controlling the concentration of electrolytes in Ag colloidal solution, the dimeric Ag nanoparticles were dominantly produced and the structures were fixed with additional silica coating (Fig. 3.7b).

Two different surface-based nanogap fabrication methods have been shown in Figs. 3.8 and 3.9, respectively. Using sequential angle evaporation technique, <5 nm gap between silver nanospheres on the surface were uniformly generated (Fig. 3.8) [36]. Vertically oriented, <10-nm gap was also achieved *via* anisotropic etching procedure on the surface (Fig. 3.9) [33].

In the above-mentioned methods, relatively uniform nanogaps in the large area were possible to generate, but the next imminent challenges remain to locate the



Fig. 3.6 Single DNA detection with gold-silver core-shell nanodumbbells. (**a**) Synthetic scheme with two different oligonucleotide-modified gold nanoparticles and additional silver shell formation for inter-particle gap engineering for single-molecule SERS. (**b**) TEM images of synthesized gold nanoparticle dimers (*upper*) and silver shell-coated nanoparticle dimers (*lower*) (Reprinted with permission from Nature Publishing Group, Ref. [55]. Copyright (2009))

target molecule of interest in the fabricated hot-spot regions and to synthesize targeted nanostructures with high structural reproducibility in a large scale.

The conductivity-based electrical DNA sensors, which were based on silver enhancing procedure, were further improved by the use of DNA-linked–CNT nanowire motifs and enzymatic metallization-based amplification [57]. In this method, a DNA detection limit of 10 fM was achieved with the ability to discriminate single, double, and triple base pair mismatches (Fig. 3.10a). Gao and coworkers reported nanogap sensor arrays for highly sensitive electrical detection and quantification of DNA [58]. The sensing process is based on bridging the nanogap upon hybridization of the two termini of a target DNA with two different surface bound capture probes, followed by a metallization step (Fig. 3.10b). Using this method, a minimum of 1.0 fM DNA was detected via two-order-of-magnitude enhancement in conductance. Above-mentioned methods validated the importance of engineered nanostructures for generating amplified optical and electrical signals, applicable for biomolecule detection with high sensitivity.

Xiao and coworkers demonstrated that engineering of biofunctionality can address many limitations of electrochemical biosensors [59]. Their electrochemical SNP (E-SNP) sensor is based on a single, redox-labeled DNA strand that incorporates three stems, which is different from redox-labeled single-stem molecular beacon [60] and double-stem pseudoknot [61] electrochemical sensors (Fig. 3.11a).



Fig. 3.7 Salt-induced silver nanoparticle grouping into dimers and fixation using silica shell formation. TEM images of synthesized silver dimers (**a**) and silica-coated dimers (**b**). (**c**) HR-TEM image of Ag dimer and its lattice plane (*inset*). (**d**) Magnified image of Ag dimer junction (Reprinted with permission from the American Chemical Society, Ref. [35]. Copyright (2008))

When complementary target oligonucleotides (17 base pairs) were present, 3'modified MB can be placed in a close contact with Au surface, eventually, increasing the faradic current ($335 \pm 7 \%$). The signal changes were large enough to detect a single base mismatch [59].

Semiconductor silicon nanowire-based field effect transistor (SiNW-FET) has been used as a biosensing platform for the real time detection of nucleic acids [62]. Based on the change in conductance of DNA-modified SiNWs, specific target DNA sequence was detected with high sensitivity (\sim 1 fM). Using this method, simultaneous and selective detection of two pathogenic strain virus DNA sequences (H1N1 and H5N1) of avian influenza was achieved (Fig. 3.11b).

Based on the similar concept, GaN nanowire(GaNNW)-based extended-gate field effect-transistor (EGFET) biosensor was developed and this platform dramatically improved sensitivity (\sim aM range) and dynamic range (10^{-19} – 10^{-6} M) for short oligonucleotides (15 base pairs) under label-free in situ conditions [63]. The wide bandgap semiconducting nature of GaN, surface-sensitivity of



Fig. 3.8 Fabrications of nanoparticle dimer array on surface using angle evaporation technique. (A) Scheme for angle evaporation technique. (B) HRTEM image (a) and simulated polarization dependence in perpendicular (b) and in parallel (c). (C) low resolution SEM image (a), TEM image (b) and HRTEM image (c) of arrayed surface (Reprinted with permission from the American Chemical Society, Ref. [36]. Copyright (2010))

the NW structures, and high transducing performance of the EGFET-design were responsible for improved sensitivity and specificity in this method, as demonstrated by the detection of single base-pair mutation in the human p53 gene. As evident from above discussion, a wide variety of engineered nanostructures have been fabricated with different materials, often displaying higher sensitivity and specificity for target DNA than solution-based methods. Although highly limited for *invivo* applications, surface-based nanostructure platforms offer the high throughput potential and relatively simple transduction modes such as electrical transduction mode and it will not be long to see commercialized version of DNA sensors for the widespread use for *in-vitro* clinical DNA analyses.



Fig. 3.9 Fabrication schemes for vertically oriented plasmonic nanogap using anisotrophic etching technique. (a) Calculated electromagnetic distribution profile, (b) Schematic description of fabrication procedures, (c) SEM images of patterned surface, (d, e) Magnified images (Reprinted with permission from the American Chemical Society, Ref. [33]. Copyright (2010))

3.4 Engineered Nanopores and Nanochannels for DNA Analysis

Nanopore or nanochannel is a nanometer-sized path through a thin layer, which selectively allows specific biomolecules (e.g., DNA) to pass, based on their size, shape and chemical nature, typically in a 'one-by-one' mode. During the process, individual molecules are detected and analyzed via optical or electrical signals. These engineered three-dimensional nanostructures along with microfluidics and nanowells have been extensively investigated as promising candidates for real-time, efficient genome-wide next generation DNA sequencing methods [64–66]. In this section, we will discuss the DNA sensing technologies developed using



Fig. 3.10 Electric nanogap device. (a) DNA-CNT nanowire network for DNA detection, (b) nanogap sensor arrays for ultrasensitive DNA analysis (Reprinted with permission from the American Chemical Society, (a) Ref. [57]. Copyright (2011), (b) Ref. [58]. Copyright (2009))



Fig. 3.11 Triple-stem DNA probe based electrochemical sensor for single nucleotide polymorphism detection (a) and silicon-nanowire-based CMOS compatible FET nanosensors for the ultrasensitive electrical detection of nucleic acids (b) (Reprinted with permission from the American Chemical Society, (a) Ref. [59]. Copyright YEAR (2009), (b) Ref. [62]. Copyright YEAR (2011))

microfluidic channels and nanochannels. Nie and coworkers reported the use of color-coded nanoparticles and dual-color fluorescence coincidence for real-time detection of native biomolecules and viruses in a microfluidic channel [67]. They



Fig. 3.12 Microfluidics and nanopore-based DNA detection and sequencing technologies. (A) Microchannels for small fragment DNA [68]. (B) Schematic of the nanopore/nanogap device integrated into an electrochemical cell. The tunneling junction is located at the exit of the nanopore. DNA is inserted in the *bottom* reservoir and electrophoretically driven through the nanopore and the tunneling junction [73]. (C) Optical trap for effective DNA stretching [75]. (D) Multiple nanopores for optical detection of two *color*-coded oligonucleotides [76] (Reprinted with permission from the American Chemical Society, (A) Ref. [68]. Copyright YEAR (2010), (B) Ref. [73]. Copyright YEAR (2010), (C) Ref. [75]. Copyright YEAR (2011), (D) Ref. [76]. Copyright YEAR (2010))

demonstrated that individual molecules of interest can be detected and identified in complex mixtures without target amplification or probe/target separation at the target concentration of as low as 20–30 fM. Recently, Wang and coworkers reported single molecule analysis of circulating fluorescently-labeled DNA (CNA) in human serum using microfluidic cylindrical illumination confocal spectroscopy and fluorescence burst size analysis as shown in Fig. 3.12A [68].

For the development and widespread use of new DNA sequencing technologies, critical issues such as cost, reliability, and high-throughput capability should be addressed. Fast spectroscopic response, easy chemistry and multiplexing capability are other important desired features for the next-generation DNA sequencing technologies. Development of software and bioinformatics tools for data analysis and sophisticated instruments are also needed to obtain reliable and accurate DNA sequencing results. However, the most important requirement is the development of engineered nanostructures such as nanowells [69], nanochannels [70–72], and nanopores [73] where DNA polymerization or efficient stretching can occur often in a single-molecule level. At the same time, engineered nanostructures play a critical role in reducing background signal, thereby, making possible to detect fluorescence signal per single polymerization reaction. An example is provided by Zero-mode waveguides (Pacific Biosciences, Menlo Park, CA, USA), with which illumination can be restricted to a zeptoliter-scale volume with a surface-tethered polymerase

such that incorporation of single nucleotide (with a fluorescent label on phosphate group) can be monitored with very low background signals [69].

Single, sub-20-nm wide and centimeter-long nanofluidic channel, fabricated by a nanoimprint mold method, can successfully stretch small DNA fragment by liquid flow in the nanochannel [74]. Kim and coworkers reported that polystyrene-*block*-poly(methyl methacrylate) copolymer having dicarboxylic acid end group(PS-*b*-PMMA-diCOOH)-coated nanochannels having a pore size of ~15 nm and depth of ~80 nm can successfully differentiate single-nucleotide polymorphisms (SNPs) in 18 base-pair oligonucleotides [72]. Due to the ultrahigh density (~6 × 10¹¹/in²) and short nanochannel length, very high resolution was achieved therein.

Recently, Edel and coworkers fabricated a device that combines a nanopore with a tunneling junction [73]. In such a configuration, the nanopore was used to "unfold" the DNA strand to a linear configuration, while a tunneling junction, precisely aligned at the pore opening, detects sequence-specific changes in the tunneling current, as shown in Fig. 3.12Ba. The experimental platform that was composed of a 5 mm \times 5 mm silicon chip (Fig. 3.12Bb) with Si₃N₄ and Au layers microelectrodes with a 2 μ m gap (50 nm thick gold with 10 nm chromium adhesion layer) was fabricated by a conventional optical lithographic technique on the front face of the wafer. An additional 300-nm thick layer of Si₃N₄ was deposited by plasma enhanced chemical vapor deposition (PECVD). Windows (5 μ m \times 5 μ m) centered at the microelectrode tips were opened in the PECVD Si_3N_4 layer (Fig. 3.12 Bc, d). The additional Si₃N₄ layer reduces the membrane capacitance and faradic currents by minimizing the electroactive area. Furthermore, the additional Si_3N_4 layer helps improve the mechanical strength of the membrane. A window in the LPCVD nitride on the back face of the wafer was opened by reactive ion etching (RIE), followed by a wet etch of the silicon in KOH solution, resulting in a 70 nm thick free-standing Si_3N_4 membrane (40 μ m × 40 μ m) aligned to the microelectrodes (Fig. 3.12Bc, d). The solid state Si₃N₄-membrane (20 nm thickness) was milled with He ions, rendering nanopore openings with typically 20-60 nm in diameter. The membrane was integrated into the fluidic cell so that the two sides of the membrane were only electrically contacted via the nanopore. Translocation dynamics of a single protein molecule attached to double stranded DNA has been studied using the nanopore force spectroscopy (Fig. 3.12C) [75]. In a different approach, Meller and coworkers reported the use of multiple nanopores and optical sensing method (Fig. 3.12D) [76]. Here, they used two color readout platform, which was coded for four DNA base, for example, 1,1 for A, 0,1 for T, 1,0 for G, 0,0 for C, and then specific hybridization to the 0 and 1 sequences was followed. The converted DNA and hybridized molecular beacons were electrophoretically threaded through a solid-state pore.

3.5 Conclusion

Improving sensitivity is the foremost important issue in bioassays without deteriorating other factors such as assay cost, time, complexity, dynamic range and reproducibility. It would be beneficial to develop a nucleic acid detection assay with PCR-like sensitivity and without the need for enzyme and temperature controlbased target amplification. This way, problems of using enzymes such as extra steps and complex instrumentation to control the conditions could be avoided. The development of new functional nanomaterials for amplification methodologies could improve or replace ELISA method for protein targets and PCR for nucleic acid targets. As discussed above, new 1,2,3-dimensional nanostructures have been extensively investigated for this purpose owing to their important physiochemical properties, and some of them are already commercialized and utilized in many laboratories for research purpose and as clinical diagnostic tools. However, the composition, size, shape and other chemical and structural features such as nanogap of nanostructures still required to be finely controlled for more accurate and sensitive DNA analysis with multiplexing capability. Recently, nanopore- or nanochannelbased DNA sequencing methods have great potential in this field, but these strategies face many challenges including selective recognition of individual nucleotide and multiplexing capability (parallel recognition) along with assay cost and time.

References

- Shim, S.-Y., Lim, D.-K., Nam, J.-M.: Ultrasensitive optical biodiagnostic methods using metallic nanoparticles. Nanomedicine 3(2), 215–232 (2008)
- 2. Peng, H.-I., Miller, B.L.: Recent advancements in optical DNA biosensors: exploiting the plasmonic effects of metal nanoparticles. Analyst **136**(3), 436–447 (2011)
- Jeon, J., Lim, D.-K., Nam, J.-M.: Functional nanomaterial-based amplified bio-detection strategies. J. Mater. Chem. 19(15), 2107–2117 (2009)
- Wittenberg, N.J., Haynes, C.L.: Using nanoparticles to push the limits of detection. Wiley Interdiscip. Rev. Nanomed. Nanobiotechnol. 1(2), 237–254 (2009)
- 5. Selhuber-Unkel, C., et al.: Quantitative optical trapping of single gold nanorods. Nano Lett. **8**(9), 2998–3003 (2008)
- Khoury, C.G., Vo-Dinh, T.: Gold nanostars for surface-enhanced Raman scattering: synthesis, characterization and optimization. J. Phys. Chem. C 112(48), 18849–18859 (2008)
- Personick, M.L., et al.: Synthesis and isolation of {110}-faceted gold bipyramids and rhombic dodecahedra. J. Am. Chem. Soc. 133(16), 6170–6173 (2011)
- Kitaygorodskiy, A., et al.: Nmr detection of single-walled carbon nanotubes in solution. J. Am. Chem. Soc. 127(20), 7517–7520 (2005)
- 9. Si, Y., Samulski, E.T.: Synthesis of water soluble graphene. Nano Lett. 8(6), 1679–1682 (2008)
- Rosi, N.L., Mirkin, C.A.: Nanostructures in biodiagnostics. Chem. Rev. 105(4), 1547–1562 (2005)
- Mirkin, C.A., et al.: A DNA-based method for rationally assembling nanoparticles into macroscopic materials. Nature 382(6592), 607–609 (1996)
- Elghanian, R., et al.: Selective colorimetric detection of polynucleotides based on the distancedependent optical properties of gold nanoparticles. Science 277(5329), 1078–1081 (1997)
- Storhoff, J.J., et al.: What controls the optical properties of DNA-linked gold nanoparticle assemblies? J. Am. Chem. Soc. 122(19), 4640–4650 (2000)
- Storhoff, J.J., et al.: One-pot colorimetric differentiation of polynucleotides with single base imperfections using gold nanoparticle probes. J. Am. Chem. Soc. 120(9), 1959–1964 (1998)
- Storhoff, J.J., et al.: Homogeneous detection of unamplified genomic DNA sequences based on colorimetric scatter of gold nanoparticle probes. Nat. Biotechnol. 22(7), 883–887 (2004)

- Ling, J., et al.: Light-scattering signals from nanoparticles in biochemical assay, pharmaceutical analysis and biological imaging. TrAC Trends Anal. Chem. 28(4), 447–453 (2009)
- Taton, T.A., Mirkin, C.A., Letsinger, R.L.: Scanometric DNA array detection with nanoparticle probes. Science 289(5485), 1757–1760 (2000)
- Cao, Y.C., Jin, R., Mirkin, C.A.: Nanoparticles with Raman spectroscopic fingerprints for DNA and RNA detection. Science 297(5586), 1536–1540 (2002)
- Park, S.-J., Taton, T.A., Mirkin, C.A.: Array-based electrical detection of DNA with nanoparticle probes. Science 295(5559), 1503–1506 (2002)
- Thompson, D.G., et al.: Ultrasensitive DNA detection using oligonucleotide silver nanoparticle conjugates. Anal. Chem. 80(8), 2805–2810 (2008)
- Lee, J.-S., et al.: Silver nanoparticle oligonucleotide conjugates based on DNA with triple cyclic disulfide moieties. Nano Lett. 7(7), 2112–2115 (2007)
- Cao, Y., Jin, R., Mirkin, C.A.: DNA-modified core-shell Ag/Au nanoparticles. J. Am. Chem. Soc. 123(32), 7961–7962 (2001)
- Lim, D.-K., Kim, I.-J., Nam, J.-M.: DNA-embedded Au/Ag core-shell nanoparticles. Chem. Commun. 42, 5312–5314 (2008)
- 24. Li, H., Rothberg, L.: Colorimetric detection of DNA sequences based on electrostatic interactions with unmodified gold nanoparticles. Proc. Natl. Acad. Sci. USA 101(39), 14036– 14039 (2004)
- Xia, F., et al.: Colorimetric detection of DNA, small molecules, proteins, and ions using unmodified gold nanoparticles and conjugated polyelectrolytes. Proc. Natl. Acad. Sci. USA 107(24), 10837–10841 (2010)
- 26. Xu, X., et al.: Homogeneous detection of nucleic acids based upon the light scattering properties of silver-coated nanoparticle probes. Anal. Chem. **79**(17), 6650–6654 (2007)
- Nam, J.-M., Thaxton, C.S., Mirkin, C.A.: Nanoparticle-based bio-bar codes for the ultrasensitive detection of proteins. Science 301(5641), 1884–1886 (2003)
- Goluch, E.D., et al.: A bio-barcode assay for on-chip attomolar-sensitivity protein detection. Lab Chip 6(10), 1293–1299 (2006)
- Stoeva, S.I., et al.: Multiplexed DNA detection with biobarcoded nanoparticle probes. Angew. Chem. Int. Ed. 45(20), 3303–3306 (2006)
- Zhang, D., Carr, D.J., Alocilja, E.C.: Fluorescent bio-barcode DNA assay for the detection of Salmonella enterica serovar Enteritidis. Biosens. Bioelectron. 24(5), 1377–1381 (2009)
- Nam, J.-M., Jang, K.-J., Groves, J.T.: Detection of proteins using a colorimetric bio-barcode assay. Nat. Protoc. 2(6), 1438–1444 (2007)
- Graham, D., et al.: Control of enhanced Raman scattering using a DNA-based assembly process of dye-coded nanoparticles. Nat. Nanotechnol. 3(9), 548–551 (2008)
- 33. Im, H., et al.: Vertically oriented sub-10-nm plasmonic nanogap arrays. Nano Lett. 10(6), 2231–2236 (2010)
- Kang, T., et al.: Patterned multiplex pathogen DNA detection by Au particle-on-wire SERS sensor. Nano Lett. 10(4), 1189–1193 (2010)
- 35. Li, W., et al.: Dimers of silver nanospheres: facile synthesis and their use as hot spots for surface-enhanced Raman scattering. Nano Lett. 9(1), 485–490 (2008)
- Theiss, J., et al.: Plasmonic nanoparticle arrays with nanometer separation for highperformance SERS substrates. Nano Lett. 10(8), 2749–2754 (2010)
- Qian, X., Zhou, X., Nie, S.: Surface-enhanced Raman nanoparticle beacons based on bioconjugated gold nanocrystals and long range plasmonic coupling. J. Am. Chem. Soc. 130(45), 14934–14935 (2008)
- Johnson, D.S., et al.: Genome-wide mapping of in vivo protein-DNA interactions. Science 316, 1497–1502 (2007)
- Wabuyele, M.B., Vo-Dinh, T.: Detection of human immunodeficiency virus type 1 DNA sequence using plasmonics nanoprobes. Anal. Chem. 77(23), 7810–7815 (2005)
- Huh, Y.S., et al.: Surface-enhanced Raman scattering based ligase detection reaction. J. Am. Chem. Soc. 131(6), 2208–2213 (2009)

- Xi, D., et al.: The detection of HBV DNA with gold-coated iron oxide nanoparticle gene probes. J. Nanopart. Res. 10(3), 393–400 (2008)
- Perez, J.M., et al.: Magnetic relaxation switches capable of sensing molecular interactions. Nat. Biotechnol. 20(8), 816–820 (2002)
- Gerion, D., et al.: Room-temperature single-nucleotide polymorphism and multiallele DNA detection using fluorescent nanocrystals and microarrays. Anal. Chem. **75**(18), 4766–4772 (2003)
- 44. Wu, Z.-S., et al.: Optical detection of DNA hybridization based on fluorescence quenching of tagged oligonucleotide probes by gold nanoparticles. Anal. Biochem. **353**(1), 22–29 (2006)
- 45. Wang, H., et al.: Combination of DNA ligase reaction and gold nanoparticle-quenched fluorescent oligonucleotides: a simple and efficient approach for fluorescent assaying of singlenucleotide polymorphisms. Anal. Chem. 82(18), 7684–7690 (2010)
- 46. Son, A., et al.: Rapid and quantitative DNA analysis of genetic mutations for polycystic kidney disease (PKD) using magnetic/luminescent nanoparticles. Anal. Bioanal. Chem. **390**(7), 1829– 1835 (2008)
- Malicka, J., Gryczynski, I., Lakowicz, J.R.: DNA hybridization assays using metal-enhanced fluorescence. Biochem. Biophys. Res. Commun. 306(1), 213–218 (2003)
- Gunnarsson, A., et al.: Single-molecule detection and mismatch discrimination of unlabeled DNA targets. Nano Lett. 8(1), 183–188 (2007)
- Shiddiky, M.J.A., Rahman, M.A., Shim, Y.-B.: Hydrazine-catalyzed ultrasensitive detection of DNA and proteins. Anal. Chem. 79(17), 6886–6890 (2007)
- 50. Verigene System, Nanosphere, Inc., Northbrook, IL, USA. www.nanosphere.us
- 51. Zhang, J., et al.: Sequence-specific detection of femtomolar DNA via a chronocoulometric DNA sensor (CDS): effects of nanoparticle-mediated amplification and nanoscale control of DNA assembly at electrodes. J. Am. Chem. Soc. 128(26), 8575–8580 (2006)
- 52. Bailey, R.C., et al.: Real-time multicolor DNA detection with chemoresponsive diffraction gratings and nanoparticle probes. J. Am. Chem. Soc. **125**(44), 13541–13547 (2003)
- Zhao, X., Tapec-Dytioco, R., Tan, W.: Ultrasensitive DNA detection using highly fluorescent bioconjugated nanoparticles. J. Am. Chem. Soc. 125(38), 11474–11475 (2003)
- 54. Qin, L., et al.: Nanodisk codes. Nano Lett. 7(12), 3849-3853 (2007)
- Lim, D.-K., et al.: Nanogap-engineerable Raman-active nanodumbbells for single-molecule detection. Nat. Mater. 9(1), 60–67 (2010)
- 56. Lim, D.-K., et al.: Highly uniform and reproducible surface-enhanced Raman scattering from DNA-tailorable nanoparticles with 1-nm interior gap. Nat. Nanotechnol. 6(7), 452–460 (2011)
- 57. Weizmann, Y., Chenoweth, D.M., Swager, T.M.: DNA-CNT nanowire networks for DNA detection. J. Am. Chem. Soc. **133**(10), 3238-3241 (2011)
- Roy, S., et al.: Mass-produced nanogap sensor arrays for ultrasensitive detection of DNA. J. Am. Chem. Soc. 131(34), 12211–12217 (2009)
- Xiao, Y., et al.: An electrochemical sensor for single nucleotide polymorphism detection in serum based on a triple-stem DNA probe. J. Am. Chem. Soc. 131(42), 15311–15316 (2009)
- 60. Lubin, A.A., et al.: Sequence-specific, electronic detection of oligonucleotides in blood, soil, and foodstuffs with the reagentless, reusable e-DNA sensor. Anal. Chem. **78**(16), 5671–5677 (2006)
- Lubin, A.A., et al.: Effects of probe length, probe geometry, and redox-tag placement on the performance of the electrochemical e-DNA sensor. Anal. Chem. 81(6), 2150–2158 (2009)
- Gao, A., et al.: Silicon-nanowire-based CMOS-compatible field-effect transistor nanosensors for ultrasensitive electrical detection of nucleic acids. Nano Lett. 11(9), 3974–3978 (2011)
- Chen, C.-P., et al.: Ultrasensitive in situ label-free DNA detection using a GaN nanowire-based extended-gate field-effect-transistor sensor. Anal. Chem. 83(6), 1938–1943 (2011)
- Shendure, J., Ji, H.: Next-generation DNA sequencing. Nat. Biotechnol. 26(10), 1135–1145 (2008)
- Hutchison, C.A.: DNA sequencing: bench to bedside and beyond. Nucleic Acids Res. 35, 6227–6237 (2007)

- Shendure, J., et al.: Advanced sequencing technologies: methods and goals. Nat. Rev. Genet. 5, 335–344 (2004)
- Agrawal, A., et al.: Counting single native biomolecules and intact viruses with color-coded nanoparticles. Anal. Chem. 78(4), 1061–1070 (2006)
- Liu, K.J., et al.: Decoding circulating nucleic acids in human serum using microfluidic single molecule spectroscopy. J. Am. Chem. Soc. 132(16), 5793–5798 (2010)
- Levene, M.J.: Zero-mode waveguides for single-molecule analysis at high concentrations. Science 299, 682–686 (2003)
- Li, S.-J., et al.: A nanochannel array-based electrochemical device for quantitative label-free DNA analysis. ACS Nano 4(11), 6417–6424 (2010)
- Menard, L.D., Ramsey, J.M.: Fabrication of sub-5 nm nanochannels in insulating substrates using focused ion beam milling. Nano Lett. 11(2), 512–517 (2010)
- 72. Yang, S.Y., et al.: DNA-functionalized nanochannels for SNP detection. Nano Lett. 11(3), 1032–1035 (2011)
- 73. Ivanov, A.P., et al.: DNA tunneling detector embedded in a nanopore. Nano Lett. 11(1), 279–285 (2010)
- 74. Liang, X., et al.: Single sub-20 nm wide, centimeter-long nanofluidic channel fabricated by novel nanoimprint mold fabrication and direct imprinting. Nano Lett. 7(12), 3774–3780 (2007)
- 75. Spiering, A., et al.: Nanopore translocation dynamics of a single DNA-bound protein. Nano Lett. **11**(7), 2978–2982 (2011)
- 76. McNally, B., et al.: Optical recognition of converted DNA nucleotides for single-molecule DNA sequencing using nanopore arrays. Nano Lett. 10(6), 2237–2244 (2010)

Chapter 4 Advanced Molecular Probes for Sequence-Specific DNA Recognition

Alessandro Bertucci, Alex Manicardi, and Roberto Corradini

Abstract DNA detection can be achieved using the Watson-Crick base pairing with oligonucleotides or oligonucleotide analogs, followed by generation of a physical or chemical signal coupled with a transducer device. The nature of the probe is an essential feature which determines the performances of the sensing device. Many synthetic processes are presently available for "molecular engineering" of DNA probes, enabling label-free and PCR-free detection to be performed. Furthermore, many DNA analogs with improved performances are available and are under development; locked nucleic acids (LNA), peptide nucleic acids (PNA) and their analogs, morpholino oligonucleotides (MO) and other modified probes have shown improved properties of affinity and selectivity in target recognition compared to those of simple DNA probes. The performances of these probes in sensing devices, and the requirements for detection of unamplified DNA will be discussed in this chapter. Chemistry and architectures for conjugation of probes to reporter units, surfaces and nanostructures will also be discussed. Examples of probes used in ultrasensitive detection of unamplified DNA are listed.

4.1 Introduction

The possibility to perform detection of specific nucleic acid sequences in biological samples and to correlate the information obtained with specific genetic traits is one of the most powerful tools available in this so called "post-genomic era" [1]. Tracking DNA and RNA instead of other analytes, such as proteins, carbohydrates, lipids or other metabolic products, has the advantage to be accessible through a

A. Bertucci • A. Manicardi • R. Corradini (🖂)

Dipartimento di Chimica Organica e Industriale, Università di Parma, Parco Area delle Scienze 17/A, Parma, Italy e-mail: alessandro.bertucci@nemo.unipr.it; alex.manicardi@unipr.it; roberto.corradini@unipr.it

G. Spoto and R. Corradini (eds.), *Detection of Non-Amplified Genomic DNA*, Soft and Biological Matter, DOI 10.1007/978-94-007-1226-3_4,

[©] Springer Science+Business Media Dordrecht 2012

rational design of probes. Oligonucleotide probes have DNA structure and can bind to target sequences via formation of the same base pairing occurring in the genetic material, i.e. Watson-Crick hydrogen bond scheme and several other simple and well-established interactions. Genosensors are often realized by coupling a set of specific oligonucleotide probes with transduction devices able to transform the molecular recognition into a physical measurable signal, mostly electrochemical, optical, mechanical or magnetic [2, 3].

Since the recognition event is the central part of the biosensing process, an ideal probe is that which binds to the target sequence with high affinity, recognizes only a specific target sequence and does not bind in the presence of mismatched bases; however, this goal is met only under appropriate conditions and by careful design of the probes, and still differences of only one base in the target sequences, corresponding to single nucleotide polymorphisms (SNPs) or point mutations, are still the most difficult to detect, so that discrimination at this level is one of the ultimate goals of genosensor design. In fact both SNPs and point mutations have great relevance in diagnostics. Mutation at a single base level are often found in connection with pathologies: hereditary diseases such as thalassemia [4] and cystic fibrosis [5] are correlated to single point mutations, and point-mutated genes (e.g. p53 or K-ras point mutation) are frequently associated to tumor onset [6, 7].

SNPs can be used as molecular markers to discriminate between species, breeds, varieties and even individuals [8], and in several applications, SNP-detecting devices can be used as tools for identification of raw materials in foods [9].

In order to achieve best performances in DNA recognition in terms of affinity and sequence specificity not only for analytical purposes, but also in the design of new drug candidates, chemists have designed and synthesized a large number of DNA mimics. Structures of oligomers bearing natural nucleobases in combination with unnatural backbones have been proposed in the last two decades, and some of the best performing structures have been used as new tools for improving the performances of genosensing devices. The optimization of probes is a crucial element especially when ultrasensitive techniques are used for enabling detection of nucleic acids at very low concentrations. Some of the most important classes of DNA analogs relevant for ultrasensitive detection will be discussed in the present chapter.

Furthermore, a series of chemical tools aimed to the modification of the molecular probes with reporter groups or sensor surfaces have been developed in the last decades, and the possibility to have probes "decorated" with the proper combination of elements has become very important in the design of specific tests. More recently conjugation with nanostructured materials such as liposomes, nanoparticles, nanotubes or graphene has become more and more important, due to the suitability of these components for optical and electrochemical detection. The use of these tools will be discussed thoroughly in another chapter of this book [10] and only chemical linking of probes to these objects will be taken into account here. Finally, examples of probes used in ultrasensitive detection of unamplified DNA will be presented.

4.2 Oligonucleotides (ON) and Modified Synthetic Analogs

The availability of synthetic oligo(deoxyribonucleotides) (ON), mostly synthesized by phosphoroamidite chemistry described by Caruthers, [11] (now commercially available from many sources) is one of the factors that make possible a large variety of applications from biotechnology to nanotechnology and is particularly useful in the design of genosensors. A careful quality control of commercial products and eventual purification steps are advisable for best performances in analytical systems.

Proper choice of DNA probes is necessary in order to design a genetic test for specific purposes and in combination with specific devices, but some limitations are intrinsic in the chemical nature of DNA. Thus, oligonucleotide mimics are currently under development in many laboratories and some of them are only available in the inventors' lab; however, best performing structures have become commercially available and can be purchased from specialized firms. Among them, most interesting are locked nucleic acids (LNA), peptide nucleic acids (PNA) and morpholino oligonucleotides (MO), which will be discussed more in detail, but also other structural variations of DNA have proven to be very effective in DNA recognition and have potentiality to become new commercially available tools. In this paragraph the performances of these probes will be discussed.

4.2.1 Performances of Probes in DNA Analysis

In order to have control of the sensing devices, the characteristics of DNA probes should be taken into account, especially for the design of ultra-sensitive techniques. Supramolecular interactions occurring in solution can be evaluated using existing models, since thermodynamics of DNA:DNA interactions have been extensively studied; surface-attached probes have however different models for their interactions and differences in affinity and selectivity from the solution data are likely to occur; this led many scientist to specifically study the probe-target interaction on solid surface and to evaluate surface effects which can be rationalized. Finally, the multiple functionalization of nanostructures, such as nanoparticles, nanorods, nanotubes give rise to binding processes which, although occurring in solution, are governed by a surface-like environment in which multiple interactions can occur. The following paragraph will discuss these issues in terms of affinity and selectivity of DNA probes.

4.2.1.1 Affinity

The possibility to perform detection of a specific DNA sequence of interest is given by the formation of a duplex with the target sequence (T) by the probe (P):

 $T + P \rightleftharpoons T : P$

The stability constant of the process is given by:

$$K_{ass} = [T:P] / [T] [P]$$
 (4.1)

and the dissociation constant is the reciprocal of this constant. i.e.

$$K_d = 1/K_{ass} \tag{4.2}$$

the latter parameter is useful in analytical applications since it is correlated to the possibility of duplex formation under very low concentration of the target sequence [12]. If an excess of P is used, the dissociation constant represents the concentration of P at which half of the target is bound to the probe. Dilution of a 1:1 complex have a dissociation curve such as that reported in (Fig. 4.1a) for a 10^{15} M⁻¹ constant. For oligonucleotide analogs, unfortunately, most of the studies on DNA binding report melting temperature (Tm) as a measure of affinity. Tm is the temperature at which, under given conditions, a DNA:DNA duplex is half dissociated, and therefore is not a direct measurement of the stability constant, but only of its temperature dependence (Fig. 4.1b). However, thermodynamic parameters can be inferred from melting profiles using several models [13].

Thermodynamic data for oligonucleotides are available in several databases and different models for predicting the DNA:DNA duplexes stability on the basis of the sequence have been proposed. Prediction of ΔG° values can be done using these models [14]. As an example, using the Santa Lucia model, a 10mer duplex of sequence 5'-GTAGATCACT-3' at 1 μ M in 0.1 M NaCl can bind its complementary sequence with a Tm of 23.47 °C and a $\Delta G^{\circ} = 36$ KJ mole⁻¹, which corresponds to K of 2.4 × 10⁶. Therefore, using DNA probes of this length, with analytes largely below micromolar concentration is not likely to succeed.

By doubling the sequence, the ΔG° is more than doubled, leading to a K of 1.1×10^{15} as calculated by the web tool available at http://mobyle.pasteur.fr/cgi-bin/portal.py?#forms::melting (for a library of Tm prediction website see http://protein.bio.puc.cl/cardex/servers/melting/sup_mat/servers_list.html).

A different case is given when the probe is immobilized onto a surface.

In this case, the concentration of the probe can be substituted by its density on the surface and T:P is formed only on the surface itself. In this case the simple equation (4.1) can be used for the evaluation of stability constant on the surface, provided that 2D concentration (i.e. probe density) is used.

Alternatively, a Langmuir model can be used by considering each probe as an "active site". A complete model for thermodynamics as well as kinetics of oligonucleotide hybridization on surfaces can be found in the work of Erikson et al. [15].

It should be noted that models are affected by deviation due to the surface localization of the probes, which forces the target molecules to remain on the surface. In the case of DNA molecules, which are negatively charged, a negative cooperative effect can be present, since the occupation of one active site by a DNA molecule would disfavor the binding of another DNA molecule on an adjacent site



Fig. 4.1 (a) Effect of dilution on T:P fraction in the case of a full matched (FM) duplex and in the case of a two order of magnitude less stable mismatched (MM) sequence; (b) effect of temperature on T:P fraction for full-match and mismatched duplexes.; (c) ranges of concentration for detection of PCR-amplified or non-amplified DNA

by electrostatic repulsion. This effect depends on probe density and on the ionic strength of the aqueous solution, but also on the nature of the probe itself. This issue has been studied in detail for SPR sensors by the group of Knoll and hybridization efficiency was shown to depend on the probe density and reached nearly 100 % for 15mer DNA fully complementary targets at a probe density of $\sim 1.2 \times 10^{12}$ molecules/cm² [16]; thus this coverage has been taken as reference by successive works in order to avoid these negative effects.

A third case is that of modified nanoparticles bearing a large number of probe molecules. Modeling of this situation can be done using the concept of multivalence developed by several authors and described in different review articles [17]. Also in this case negative cooperativity can be present for DNA. A simplified model can be used by considering the total concentration of probes as the concentration of nanoparticles multiplied by the mean number of probes per nanoparticle. Again, this model is valid only if the probe density on a single nanoparticle is sufficiently low to prevent cross-talking.

As a term of reference, in Fig. 4.1c the concentrations of DNA targets in analytical applications are depicted. In normal applications in which a PCR protocol is used, the concentration of the target is at least in the nanomolar range and

therefore stability constants of 10^9 would be recommended in order to have sufficient complex formation for an efficient detection. For PCR-free protocols, very high stability constants of 10^{15} – 10^{18} should be used for good hybridization.

However, it should be considered that, if the detection method is very sensitive and it is able to distinguish the target in the free and in the bound state, even very few copies of the complex can be detectable, thus allowing to considerably lower the detection limit even if the probe has dissociation constants largely above the concentration of the target.

Self-complementarity and probe folding are two of the characteristics of the probes to be strictly evaluated (and avoided) in the design of probes for sensing technologies, as illustrated by Minunni and coworkers in a recent paper using freely available software for secondary structure evaluation [18].

4.2.1.2 Sequence Selectivity

Sequence selectivity is also a very important issue in DNA detection, in fact, DNA probes do not display the ideal all-or-none behavior, since partial complexation is possible for sequences differing from the target for only few bases. The most difficult case is that of a single-base difference (single mismatch), for which the stability constant and melting temperature could be very close to that of a fully matched duplex; however since differences in a single base represent clinically interesting cases involving oncogenes (e.g. *K-ras*) [7] or genetic diseases (e.g. thalassemia [4] or cystic fibrosis [5]), and single nucleotide polymorphisms (SNPs) detection can be a very powerful tool for identification purposes, pushing the recognition properties of the probes to the maximum efficiency in mismatch recognition is one of the main goals in probe design.

Sequence selectivity can be expressed in sensory systems as the ratio between the signal obtained with a specific probe-sensor combination, using the full-matched DNA and that obtained using mismatched sequences. The most stringent conditions for this evaluation are those relative to single-base mismatch. It should be noted that thermodynamics of mismatch binding are dependent on the length of the T:P duplex and on the type of mismatch. For DNA probes a full set of mismatch data are available from databases reporting thermodynamic data for oligonucleotides [14].

Usually, both the increase in temperature and the dilution of samples have the effect of increasing sequence selectivity, since conditions for dissociation of mismatched targets are met before those of the full-matched sequence (Fig. 4.1a, b). A detailed study of temperature dependence of fluorescent signal has been carried out for beacon probes (see Sect. 4.3.2), and it was shown that proper choice of temperature was essential in order to have maximum selectivity for single mismatches. However, high temperatures are not always compatible with sensing devices. Dilution of both probe and targets have the effect of dissociating the less-stable mismatched sequence (Fig. 4.1a), but this causes also a decrease in the sensitivity of the method; this strategy is therefore much more efficient when ultrasensitive techniques are used for detection; working at very low concentration is normally accompanied by very high sequence selectivity.

According to these concepts, at a fixed temperature and concentration, the increase of the T:P length induces an increase in stability (and hence in sensitivity if the target is at very low concentration), but a loss in sequence-selectivity. Thus the only way to further push the performances of an optimized DNA sensory systems is to change the chemical nature of the probes used.

4.2.2 Oligonucleotide Mimics

During the past decades, the limitations in the use of ON probes were partially overcome using newly designed DNA mimics, in which the essential features of DNA structure are conserved, in particular the presence of nucleobases able to perform Watson-Crick-like base pairing, whereas the other chemical groups are drastically changed. The quest for new, highly efficient, DNA binding molecules has been pursued for the last two decades very intensely and it is still a subject of many studies. Chemists have changed all the essential features of the DNA model: the phosphate group, the sugar moiety and nucleobases. The effect of these modifications have been extensively studied, [19] especially in connection with antisense technology [20]. The most relevant modifications which proved to be successful in sensory systems are those related to changes in the backbone, using locked nucleic acids (LNA), morpholino oligonucleotides (MO), peptide nucleic acids (PNA) and their analogs, (Fig. 4.2), and several other miscellaneous probes which will be discussed in the next paragraphs.

4.2.2.1 Locked Nucleic Acids (LNA)

LNA (Locked nucleic acid, Fig. 4.2a) was first described by Singh et al. in 1998 as a new class of conformational restricted oligonucleotide analogs [21]. LNAs are oligonucleotides containing at least one LNA monomer, i.e., one 2'-O,4'-*C*-methylene- β -D-ribofuranosyl nucleotide (β -D-LNA), where a ribonucleoside is linked between the 2'-oxygen and the 4'-carbon atom with a methylene bridge [21, 22]. In LNA monomers, the sugar is locked in a C3'-endo conformation (N-type and RNA-like) by the methylene bridge [21, 23] and structural studies show that LNA is a RNA mimic, fitting seamlessly into an A-type duplex geometry [24]. α -L-LNA (α -L-ribo configuration), stereoisomeric analogs of LNA have also been extensively studied [25].

LNA oligonucleotides can be synthesized using conventional automated phosphoramidite chemistry and LNA monomers are compatible with other monomers, e.g. DNA, RNA, 2'-O-Me-RNA, and with phosphorothioate (PS) and phosphodiester linkages [26, 27]. The interest in LNA applications has steadily grown up in last years: these nucleic acids analogs in fact show a very efficient binding to



Fig. 4.2 Structures of (**a**) DNA (**b**) locked nucleic acids (*LNA*) (**c**) morpholino phosphoamidate oligonucleotides (*MO*) (**d**) peptide nucleic acids (*PNA*). (**e**) Hoogsteen base pairing occurring in triplex formation; (**f**) Mechanism of action of PNA openers via triplex formation

complementary nucleic acids and a high potency as antisense molecules *in vitro* and *in vivo*, together with a commercial availability as oligonucleotides (fully modified, mix-mers with DNA, RNA, other monomers and various modification) and as phosphoramidite building blocks [28].

LNAs act as natural nucleic acids with respect to Watson-Crick base pairing and fully modified LNA oligomers induce very high thermal stability of duplexes toward complementary RNA or DNA [26]. In particular, a large number of LNA:DNA hybrids have been structurally characterized by NMR spectroscopy and X-ray crystallography: these duplexes showed the common features for native nucleic acid duplexes, i.e. Watson–Crick base pairing, *anti* orientation of nucleobases, base stacking, and right-handed helical conformation [29].

The most conclusive feature of LNAs is the unprecedented hybridization to complementary nucleic acids, a property that made LNAs become general and versatile tools for specific high-affinity recognition of single-stranded DNA (ssDNA) and, due to their forced C3'-endo conformation, particularly suited for the binding of RNA.

Single-stranded LNA oligonucleotides can also interact with double-stranded DNA (dsDNA) by two possible types of hybridization, i.e. binding to the dsDNA by the formation of a triple-helical complex or binding to the dsDNA by single-strand targeting, thus displacing the other DNA strand by a so-called "strand invasion" process. In particular, gene expression could be up- or downregulated at the level of transcription via selective triple-helix formation, preferably at promoter sites, emphasizing new appealing approaches [30]. The properties of LNAs for targeting DNA thus make LNAs exceptional tools for nucleic acid probes and absolutely

crucial in this context is the freedom in design with excellent cooperativity between LNA monomers and other classes of monomers and labels. Then, a wide range of LNA-based approaches have been evaluated in the field of ultrasensitive detection of DNA, especially in developing PCR-free techniques.

LNA probes modified with pyrene units were used as "invaders" for dsDNA (vide infra) [31]. LNA capture probes have also been used for the development of electrochemical hybridization assays. For example LNA probes was employed in an electrochemical assay in which streptavidin-coated magnetic micro-beads were used as solid phase to immobilize the biotinylated capture probes. The target biotinylated DNA sequences were then recognized via hybridization with the LNA capture probe. After labeling the biotinylated hybrid with streptavidin–Horse Radish Peroxidase (HRP) conjugate, the electrochemical detection of the enzymatic product was performed onto the surface of a disposable screen-printed electrochemical cell (SPEC) [32].

In another approach, LNA oligonucleotides have been used in the development of sensitive and selective molecular beacons (MB) [33].

Thus, hybridization properties displayed by LNAs have made them very efficient tools for many applications in detecting DNA targets in various fields of research, especially diagnostics, genomics and biotechnology.

4.2.2.2 Morpholino Oligonucleotides

Morpholino oligonucleotides (MOs, Fig. 4.2c) are another class of very popular oligonucleotide analogs, first described by Summerton in 1989 [34]. MOs have standard nucleic acid bases bound to morpholine rings which are linked through phosphorodiamidate groups [35].

Replacement of anionic phosphates with the uncharged phosphorodiamidate groups eliminates ionization in the usual physiological pH range, so MOs in organisms or cells are present as uncharged molecules; in addition, the lack of charge makes them especially useful for electrostatic transduction. Furthermore, this particular structure makes MOs extremely resistant to nucleases and other degradative factors in biological systems [24]. Finally, it is worth to say that MOs excel in the properties of stability, efficacy, long-term activity, water solubility, low toxicity and specificity. The morpholino structure was devised in order to circumvent some cost-problems associated with other DNA mimics: in fact, MOs can be synthesized starting from cheap ribonucleosides, to which an amine group is introduced via a relatively simple ribose-to-morpholine transformation. Hence, the second step is represented by the oligo assembly of activated subunits into oligos of defined sequence, through coupling and deprotection reactions [24, 36].

MOs have been used as antisense to direct splicing of pre-mRNA [37], blocking of selected gene expression [38] and detecting RNA sequences [39]; in addition, in recent years, an increasing number of works have been published about the use of MOs systems for sensing DNA. In particular, MOs have been frequently employed in the field of surface hybridization technologies, especially in assays performed on solid supports on which immobilized MO probes are able to react


Fig. 4.3 (a) Schematic illustration of direct DNA detection using a morpholino capture probe and silicon nanowires (*SiNW*); (b) Real-time conductance response for a Morpholino-functionalized SiNW sensor during injection of 1 nM complementary, non-complementary and one-base mismatched DNAs (Reprinted from Ref. [44], Copyright 2010. With permission from Elsevier)

with analyte targets from solution. Kinetic and thermodynamic aspects of surface hybridization in MO-DNA systems have been studied [40, 41]. One example of an application in this field is represented by the work carried out by Tercero et al. [42], based on a label-free analysis of DNA by surface hybridization employing thiolated MO monolayers on a gold surface. Hybridization of uncharged morpholino monolayers with charged nucleic acid targets alters the dielectric and electrostatic characteristics of the surface environment, so that the signatures of hybridization can be monitored through the surface differential capacitance C_d . Based on a 10:1 signal-to-noise criterion, C_d measurements limits of quantification corresponding to several picograms of material per square millimeter were attained.

In another approach, an ultrasensitive electrochemical biosensor employing a MO oligomer as capture probe and a cationic redox polymer as signal generator for direct detection of DNA was proposed, [43] based on the immobilization of the morpholino oligomer on an indium-tin oxide (ITO) electrode and amperometric detection of target DNA by forming a DNA/cationic redox polymer bilayer on the electrode. The cationic redox polymer is introduced to the ITO electrode via electrostatic interaction with the hybridized DNA and it exhibits excellent electrocatalytic activity towards the oxidation of ascorbic acid, allowing direct voltammetric and amperometric detection of DNA.

MO-DNA hybridization was also exploited by Zhang et al. [44] in the achievement of morpholino-functionalized silicon nanowires as novel gene chip platforms for the sequence-specific, label free detection of DNA. In this case MO were covalently attached to functionalized silicon nanowires, and subsequent MO-DNA hybridization on silicon surface was monitored through time-dependent conductance measurements, thus obtaining limits of detection in the femtomolar range (Fig. 4.3).

All these properties make MO promising, though still not largely used, probes for sensing applications, especially in the field of surface optical and electrochemical sensors.

4.2.2.3 Peptide Nucleic Acids (PNA)

Peptide Nucleic Acids (PNAs) are DNA mimics in which the negatively charged sugar-phosphate backbone is replaced by a neutral polyamide backbone composed of N-(2-aminoethyl)glycine units (Fig. 4.2) [45]. They can be considered very promising tools due to their unique properties, since they can bind complementary DNA or RNA sequences following standard Watson-Crick rules [46], but, due to the lack of electrostatic repulsion, PNA/DNA and PNA/RNA complexes have improved thermal stability as compared to DNA/DNA and DNA/RNA duplexes [47]. The high affinity for the DNA target allows to decrease the limit of detection in many applications devoted to DNA recognition. Polypyrimidine PNA can form very stable triplexes with polypurine DNA of the type PNA/DNA/PNA, via both Watson-Crick and Hoogsteen base pairing (Fig. 4.2e). These structures are so stable that dsDNA undergoes displacement of the non-complementary strand, thus forming a D-loop structure. Though limited by sequence restriction, this process allows to directly targeting dsDNA, a goal which is intrinsically difficult with most of the other oligonucleotide probes. The formation of triplex structures is favoured by replacing cytosine with pseuisocytosine (J-base), which can form Hoogsteen hydrogen bonds at higher pHs. Thus, PNA-openers can be obtained (Fig. 4.2f), and their use allows to make one of the two strands available for recognition by a specific probe [49].

Another interesting feature of PNA probes is their selectivity: a single basepairing mismatch in PNA/DNA duplexes is more destabilizing than in DNA/DNA complexes of the same length [46]. Moreover, since the increased affinity allows to obtain stable PNA-DNA and PNA-RNA complexes even with short PNA probes (10-mer or less), as compared to DNA probes, the possibility to use short sequences in post-PCR applications favours improved selectivity, which in some applications allows for an easy detection of single-point mutations and single nucleotide polymorphisms [50, 51].

PNA-DNA hybridization is less sensitive to the presence of salts, which are necessary to attenuate electrostatic repulsions in duplex DNA. Actually, PNA-DNA binding can be efficiently achieved even under very low salt concentrations, a condition that promotes the destabilization of RNA and DNA secondary structures resulting in an improved access to target sequences [52]. Low ionic strength conditions are essential when targeting a double-stranded DNA, in order to disfavour DNA-DNA duplex formation, and to allow PNA probes to invade the double helix, displacing the homologous DNA strand. The relative independence of their performances to the environment makes the analytical procedures more robust, especially in the case of the analysis of complex matrices such as biological fluids and foods.

Finally, an interesting property of PNAs, which is useful in many biological applications, is their stability to both nucleases and peptidases, since their "unnatural" skeleton prevents recognition by natural enzymes, making them highly persistent in biological fluids [53]. A major drawback is that enzymatic reactions, which are often used in combination with DNA probes, are not possible using PNA substitutes. Therefore, detection schemes involving e.g. DNA-ligase or DNA-polymerases

cannot be performed with PNAs. However, PNAs can easily be functionalized with recognition elements for proteins (such as biotin) and then coupled with enzymatic assays [54]. As a further consequence of their enzymatic and chemical stability, kits and sensory systems produced using PNA probes are superior to DNA-based ones for long-term storage.

For these reasons, PNAs have been used as analytical tools in PCR-base methods and in a wide variety of sensing techniques, including microarrays, electrochemical-, surface plasmon resonance (SPR)-biosensors, FISH, and single molecule detection (see the following paragraphs) [55, 56]. Ultrasensitive detection of DNA could be obtained using DNA capture by PNA and hybridization with Au nanoparticles in surface plasmon resonance imaging (SPRI) [57] with selectivity at the single base-mismatch level. The efficient capture of genomic DNA was accomplished and detection of as low as 11 DNA copies of unamplified genomic DNA from biological samples could be performed [58].

The excellent hybridization properties of PNA oligomers, combined with their unique chemistry, has been exploited in a variety of molecular biology tools, biomedical applications and diagnostic techniques [59]. Custom-made PNA probes are now commercially available, on account of the easily accomplished synthesis by the standard procedures usually adopted in peptide chemistry. The high chemical stability of PNAs allows the use of several different protecting groups for the terminal nitrogen (including Fmoc and Boc) and for the nucleobases (amides, Cbz or acid labile protecting groups), fully exploiting the diverse strategies developed in the last years in peptide synthesis [60].

4.2.2.4 Peptide Nucleic Acids Analogs

The PNAs scaffold has served as a model for the design of new compounds able to perform DNA recognition. Many modifications of the basic PNAs structure have been proposed, in order to improve their performances in term of affinity and specificity towards complementary oligonucleotides. A modification introduced in the PNAs structure can improve their performances as probe in three different ways: (i) improving DNA binding affinity; (ii) improving sequence specificity, in particular for directional preference (antiparallel vs parallel) and mismatch recognition; (iii) improving signal transduction.

Several reviews have covered the literature concerning new chemically modified PNAs [61, 62]. Base modification has been a very powerful tool for improving the DNA recognition process by PNA. One important variation of PNA which has proven to be very useful is the modification of adenine and thymine residues with pseudo-complementary bases 2,6-diaminopurine (Fig. 4.4a) and 2-thiouracil (Fig. 4.4b), which can form base pairs with thymine and adenine respectively, but are not able to bind to each other [63]. Using these bases, double-duplex invasion can be obtained (Fig. 4.4c), which enable to separate the two strands of a dsDNA; if properly designed, this process can be use to generate single-stranded tracts which can be recognized by specific probes or enzymes [48].



Fig. 4.4 Modified PNA structural variations. (a) 2,6-diaminopurine (D) paired with thymine; (b) 2-thiouracil (tU) paired with adenine. (c) Double duplex invasion process using D and tU containing PNA. (d) type of modifications reported in the literature for PNA backbone. (e) Chiral acyclic PNA bearing functional groups either on C2, C5, or both. (f) prolyl-(ACPC)-PNA. (g) pyrrodinyl-PNA

A very large amount of data is available for PNA backbone variation (Fig. 4.4d). Structure-activity relationships showed that the original design containing a 6-atom repeating unit and a 2-atom spacer between backbone and the nucleobase was optimal for DNA recognition [64]. Introduction of different functional groups with different charges/polarity/flexibility have been described and are extensively reviewed elsewhere [65]. PNAs have been constantly improved during the years, using the concept of "preorganization", i.e. the ability to adopt a conformation which is most suitable for DNA binding, thus minimizing the entropy loss of the binding process. The main strategies which have been used for achieving this goal are: (a) preorganization by cyclization of the PNA backbone (in the aminoethyl side or in the glycine side), (b) addition of substituents at C2 or C5 carbon of the monomers which has the effect of shifting the PNA preference towards a righthanded or left-handed helical conformation, according to the configuration of the new stereogenic centers, in turn affecting the stability of the PNA-DNA duplex through a control of the helix handedness. Many of these modifications included the presence of one or more stereogenic centers, and it was thoroughly demonstrated that there is a dramatic effect of chirality on DNA recognition [66, 67].

Using the linear N-(2-aminoethyl)glycine as a starting point, several PNA derivatives were obtained by insertion of side chains either at the C2 (α) or

C5 (γ) carbon atoms (Fig. 4.4e). These modifications have the effect of modulating the conformational properties of PNA. If the constraint is appropriate for the conformation required for DNA binding, this can actually result in improved DNA binding properties, whereas if not, a detrimental effect is obtained.

The insertion of stereogenic centers in one of the PNA strands induces a predominant helix handedness [68, 69]; lysine-based chiral PNAs, and in particular the "chiral box" PNA (with three chiral monomers inserted in the middle of the probe sequence) showed an increase of the sequence selectivity, both in terms of direction control and of recognition of a single base mismatch [70]. Therefore, this type of structure seems to be ideal for targeting point mutations in genes with therapeutic interest. For standard achiral decameric PNAs, the difference in melting temperature between antiparallel and parallel complexes is about 10 °C. The insertion of D-lysine side chains in the PNA strand had the effect of amplifying this difference in stability, an effect which is magnified by the above mentioned D-Lys "chiral box" PNA, which was shown to bind the complementary antiparallel-, but not the parallel-DNA, as evaluated by UV and CD melting and by ESI-MS spectra [70, 71]. Mismatch recognition was a property particularly pronounced in the "chiral box" containing PNAs and this was found to be true for different sequences. Using this model of PNA modification it was possible to detect very efficiently the single point mutations W1282X and R553X causing cystic fibrosis with surface plasmon resonance (SPR) biosensors [72] and capillary electrophoresis [73] respectively.

C5-substituted PNA could be used as very convenient scaffolds for linking functional groups to the PNA backbone [74]. In several different publications C5-modified PNA monomers have been used for the introduction of functional groups such as thiols [75] or carboxylates [76]. C5-substituted PNAs bearing small side chains derived from L-alanine and L-serine have strong preference for right-handed helicity, which favors DNA binding [77].

A systematic study performed on PNA containing one chiral monomer derived from D- or L- Lysine revealed that both the D-configuration at the C2 and the L-configuration at the C5-position induce right handedness, whereas the opposite stereochemistry induces a preference for left-handedness and therefore less DNA affinity [78]. Thus, the DNA binding affinity can be modulated from none to very strong by tuning the stereochemistry of a single monomer in a PNA decameric strand. On the base of these results, chiral PNA probes were designed for microarrays for recognizing ApoE genotypes related to the Alzheimer disease [79]. The "chiral box" model was tested using either only 2D or only 5L monomers derived from arginine [80], and an "extended chiral box" PNA containing both 2D and 5L arginine monomers [81].

Cyclic structures have been described with increasing binding affinity and sequence specificity [82]. In particular, excellent properties of DNA binding could be achieved using five-membered ring structures such as prolyl-(ACPC)-PNA (Fig. 4.4f) [83, 84] or pyrrolidinyl-PNA (Fig. 4.4g) [85], which were used in sensory applications. Selectivity for DNA over RNA [86] and for RNA over DNA [87] were also reported using cyclic PNA analogs.



Fig. 4.5 Other DNA analogs and modified bases with remarkable properties. (a) Hexitol nucleic acids (*HNA*); (b) altritol nucleic acids (*ANA*); (c) threose nucleic acids (*TNA*); (d) tricyclo-DNA; (e) the guanidium G-clamp modified nucleobase, which can improve both binding and selectivity and has been used on ON and on PNA

4.2.2.5 Other Modified Oligonucleotides

Many other oligonucleotide analogs, though not as widely used as the four classes described above, have been described in the literature and tested in antisense technology, with binding affinities and selectivity which could be exploited for diagnostics.

A variety of sugar substitutions have been evaluated by Eschenmoser for understanding ON structural features [88]. Threose nucleic acids (TNA, Fig. 4.5c) are a simplified version of DNA in which the sugar moiety has been reduced to a simple four-atom unit. Despite having a repeat unit one atom shorter than natural nucleic acids, TNA is able to base pair with RNA, DNA and itself [89]; therefore TNA oligos have been proposed as ancestors of DNA in prebiotic chemistry. TNA are very interesting also because they can act as template for DNA synthesis and can be obtained from a DNA template by polymerases [90].

Hexitol nucleic acids (HNA) and altritol nucleic acids (ANA) are DNA mimics with six-membered rings (Fig. 4.5) developed as potential antisense drugs and recently tested as components of arrays for SNP detection [91].

Tricyclo-DNA are constrained oligonucleotide analogs which have been successfully used in antisense research to redirect pre-mRNA splicing ONs and ON analogs containing modified nucleobases have also been a subject of intense research. Most notably, the introduction of "engineered" nucleobases led to an increase of both affinity and selectivity in antisense research. Nucleobases with extended aromatic cores have been shown to improve stacking interactions, thus stabilizing the double strand formed with complementary oligonucleotides. Combination of base stacking, properly designed hydrogen bonds, and electrostatic interactions led to very interesting structures, such as the G-clamp residue, able to bind strongly and selectively to guanine by simultaneous occurrence of all the above mentioned interactions (Fig. 4.5e) [92].

The research on nucleobase modification is centrally related to sequence selectivity and affinity. Benner and coworkers have developed a whole set of nucleobasesubstitutes, expanding the natural genetic code, called artificially expanded genetic information system (AEGIS) that has as many as 12 nucleotides and as many as 6 base pairs. AEGIS nucleotides are already being applied as orthogonal binding elements in human diagnostic tools, including in FDA-approved assays for HIV, hepatitis B, and hepatitis C viruses. Kool and coworkers have introduced extended bases as tools for nucleic acid recognition [93].

Specific nucleobase substitutes are very powerful tools, and fine tuning of specificity can be obtained, as demonstrated by the design of special nucleobases allowing the detection of specific oxidative lesions such as 8-oxo-guanine [94].

Substitution of the nucleobase with aromatic rings has the effect of abolishing base specificity, thus allowing a particular position to act as an universal base, which can be useful in some applications. If the aromatic moiety is fluorescent, a change in fluorescence can be obtained upon forced intercalation, thus leading to a direct read-out of the recognition process [95].

Not all the above reported variations in the chemical structure of oligonucleotides have been so far used in diagnostics. However, the toolbox for development of new, more efficient and specific sensory systems is very rich and it is continuously improving.

4.3 Chemistry and Architectures of Probes in DNA Sensing

The design of probes for a specific sensing application cannot be done without considering the type of modifications and chemistry needed for the conjugation with reporting elements, sensor surfaces and nanostructures. Furthermore, ONs or ON analogs are often used in complex molecular architectures, where the design of the probe requires the presence of special recognition elements.

4.3.1 Chemistry for Bioconjugation and Probe Immobilization

The oligonucleotides and their analogs in most cases have to be linked either to one or more reporting units or to a surface. This is normally achieved using an appropriate spacer and appropriate chemistry which is compatible with the oligonucleotide and with the groups to be linked.

Bioconjugation of oligonucleotides and their analogs is carried out with essentially three different strategies: (i) use of conjugated monomers bearing functional groups (fluorophores, electroactive moieties etc.) either on the backbone or on the nucleobase; (ii) modification during oligomer synthesis, (iii) post-synthetic conjugation using specific reaction schemes, in particular click chemistry [96]. Many types of fluorophores have been linked to oligonucleotide backbone, mainly through 2'-modification of a ribose unit. Variations of the LNA structure with the introduction of a nitrogen atom also allow to introduce substituents which can carry active reporting moieties. Peptide nucleic acids derivatives are very versatile structures for this goal, since the introduction of functional groups can be achieved through modification at C2 or C5 positions of the backbone.

Nucleobase modification is most easily obtained by modification of C5 or C6 of pyrimidines and C2 of adenine, though other substitutions have been reported.

Redox-active functional groups, e.g. pyrene, phenothiazine, porphyrin, Ru(II) (bpy)₃ or Os(II)(bpy)₃ complexes, dimethylaminonaphthalene, TAMRA, glycosides have been introduced using this strategy [97]

Several chemistries are available for linking probes to a solid surface, and many of them have been developed in the field of microarrays fabrication [98]. General strategies are for example: adsorption, covalent linking, and non-covalent interactions (mainly biotin-streptavidin). Adsorption by electrostatic interactions is one of the methods used, since DNA can bind to positively charged matrices, such as chitosan, or charged polymers such as poly(allylamine)-hydrochloride (PAAH)/sodium poly(styrenesulfonate) (PSS)/PAAH films. Adsorption on carbon and electroadsorption can also be performed using DNA or other probes. Covalent linking is best obtained using high-yield reactions forming stable bonds. Some of the most used reactions are reported in Fig. 4.6.

Thiol-modified probes can be linked to gold surfaces directly (Fig. 4.6a). On the contrary, maleimide-modified probes can react with thiols by conjugated addition (Fig. 4.6b).

Isothiocyanate and N-hydroxysuccinimidyl active ester groups can easily be coupled with nucleophilic groups, especially amines (Fig. 4.6c, d). Aldehyde moieties give rise to Schiff bases and oximes, the latter being more robust chemically. Reduction of Schiff bases to amines is a good way to increase the stability of biointerphases (Fig. 4.6e) [99]. Again click chemistry can efficiently be used in this context. Azido and alkyne groups can be used to form new bonds via copper catalyzed Huisgens 1,3-cycloaddition.

Linkage of LNA is essentially the same as oligonucleotides, due to similarity in their chemical structure. PNAs have been immobilized using Huisgen cycloaddition between azide surfaces and acetylene-carrying PNA [100], thiolated PNAs on gold or maleimide surfaces, [101–103] amino-capped PNAs on isothiocyanate, [104] succinimidyl ester, or aldehyde surfaces [99], and on surfaces partially prefunctionalized with cDNA [105].



Fig. 4.6 Main chemistries used for probe deposition. (a) Thiols linked to gold surfaces; (b) thiolated residues on the surface reacted with maleimide-conjugated probe; (c) amide formation from carboxylic acid and a condensing agent (*left*) or from N-hydroxysuccinimidyl ester (*right*); (d) reaction of an amine-modified probe with a isothiocyanate group on the surface; (e) reaction of amine-modified probe with aldehyde residues forming a Schiff base (*center*) which can then be reduced using borohydride derivatives to a more stable amine group; (f) an example of click chemistry with alkyne-modified probe and azide-modified surface by a copper(I) catalyzed Huisgen cycloaddition

MOs modified with terminal amino groups have been coupled to sensor surfaces using reaction with aldehyde groups.

The same type of chemistries could be used for the covalent linking of ON analogs to nanostructures such as nanoparticles, nanorods, and quantum dots. However, in many cases, non covalent affinity binding can be used [106]. Adsorption of streptavidine on surfaces allows to use biotinylated probes, which can then be directly linked to the surface thanks to the very high stability constant of the biotin-streptavidin complex ($K = 10^{15}$).

4.3.2 Fluorescent Probes for In Situ and Single Molecule Detection

The most direct method for visualizing the presence of a specific sequence within a sample of interest is to hybridize its genomic DNA with fluorescently labeled probes and directly observing the signal obtained after removal of the excess of unbound probe. This is particularly interesting in the observation of tissues and single cells and is the basis of *in situ* hybridization techniques. Since targets can be very diluted in these analysis, it is not surprising that either long ON probes or ON analogs showing very high DNA affinity, such as PNA, have been used in these applications.

Fluorescence in situ hybridization (FISH) is a technique that uses fluorescent probes able to bind to only those parts of the sequence target with which they show a high degree of sequence complementarity. Fluorescence microscopy can subsequently be used for detection, after washing of unhybridized probes [107]; high-sensitivity detection, simultaneous assay of multiple species and automated data collection and analysis have advanced this field significantly [108]. Thus, FISH has become a key investigation procedure in the analysis of nucleic acids, with a significant impact on diagnostics, therapeutics and biomedical research in general. In particular, last years have seen important efforts in developing novel approaches in ultrasensitive detection of DNA sequence targets. For example, some research groups exploited the possibility of using PNA probes in order to realize PNA-FISH systems, taking advantage from PNA compatibility with a wide range of fluorochromes and its capacity to hybridize with high affinity and specificity to complementary sequences. In such a context, Agerholm et al. [108] proposed sequential FISH analyses of human blastomeres using PNA probes. After fixation of the blastomeres on poly-L-lysine-coated glass slides, FISH was carried out using labeled PNA probes and then monitoring the signal by fluorescence microscopy, pointing out the possibility to make up to five sequential FISH cycles without loss of signal. An analogue system based on PNA-FISH was also proposed for the analysis of human oocytes, polar bodies and preimplantation embryos [109].

PNA probes targeting specific rRNA sequences of yeast and bacteria with clinical, environmental, and industrial value have recently been developed and applied to a variety of rapid assay formats [110].

In a different approach, FISH can be combined with flow cytometry, in the socalled Flow-FISH technique. Flow cytometry consists in a procedure for counting and examining microscopic particles, such as cells and chromosomes, by suspending them in a stream of fluid and passing them by an electronic detection apparatus: in Flow-FISH, this approach is combined with the use of labeled probes in order to collect data in fluorescence spectroscopy through the probe-target hybridization. In particular, many works have been published about the use of labeled PNA probes in such a procedure, as reported, for example, by Brind'Amour et al. [111] who exploited the use of a PNA-based Flow-FISH in the analysis of repetitive DNA in chromosome and by Baerloche et al. [112] who reported a similar protocol to measure the average length of telomere repeats in cells. Recently, Robertson et al. [113] suggested the use of LNA-modified oligonucleotide probes, instead of the typical PNA probes, describing a method based on flow–FISH for the specific, high-sensitivity, low-background detection of mRNA, emphasizing the applicability of LNA-FISH techniques in detecting target sequences.

Alternatively to Flow-FISH and traditional procedures performed after fixation of the sample on a slide, FISH can also be carried out in suspension (S-FISH), with the sample placed on a polished concave slide just as the final step of the procedure [114]. For example, Vandewoestyne et al. [115] made use of S-FISH combined with automatic detection and laser capture microdissection (LCM) for profiling of male cells in male/female mixtures. In another approach, FISH can be performed exploiting the properties of quantum dots (QDs) [116]. In particular, various method for direct QD labeling of oligonucleotide probes have been developed, in order to make possible FISH approaches, for example, in detection of breast-cancer cells, in which nanocrystal fluorophores [117]. QDs have also been used as FISH labels for a Y-chromosome-specific sequence in human-sperm cells [118] and for the construction of QD-labeled DNA probes for detecting specific sequences in *E.coli* plasmids [119].

Finally, it is worth to say that FISH can also be performed through another protocol, the so-called Fiber-FISH, which allows high resolution mapping chromatine or DNA genes and chromosomal regions on fibers. In this way it is possible to obtain high resolution mappings from some hundreds to the limit of 1 kb [120, 121].

Single molecule fluorescence spectroscopy (SMFS) can be performed using highly localised light excitation of fluorophores, which can be obtained by confocal microscopy with very high numerical aperture (NA), by total internal reflection microscopy (TIRM) or by zero-mode waveguides (ZMW). SMFS allows to measure not only fluorescence intensity but also other parameters connected with fluorescence. These techniques have been reviewed recently and an excellent recent paper by Ranasinghe and Brown describes apparatus, materials and strategies for DNA analysis by single molecule detection [122].

One problem of these techniques is that using very small volumes for detection, relatively high concentrations are needed in order to have even a single molecule in the measuring volume, and high measuring times are required in order to sample a sufficient number of molecules for statistically significant results. The combination of these measurement with microfluidics and recycling of the solutions can greatly reduce the time required and lower the detection limits; recirculation of the sample through a nL chamber and micro-evaporation led to detection of ssDNA at initial concentrations as low as 50 aM in an acquisition time of 100 s after a preconcentration step [123].

In order to avoid read-out of fluorescent probes, single molecule two-colour coincidence detection (TCCD), in which two fluorophores are simultaneously detected as a function of time and only coincident burst are counted [124]. Ultrasensitive detection of oligonucleotides with 50 fM limit of detection even in the presence of 1,000 fold excess of unrelated DNA was achieved using two color (Rhodamine green and Alexa Fluor 647) single molecule fluorescence coincidence detection



Fig. 4.7 (a) Result of the PNA-FISH technique applied to a male blastomere. Three different types of chromosomes are colored in *red, green* and *blue* by using specific PNA probes labeled with different fluorophores (Ref. [108], by permission of Oxford University Press) (b) Fluorescence bursts (expansion between 3,000 and 5,000 ms) on both the *blue*- and *red*-excited channels for a 50 pM solution of dual-labeled 40-base duplex DNA. Coincident bursts on both channels are marked with an *asterisk* (Reprinted with permission from Ref. [125]. Copyright 2001 American Chemical Society)

[125]. Hybridization detection was achieved by the simultaneous presence of signal burst in both channels (Fig. 4.7b), which is unlikely to occur by coincidence in very dilute solution (single molecule regime) for unpaired ON. In this case the target DNA was labeled, but the most convenient way for detection of non-fluorescent nucleic acids targets is to use of two fluorescent probes which co-localize upon hybridization to the same unlabelled target molecule, or give rise to FRET effect. Castro and Williams first described this approach for amplification-free detection of genomic DNA, using two 15mer peptide nucleic acid (PNA) probes [126, 127]. This procedure was then extended to the detection of cDNA produced by reverse transcription [128], or to mRNA [129].

In order to remove background fluorescence, an excess of quencher-labelled oligonucleotides can be used, as shown in the detection of microRNA (miR) using LNA/DNA mixmers [130].

Other properties can be used to distinguish between free and bound probes, as recently demonstrated by Hook and coworkers in the detection of unlabelled DNA using residence time of oligonucleotide-modified unilammelar vesicles in a sandwich-type assay with TIR fluorescence microscopy (TIRFM) [131].

These advanced fluorescent-based techniques become more efficient when using self-reporting probes described in the next paragraph [132, 133].

4.3.3 Self-Reporting Probes

Molecular beacons (MBs) are single-stranded oligonucleotide probes that contain a loop-and-stem structure [134]. The loop portion of the beacon contains the



Fig. 4.8 (a) Example of a PNA molecular beacon (*PNA MB*) used in flow cytometry for the detection of E. Coli, in comparison with a DNA beacon (*DNA MB*) of the same size (Reprinted with permission from Ref. [136]. Copyright 2003 American Society for Microbiology). (b) Schematic illustration of the LNA-MBs-based technique described by Martinez et al. When the complementary target binds to the beacon, the fluorophore is separated from the quencher and its fluorescence is restored (Reprinted with permission from Ref. [33]. Copyright 2009 American Chemical Society)

sequence complementary to a target (usually 15–30 bases), whereas the two stems are complementary to each other (5–7 bases). A quencher is covalently attached at the end of one stem, while a fluorophore is covalently attached to the end of the other. DABCYL and the more effective black hole quencher (BHQ) can be used, but also gold nano-particles and gold surfaces have been successfully applied. Because the fluorophore and quencher are in close proximity, MBs do not fluoresce until hybridization with the target nucleic acid sequence. In the presence of a target, the hairpin structure of MBs undergoes conformational changes that force the fluorophore and quencher moieties to separate, thus restoring fluorescence intensity. The use and design of MB probes in optical detection of DNA are described in details in another chapter of this book [135], and only few relevant examples involving modified DNA structures will be discussed here.

PNA beacons were described by several groups using a stemless design, in order to avoid excessive stability of the stem tract, since PNA:PNA duplexes are far more stable than DNA:DNA ones. Frank-Kamenetskii and co-workers described the performances of these beacons and compared them with ON beacons, showing improved performances. PNA beacons were used in FISH, flow cytometry [136] (Fig. 4.8a) and HPLC, [137] to generate specific signals.

Martinez et al. [33] recently experimented LNA-based beacons attached on a glass surface, obtaining detection limits for the DNA sequence targets in the low nanomolar range.

Smart probes use guanine as quencher and therefore incorporation of a single fluorophoric unit is necessary in the synthetic process [138]. Other switching mechanism can be used, in particular, fluorescence resonance energy transfer



Fig. 4.9 On the *left*, a schematic illustration of the formation of the probe-target duplexes; the term "energetic hotspot" refers to the particular zipper arrangement of the probe. On the *right*, 2'-N-(pyren-1-yl)methyl-2'-amino- α -L-LNA used as intercalator-modified-LNA monomer

(FRET), excimer formation, and forced intercalation [139], and DNA-templated reactions can generate a fluorescent signal [140].

Pyrene units are well known for their tendency to form excimers, i.e. excited state dimers, which emit light to longer wavelength than the isolated fluorophore, and the pyrene unit has been extensively used in the design of ON derivatives [141].

In this context, Sau et al. [31] recently proposed *invader* LNA probes, i.e. DNA duplexes with *"interstrand zipper arrangements"* of intercalator-functionalized 2'-amino- α -L-LNA monomers, demonstrating their ability to recognize short sequence dsDNA targets, precisely directing the appendend intercalators into the core of DNA duplexes. These invader LNA probes take advantage of the "nearest neighbour exclusion principle", i.e. intercalating units of invader LNA monomers are poorly accomodated in probe duplexes but extremely well tolerated in probe-target duplexes, whose formation is thus favored by thermodynamic reasons. The detection of probe-target duplexes was eventually carried out by steady-state fluorescence spectroscopy, thanks to the pyrene substituents used as intercalators. Infact, during the recognition of the dsDNA targets, the pyrene moieties of the probes are forced apart, causing a decay of the intensity of the excimer signals at their typical emission wavelength (Fig. 4.9).

Intercalating probes in which a fluorophore is either switched on or off by interaction of the oligonucleotide with the DNA target can also be used. Thiazole orange is particularly suited, since its fluorescent properties change dramatically upon interaction with nucleic acids [142]. PNA probes bearing thiazole orange tethered at their end have been described by Kubista et al. and proven to be very efficient: these probes are commercialized as Light-up probes [143].

Reacting probes have been described, using DNA template reactions to generate a fluorescent signal [140]. These probes have the advantage to accumulate signal during time, if the reaction is designed in order to have catalytic activity by DNA or RNA, but of course is not reversible and is better suited for in vivo studies or FISH than for sensor technologies [144].

4.3.4 Surface and Nanostructure Architectures Using Advanced Probes

Probes must be designed also according to the devices used in sensory systems, and appropriate linking group and linking strategies has to be considered. Thus a brief description of most common sensor architecture is provided in this paragraph.

The principal types of formats used for sensor development rely on surface chemistry and architectures, and are depicted in Fig. 4.10. A sensing surface can act as simple support (as in the case of microarrays) or as an active sensing element (which is the case in most sensory systems). Probes can be adsorbed on the surface, and this approach has been sometimes used on glass surfaces, especially in the field of optical fibers. Adsorbtion of streptavidine allows to obtain activated surfaces which can be linked to biotinylated probes with non-covalent, yet very stable, bonds. If the probes are chemically attached on the surface, the latter must be suitable for linking of specific functional groups or must be activated by introduction of appropriate functional groups. Probes can be directly linked to the surface, but normally a 15-25 atom spacer should be added to the tail of the probe in order to allow sufficient space for hybridization, as tested for PNA microarrays [145]. Direct linking is necessary when the direct readout of the hybridization is necessary, such as in carbon nanotube electrochemical devices allowing the detection of single molecule target DNA [146]. In order to avoid probe overcrowding, dilute solutions are used for deposition, and normally the empty space on the surface is filled with a diluter molecule, thus forming a membrane-like monolayer (Fig 4.10a). 2D architectures can be obtained by covering a self assembled monolayer (SAM) with a second layer bearing the probes, thus obtaining a spatially controlled disposition in which the spacing between the surface and the probes can be controlled at the nanometer scale, thus allowing complete prediction of the physical properties of the surface assembly (Fig. 4.10b). A typical example of this approach is described by Knoll and coworkers in another chapter of this book using biotinylated thiols and a second layer of streptavidin molecules, which in turn are linked to biotinylated probes [147]. In alternative, a binding matrix bearing suitable functional groups can be linked to the surface, and probes can be attached to this matrix. This approach allows to have 3D instead of 2D architectures, in which the number of captured targets is enhanced without overcrowding (Fig. 4.10c). Polymers and polysaccharides (e.g. dextran, chitosan) can be used as surface layers. These matrices should be highly hydrophilic in order to be permeable to aqueous phase in which the target DNA molecules are dissolved. Hydrogels are especially suited as 3D matrices, since they are able to swell and shrink by change of the solvent, with a concentration effect which can enhance sensitivity considerably [148].

Gold surfaces are particularly suitable as sensor surface not only for their optical or electrochemical properties, but also for their ability to bind thiolated molecules. Thiolated probes can be deposited, followed by deposition of a short alkanethiol such as mercaptohexanol (MCH) which can prevent non-specific Au–DNA interaction. The introduction of a third "backfiller" bearing a carboxylic acid group has been claimed to greatly enhance the performances of sensors [149].



Fig. 4.10 Schematic representation of surface architectures used in sensor technology. (a) Spacerbearing probes directly linked to the surface with a diluter molecule; (b) Probes attached to a functional element at the end of a self-assembled monolayer (*SAM*); (c) probes linked to a 3D matrix enabling higher loading; (d) two examples of read-out: on the *left* the recognition element is directly read, on the *right* a sandwich-type assay in which the captured DNA is bound to a second oligonucleotide linked to a nanostructure for signal amplification

Glass surfaces can be modified using trialkoxysilanes with suitable amino or thiol groups, and the activated surface is then reacted with bifunctional molecules to which the probes can be attached. Silicon surfaces are first oxidized to produce silanolic groups and then modified as silica surfaces. Carbon-based layers, such as carbon nanotubes and graphene can be obtained with appropriate (i.e. carboxylic) groups by oxidation of the material itself or by transformation of precursor materials (e.g. chemically reduced graphene oxide or electrochemically reduced graphene oxide) [150].

Paper and polymeric materials are also interesting surfaces, since the development of low cost sensing devices can be envisaged, [151] and many chemical methods allow proper derivatization of these materials.

Finally, the readout process can be direct or the target molecule can bind polymeric structures or probe-labeled nanostructures (nanoparticles, quantum dots, dendrimers, unilamellar vesicles, SWCN, etc.), in a sandwich-type assay, which also can employ the specific probes above reported as a second recognition element (Fig. 4.10d). These decoration strategies will be described in other chapters of this book [10, 152].

4.4 Currently Used Probes in Ultrasensitive Techniques and PCR-Free Detection of Genomic DNA

Although the above reported examples of probes and probe-surface architectures have been described for oligonucleotide mimicking the DNA target even at very low copy number, or single molecule, only a few works have been published showing actual genomic DNA to demonstrate the use of the techniques for direct detection without amplification. One of the reason for this is the difficulty to target double stranded DNA by displacing one of the two strands, though strategies for overcome this problem have been developed, several examples reported in the text together with further examples are reported in Table 4.1 [153–161] for DNA probes and in Table 4.2 [162–164] for other probes.

DNA molecules have been the working horse for the development of most genosensor technologies and thus the majority of methods rely on these type of probes. The possibility to have enzymatic degradation of DNA:RNA hybrids have been the bases of a very efficient sensory system based on depletion of surface RNA probes, as described by Corn and coworkers [162]. Artificial probes cannot be easily used in combination with enzymatic reactions.

probes	10003				
Probe	Method	Limit of detection	Ref.		
DNA	FISH	See references	[107] and references therein		
DNA	Single molecule	See references	[122–125, 131–133]		
DNA	Piezoelectric sensing	40,000 copies	[153]		
DNA	Fluorescence	20 copies, 3 zM	[154]		
DNA	QCM	50 copies 0.17 aM	[155]		
DNA	Total internal reflection ellipsometry	10-1,000 μg/mL for genomic 1 fg/mL for single gene	[156]		
DNA	Gold nanoparticle-based detection By evanescent wave light	7.5×10^7 target copies	[157]		
DNA	Colorimetric scatter of gold nanoparticle probes	66 pg/μL, 20,000 copies (33 zmoles)	[158]		
DNA	Colorimetric scatter of gold nanoparticles	150,000 copies (500 ng of total genomic DNA)	[159]		
DNA	Colorimetric scatter of gold nanoparticle probe	2.9 amoles0.5 μg of unamplified RNA	[160]		
DNA	Nanoparticle-amplified SPR imaging-polyadenylation reactions	10 fM (5 amoles)	[161]		

 Table 4.1
 Selected examples of unamplified genomic DNA/RNA detection using DNA probes

Probe	Method	Limit of detection	Ref.
RNA	SPR imaging with ablation of the probes by RNase H	1 fM (oligonucleotides) 7 fM (genomic) or 5.8 μg/mL	[162]
LNA	Flow-FISH	n.r.	[113]
LNA	Single-molecule detection	500 fM	[130]
PNA	QCM with RecA enhancement	8.6 pg/L	[163]
PNA	Single-molecule detection	Minimum 1,200 molecules See references	[126–129]
PNA	Surface acoustic wave	1.21 pg/L	[164]
PNA	SPR imaging with nanoparticles	41 zM unamplified DNA	[57, 58]
PNA	FISH	See references	[108–111]
PNA	Enzymatic with colorimetric detection	10 zmole	[54]

Table 4.2 Selected examples of unamplified genomic DNA/RNA detection using other probes

n.r not reported

However, in sensory systems, the use of modified oligonucleotides has been increasingly important and some of these has been proved to be suitable for unamplified DNA detection. PNAs and LNAs are good candidates for this type of strategy, and this justifies the appearance of more and more examples of methods using these probes instead of DNA. MOs have also been demonstrated to be efficient probes, though their use has not be reported so far for real genomic DNA detection.

4.5 Conclusion

Among the factors which allowed ever improving performances of sensing techniques, the use of chemically modified and properly modified probes is of major importance. Though most of the works reported in early studies took advantage of the availability of oligonucleotide probes, recent works more and more rely on improved probe structures. The combination of advanced techniques and highly selective molecular recognition events, such as those obtained by PNA, LNA and MO, will probably allow DNA sensing techniques to surpass the required threshold for becoming widespread large-scale items, thus enabling DNA detection to be used in simple, affordable, and delocalized analysis for personalized and point-of-care diagnostics, even in limited resources areas.

The quest for even higher affinities and specificity by chemically modification of probes is still open and the so far obtained structures have still not be fully explored in diagnostics. Thus, research in this field is one of the most interesting challenges for application of our ability to synthesize bio-inspired biomaterials.

References

- Landegren, U., Dahl, F., Nilsson, M., Fredriksson, S., Banér, J., Gullberg, M., Jarvius, J., Gustafsdottir, S., Söderberg, O., Ericsson, O., Stenberg, J., Schallmeiner, E.: Padlock and proximity probes for *in situ* and array-based analyses: tools for the post-genomic era. Comp. Funct. Genom. 4, 525–530 (2003)
- Bashir, R.: BioMEMS: state-of-the-art in detection, opportunities and prospects. Adv. Drug Deliv. Rev. 56, 1565–1586 (2004)
- Khanna, V.K.: Existing and emerging detection technologies for DNA (Deoxyribonucleic Acid) finger printing, sequencing, bio- and analytical chips: a multidisciplinary development unifying molecular biology, chemical and electronics engineering. Biotechnol. Adv. 25, 85–98 (2007)
- 4. Li, Y., Di Naro, E., Vitucci, A., Zimmermann, B., Holzgreve, W., Hahn, S.: Detection of paternally inherited fetal point mutations for β-thalassemia using size-fractionated cell-free DNA in maternal plasma. J. Am. Med. Assoc. 293, 843–849 (2005)
- Audrezet, M.P., Dabricot, A., Le Marechal, C., Ferec, C.: Validation of high-resolution DNA melting analysis for mutation scanning of the cystic fibrosis transmembrane conductance regulator (CFTR) gene. J. Mol. Diagn. 10, 424–434 (2008)
- Garcea, G., Neal, C.P., Pattenden, C.J., Steward, W.P., Berry, D.P.: Molecular prognostic markers in pancreatic cancer: a systematic review. Eur. J. Cancer 41, 2213–2236 (2005)
- Greenman, C., et al.: Patterns of somatic mutation in human cancer genomes. Nature 446, 153–158 (2007)
- Homer, N., Szelinger, S., Redman, M., Duggan, D., Tembe, W., Muehling, J., Pearson, J.V., Stephan, D.A., Nelson, S.F., Craig, D.W.: Resolving individuals contributing trace amounts of DNA to highly complex mixtures using high-density SNP genotyping microarrays. PLoS Genet. 4, e1000167 (2008)
- Suekawa, Y., Aihara, H., Araki, M., Hosokawa, D., Mannen, H., Sasazaki, S.: Development of breed identification markers based on a bovine 50 K SNP array. Meat Sci. 85, 285–288 (2010)
- Lim, D.K., Kumar, A., Nam, J.M.: Engineered nanostructures for the ultrasensitive DNA detection, Chapter 3. In: Spoto, G., Corradini, R. (eds.) Detection of Non-amplified Genomic DNA, pp. 67–87. Springer, Dordrecht (2012)
- 11. Caruthers, M.H.: Chemical synthesis of DNA and DNA analogs. Acc. Chem. Res. 24, 278–284 (1991)
- Demidov, V.V., Frank-Kamenetskii, M.D.: Two sides of the coin: affinity and specificity of nucleic acid interactions. Trends Biochem. Sci. 29, 62–71 (2004)
- Marky, L.A., Breslauer, K.J.: Calculating thermodynamic data for transitions of any molecularity from equilibrium melting curves. Biochemistry 26, 1601–1620 (1987)
- SantaLucia Jr., J., Hicks, D.: The thermodynamics of DNA structural motifs. Annu. Rev. Biophys. Biomol. Struct. 33, 415–440 (2004)
- Erickson, D., Li, D., Krull, J.: Modeling of DNA hybridization kinetics for spatially resolved biochips. Anal. Biochem. 317, 186–200 (2003)
- Yu, F., Yao, D., Knoll, W.: Oligonucleotide hybridization studied by a surface plasmon diffraction sensor (SPDS). Nucleic Acids Res. 32, e752 (2004)
- Scrimin, P., Prins, L.J.: Sensing through signal amplification. Chem. Soc. Rev. 40, 4499–4505 (2011)
- Ermini, M.L., Scarano, S., Bini, R., Banchelli, M., Berti, D., Mascini, M., Minunni, M.: A rational approach in probe design for nucleic acid-based biosensing. Biosens. Bioelectron. 26, 4785–4790 (2011)
- Corradini, R., Sforza, S., Tedeschi, T., Marchelli, R.: Chirality as a tool in nucleic acid recognition: principles and relevance in biotechnology and in medicinal chemistry. Chirality 19, 269–294 (2007)

4 Advanced Molecular Probes for Sequence-Specific DNA Recognition

- Tamm, I.: Antisense therapy in malignant diseases: status quo and quo vadis? Clin. Sci. 110, 427–442 (2006)
- Singh, S.K., Koshkin, A.A., Wengel, J., Nielsen, P.: LNA (locked nucleic acids): synthesis and high-affinity nucleic acid recognition. Chem. Commun. 455–456 (1998)
- Koshkin, A.A., Singh, S.K., Nielsen, P., Rajwanshi, V.K., Kumar, R., Meldgaard, M., Olsen, C.E., Wengel, J.: LNA (locked nucleic acids): synthesis of the adenine, cytosine, guanine, 5-methylcytosine, thymine and uracil bicyclonucleoside monomers, oligomerisation, and unprecedented nucleic acid recognition. Tetrahedron 54, 3607–3630 (1998)
- Obika, S., Nanbu, D., Hari, Y., Morio, K., In, Y., Ishida, T., Imanishi, T.: Synthesis of 2'-O,4'-C-methyleneuridine and -cytidine. Novel bicyclic nucleosides having a fixed C3'-endo sugar puckering. Tetrahedron Lett. 38, 8735–8738 (1997)
- Karkare, S., Bhatnagar, D.: Promising nucleic acid analogs and mimics: characteristic features and applications of PNA, LNA, and morpholino. Appl. Microbiol. Biotechnol. 71, 575–586 (2006)
- 25. Sørensen, M.D., Kværnø, L., Bryld, T., Håkansson, A.E., Verbeure, B., Gaubert, G., Herdewijn, P., Wengel, J.: α-l-ribo-configured locked nucleic acid (α-L-LNA): synthesis and properties. J. Am. Chem. Soc. **124**, 2164–2176 (2002)
- Wengel, J.: Synthesis of 3'-C- and 4'-C-branched oligodeoxynucleotides and the development of locked nucleic acid (LNA). Acc. Chem. Res. 32, 301–310 (1999)
- Kumar, R., Singh, S.K., Koshkin, A.A., Rajwanshi, V.K., Meldgaard, M., Wengel, J.: The first analogues of LNA (Locked Nucleic Acids): phosphorothioate-LNA and 2'-thio-LNA. Bioorg. Med. Chem. Lett. 8, 2219–2222 (1998)
- Vester, B., Wengel, J.: LNA (locked nucleic acid): high-affinity targeting of complementary RNA and DNA. Biochemistry 43, 13233–13241 (2004)
- 29. Petersen, M., Wengel, J.: LNA: a versatile tool for therapeutics and genomics. Trends Biotechnol. 21, 74–81 (2003)
- Buchini, S., Leumann, C.J.: Recent improvements in antigene technology. Curr. Opin. Chem. Biol. 7, 717–726 (2003)
- Sau, S.P., Kumar, T.S., Hrdlicka, P.J.: Invader LNA: efficient targeting of short double stranded DNA. Org. Biomol. Chem. 8, 2028–2036 (2010)
- Laschi, S., Palchetti, I., Marrazza, G., Mascini, M.: Enzyme-amplified electrochemical hybridization assay based on PNA, LNA and DNA probe-modified micro-magnetic beads. Bioelectrochemistry 76, 214–220 (2009)
- Martinez, K., Estevez, M.-C., Wu, Y., Phillips, J.A., Medley, C.D., Tan, W.: Locked nucleic acid based beacons for surface interaction studies and biosensor development. Anal. Chem. 81, 3448–3454 (2009)
- 34. Stein, D., Foster, E., Huang, S.B., Weller, D., Summerton, J.: A specificity comparison of four antisense types: morpholino, 2'-O-methyl RNA, DNA, and phosphorothioate DNA. Nucleic Acid Drug Dev. 7, 151–157 (1997)
- Summerton, J., Weller, D.: Morpholino antisense oligomers: design, preparation, and properties. Antisense Nucleic Acid Drug Dev. 7, 187–195 (1997)
- Ouyang, X., Shestopalov, I.A., Sinha, S., Zheng, G., Pitt, C.L.W., Li, W., Olson, A.J., Chen, J.K.: Versatile synthesis and rational design of caged morpholinos. J. Am. Chem. Soc. 131, 13255–13269 (2009)
- Parra, M.K., Gee, S., Mohandas, L., Conboy, J.G.: Efficient *in vivo* manipulation of alternative pre-mRNA splicing events using antisense morpholinos in mice. J. Biol. Chem. 286, 6033– 6039 (2010)
- 38. Heasman, J.: Morpholino oligos: making sense of antisense? Dev. Biol. 24, 209–214 (2002)
- Shestopalov, I.A., Chen, J.K.: Oligonucleotide-based tools for studying zebrafish development. Zebrafish 7, 31–40 (2010)
- Liu, Y., Irving, D., Qiao, W., Ge, D., Levicky, R.: Kinetic mechanisms in morpholino–DNA surface hybridization. J. Am. Chem. Soc. 133, 11588–11596 (2011)
- Gong, P., Wang, K., Liu, Y., Shepard, K., Levicky, R.: Molecular mechanisms in morpholino DNA surface hybridization. J. Am. Chem. Soc. 132, 9663–9671 (2010)

- 42. Tercero, N., Wang, K., Gong, P., Levicky, R.: Morpholino monolayers: preparation and labelfree DNA analysis by surface hybridization. J. Am. Chem. Soc. **131**, 4953–4961 (2009)
- Gao, Z., Ting, B.P.: A DNA biosensor based on a morpholino oligomer coated indium-tin oxide electrode and a cationic redox polymer. Analyst 134, 952–957 (2009)
- 44. Zhang, G., Luo, Z.H.H., Huang, M.J., Tay, G.K.I., Lim, E.A.: Morpholino-functionalized silicon nanowire biosensor for sequence-specific label-free detection of DNA. Biosens. Bioelectron. 25, 2447–2453 (2010)
- Nielsen, P.E., Egholm, M., Berg, R.H., Buchardt, O.: Sequence-selective recognition of DNA by strand displacement with a thymine-substituted polyamide. Science 254, 1497–1500 (1991)
- 46. Egholm, M., Buchardt, O., Christensen, L., Behrens, C., Freier, S.M., Driver, D.A., Berg, R.H., Kim, S.K., Norden, B., Nielsen, P.E.: PNA hybridizes to complementary oligonucleotides obeying the Watson-Crick hydrogen-bonding rules. Nature 365, 566–568 (1993)
- Ratilainen, T., Holmén, A., Tuite, E., Nielsen, P.E., Nordén, B.: Thermodynamics of sequence-specific binding of PNA to DNA. Biochemistry 39, 7781–7791 (2000)
- Komiyama, M., Aiba Y., Yamamoto, Y., Sumaoka J.: Artificial restriction DNA cutter for site-selective scission of double-stranded DNA with tunable scission site and specificity, Nat. Protoc. 3, 655–662 (2008)
- 49. Demidov, V.V.: PNA and LNA throw light on DNA. Trends Biotechnol 21, 4-7 (2003)
- Ren, B., Zhou, J.M., Komiyama, M.: Straightforward detection of SNPs in double-stranded DNA by using exonuclease III/nuclease S1/PNA system. Nucleic Acids Res. 32, e42 (2004)
- Gaylord, B.S., Massie, M.R., Feinstein, S.C., Bazan, G.C.: SNP detection using peptide nucleic acid probes and conjugated polymers: applications in neurodegenerative disease identification. Proc. Natl. Acad. Sci. USA 102, 34–39 (2005)
- 52. Tomac, S., Sarkar, M., Ratilainen, T., Wittung, P., Nielsen, P.E., Nordén, B., Graslund, A.: Ionic effects on the stability and conformation of peptide nucleic acid complexes. J. Am. Chem. Soc. 118, 5544–5552 (1996)
- Demidov, V.A., Potaman, V.N., Frank-Kamenetskii, M.D., Egholm, M., Buchardt, O., Sonnichsen, S.H., Nielsen, P.E.: Stability of peptide nucleic-acids in human serum and cellular-extracts. Biochem. Pharm. 48, 1310–1313 (1994)
- Zhang, N., Appella, D.H.: Colorimetric detection of anthrax DNA with a peptide nucleic acid sandwich-hybridization assay. J. Am. Chem. Soc. 129, 8424–8425 (2007)
- Zhang, N., Appella, D.H.: Advantages of peptide nucleic acids as diagnostic platforms for detection of nucleic acids in resource-limited settings. J. Infect. Dis. 201, S42–S45 (2010)
- Sforza, S., Corradini, R., Tedeschi, T., Marchelli, R.: Food analysis and food authentication by peptide nucleic acid (PNA)-based technologies. Chem. Soc. Rev. 40, 221–232 (2011)
- D'Agata, R., Corradini, R., Grasso, G., Marchelli, R., Spoto, G.: Ultrasensitive detection of DNA by PNA and nanoparticle-enhanced surface plasmon resonance imaging. Chembiochem 9, 2067–2070 (2008)
- 58. D'Agata, R., Corradini, R., Ferretti, C., Zanoli, L., Gatti, M., Marchelli, R., Spoto, G.: Ultrasensitive detection of non-amplified genomic DNA by nanoparticle-enhanced surface plasmon resonance imaging. Biosens. Bioelectron. 25, 2095–2100 (2010)
- Sahu, N., Shilakari, G., Nayak, A., Kohli, D.H.: Antisense technology: a selective tool for gene expression regulation and gene targeting. Curr. Chem. Biol. 2, 110–121 (2008)
- 60. Nielsen, P.E., Egholm, M.: Peptide Nucleic Acids: Protocols and Applications, 2nd edn. Horizon Press, Wymondham (2004)
- Ganesh, K.N., Nielsen, P.E.: Peptide nucleic acids: analogs and derivatives. Curr. Org. Chem. 4, 931–943 (2000)
- Falkiewicz, B.: Peptide nucleic acids and their structural modifications. Acta Biochim. Pol. 46, 509–529 (1999)
- Lohse, J., Dahl, O., Nielsen, P.E.: Double duplex invasion by peptide nucleic acid: a general principle for sequence-specific targeting of double-stranded DNA. Proc. Natl. Acad. Sci. USA 96, 11804–11808 (1999)

- 64. Nielsen, P.E., Haaima, G.: Peptide Nucleic acid (PNA). A DNA mimic with a pseudopeptide backbone. Chem. Soc. Rev. 26, 73–78 (1997)
- Corradini, R., Sforza, S., Tedeschi, T., Marchelli R.: Peptide nucleic acids (PNAs) and their analogs: synthesis and applications. Seminars in Organic Synthesis. Società Chimica Italiana 41–70 (2003)
- 66. Sforza, S., Galaverna, G., Dossena, A., Corradini, R., Marchelli, R.: Role of chirality and optical purity in nucleic acid recognition by PNA and PNA analogs. Chirality 14, 591–598 (2002)
- Menchise, V., De Simone, G., Tedeschi, T., Corradini, R., Sforza, S., Marchelli, R., Capasso, D., Saviano, M., Pedone, C.: Insights into peptide nucleic acid (PNA) structural features: the crystal structure of a D-lysinebased chiral PNA-DNA duplex. Proc. Natl. Acad. Sci. USA 100, 12021–12026 (2003)
- Sforza, S., Haaima, G., Marchelli, R., Nielsen, P.E.: Chiral peptide nucleic acids (PNAs): helix handedness and DNA recognition. Eur. J. Org. Chem. 197–204 (1999)
- Tedeschi, T., Sforza, S., Corradini, R., Dossena, A., Marchelli, R.: Lysinebased peptide nucleic acids (PNAs) with strong chiral constraint: control of the helix handedness and DNA binding by chirality. Chirality 17, S196–S204 (2005)
- 70. Sforza, S., Corradini, R., Ghirardi, S., Dossena, A., Marchelli, R.: DNA binding of a D-lysine-based chiral PNA: direction control and mismatch recognition. Eur. J. Org. Chem. 2905–2913 (2000)
- Sforza, S., Tedeschi, T., Corradini, R., Dossena, A., Marchelli R.: Direction control in DNA binding of chiral D-lysine-based peptide nucleic acid (PNA) probed by electrospray mass spectrometry. Chem. Commun. 1102–1103 (2003)
- 72. Corradini, R., Feriotto, G., Sforza, S., Marchelli, R., Gambari, R.: Enhanced recognition of cystic fibrosis W1282X DNA point mutation by chiral peptide nucleic acids probes by a surface plasmon resonance biosensor. J. Mol. Recognit. 17, 76–84 (2004)
- Tedeschi, T., Chiari, M., Galaverna, G., Sforza, S., Cretich, M., Corradini, R., Marchelli, R.: Detection of the R553X DNA single point mutation related to cystic fibrosis by a "chiral box" D-lysine-peptide nucleic acid probe by capillary electrophoresis. Electrophoresis 26, 4310–4316 (2005)
- 74. Englund, E.A., Appella, D.H.: γ-Substituted peptide nucleic acids constructed from L-lysine are a versatile scaffold for multifunctional display. Angew. Chem. Int. Ed. 46, 1414–1418 (2007)
- de Koning, M.C., Petersen, L., Weterings, J., Overhand, J.M., van der Marel, G.A., Filippov, D.V.: Synthesis of thiol-modified peptide nucleic acids designed for post-assembly conjugation reactions. Tetrahedron 62, 3248–3258 (2006)
- 76. Huang, H., Joe, G.H., Choi, S.R., Kim, S.N., Kim, Y.T., Pak, C.S., Hong, J.H., Lee, W.: Synthesis of enantiopure γ-glutamic acid functionalized peptide nucleic acid monomers. Bull. Korean Chem. Soc. **31**, 2054–2056 (2010)
- 77. Dragulescu-Andrasi, A., Rapireddy, S., Frezza, B.M., Gayathri, C., Gil, R.R., Ly, D.H.: A simple γ-backbone modification preorganizes peptide nucleic acid into a helical structure. J. Am. Chem. Soc. **128**, 10258–10267 (2006)
- Sforza, S., Tedeschi, T., Corradini, R., Marchelli, R.: Induction of helical handedness and DNA binding properties of peptide nucleic acids (PNAs) with two stereogenic centres. Eur. J. Org. Chem. 2007, 5879–5885 (2007)
- Calabretta, A., Tedeschi, T., Di Cola, G., Corradini, R., Sforza, S., Marchelli, R.: Argininebased PNA microarrays for APOE genotyping. Mol. Biosyst. 5, 1323–1330 (2009)
- Manicardi, A., Calabretta, A., Bencivenni, M., Tedeschi, T., Sforza, S., Corradini, R., Marchelli, R.: Affinity and selectivity of C2- and C5-substituted "chiral-box" PNA in solution and on microarrays. Chirality 22, E161–E172 (2010)
- Calabretta, A., Tedeschi, T., Corradini, R., Marchelli, R., Sforza, S.: DNA and RNA binding properties o fan arginine-base "extended chiral box" peptide nucleic acid. Tetrahedron Lett. 52, 300–304 (2011)

- Kumar, V.A., Ganesh, K.N.: Conformationally constrained PNA analogues: structural evolution towards DNA/RNA binding selectivity. Acc. Chem. Res. 38, 404–412 (2005)
- Ananthanawat, C., Vilaivan, T., Hoven, V.P.: Synthesis and immobilization of thiolated pyrrolidinyl peptide nucleic acids on gold-coated piezoelectric quartz crystals for the detection of DNA hybridization. Sens. Actuators B Chem. 137, 215–221 (2009)
- Ananthanawat, C., Vilaivan, T., Mekboonsonglarp, W., Hoven, V.P.: Thiolated pyrrolidinyl peptide nucleic acids for the detection of DNA hybridization using surface plasmon resonance. Biosens. Bioelectron. 24, 3544–3549 (2009)
- Vilaivan, T., Suprarpprom, C., Harnyyuttanakorn, P., Lowe, G.: Synthesis and properties of novel pyrrolidinyl PNA carrying β-amino acid spacers. Tetrahedron Lett. 42, 5533–5536 (2001)
- Vilaivan, T., Lowe, G.A.: Novel pyrrolidinyl PNA showing high sequence specificity and preferential binding to DNA over RNA. J. Am. Chem. Soc. 124, 9326–9327 (2002)
- 87. Govindaraju, T., Madhuri, V., Kumar, V.A., Ganesh, K.N.: Cyclohexanyl peptide nucleic acids (chPNAs) for preferential RNA binding. Tuning of dihedral angle β in PNAs for DNA/RNA discrimination. J. Org. Chem. **71**, 14–21 (2006)
- 88. Hall, N.: The quest for the chemical roots of life. Chem. Commun. 1247-1252 (2004)
- Schoning, K., Scholz, P., Guntha, S., Wu, X., Krishnamurthy, R., Eschenmoser, A.: Chemical etiology of nucleic acid structure: the alpha-threofuranosyl-(3'-2') oligonucleotide system. Science 290, 1347–1351 (2000)
- Horhota, A., Zou, K., Ichida, J.K., Yu, B., McLaughlin, L.W., Szostak, J.W., Chaput, J.C.: Kinetic analysis of an efficient DNA-dependent TNA polymerase. J. Am. Chem. Soc. 127, 7427–7434 (2005)
- Abramov, M., Schepers, G., Van Aerschot, A., Van Hummelenb, P., Herdewijn, P.: HNA and ANA high-affinity arrays for detections of DNA and RNA single-base mismatches. Biosens. Bioelectron. 23, 1728–1732 (2008)
- 92. Flanagan, W.M., Wolf, J.J., Olson, P., Grant, D., Lin, K.Y., Wagner, R.W., Matteucci, M.D.: A cytosine analog that confers enhanced potency to antisense oligonucleotides. Proc. Natl. Acad. Sci. USA 96, 3513–3518 (1999)
- Krueger, A.T., Kool, E.T.: Model systems for understanding DNA base pairing. Curr. Opin. Chem. Biol. 11, 588–594 (2007)
- Nakagawa, O., Ono, S., Li, Z., Tsujimoto, A., Sasaki, S.: Specific fluorescent probe for 8-oxoguanosine. Angew. Chem. Int. Ed. 46, 4500–4503 (2007)
- 95. Loakes, D.: The application of universal DNA base analogue. Nucleic Acids Res. 29, 2437–2447 (2001)
- 96. El-Sagheer, A.H., Brown, T.: Click chemistry with DNA. Chem. Soc. Rev. 39, 1388–1405 (2010)
- Hocek, M., Fojta, M.: Nucleobase modification as redox DNA labelling for electrochemical detection. Chem. Soc. Rev. 40, 5802–5814 (2011)
- Sassolas, A., Leca-Bouvier, B.D., Blum, L.J.: DNA biosensors and microarrays. Chem. Rev. 108, 109–139 (2008)
- Calabretta, A., Wasserberg, D., Posthuma-Trumpie, G.A., Subramaniam, V., van Amerongen, A., Corradini, R., Tedeschi, T., Sforza, S., Reinhoudt, D.N., Marchelli, R., Huskens, J., Jonkheijm, P.: Patterning of peptide nucleic acids using reactive microcontact printing. Langmuir 27, 1536–1542 (2011)
- 100. Lim, S.Y., Chung, W.Y., Lee, H.K., Park, M.S., Park, H.G.: Direct and nondestructive verification of PNA immobilization using click chemistry. Biochem. Biophys. Res. Commun. 376, 633–636 (2008)
- 101. Chelmowski, R., Koster, D., Kerstan, A., Prekelt, A., Grunwald, C., Winkler, T., Metzler-Nolte, N., Terfort, A., Woll, C.: Peptide-based SAMs that resist the adsorption of proteins. J. Am. Chem. Soc. 130, 14952–14953 (2008)
- 102. Mateo-Martí, E., Briones, C., Roman, E., Briand, E., Pradier, C.M., Martín-Gago, J.A.: Selfassembled monolayers of peptide nucleic acids on gold surfaces: a spectroscopic study. Langmuir 21, 9510 (2005)

- 103. Cattani-Scholz, A., Pedone, D., Blobner, F., Abstreiter, G., Schwartz, J., Tornow, M., Andruzzi, L.: PNA-PEG modified silicon platforms as functional bio-interfaces for applications in DNA microarrays and biosensors. Biomacromolecules **10**, 489–496 (2009)
- 104. Sun, C., Gaylord, B.S., Hong, J.W., Liu, B., Bazan, G.C.: Application of cationic conjugated polymers in microarrays using label-free DNA targets. Nat. Protoc. 2, 2148–2151 (2007)
- 105. Paul, A., Bezer, S., Venkatramani, R., Kocsis, L., Wierzbinski, E., Balaeff, A., Keinan, S., Beratan, D.N., Achim, C., Waldeck, D.H.: Role of nucleobase energetics and nucleobase interactions in single-stranded peptide nucleic acid charge transfer. J. Am. Chem. Soc. 131, 6498 (2009)
- Akbulut, O., Yu, A.A., Stellacci, F.: Fabrication of biomolecular devices via supramolecular contact-based approaches. Chem. Soc. Rev. 39, 30–37 (2010)
- Levsky, J.M., Singer, R.H.: Fluorescence in situ hybridization: past, present and future. J. Cell Sci. 116, 2833–2838 (2003)
- Agerholm, I.E., Ziebe, S., Williams, B., Berg, C., Crüger, D.G., Bruun Petersen, G., Kølvraa, S.: Sequential FISH analysis using competitive displacement of labelled peptide nucleic acid probes for eight chromosomes in human blastomeres. Hum. Reprod. 20, 1072–1077 (2005)
- 109. Paulasova, P., Andréo, B., Diblik, J., Macek, M., Pellestor, F.: The peptide nucleic acids as probes for chromosomal analysis: application to human oocytes, polar bodies and preimplantation embryos. Mol. Hum. Reprod. 10, 467–472 (2004)
- Stender, H., Fiandaca, M., Hyldig-Nielsen, J.J., Coull, J.: PNA for rapid microbiology. J. Microbiol. Methods 48, 1–17 (2002)
- 111. Brind'Amour, J., Lansdorp, P.M.: Analysis of repetitive DNA in chromosomes by flow cytometry. Nat. Methods **8**, 484–486 (2011)
- 112. Baerlocher, G.M., Vulto, I., De Jong, G., Lansdorp, P.M.: Flow cytometry and FISH to measure the average length of telomeres (flow FISH). Nat. Protoc. 1, 2365–2376 (2006)
- 113. Robertson, K.L., Thach, D.C.: LNA flow–FISH: a flow cytometry–fluorescence in situ hybridization method to detect messenger RNA using locked nucleic acid probes. Anal. Biochem. **390**, 109–114 (2009)
- 114. Steinhaeuser, U., Starke, H., Nietzel, A., Lindenau, J., Ullmann, P., Claussen, U., Liehr, T.: Suspension (S)-FISH, a new technique for interphase nuclei. J. Histochem. Cytochem. 50, 1697–1698 (2002)
- 115. Vandewoestyne, M., Van Hoofstat, D., Van Nieuwerburgh, F., Deforce, D.: Suspension fluorescence in situ hybridization (S-FISH) combined with automatic detection and laser microdissection for STR profiling of male cells in male/female mixtures. Int. J. Legal Med. 123, 441–447 (2009)
- Algar, W.R., Massey, M., Krull, U.J.: The application of quantum dots, gold nanoparticles and molecular switches to optical nucleic-acid diagnostics. Trends Anal. Chem. 28, 292–306 (2009)
- 117. Xiao, Y., Barker, P.E.: Semiconductor nanocrystal probes for human metaphase chromosomes. Nucleic Acids Res. **32**, e28 (2004)
- 118. Pathak, S., Choi, S., Arnheim, N., Thompson, M.E.: Hydroxylated quantum dots as luminescent probes for in situ hybridization. J. Am. Chem. Soc. **123**, 4103–4104 (2001)
- 119. Wu, S., Zhao, X., Zhang, Z., Xie, H., Tian, Z., Peng, J., Lu, Z., Pang, D., Xie, Z.: Quantumdot-labeled DNA probes for fluorescence in situ hybridization (FISH) in the microorganism *Escherichia coli*. Chem. Phys. Chem. **7**, 1062–1067 (2006)
- Volpi, E.V., Bridger, J.M.: FISH glossary: an overview of the fluorescence in situ hybridization technique. Biotechniques 45, 385–409 (2008)
- 121. Tsuchiya, K.D.: Fluorescence in situ hybridization. Clin. Lab. Med. 31, 525-542 (2011)
- Ranasinghe, R.T., Brown, T.: Ultrasensitive fluorescence-based methods for nucleic acid detection: towards amplification-free genetic analysis. Chem. Commun. 47, 3717–3735 (2011)
- 123. Puleo, C.M., Wang, T.H.: Microfluidic means of achieving attomolar detection limits with molecular beacon probes. Lab Chip **9**, 1065–1072 (2009)

- 124. Orte, A., Clarke, R., Klenerman, D.: Single-molecule two-colour coincidence detection to probe biomolecular associations. Biochem. Soc. Trans. **38**, 914–918 (2010)
- 125. Li, H., Ying, L., Green, J.J., Balasubramanian, S., Klenerman, D.: Ultrasensitive coincidence fluorescence detection of single DNA molecules. Anal. Chem. **75**, 1664–1670 (2003)
- Castro, A., Williams, J.G.K.: Single-molecule detection of specific nucleic acid sequences in unamplified genomic DNA. Anal. Chem. 69, 3915–3920 (1997)
- 127. Castro, A., Okinaka, R.T.: Ultrasensitive, direct detection of a specific DNA sequence of Bacillus anthracis in solution. Analyst **125**, 9–11 (2000)
- 128. Camacho, A., Korn, K., Damond, M., Cajot, J., Litborn, E., Liao, B., Thyberg, P., Winter, H., Honegger, A., Gardellin, P., Rigler, R.: Direct quantification of mRNA expression levels using single molecule detection. J. Biotechnol. **107**, 107–114 (2004)
- 129. Korn, K., Gardellin, P., Liao, B., Amacker, M., Bergstrom, A., Bjorkman, H., Camacho, A., Dorhofer, S., Dorre, K., Enstrom, J., Ericson, T., Favez, T., Gosch, M., Honegger, A., Jaccoud, S., Lapczyna, M., Litborn, E., Thyberg, P., Winter, H., Rigler, R.: Gene expression analysis using single molecule detection. Nucleic Acids Res. **31**, 89e (2003)
- Neely, L., Patel, S., Garver, J., Gallo, M., Hackett, M., McLaughlin, S., Nadel, M., Harris, J., Gullans, S., Rooke, J.: A single-molecule method for the quantitation of microRNA gene expression. Nat. Methods 3, 41 (2006)
- 131. Gunnarsson, A., Johnsson, P., Marie, R., Tegenfeldt, J.O., Hook, F.: Single-molecule detection and mismatch discrimination of unlabeled DNA targets. Nano Lett. 8, 183–188 (2008)
- 132. Földes-Papp, Z., Kinjo, M., Tamura, M., Birch-Hirschfeld, E., Demel, U., Tilz, G.P.: A new ultrasensitive way to circumvent PCR-based allele distinction: direct probing of unamplified genomic DNA by solution-phase hybridization using two-color fluorescence cross-correlation spectroscopy. Exp. Mol. Pathol. 78, 177 (2005)
- Friedrich, A., Hoheisel, J.D., Marmé, N., Knemeyer, J.P.: DNA-probes for the highly sensitive identification of single nucleotide polymorphism using single-molecule spectroscopy. FEBS Lett. 581, 1644–1648 (2007)
- Tyagi, S., Kramer, F.R.: Molecular beacons: probes that fluoresce upon hybridization. Nat. Biotechnol. 14, 303–308 (1996)
- 135. Li, D., Fan, C.: Optical detection of non-amplified genomic DNA, Chapter 6. In: Spoto, G., Corradini, R. (eds.) Detection of Non-amplified Genomic DNA, pp. 153–183. Springer, Dordrecht (2012)
- 136. Xi, C., Balberg, M., Boppart, S.A., Raskin, L.: Use of DNA and peptide nucleic acid molecular beacons for detection and quantification of rRNA in solution and in whole cells. Appl. Environ. Microbiol. 69, 5673–5678 (2003)
- 137. Totsingan, F., Rossi, S., Corradini, R., Tedeschi, T., Sforza, S., Juris, A., Scaravelli, E., Marchelli, R.: Label-free selective DNA detection with high mismatch recognition by PNA beacons and ion exchange HPLC. Org. Biomol. Chem. 6, 1232–1237 (2008)
- Song, C., Zhang, C., Zhao, M.: Singly labeled smart probes for real-time monitoring of the kinetics of dNTP misincorporation and single nucleotide extension in DNA intra-molecular polymerization. Biosens. Bioelectron. 25, 301–305 (2009)
- 139. Kummer, S., Knoll, A., Socher, E., Bethge, L., Herrmann, A., Seitz, O.: Fluorescence imaging of influenza H1N1 mRNA in living infected cells using single-chromophore FIT-PNA. Angew. Chem. Int. Ed. 50, 1931–1934 (2011)
- Grossmann, T.N., Seitz, O.: Nucleic acid templated reactions: consequences of probe reactivity and readout strategy for amplified signaling and sequence selectivity. Chem. Eur. J. 15, 6723–6730 (2009)
- 141. Østergaard, M.E., Hrdlicka, P.J.: Pyrene-functionalized oligonucleotides and locked nucleic acids (LNAs): tools for fundamental research, diagnostics, and nanotechnology. Chem. Soc. Rev. 40, 5771–5788 (2011)
- 142. Jarikote, D.V., Krebs, N., Tannert, S., Röder, B., Seitz, O.: Exploring base-pair-specific optical properties of the DNA stain thiazole orange. Chem. Eur. J. **13**, 300–310 (2007)
- Leijon, M., Mousavi-Jazi, M., Kubista, M.: LightUp probes in clinical diagnostics. Mol. Aspects Med. 27, 160–175 (2006)

- 144. Pianowski, Z., Gorska, K., Oswald, L., Merten, C.A., Winssinger, N.: Imaging of mRNA in live cells using nucleic acid-templated reduction of azidorhodamine probes. J. Am. Chem. Soc. 131, 6492–6497 (2009)
- 145. Germini, A., Rossi, S., Zanetti, A., Corradini, R., Fogher, C., Marchelli, R.: Development of a PNA array platform for the detection of genetically modified organisms in food. J. Agric. Food Chem. 53, 3958–3962 (2005)
- 146. Roy, S., Vedala, H., Roy, A.D., Kim, D.H., Doud, M., Mathee, K., Shin, H.K., Shimamoto, N., Prasad, V., Choi, W.: Direct electrical measurements on single-molecule genomic DNA using single-walled carbon nanotubes. Nano Lett. 8, 26–30 (2008)
- 147. Knoll, W., Liu, J., Niu, L., Nielsen, P.E., Tiefenauer, L.: Parallel optical and electrochemical DNA detection, Chapter 10. In: Spoto, G., Corradini, R. (eds.) Detection of Non-amplified Genomic DNA, pp. XX–XX. Springer, Dordrecht (2012)
- 148. Wang, Y., Brunsen, A., Jonas, U., Dostalek, J., Knoll, W.: Prostate specific antigen biosensor based on long range surface plasmon-enhanced fluorescence spectroscopy and dextran hydrogel binding matrix. Anal. Chem. 81, 9625–9632 (2009)
- 149. Henry, O.Y.F., Gutierrrez Perez, J., Lluis Acero Sanchez, J., O'Sullivan, C.K.: Electrochemical characterisation and hybridisation efficiency of co-assembled monolayers of PEGylated ssDNA and mercaptohexanol on planar gold electrodes. Biosens. Bioelectron. 25, 978–983 (2010)
- Bonanni, A., Ambrosi, A., Pumera, M.: Nucleic acid functionalized graphene for biosensing. Chem. Eur. J. 18, 1668–1673 (2012)
- 151. Ellerbee, A.K., Phillips, S.T., Siegel, A.C., Mirica, K.A., Martinez, A.W., Striehl, P., Jain, N., Prentiss, M., Whitesides, G.M.: Quantifying colorimetric assays in paper-based microfluidic devices by measuring the transmission of light through paper. Anal. Chem. 81, 8447–8452 (2009)
- 152. Lv, F., Wang, S.: Charged conjugated polymers, Chapter 5. In: Spoto, G., Corradini, R. (eds.) Detection of Non-amplified Genomic DNA, pp. XX–XX. Springer, Dordrecht (2012)
- 153. Minunni, M., Tombelli, S., Fonti, J., Spiriti, M.M., Mascini, M., Bogani, P., Buiatti, M.: Detection of fragmented genomic DNA by PCR-free piezoelectric sensing using a denaturation approach. J. Am. Chem. Soc. **127**, 7966–7967 (2005)
- 154. Ho, H.A., Doré, K., Boissinot, M., Bergeron, M.G., Tanguay, R.M., Boudreau, D., Leclerc, M.: Direct molecular detection of nucleic acids by fluorescence signal amplification. J. Am. Chem. Soc. 127, 12673–12676 (2005)
- Mo, Z.-H., Wei, X.-L.: Toward hybridization assays without PCR using universal nanoamplicons. Anal. Bioanal. Chem. 386, 2219–2223 (2006)
- 156. Nabok, A., Tsargorodskaya, A., Davis, F., Higson, S.P.J.: The study of genomic DNA adsorption and subsequent interactions using total internal reflection ellipsometry. Biosens. Bioelectron. 233, 377–383 (2007)
- 157. Storhoff, J.J., Marla, S.S., Bao, P., Hagenow, S., Mehta, H., Lucas, A., Garimella, V., Patno, T., Buckingham, W., Cork, W., Müller, U.R.: Gold nanoparticle-based detection of genomic DNA targets on microarrays using a novel optical detection system. Biosens. Bioelectron. 19, 875–883 (2004)
- Storhoff, J.J., Lucas, A.D., Garimella, V., Bao, Y.P., Muller, U.R.: Homogeneous detection of unamplified genomic DNA sequences based on colorimetric scatter of gold nanoparticle probes. Nat. Biotechnol. 22, 883–887 (2004)
- 159. Bao, Y.P., Huber, M., Wei, T., Marla, S.S., Storhoff, J.J., Muller, U.R.: SNP identification in unamplified human genomic DNA with gold nanoparticle probes. Nucleic Acids Res. 33, e15 (2005)
- 160. Huber, M., Wei, T., Muller, U.R., Lefebvre, P.A., Marla, S.S., Bao, Y.P.: Gold nanoparticle probe-based gene expression analysis with unamplified total human RNA. Nucleic Acids Res. 32, e137 (2004)
- 161. Fang, S., Lee, H.J., Wark, A.W., Corn, R.M.: Attomole microarray detection of MicroRNAs by nanoparticle-amplified SPR imaging measurements of surface polyadenylation reactions. J. Am. Chem. Soc. **128**, 14044–14046 (2006)

- 162. Goodrich, T.T., Lee, H.J., Corn, R.M.: Direct detection of genomic DNA by enzymatically amplified SPR imaging measurements of RNA microarrays. J. Am. Chem. Soc. 126, 4086–4087 (2004)
- 163. Yao, C., Zhu, T., Tang, J., Wu, R., Chen, Q., Chen, M., Zhang, B., Huang, J., Fu, W.: Hybridization assay of hepatitis B virus by QCM peptide nucleic acid biosensor. Biosens. Bioelectron. 23, 879–885 (2008)
- 164. Wang, Y., Chen, M., Zhang, L., Ding, Y., Luo, Y., Xu, Q., Shi, J., Cao, L., Fu, W.: Rapid detection of human papilloma virus using a novel leaky surface acoustic wave peptide nucleic acid biosensor. Biosens. Bioelectron. 24, 3455–3460 (2009)

Chapter 5 Charged Conjugated Polymers

Fengting Lv and Shu Wang

Abstract Signal amplification property and versatility in molecular design of conjugated polymers make them attractive as signal transduction materials. Cationic conjugated polymers (CCP) have demonstrated great potential as DNA-sensing materials since electrostatic interactions between CCP and negatively charged DNA have been well developed to modify optical properties and thereby detect hybridization states. In this chapter, we focus on the state-of-art of CCP-based non-amplified genomic DNA detection in homogenous solution as well as on microarray format. Representative applications of poly(thiophene)-based DNA sensors which display colorimetric output due to conformational changes, and poly(fluorene)-based DNA sensors which utilize CCP as energy donors are highlighted. The developments of amplified single nucleotide polymorphisms and DNA methylation detection employing CCP as sensory signal are briefly discussed. Integrated signal amplification effect of CCP with high-throughput DNA microarray format will yield superior platform for genomic DNA detection.

5.1 Introduction

Simple, sensitive, and specific chemical and biological sensing is becoming ever more dependent upon the development of novel materials. Polymers, quantum dots, crystals, and nanostructures have made a profound impact and have endowed modern sensory systems with superior performance. Conjugated polymers (CPs), characterized by a backbone with a delocalized electronic structure, have emerged as one of the most important classes of transduction materials. For the past three decades, conjugated polymers have attracted much attention as an active component

F. Lv • S. Wang (🖂)

Beijing National Laboratory for Molecular Science, Key Laboratory of Organic Solids, Institute of Chemistry, Chinese Academy of Sciences, Beijing, China e-mail: lvft@iccas.ac.cn; wangshu@iccas.ac.cn

G. Spoto and R. Corradini (eds.), *Detection of Non-Amplified Genomic DNA*, Soft and Biological Matter, DOI 10.1007/978-94-007-1226-3_5, © Springer Science+Business Media Dordrecht 2012

for their potential applications such as light-emitting diodes (LEDs), photovoltaic cells, field effect transistors (FETs), and chemical and biological sensors [1]. The chemical structures of conjugated polymers offer several advantages as the responsive basis for chemical and biological detection schemes based on optical methods. The conjugated units in close proximity of the polymer backbone allows for efficient electronic coupling and therefore fast intra- and interchain energy transfer, which indicated that an environmental change at a single site can affect the properties of the collective system, producing large signal amplification [2–11]. Trace detection of analytes including ions, biomacromolecules, and explosives have been successfully accomplished by utilizing this amplification mechanism [12–22]. A wide range of conjugated polymers, for example, poly(thiophene)s (PT), poly-(*p*-phenylenevinylene)s (PPV), poly(*p*-phenyleneethynylene)s (PPE), and poly(fluorene)s (PF) have been reported in the literature as promising materials for these applications.

Because the interchain electron transfer interactions of conjugated polymers are relatively strong compared with the van der Waals and hydrogen bonding interchain interactions, conjugated polymers tend to be insoluble. Water solubility, a prerequisite for interrogating biological substrates, was typically achieved by incorporating charged pendant groups or polar groups onto the CP backbone. The introduction of various substitutions along the conjugated backbone can not only improve the processability of these aromatic polymers but can also modify their physical properties, which can lead to physical phenomena that are not found in the parent unsubstituted polymers. Conjugated polymers bearing ionic pendant groups are also called conjugated polyelectrolytes (CPEs) and these water-soluble materials combine the properties of polyelectrolytes, which coordinate electrostatic forces with oppositely charged analytes, with the attractive optical and electronic properties of conjugated polymers [6]. Conjugated polyelectrolytes have demonstrated great potential as biosensory materials, especially in DNA-sensing applications, which are driven by the formation of stable aggregates with cationic conjugated polymer through the strong electrostatic interactions. The primary advantage of this strategy is that it relies upon noncovalent interactions and therefore does not require covalent attaching of single-stranded DNA (ss-DNA) probe onto conjugated polymer backbone.

Reliable and sensitive DNA detection is of vast scientific and economic importance, and novel techniques and materials that determine DNA hybridization and sequence characterization are under intense investigation [23–26] for applications, such as medical diagnostics, identification of genetic mutations, gene delivery monitoring, and specific genomic techniques. In recent years, much effort has been devoted to improve the selectivity, sensitivity, and miniaturization to allow detection in small sample volumes. Gene chips and DNA microarrays which perform highthroughput screening of hundreds to thousands of targets in a single experiment are dominating the market as one of the most powerful fluorescent DNA detection techniques [27–29]. However, most available methods involve polymerase chain reaction (PCR) protocols to increase the concentration of specific nucleic acid sequences, which tend to complicate the procedures and therefore prolong the



Fig. 5.1 The molecular structures of typical water-soluble CPEs

overall detection time. In response to this limitation, signal-amplification systems constructed by water-soluble cationic conjugated polyelectrolytes have been developed by taking advantage of their unique optical properties. The molecular structures of typical water-soluble CPEs are shown in Fig. 5.1.

The purpose of this review is to highlight some of the fundamental properties of cationic conjugated polymers and their remarkable applications in optical non amplified DNA sensory processes. Since there has been great progress made in the detection of single nucleotide polymorphisms (SNP) and DNA methylation with PCR-associated target preparation utilizing cationic conjugated polymers as sensory signal, all these developments are also addressed.

5.2 Synthesis, Optical Properties, and Sensing Mechanisms

5.2.1 Synthesis

The opportunity to synthesize new conducting polymers with improved and desired properties began to attract the attention of synthetic chemists in the 1980s. In this review, typical synthetic approaches of some classical water soluble conjugated polymers are provided. The reader interested in the exhaustive list of published procedures should refer to the literature [5, 14, 30].

Poly(thiophene)s with ionic side chains have been widely used for the specific detection of DNA sequences which relies upon conformational changes of the conjugated polymer backbone upon ssDNA hybridization with complementary ssDNA. The synthesis of water soluble poly(thiophene) was initiated by Wudl group in 1987, where the corresponding methyl sulfonate monomers were electropolymerized first and then converted into the sodium salts [31]. Recently, the Leclerc [32] and Inganäs [33] groups have also developed new water soluble poly(thiophene) derivatives, which were prepared from an oxidation polymerization of monomers in chloroform with FeCl₃ as the oxidizing agent. Figure 5.2a illustrates the synthesis of poly[3-(3'-N,N,N-triethylamino-1'-propyloxy)-4-methyl-2,5-thiophene hydrochloride] (PMNT) [32].

By virtue of high fluorescence quantum yields in water, PPE and their derivatives become one of the most studied classes of conjugated polymers in fluorescent biosensor. The Sonagashira method is widely employed to prepare high molecular weight PPEs by coupling a dihaloarene and a diethynyl-arene in a DMF/water mixture with Pd(PPh₃)₄/CuI as catalyst [34]. The overall synthetic route of PPEs is illustrated in Fig. 5.2b. The groups of Schanze, Swager and Bunz have synthesized a series of water soluble PPEs employing the similar method [4, 35, 36].

Wudl [37] and coworkers reported the synthesis of water soluble PPV by using Wessling polymerization procedure. As depicted in Fig. 5.2c, the monomer was first polymerized in the presence of base, and further hydrolyzed in DMF/water mixture under reflux to get the non-conjugated polymer. Upon heating or acid treatment, the fully conjugated poly(2,5-methoxy-propyloxysulfonate phenylene vinylene) (MPS-PPV) was obtained.

Cationic PFs materials serve as donors in fluorescence resonance energy transfer (FRET) DNA assays which display signal amplification. The palladium-catalyzed Suzuki coupling reaction has gained considerable popularity for synthesis of water-soluble PFs due to the mild reaction conditions. Bazan and co-workers [38] initially prepared poly[9,9-bis(6'-[N,N,N-trimethylammonium]hexyl)fluoreneco-phenylene dibromide] (PFP) by quaternization with methyl iodide of pendant amine groups on a neutral precursor. Since it is difficult to purify the monomer containing amine groups, they developed an improved synthetic approach by using monomers containing bromo groups, as shown in Fig. 5.2d. Subsequent quaternization proceeds with excellent yield via the reaction of the precursor polymer with trimethylamine in THF/water, followed by precipitation from acetone.

5.2.2 Optical Properties

When a CPE exists in solution in a non-aggregated state, the photophysical properties (fluorescence quantum yields, absorption and emission maximum, and lifetimes) of this material are remarkably similar to those of the neutral analogue dissolved in non-polar organic solvent [39]. In fact, photophysical studies provide



Fig. 5.2 The synthetic scheme of PMNT (a) [32], PPEs (b) [34], MPS-PPV (c) [37], PFP (d) [38]

very clear evidence that in aqueous solution CPEs have a very strong tendency to aggregate, and this aggregation phenomenon has a substantial influence on the photophysical properties of the polymers. The number of charges per repeat unit and the nature of the linkers between the backbone and the charged groups influence solubility and thus aggregation tendency. Understanding the correlation between structure and solution behavior of CPEs will provide unique opportunities for fine-tuning of CPEs structures and therefore improving detection sensitivity and selectivity.

Bazan and co-workers studied the solvent-dependent aggregation properties of cationic poly(fluorene) (P1, chemical structure shown in Fig. 5.1) [40]. The backbone and alkyl side chain are hydrophobic moieties, while the cationic charged quaternary amines control electrostatic interactions. Dissociation of the charged ionogenic groups requires polar solvents, while the hydrophobic segments are better accommodated in non-polar solvents. There exist two different types of aggregates depending on the solvent medium. Single chain behavior or minor aggregation occurs when the THF content is in the range from 30 to 80%. In pure water, the polymer likely forms tight aggregates with chains coming together and forming the inner core, which lead to lower emission intensities due to π - π stacking and self-quenching. Upon addition of THF to aqueous solutions, the aggregates are broken up and therefore self-quenching is reduced. With further increasing the THF content higher than 80%, the ionic interactions of charged groups with the nonpolar medium lead to the groups becoming buried within a new aggregate structure. The overall aggregation driving force is the amphiphilic nature of the cationic PF. The obtained solvent-dependent aggregated information of cationic PFs is beneficial to gain insight into their signal transduction mechanism.

Compared with their neutral analog PPV, the red-shift of fluorescence emission and significantly decreased fluorescence quantum yield for MPS-PPV were reported by Whitten and co-workers [41]. Similarly, Schanze et al. [42] reported the solventdependence of the absorption and fluorescence of PPE type CPE, which also revealed that aggregation has a strong effect upon the photophysics of CPEs.

Another work carried out in Bazan group demonstrated that increasing the counter anion size can decrease the interchain contacts of PFBT-X and lead to a substantial increase of quantum yield in the bulk [43]. The main backbone of PFBT-X (P2, chemical structure shown in Fig. 5.1) is composed of alternating fluorene and benzothiadiazole (BT) units, and dynamic light scattering analysis indicated that the suppression of aggregation of PFBT-X in water by the large and hydrophobic anion, $B(3,5-(CF_3)_2C_6H_3)_4$. Simple ion exchange methods can be employed to generate cationic conjugated polymers with different charge compensating ions, which make regulating their optoelectronic properties for specific applications easier.

Water-soluble dendritic poly(fluorene) with positively charged amine groups on the exterior were synthesized by Wang and co-workers [44]. By virtue of the higher cationic charge density, this novel cationic conjugated polymer is less aggregated in aqueous solution and exhibits an improvement of fluorescent DNA assays signal, whereas higher generation is not easily available due to the steric hindrance of the side chain and the lower yield.

Besides tuning polymer properties by rational design and synthesis at the level of the conjugated repeat unit, the groups of Whitten [45] and Bazan [38] demonstrate that by complexing with appropriate surfactants, one may control the geometric conformation of the polymer, and thereby tune its optical and chemical properties.

5.2.3 Signal Transduction Mechanism

CPE-based DNA sensing systems in homogeneous media and heterogeneous state have been successfully accomplished by utilizing two main signal transduction mechanisms. One is DNA-induced conformational or aggregation change of CPEs, and another mechanism is quenching the fluorescence of CPEs through electron transfer or energy transfer.

Cationic polythiophenes transduce DNA hybridization into a colorimetric output based on conformational changes of the polymer upon interaction with singlestranded DNA or double-stranded DNA. Polythiophenes are known to exhibit interesting chromic features (change of color induced by a conformational change of the conjugated backbone) in the presence of different stimuli [10]. In many cases, such chromic effects also have a strong influence on the fluorescence properties of these conjugated polymers. The optical features afforded by polythiophenes provide the basis for specific detection of non-amplified DNA sequences. Specifically, an aqueous solution of cationic polythiophenes is yellow with an absorption wavelength near 400 nm which is related to a random-coil (non-planar) conformation of the polythiophene derivative. After complex with single-stranded DNA, the highly conjugated and planar conformation, which exhibit relatively red-shifted absorption wavelength, is predominant. A fluorometric detection of oligonucleotide hybridization is accessible since the fluorescence of polythiophenes is quenched in the planar, aggregated form. This change in fluorescence intensity is mainly due to a modification in the delocalization of π electrons along the carbon chain backbone that occurs when switching between the two conformations.

Water-soluble CPEs exhibit strong photoluminescence and rapid transport of the singlet excitation throughout the conjugated backbone. An important characteristic of these materials is the efficient quenching of the CPE fluorescence that can be achieved by quencher-labeled target strand DNA through either electron transfer or energy transfer. This amplified quenching ("superquenching") can be attributed to a combination of Coulombic and hydrophobic interactions in aqueous media and to rapid migration of excitons through the polymer following photoexcitation. PPE derivatives as one of the representative CPEs were employed to develop DNA sensors by this superquenching mechanism [46–48].

The quencher can be designed as an energy acceptor to allow for efficient fluorescence resonance energy transfer from the CPEs to the acceptor. The unique characteristic of CPEs is their ability to amplify signals of dyes labeled on DNA; therefore lower concentration of analytes can be easily detected, in comparison to methods using direct excitation of small molecular dyes. In comparison to intensity-based methods, these techniques provide large changes in emission profiles and open the opportunity for ratiometric fluorescence measurements [49]. Furthermore, assays of this type are less prone to false positives since non-specific interactions between the CPEs and DNA could be effectively eliminated. In FRET-based DNA assays, the light harvesting CPE and a fluorophore capable of introduction into a probe structure are generally designed to function as the donor and the acceptor,

respectively. According to Förster theory [49], FRET is a long-range energy transfer which arises from dipole-dipole interactions. FRET rate (k_t) is influenced by three factors which are the donor-acceptor distance (r), the orientation factor (k), and the overlap integral (J) between the emission of the CPEs and the absorption of the acceptor. The overlap integral ($J(\lambda)$) expresses the degree of spectral overlap between the donor emission (described by emission intensity, $F_D(\lambda)$) and the acceptor absorption (described by molar absorbance, ε_A (λ)), as described in Eq. 5.1. The components of the DNA sensor can be carefully chosen so that their optical properties meet this requirement and then the FRET process is optimized. Cationic poly(fluorene-co-phenylene)s based DNA assays involving FRET have been extensively developed and reviewed [5, 7].

$$k_{t(r)} \propto \frac{1}{r^6} \cdot \kappa^2 \cdot J(\lambda)$$
$$J(\lambda) = \int_0^\infty F_D(\lambda) \varepsilon_{\rm A}(\lambda) \lambda^4 d\lambda$$
(5.1)

5.3 Signal-Amplified DNA Detection

5.3.1 PT-Based DNA Detection

The Leclerc group pioneered DNA detection with conjugated polymers in homogenous solution. In 2002, they reported new water-soluble cationic polythiophene derivatives, which can easily transduce oligonucleotide hybridization with a specific 20-mer capture probe into a clear optical (colorimetric or fluorometric) output [32]. Three types of negatively charged oligonucleotides were used: a captureprobe sequence (X1: 5'-CATGATTGAACCATCCACCA-3'), a perfect complementary target (Y1: 3'-GTACTAACTTGGTAGGTGGT-5'), X1 and Y1 are a specific oligonucleotide pair for Candida albicans. A two-mismatch complementary target (Y2: 3'-GTACTAACTTCGAAGGTGGT-5'), and a one-mismatch complementary target (Y3: 3'-GTACTAACTTCGTAGGTGGT-5'). Y2 is a specific oligonucleotide for Candida dubliniensis, and Y3 is a variant designed to have single mismatches with either C. albicans or C. dubliniensis sequences. An aqueous solution of cationic poly(3-alkoxy-4-methylthiophene)s (P3, chemical structure shown in Fig. 5.1) is yellow with an absorption wavelength near 400 nm (Fig. 5.3 left panel curve a) which is related to a random-coil conformation. Upon addition of capture oligonucleotide X1, the mixture becomes red ($\lambda_{max} = 527$ nm) (Fig. 5.3 left panel curve b), the highly conjugated and planar conformation dominated due to the formation of a so-called duplex between the polythiophene and the oligonucleotide probe (see Fig. 5.4). After mixing with the complementary oligonucleotide Y1, the solution becomes yellow ($\lambda_{max} = 421 \text{ nm}$) (Fig. 5.3 left panel curve c); presumably caused by the formation of a new complex termed a triplex (c.f. Fig. 5.4), formed



Fig. 5.3 *Left panel*: UV-vis absorption spectra of a solution of (**a**) poly(3-alkoxy-4-methylthiophene)s, (**b**) poly(3-alkoxy-4-methylthiophene)s/X1 duplex, (**c**) poly(3-alkoxy-4-methylthiophene)s/X1/Y1 perfect match triplex, (**d**) poly(3-alkoxy-4-methylthiophene)s/X1/Y2 mixture with two mismatches, and (**e**) poly(3-alkoxy-4-methylthiophene)s/X1/Y3 mixture with one mismatch. *Right panel*: Fluorescence spectra corresponding to the different assays in *left panel* (Reprinted with permission from Ref. [8]. Copyright (2004) American Chemical Society)



Fig. 5.4 Schematic description of the formation polythiophene/ss-DNA duplex and polythiophene/ds-DNA triplex complex (Reprinted with permission from Ref. [8]. Copyright (2008) American Chemical Society)

by complexation of the polymer with the hybridized nucleic acids. A schematic description of duplex and triplex formation is shown in Fig. 5.4. They also investigated the specificity of this polymeric optical transducer in the presence of imperfect or incomplete hybridizations. As shown in Fig. 5.3 left panel curve d, a distinct absorption spectrum is observed in the case of polymer/X1 complexed with
oligonucleotide targets with two mismatches (Y2). It is also possible to distinguish only one mismatch (Fig. 5.3 left panel curve e). It is interesting to note that the spectra of duplex and triplex forms show an isosbestic point, which indicates the presence of only two distinct conformational structures for the polymeric transducer. The detection limit of this colorimetric method is about 1×10^{13} molecules of 20mers oligonucleotide (a concentration of 2×10^{-7} M) in a total volume of 100 µL.

A fluorometric detection of oligonucleotide hybridization is possible since the fluorescence of polythiophenes is guenched in the planar, aggregated form. The yellow form of poly(3-alkoxy-4-methylthiophene)s is fluorescent (quantum yield of 0.03 with a maximum of emission at 530 nm, c.f. Fig. 5.3 right panel curve a). Upon addition of capture oligonucleotide probe X1, the fluorescence intensity decreases and the maximum of emission is slightly red-shifted (Fig. 5.3 right panel curve b). When hybridized with the complementary strand Y1, the formation of a polymeric triplex leads to a five-fold rise in fluorescence intensity (Fig. 5.3 right panel curve c). Upon addition of the oligonucleotide target having two mismatches (Y2), the fluorescence intensity is weak and not significantly modified (Fig. 5.3 right panel curve d). Moreover, it is possible to distinguish oligonucleotides with one mismatch (Y3) (Fig. 5.3 right panel curve e). By utilizing the fluorometric output of poly(3-alkoxy-4-methylthiophene)s, the detection limit of the perfect complementary oligonucleotide (20-mers) is as few as 3×10^6 molecules in a volume of 200 μ L, which is a concentration of 2 × 10⁻¹⁴ M. Similar results have been obtained with other polythiophene derivatives [50].

Following the aforementioned work, the same group developed a DNA detection platform by using a custom-built fluorimeter based on a high-intensity blue diode (as the excitation source) and a nondispersive interference filter [51]. A few hundred copies of either DNA or RNA can be specifically detected. The possibility to follow hybridization reactions in real time at such low concentrations could be used to study kinetics and thermodynamics of nucleic acids in various biological systems. In addition, similar selectivity and sensitivity was obtained when the fluorescent polymeric hybridization transducer was grafted on magnetic microbeads, which allowed the possibility of performing both preconcentration and detection steps simultaneously [52].

This simple, rapid, and versatile methodology based on different electrostatic interactions and conformational structures between cationic polythiophenes, and single-stranded oligonucleotides or double-stranded (hybridized) nucleic acids is greatly advantageous because this assay does not require any chemical reaction of the probes or the analytes.

Building on this initial concept, in 2005 they reported the detection of DNA targets in aqueous solutions at the zeptomolar range using a physical amplification method described as fluorescence chain reaction (FCR), which is a combination of electrostatic interactions, chromism, and a Förster resonance energy transfer mechanism, leading to novel fluorescence signal amplification detection [53]. Neutral duplexes made from a cationic polythiophene, serves as a potential donor, and oligonucleotide capture probes labeled with Alexa Fluor 546 (acceptor). The choice of Alexa Fluor 546 depends on the need to get a good overlap between

its excitation spectrum and the emission spectrum of the polymer donor, and an excellent fluorescence quantum yield of the acceptor. In the starting conditions, the stoichiometric duplex gives a red color and fluorescence quenching. After hybridization with the perfectly complementary DNA target, a new absorption feature appears at 420 nm, which is related to the formation of a triplex. Upon irradiation at 420 nm, the fluorescence band at 530 nm of the resulting triplex (the donor in this FRET scheme) overlaps neatly with the absorption spectrum of Alexa Fluor 546, and efficient energy transfer to the neighboring acceptor (emission maximum at 572 nm) is observed. Upon addition of noncomplementary or mismatched DNA to the duplex, the complex remains in the red non-fluorescent form, preventing the FRET mechanism from occurring. Consequently, the fluorescence intensity measured under the same experimental conditions with the perfect complementary ss-DNA strand is always higher than that obtained with targets having mismatch. The detection limit can be further enhanced by increasing the number of duplex probes, in the case of 10^{10} copies of duplex probes as starting materials, 30 copies of 20-mer target oligonucleotides are easily detected from a volume of 3 mL, and perfectly complementary targets can still be distinguished at such low concentrations from sequences having two or even one mismatch. The detection limit is five copies in 3 mL, or 3 zM, with a custom blue-LED fluorimeter, whereas a somewhat higher detection limit of 30 copies in 3 mL, or 18 zM, was obtained using a commercial spectrofluorometer. Moreover, this novel signal amplification detection strategy can also be used to direct detect ds-DNA at ultralow concentration levels. Because the probe-to-target hybridization reaction is in competition with rehybridization of the ds-DNA, most previously reported direct DNA detection techniques rely on the availability of the target sequence as ss-DNA. Since the polythiophene has a higher affinity toward ds-DNA compared to that of the ss-DNA probes, the amplified quenching of the signal in previous studies [32, 51] may lead to false positive signals, whereas this novel approach, involving "turn-on" signal amplification, is promising. The massive intrinsic signal amplification provided by this novel integrated molecular system allows the specific detection of as little as five dsDNA copies in a 3 mL sample volume in only 5 min, without the need for prior PCR amplification or chemical tagging of the genetic target. All these features provided by this novel sensing platform will lead to major advances in genomics.

For the purpose of detecting multitarget simultaneously, they extended the FCR detection scheme to microarray-based multiplex detection by hybrid self assembled polythiophene/DNA probe complexes for the direct and specific detection of DNA [54].

Two capture probes were used for DNA detection: one single-stranded DNA probe labeled with Cy3 chromophore, 5'-NH₂-C6-CATGATTGAACCATCCACCA-Cy3-3' (NH₂-P1), and one unlabeled DNA probe, 5'-NH₂-C6-CATGATTGAACCA TCCACCA-3' (NH₂-P2). Two DNA targets, one perfectly matched to the DNA probes, 3'-GTACTAACTTGGTAGGTGGT-5' (T1), which corresponds to a conserved region of the Candida albicans yeast genome, and one target having one mismatched base, 3'-GTACTAACTT<u>C</u>GTAGGTGGT-5' (T2), were investigated. The average polythiophene/DNA aggregate diameter is around

200–250 nm, while the height is around 20–30 nm. The diameter of the spots is about 1.5–1.7 mm, each including about 1×10^{12} probes. Figure 5.5b exhibits fluorescence images from a spotted slide after hybridization with the perfect complementary target oligonucleotide and a target with one mismatch for concentrations ranging from 1×10^{-15} to 1×10^{-6} M. The image measured at the fluorescence wavelength specific to the Cy3 chromophore shows a clear contrast between perfect complementary targets and those having one mismatch at concentrations greater than 10^{-14} M. As shown in Fig. 5.5b, the fluorescence intensity at 530 nm (fluorescence from the polythiophene itself) is very weak, even for the perfect complementary target. This observation indicates that the FRET mechanism is highly efficient. For a perfect complementary target oligonucleotide, a LOD of 5×10^{-16} M from a sample volume of 0.4 µL, corresponding to 300 DNA molecules. The detection sensitivity obtained with the system was improved by a factor of ca. 1,500 over that obtained using an unlabeled duplex (NH₂-P2/polythiophene complexes).

5.3.2 PPE-Based DNA Detection

Whitten and co-workers developed DNA sensors employing PPE derivatives with advantages of high fluorescence yields in both solution and absorbed states coupled with lower nonspecific effects in solid phase. In 2002, they reported an assay for a target single strand 20-base sequence of DNA coding for the anthrax lethal factor (ALF), which based on fluorescent polymer superquenching [46]. The sensor was constructed using a carboxylic acid functionalized polystyrene microsphere covalently functionalized with streptavidin. Biotinylated PPE (PPE-B, P4 chemical structure shown in Fig. 5.1) and DNA capture strands were simultaneous coated on polystyrene microsphere. Addition of the target strand would not attenuate the fluorescence of the microsphere, whereas addition of a quencher-labeled target strand (DNA-QTL) would reduce the fluorescence of the sensor. Target strand nucleic acids compete with a DNA-QTL for a capture strand associated closely with a fluorescent polyelectrolyte. The direct competition assay did not provide a sensitive assay, so they developed a two-step assay, in which the sensor was first incubated with target DNA and DNA-QTL was added in a second step, providing a sensitive quenching inhibition (turn-on) assay. The level of quenching decreases with increasing concentration of target DNA. An assay with similar sensitivity was obtained by pre-incubating the biotinylated capture strand with a mixture containing a constant amount of DNA-QTL and variable amounts of target DNA and then adding the mixture to microspheres pre-treated with only PPE-B.

Another improved assay for target DNA was developed using a modified sensor and replacing the biotinylated capture DNA with a biotinylated peptide nucleic acid (PNA) capture strand [47]. In this case, the polystyrene microspheres were functionalized with quaternary ammonium groups, PPE-B was precomplexed with Neutravidin at an empirically ratio that provided the strongest superquenching



Fig. 5.5 (a) Schematic description of the specific recognition of target ssDNA by duplex aggregates onto glass slides with visualization of signal amplification detection mechanism based on the conformational change of cationic polythiophene and fast energy transfers within the micelle. (b) Fluorometric detection of hybridization of 20-mer DNA oligomers on arrays. *Left:* Excitation at 408 nm, emission at 570 nm, and *right:* excitation at 408 nm, emission at 530 nm (Reprinted with permission from Ref. [54]. Copyright (2006) American Chemical Society)

effect, and the complex was added to the ammonium functionalized microspheres. It was found that a 15 base capture PNA strand gave excellent results in sensing compared to a 20-base capture DNA. This assay was carried out by first incubating the biotinylated capture strand with the microsphere sensor described above, subsequently adding variable amounts of target DNA and then adding the DNA-QTL and measuring the intensity of the fluorescence. Thus the assay functions in a fluorescence "turn-on" mode. Additionally, the authors evaluated the sensor to discriminate single nucleotide mismatches, and the experimental data demonstrated that the sensor was capable of detecting single nucleotide mismatches within 20 min for subpicomolar amounts of the DNA target at 40 °C. These assays offer comparable or improved sensitivity to other conventional fluorescence microarray-based assays for DNA [55, 56], which do not involve amplification by a polymerase chain reaction.

Schanze and co-workers constructed a novel molecular beacon by linking oligonucleotide to PPE [48]. The oligonucleotide took part in the polymerization process and was incorporated into the PPE chain as an end capping unit. Molecular-beacon sequence was 5'-PPE-CCTAGCTCTAAATCACTATGGTCGCGCTAGG-DABCYL-3', a universal quencher DABCYL (4-(4-(dimethylamino)phenylazo) benzoic acid) was selected as the quencher, PPE chain as the fluorophore to amplify the fluorescence signal. When the molecular beacon is in its closed state, the polymer chain will be brought close to the quencher, and the fluorescence of PPE will be strongly suppressed because of the superquenching property of the conjugated polymer. After target-DNA binding, the fluorescence of the PPE would be restored as a result of the increased separation distance between the conjugated polymer and the quencher. Selective label-free detection of target oligonucleotides with an amplified fluorescence signal was successfully achieved by using this novel molecular beacon.

5.3.3 PF-Based DNA Detection

Bazan and co-workers have shown in a series of publications that effective FRET transduction mechanism can be employed for the detection of DNA. Much of the FRET work was stimulated by the demonstration of specific DNA sequence detection using a dye-labeled PNA probe and cationic poly-(fluorene*alt*-phenylene) (PFP) [57]. As shown in Fig. 5.6 left panel, a PNA probe (5'-CAGTCCAGTGATACG-3') with fluorescein at the 5' position was used as PNA-C*, and electrostatic attraction brings the donor (PFP) and anionic DNA into close proximity. If the ssDNA target hybridizes to a complementary PNA-Fl probe, PFP and acceptor (Fl) are close enough that excitation of PFP results in FRET and emission from fluorescein. If the target is not complementary to the PNA probe, the polymer to fluorescein distance is large, and no fluorescein emission is observed (Fig. 5.6 right panel). Signal amplification by the PFP provides a fluorescein emission that is over 25 times higher than that obtained by the direct excitation of



Fig. 5.6 *Left panel*: Schematic representation for the use of a water-soluble CP with a specific PNA-C* optical reporter probe to detect a complementary ssDNA sequence. *Right panel*: Emission spectra of PNA-Fl in the presence of complementary. (**a**) and noncomplementary, (**b**) DNA by excitation of PFP. The spectra are normalized with respect to the emission of PFP (Reprinted with permission from Ref. [57]. Copyright (2002) National Academy of Sciences, USA)

the dye at its absorption maximum. This amplification reflects the higher optical density of the polymer, and the very efficient FRET step.

By adding an S1 nuclease enzyme, they developed a new strategy that is sensitive to single nucleotide polymorphisms [58]. The recognition is accomplished by sequence-specific hybridization between the uncharged PNA-Fl probe and the ssDNA sequence of interest. Treatment with S1 nuclease leads to digestion of DNA, except for those regions "protected" by a perfectly matching hybridized PNA. PFP only works as excitation donor to Fl for PNA/ssDNA duplex. The overall method was tested by measuring the ability of the sensor system to detect wild type human DNA sequences, as opposed to sequences containing a single base mutation.Specifically, the PNA probe was complementary to a region of the gene encoding the microtubule associated protein Tau. The probe sequence covers a known point mutation implicated in a dominant neurodegenerative dementia known as FTDP-17, which has many clinical and molecular similarities to Alzheimer's disease. Using an appropriate PNA probe, unambiguous FRET signaling is achieved only for the wild type DNA and not the mutant sequence harboring the single nucleotide polymorphism.

The sensing strategy in Fig. 5.6 was modified to replace the PNA-Fl probes with more readily available ssDNA-Fl as the signaling reporter because DNA/DNA interactions are more common in biology and are therefore better catalogued and understood [59]. The higher local charge density of the double strand results in a stronger electrostatic attraction between the dsDNA-Fl and the conjugated polyelectrolyte relative to the situation with ssDNA, upon adding a complementary strand ssDNA, the hybridized strand will result in an efficient FRET from PFP to Fl.

In the case of a noncomplementary strand, the nonhybridized strand will interfere with the ssDNA-Fl-PFP interactions and FRET from PFP to Fl does not occur.

Liu et al. [60] reported a strategy of using a probe-immobilized silica nanoparticle (NP) surface as a platform for DNA detection which took advantage of the optical amplification of cationic conjugated polymers and easy separation of NPs. The approach begins with probe immobilized NPs in solution to capture the target labeled with fluorescein and bring fluorophores close to the surface, after separation, PFP was added to the system. Only when the complementary sequence was present in the initial solution would PFP and fluorescein be present in the final mixture, allowing intense acceptor emission upon polymer excitation. Excess DNA probes on the NP surface complexed with PFP to increase the local concentration of donor units and fluorescein self-quenching was minimized, leading to over 110 fold signal amplification. The nearly homogeneous solution of silica NPs facilitates DNA hybridization and meanwhile provides the advantage of solid state sensors that allow separation and base mismatch detection.

Ethidium bromide (EB) intercalate within the internally stacked bases of dsDNA, resulting in an increase in its fluorescence quantum yield. The scheme for DNA detection by utilizing the FRET between PFP and EB was designed by Bazan group [61]. Actually, it turns out that the FRET efficiency between PFP and EB is not high when complementary ssDNA was introduced to the sensing system, especially compared to previously reported studies with CCPs and fluorescein-labeled DNA or PNA, which may be attributed to nonoptimized transition dipole alignment and electrostatic repulsion between PFP and EB. In order to enhance the FRET efficiency to gain higher signal, fluorescein was attached to ssDNA terminus, which function as a fluorescence resonance gate for transferring conjugated polymer excitations to dyes intercalated within dsDNA. FRET from PFP to Fl, followed by a second FRET to EB, becomes energetically feasible. The overall process provides a substantial improvement of selectivity over previous homogeneous conjugated polymer based DNA sensors. DNA conformational transition can be detected by utilizing this two-step FRET principle [62].

Our group has developed a DNA detection method which combines optical amplification properties of cationic conjugated polyelectrolytes with highly selective target-induced DNA strand displacement [63]. Efficient strand displacement allowed for formation of a hairpin shape DNA which could be recognized and cleaved by Hae III endonuclease, or by formation of an active DNAzyme which was inhibited before strand displacement. The occurrence of cleavage after strand displacement afforded a weaker FRET between PFP and the cleavage product. Since the existence of a mismatch in the target DNA could lower the efficiency of the displacement reaction, the assay was able to detect even a single mismatch in the target DNA. The detection limits of the target DNA are 2.14 pmol for the Hae III cleavage system and 0.75 pmol for the DNAzyme system.

Very recently, our group developed a dual-amplified fluorescence DNA detection scheme taking advantages of signal amplification property of conjugated polymers and recyclable autocatalytic hybridization amplification of DNA [64]. The detection principle is illustrated in Fig. 5.7. The hairpin DNA_A (probe A) and DNA_B labeled



Fig. 5.7 Schematic representation of the dual-amplification DNA detection. DNA_C is complementary to probe A, DNA_{NC} is non-complementary to probe A. HaeIII is the restriction endonuclease for probe A/probe B duplex cleavage and the cleavage sites are indicated by *arrows* (Reprinted from Ref. [64]. With permission by the Royal Society of Chemistry)

with a fluorescein dye at 50-terminus (probe B) act as the signaling probes. Hairpin probe A and probe B co-exist metastably in the absence of target DNA. The autocatalytic hybridization process of probe A and probe B is initiated by adding the target DNA (DNA_C) which is complementary to probe A. The formed probe A/DNA_C duplex with a 3' single strand can then hybridize with the 5' single strand of probe B to simultaneously release the target DNA_C. The released DNA_C can then hybridize with another hairpin probe A and the cycle starts anew. Thus, the autocatalytic hybridization of DNA induced by the target DNA_C can produce large amounts of probe A/probe B duplex labeled with fluorescein at its 5'-terminus. The probe A/probe B duplex can then be cleaved away by the endonuclease HaeIII. Owing to the two amplification processes, the new detection approach achieves high detection sensitivity with a 10 fmol detection limit, thus greatly improving the sensitivity of DNA detection using cationic conjugated polyelectrolytes.

Aggregation-induced interchain energy transfer in polymers of PFPB, 5% of the phenylene units in the typical poly(fluorine-*co*-phenylene) structure substituted with BT units chain (P5, chemical structure shown in Fig. 5.1), can be used to design DNA sensor [65, 66]. The emission of PFPB is predominantly blue in dilute conditions, and emit green light for concentrated solutions. A similar blue to green color change occurs upon addition of either ssDNA or dsDNA, and these color

changes can be used to generate calibration curves that can be used to determine the concentration of DNA in solution [66]. When combined with a Cy5-labeled PNA probe (PNA-Cy5), the polymer can be used to design a three-color DNA detection assay. PFPB/PNA-Cy5 emits blue color in the absence of ssDNA, and it emits green light due to the interchain FRET characteristic of PFPB when noncomplementary DNA is present (ssDNAN_C). When bound to the complementary ssDNA_C, PFBT/PNA-Cy5 emits a red color due to the intramolecular FRET of PFBT followed by a subsequent FRET step to Cy5.

Bazan and co-workers extended the well developed homogenous cationic conjugated polymers (CCPs)-PNA/DNA sensory method to solid platforms [67]. They utilized surface-bound PNA probe labeled with Cy5 to hybridize complementary ssDNA (ssDNAc), after addition of synthesized poly[9,9'-bis-(6"-(N.N.N-trimethylammonium) hexyl) fluorene-co-alt-4.7-(2.1.3.-benzothiadiazole) dibromide] (PFBT) to ssDNAc/PNA-Cy5, an intense red emission from Cy5 was observed. There was no energy transfer for noncomplementary ssDNA/PNA-Cy5 under the same experimental conditions. In order to demonstrate that the lightharvesting properties of PFBT could be incorporated into platforms suitable for microarray technologies, they used unlabeled PNA chemically binding to surfacemodified glass surface, and the target ssDNA was labeled with Cy5, when PNA hybridized with complementary ssDNA, excitation of PFBT resulted in FRET to Cy5, while no Cy5 emission was observed when the surface was treated with noncomplementary ssDNA-Cy5. Moreover, they designed a solid-state assay that did not require labeled target ssDNA. The design strategy was that ssDNA hybridization to immobilized PNA resulted in a change in surface charge from neutral to negative. Exposure of the surface to the cationic PFBT resulted in electrostatic binding only on those surfaces that bear the duplex structures. By measuring the PFBT emission, it is possible to detect the target DNA. The selectivity of this method relies on the successful removal of CPE from nonhybridized PNA surfaces.

Building on the above work, in 2007 Bazan et al. [68] created a PNA microarray to detect unlabeled target DNA utilizing PFBT as optical transducer where the basic sensing strategy was depicted in Fig. 5.8a. Since PFBT has two separated absorption bands, at 330 and 455 nm, and that it also has absorption at 488 nm both in solution and in solid state. By exciting the conjugated polymer at 488 nm on a commercial microarray scanner, the presence of the target is directly indicated by the fluorescence emission of the polymer. The long target complementary DNA derived from the pathogenic bacteria was also detected with this protocol. A representative image and related information are shown in Fig. 5.8b. Each spot has a diameter of ca. 200 mm. The probe sequences in Fig. 5.8b correspond to segments found in Staphylococcus aureus and Staphylococcus epidermidis. As expected, higher signals were obtained using completely matched DNA sequence. Furthermore, the sensitivity was also improved in comparison to previous reported label-free DNA microarrays based on cationic polythiophene [69]. All these features including high sensitivity, eliminating the necessity of target labeling, and detection on a commercial microarray scanner directly made this simple and rapid methodology highly promising for the future of genomics research.



Fig. 5.8 (a) Label-free ssDNA detection using immobilized PNA and PFBT. *a*: Surface bound PNA (shown in *yellow*), *b*: hybridization with ssDNA (shown in *blue*), *c*: electrostatic adsorption of PFBT onto the PNA/ssDNA surface (Reprinted with permission from Ref. [5]. Copyright 2008 Springer). (b) Array results obtained from hybridization with a mixture of two unlabeled oligomeric targets, 5'-TGCGAAAAAA GTCTGAATTGCAAGGGAGCAG-3' (probe 2 match) and 5'-TGCAACTCATCT GGTTTAGGATCTGGTTGA-3' (probe 4 match). *Left*: Layout of PNA probes. *1*: negative control, 5'-ACTTGTAAATGGTAG-3'; *2*: positive control, 5'-CTGCTCCCTGCAATT-3'; *3*: negative control, 5'-GATATTCGCACTGA-3'; *4*: positive control, 5'-TCAACCAGATCCTA-3'; and *5*: negative control, 5'-TCAAGCAAATCCCAT-3'. *Right*: Scanned array image of polymer emission (Reprinted with permission from Ref. [68]. Copyright 2007 Nature Publishing Group)

5.3.4 PCR-Associated PF-Based DNA SNP and Methylation Detection

SNP is the most common sequence variation in the human genome. SNPs can be used as genetic markers for mapping genes, defining population structure, and performing gene association studies, as well as a fundamental tool in drug discovery and identification of genetic and inherited diseases [70–72]. Our recent studies showed that conjugated polyelectrolytes can be used for detecting SNP, and the detection model for SNPs is indicated in Fig. 5.9 based on FRET between PFP and fluorescein using a single-base extension (SBE) reaction [73]. The target DNA



Fig. 5.9 Schematic representation of SNP genotyping based on CCP and allelic-specific primer extension, and DNA methylation detection (Reprinted with permission from Ref. [16]. Copyright (2010) American Chemical Society)

fragment is part of p53 exon8 containing a polymorphic site, where the nucleotide G in a wild-type target is replaced by A in a mutant. The 3'-terminal base of the probe is T that is complementary to mutant type target sequence and is not complementary to the wild type target at the 3'-terminus. For the mutant target, the probe is fully complementary, and the fluorescein-labeled base G is incorporated into the probe by an extension reaction in the presence of Taq DNA polymerase. Upon adding PFP, strong electrostatic interactions between DNA and PFP bring the fluorescein close to PFP, and efficient FRET from PFP to fluorescein occurs. For the wild target, the 3'-terminal base of the probe is not complementary, thus the base extension reaction is blocked and the dGTP-FI remains in the solution. In this case, upon adding the PFP, FRET from PFP to fluorescein is inefficient. The method has high sensitivity and can detect as low as 2% allele frequency. The sensing platform was employed to discriminate the SNP genotypes of 76 individuals of Chinese ancestry [74].

Replacing the single base extension reaction by incorporating dGTP-Fl and dUTP-Fl during the PCR extension, one step SNP detection was achieved which can be used to detect genomic DNA. In this approach, when PCR amplification occurs, dGTP-Fl and dUTP-Fl are incorporated into formed PCR amplicons. By detecting the existence of PCR product with the FRET signaling from PFP to fluorescein-labeled PCR amplicons, the SNP genotype can be identified. The genotyping of 50 ng of genomic DNA from a human lung cancer cell is easily detected using this method [75].

Along these lines, we designed an improved method for multi-color and one-tube SNP genotyping assays based on PFP/DNA assemblies using fluorescein-modified



Fig. 5.10 (a) Emission spectra from solutions containing PFP and extension products of homozygous A, heterozygous G/A, and homozygous G. No-template control (NTC) was used as the blank. The excitation wavelength is 380 nm. Inset in Figure is the images of extension products under a 300 nm UV light with a 550–650 nm band-pass optical filter (Reprinted from Ref. [76]. With permission by the Royal Society of Chemistry). (b) FRET ratio (I_{530nm}/I_{422nm}) from solutions containing PFP and single nucleotide base extension products of methylated plasmid and nonmethylated plasmid using methylation specific probes in HEPES buffer solution. The probes used in extension reactions were p16-1m for site 1, p16-2m for site 2, and p16-3m for site 3 (Reprinted with permission from Ref. [81]. Copyright (2008) American Chemical Society)

dUTP and Cy3-modified dCTP in single base extension reactions [76]. A sequence on exon 8 of human p53 containing a polymorphic site is used as a DNA target in which the nucleotide G in the wild type is replaced by A in the mutant target. This G-A point mutant alters the coding of an amino acid (Arg-Trp) [77]. Thus three kinds of SNP genotypes are possible: homozygous G, heterozygous G/A, and homozygous A. The dye incorporated to probe DNA by polymerase is dependent on the template DNA and whether it is homozygous or heterozygous. Multi-step FRET is present if both dyes are incorporated to the probe DNA in one tube, the corresponding emission spectra and images of extension products with PFP dropped onto a glass slide under 300 nm UV light with transmission mode are shown in Fig. 5.10a. As a high throughput system (HTS)-compatible and label-free technique, the method is promising for the sensitive detection of practical SNP samples.

DNA methylation is an important component of epigenetic regulation that can be used as a biomarker for early cancer diagnosis and determination of specific cancer types [78–80]. Our recent studies showed that conjugated polyelectrolytes can be involved to detect DNA methylation [81]. The system relies on a highly selective single base extension reaction and significant optical amplification of conjugated polymers, which is illustrated in Fig. 5.9. Bisulfite treatment changes nonmethylated C into U but does not influence the methylated C, and the resulting U is substituted by T after PCR amplification [82]. Thus, the DNA methylation status can convert into a C/T polymorphism. After single nucleotide base extension with Taq polymerase in the presence of dGTP-Fl and methylated DNA, but not for nonmethylated DNA. Upon addition of PFP, strong electrostatic interactions between DNA and PFP bring the fluorescein close to PFP, and efficient FRET from PFP to fluorescein occurs for the methylated DNA but not for nonmethylated DNA.

The methylation status of a specific CpG site can be monitored by fluorescence spectra of the PFP or fluorescein emission changes in aqueous solutions. We established this platform by using a pUC57 plasmid containing a 283-bp DNA fragment from the promoter region of a tumor-suppressor gene (p16) as a model. Plasmid pUC57 that carried the sequence was fully methylated by methylase (M.SssI). The FRET ratios (I_{530nm}/I_{422nm}) for specific extension are three to five times higher than that of the nonspecific extension (*c.f.* Fig. 5.10b), which demonstrates the good selectivity of this assay method for the detection of the methylation status of specific CpG sites. The improved fluorescein emission by FRET shows the optical amplification by PFP, and analyte DNA at the picomolar level can be easily detected. Based on the detection model, methylation status of three CpG sites in the p16 promoter region of the human colon cancer cell line HT29 has been detected. As expected, significant FRET signaling for the methylation-specific probe implies that hypermethylation occurs in the p16 promoter region of HT29 cells.

Utilizing the novel detection strategy, the methylated statuses of the p16, HPP1 and GALR2 promoters of five cancer cell lines (HT29, HepG2, A498, HL60 and M17) were assayed. Genomic DNA from cancer cells is pretreated with a methylation-sensitive restriction endonuclease, followed by PCR amplification in the presence of fluorescein-labeled dNTP and Taq polymerase. The PCR only occurs for methylated DNA, and DNA methylation of the gene sequence of interest is detected as a result of FRET between CCP and fluorescein that is incorporated into DNA. The obtained assay results provide an association study between the cancers and susceptibility genes, which shows great potential for early cancer diagnosis [83].

5.4 Summary and Outlook

In summary, the recent reports presented in this chapter show that substantial efforts have been placed in developing homogeneous and heterogeneous non-amplified genomic DNA assays that take advantage of the optical amplification afforded by CPEs. Cationic conjugated polymers have great potential to be incorporated into the widely used microarray technology for simplified process with improved detection sensitivity, and the combination will open unique possibilities in efficient DNA detection. Although we have described a number of possible applications in the present chapter, there is considerable room for improvement in the intrinsic emissive polymer properties to enhance the detection sensitivity. Therefore, rationally design and synthesize functionalized conjugated polymers is still one of the challenges in polymer science, and optimizing CP structure with a better understanding of the forces that control the association between conjugated polyelectrolytes and DNA will yield superior detection platforms. Moreover, widely use of an optimized fluorescence detection procedure based on a high-intensity blue diode as the excitation source should display more sensitive and specific detection capability. Furthermore,

convenient methods that are suitable for rapid assessment of the identity of SNP, and DNA methylation without the need of DNA amplification are highly desirable. We believe that there is great promise for the use of conjugated polymers-based fluorescence sensors to accomplish this objective in future investigations, which will broaden the application field of conjugated polymers and offer new assay strategies for detecting genomic disorders.

References

- 1. Heeger, A.J.: Semiconducting and metallic polymers: the fourth generation of polymeric materials (nobel lecture). Angew. Chem. Int. Ed. 40, 2591–2611 (2001)
- Swager, T.M.: The molecular wire approach to sensory signal amplification. Acc. Chem. Res. 31, 201–207 (1998)
- McQuade, D.T., Pullen, A.E., Swager, T.M.: Conjugated polymer-based chemical sensors. Chem. Rev. 100, 2537–2574 (2000)
- 4. Thomas, S.W., Joly, G.D., Swager, T.M.: Chemical sensors based on amplifying fluorescent conjugated polymers. Chem. Rev. **107**, 1339–1386 (2007)
- 5. Bernards, D.A., Malliaras, G.G.: Organic Semiconductors in Sensor Applications, vol. 107. Springer, Berlin (2008)
- Achyuthan, K.E., Bergstedt, T.S., Chen, L., Jones, R.M., Kumaraswamy, S., Kushon, S.A., Ley, K.D., Lu, L., McBranch, D., Mukundan, H., Rininsland, F., Shi, X., Xia, W., Whitten, D.G.: Fluorescence superquenching of conjugated polyelectrolytes: applications for biosensing and drug discovery. J. Mater. Chem. 15, 2648–2656 (2005)
- 7. Liu, B., Bazan, G.C.: Homogeneous fluorescence-based DNA detection with water-soluble conjugated polymers. Chem. Mater. 16, 4467–4476 (2004)
- Ho, H.A., Najari, A., Leclerc, M.: Optical detection of DNA and proteins with cationic polythiophenes. Acc. Chem. Res. 41, 168–178 (2008)
- 9. Zhou, Q., Swager, T.M.: Method for enhancing the sensitivity of fluorescent chemosensors: energy migration in conjugated polymers. J. Am. Chem. Soc. **117**, 7017–7018 (1995)
- 10. Leclerc, M.: Optical and electrochemical transducers based on functionalized conjugated polymers. Adv. Mater. **11**, 1491–1498 (1999)
- Duarte, A., Pu, K.-Y., Liu, B., Bazan, G.C.: Recent advances in conjugated polyelectrolytes for emerging optoelectronic applications. Chem. Mater. 23, 501–515 (2010)
- Albert, K.J., Walt, D.R.: High-speed fluorescence detection of explosives-like vapors. Anal. Chem. 72, 1947–1955 (2000)
- He, F., Tang, Y., Wang, S., Li, Y., Zhu, D.: Fluorescent amplifying recognition for DNA Gquadruplex folding with a cationic conjugated polymer: a platform for homogeneous potassium detection. J. Am. Chem. Soc. 127, 12343–12346 (2005)
- Bunz, U.H.F.: Poly(aryleneethynylene)s: syntheses, properties, structures, and applications. Chem. Rev. 100, 1605–1644 (2000)
- Ho, H.-A., Béra-Abérem, M., Leclerc, M.: Optical sensors based on hybrid DNA/conjugated polymer complexes. Chem. Euro. J. 11, 1718–1724 (2005)
- Duan, X., Liu, L., Feng, F., Wang, S.: Cationic conjugated polymers for optical detection of DNA methylation, lesions, and single nucleotide polymorphisms. Acc. Chem. Res. 43, 260–270 (2010)
- Feng, X., Liu, L., Wang, S., Zhu, D.: Water-soluble fluorescent conjugated polymers and their interactions with biomacromolecules for sensitive biosensors. Chem. Soc. Rev. 39, 2411–2419 (2010)
- Feng, F., He, F., An, L., Wang, S., Li, Y., Zhu, D.: Fluorescent conjugated polyelectrolytes for biomacromolecule detection. Adv. Mater. 20, 2959–2964 (2008)

- 19. Lee, J., Jun, H., Kim, J.: Polydiacetylene-liposome microarrays for selective and sensitive mercury(II) detection. Adv. Mater. 21, 3674–3677 (2009)
- Lee, J., Kim, H.J., Kim, J.: Polydiacetylene liposome arrays for selective potassium detection. J. Am. Chem. Soc. 130, 5010–5011 (2008)
- Lee, K., Rouillard, J.M., Pham, T., Gulari, E., Kim, J.: Signal-amplifying conjugated polymer-DNA hybrid chips. Angew. Chem. Int. Ed. 46, 4667–4670 (2007)
- 22. Yang, J.-S., Swager, T.M.: Porous shape persistent fluorescent polymer films: an approach to TNT sensory materials. J. Am. Chem. Soc. **120**, 5321–5322 (1998)
- Dubertret, B., Calame, M., Libchaber, A.J.: Single-mismatch detection using gold-quenched fluorescent oligonucleotides. Nat. Biotech 19, 365–370 (2001)
- 24. Wang, J.: Survey and summary. Nucleic Acids Res. 28, 3011-3016 (2000)
- Daar, A.S., Thorsteinsdottir, H., Martin, D.K., Smith, A.C., Nast, S., Singer, P.A.: Top ten biotechnologies for improving health in developing countries. Nat. Genet. 32, 229–232 (2002)
- 26. Tumor Analysis Best Practices Working Group: Expression profiling—best practices for data generation and interpretation in clinical trials. Nat. Rev. Genet. **5**, 229–237 (2004)
- 27. Gershon, D.: Microarray technology: an array of opportunities. Nature 416, 885–891 (2002)
- Heller, M.J.: DNA microarray technology: devices, systems, and applications. Annu. Rev. Biomed. Eng. 4, 129–153 (2002)
- Sassolas, A., Leca-Bouvier, B.D., Blum, L.J.: DNA biosensors and microarrays. Chem. Rev. 108, 109–139 (2008)
- Pinto, M.R., Schanze, K.S.: Conjugated polyelectrolytes: synthesis and applications. Synthesis 9, 1293–1309 (2002)
- Patil, A.O., Ikenoue, Y., Wudl, F., Heeger, A.J.: Water soluble conducting polymers. J. Am. Chem. Soc. 109, 1858–1859 (1987)
- 32. Ho, H.-A., Boissinot, M., Bergeron, M.G., Corbeil, G., Doré, K., Boudreau, D., Leclerc, M.: Colorimetric and fluorometric detection of nucleic acids using cationic polythiophene derivatives. Angew. Chem. Int. Ed. 41, 1548–1551 (2002)
- Nilsson, K.P.R., Rydberg, J., Baltzer, L., Inganos, O.: Self-assembly of synthetic peptides control conformation and optical properties of a zwitterionic polythiophene derivative. Proc. Natl. Acad. Sci. U.S.A. 100, 10170–10174 (2003)
- 34. McQuade, D.T., Hegedus, A.H., Swager, T.M.: Signal amplification of a turn-on sensor: harvesting the light captured by a conjugated polymer. J. Am. Chem. Soc. **122**, 12389–12390 (2000)
- 35. Bunz, U.H.F.: Poly(p-phenyleneethynylene)s by alkyne metathesis. Acc. Chem. Res. 34, 998–1010 (2001)
- 36. Levitsky, I.A., Kim, J., Swager, T.M.: Energy migration in a poly(phenylene ethynylene): determination of interpolymer transport in anisotropic Langmuir Blodgett films. J. Am. Chem. Soc. 121, 1466–1472 (1999)
- Shi, S., Wudl, F.: Synthesis and characterization of a water-soluble poly(p-phenylenevinylene) derivative. Macromolecules 23, 2119–2124 (1990)
- Stork, M., Gaylord, B.S., Heeger, A.J., Bazan, G.C.: Energy transfer in mixtures of watersoluble oligomers: effect of charge, aggregation, and surfactant complexation. Adv. Mater. 14, 361–366 (2002)
- 39. Kim, S., Jackiw, J., Robinson, E., Schanze, K.S., Reynolds, J.R., Baur, J., Rubner, M.F., Boils, D.: Water soluble photo- and electroluminescent alkoxy-sulfonated poly(p-phenylenes) synthesized via palladium catalysis. Macromolecules **31**, 964–974 (1998)
- Gaylord, B.S., Wang, S., Heeger, A.J., Bazan, G.C.: Water-soluble conjugated oligomers: effect of chain length and aggregation on photoluminescence-quenching efficiencies. J. Am. Chem. Soc. 123, 6417–6418 (2001)
- Chen, L., McBranch, D.W., Wang, H.-L., Helgeson, R., Wudl, F., Whitten, D.G.: Highly sensitive biological and chemical sensors based on reversible fluorescence quenching in a conjugated polymer. Proc. Natl. Acad. Sci. U.S.A. 96, 12287–12292 (1999)
- 42. Tan, C., Pinto, M.R., Schanze, K.S.: Photophysics, aggregation and amplified quenching of a water-soluble poly(phenylene ethynylene). Chem. Commun. **5**, 446–447 (2002)

- 43. Yang, R., Garcia, A., Korystov, D., Mikhailovsky, A., Bazan, G.C., Nguyen, T.-Q.: Control of interchain contacts, solid-state fluorescence quantum yield, and charge transport of cationic conjugated polyelectrolytes by choice of anion. J. Am. Chem. Soc. 128, 16532–16539 (2006)
- 44. Yu, M., Tang, Y., He, F., Wang, S., Zheng, D., Li, Y., Zhu, D.: Synthesis of water-soluble dendritic conjugated polymers for fluorescent DNA assays. Macromol. Rapid Commun. 27, 1739–1745 (2006)
- Chen, L., Xu, S., McBranch, D., Whitten, D.: Tuning the properties of conjugated polyelectrolytes through surfactant complexation. J. Am. Chem. Soc. 122, 9302–9303 (2000)
- Kushon, S.A., Ley, K.D., Bradford, K., Jones, R.M., McBranch, D., Whitten, D.: Detection of DNA hybridization via fluorescent polymer superquenching. Langmuir 18, 7245–7249 (2002)
- Kushon, S.A., Bradford, K., Marin, V., Suhrada, C., Armitage, B.A., McBranch, D., Whitten, D.: Detection of single nucleotide mismatches via fluorescent polymer superquenching. Langmuir 19, 6456–6464 (2003)
- Yang, C.J., Pinto, M., Schanze, K., Tan, W.: Direct synthesis of an oligonucleotidepoly(phenylene ethynylene) conjugate with a precise one-to-one molecular ratio. Angew. Chem. Int. Ed. 44, 2572–2576 (2005)
- 49. Lakowicz, J.R.: Principles of Fluorescence Spectroscopy, 2nd edn. Kluwer Academic/Plenum Publishers, New York (1999)
- 50. Nilsson, K.P.R., Inganas, O.: Chip and solution detection of DNA hybridization using a luminescent zwitterionic polythiophene derivative. Nat. Mater. 2, 419–424 (2003)
- Dore, K., Dubus, S., Ho, H.-A., Levesque, I., Brunette, M., Corbeil, G., Boissinot, M., Boivin, G., Bergeron, M.G., Boudreau, D., Leclerc, M.: Fluorescent polymeric transducer for the rapid, simple, and specific detection of nucleic acids at the zeptomole level. J. Am. Chem. Soc. 126, 4240–4244 (2004)
- 52. Dubus, S., Gravel, J.-F., Le Drogoff, B., Nobert, P., Veres, T., Boudreau, D.: PCP-free DNA detection using a magnetic bead-supported polymeric transducer and microelectromagnetic traps. Anal. Chem. **78**, 4457–4464 (2006)
- Ho, H.A., Dore, K., Boissinot, M., Bergeron, M.G., Tanguay, R.M., Boudreau, D., Leclerc, M.: Direct molecular detection of nucleic acids by fluorescence signal amplification. J. Am. Chem. Soc. 127, 12673–12676 (2005)
- Najari, A., Ho, H.A., Gravel, J.F., Nobert, P., Boudreau, D., Leclerc, M.: Reagentless ultrasensitive specific DNA array detection based on responsive polymeric biochips. Anal. Chem. 78, 7896–7899 (2006)
- Dueymes, C., Décout, J.L., Peltié, P., Fontecave, M.: Fluorescent deazaflavin–oligonucleotide probes for selective detection of DNA. Angew. Chem. Int. Ed. 41, 486–489 (2002)
- 56. Dudley, A.M., Aach, J., Steffen, M.A., Church, G.M.: Measuring absolute expression with microarrays with a calibrated reference sample and an extended signal intensity range. Proc. Natl. Acad. Sci. U.S.A. 99, 7554–7559 (2002)
- Gaylord, B.S., Heeger, A.J., Bazan, G.C.: DNA detection using water-soluble conjugated polymers and peptide nucleic acid probes. Proc. Natl. Acad. Sci. U.S.A. 99, 10954–10957 (2002)
- Gaylord, B.S., Massie, M.R., Feinstein, S.C., Bazan, G.C.: SNP detection using peptide nucleic acid probes and conjugated polymers: applications in neurodegenerative disease identification. Proc. Natl. Acad. Sci. U.S.A. **102**, 34–39 (2005)
- Gaylord, B.S., Heeger, A.J., Bazan, G.C.: DNA hybridization detection with water-soluble conjugated polymers and chromophore-labeled single-stranded DNA. J. Am. Chem. Soc. 125, 896–900 (2003)
- Wang, Y., Liu, B.: Silica nanoparticle assisted DNA assays for optical signal amplification of conjugated polymer based fluorescent sensors. Chem. Commun. 34, 3553–3555 (2007)
- Wang, S., Gaylord, B.S., Bazan, G.C.: Fluorescein provides a resonance gate for fret from conjugated polymers to DNA intercalated dyes. J. Am. Chem. Soc. 126, 5446–5451 (2004)
- 62. He, F., Tang, Y., Yu, M., Feng, F., An, L., Sun, H., Wang, S., Li, Y., Zhu, D., Bazan, G.C.: Quadruplex-to-duplex transition of g-rich oligonucleotides probed by cationic water-soluble conjugated polyelectrolytes. J. Am. Chem. Soc. **128**, 6764–6765 (2006)

- He, F., Feng, F., Duan, X., Wang, S., Li, Y., Zhu, D.: Selective and homogeneous fluorescent DNA detection by target-induced strand displacement using cationic conjugated polyelectrolytes. Anal. Chem. 80, 2239–2243 (2008)
- 64. Feng, X., Liu, L., Yang, Q., Wang, S.: Dual-amplified sensitive DNA detection based on conjugated polymers and recyclable autocatalytic hybridization of DNA. Chem. Commun. 47, 5783–5785 (2011)
- 65. Liu, B., Bazan, G.C.: Interpolyelectrolyte complexes of conjugated copolymers and DNA platforms for multicolor biosensors. J. Am. Chem. Soc. **126**, 1942–1943 (2004)
- Hong, J.W., Hemme, W.L., Keller, G.E., Rinke, M.T., Bazan, G.C.: Conjugated-polymer/DNA interpolyelectrolyte complexes for accurate DNA concentration determination. Adv. Mater. 18, 878–882 (2006)
- Liu, B., Bazan, G.C.: Methods for strand-specific DNA detection with cationic conjugated polymers suitable for incorporation into DNA chips and microarrays. Proc. Natl. Acad. Sci. U.S.A. 102, 589–593 (2005)
- Sun, C., Gaylord, B.S., Hong, J.W., Liu, B., Bazan, G.C.: Application of cationic conjugated polymers in microarrays using label-free DNA targets. Nat. Protocols 2, 2148–2151 (2007)
- 69. Raymond, F., Ho, H.-A., Peytavi, R., Bissonnette, L., Boissinot, M., Picard, F., Leclerc, M., Bergeron, M.: Detection of target DNA using fluorescent cationic polymer and peptide nucleic acid probes on solid support. BMC Biotech. 5, 10 (2005)
- Kim, S., Misra, A.: SNP genotyping: technologies and biomedical applications. Annu. Rev. Biomed. Eng. 9, 289–320 (2007)
- Ding, C.: Other applications of single nucleotide polymorphisms. Trends Biotechnol. 25, 279–283 (2007)
- 72. The International HapMap Consortium: A haplotype map of the human genome. Nature **437**, 1299–1320 (2005)
- Duan, X., Li, Z., He, F., Wang, S.: A sensitive and homogeneous SNP detection using cationic conjugated polymers. J. Am. Chem. Soc. 129, 4154–4155 (2007)
- 74. Duan, X., Yue, W., Liu, L., Li, Z., Li, Y., He, F., Zhu, D., Zhou, G., Wang, S.: Single-nucleotide polymorphism (SNP) genotyping using cationic conjugated polymers in homogeneous solution. Nat. Protocols 4, 984–991 (2009)
- Duan, X., Liu, L., Wang, S.: Homogeneous and one-step fluorescent allele-specific PCR for SNP genotyping assays using conjugated polyelectrolytes. Biosens. Bioelectron. 24, 2095–2099 (2009)
- Duan, X., Wang, S., Li, Z.: Conjugated polyelectrolyte-DNA complexes for multi-color and one-tube SNP genotyping assays. Chem. Commun. 21, 1302–1304 (2008)
- 77. Li, Z.P., Tsunoda, H., Okano, K., Nagai, K., Kambara, H.: Microchip electrophoresis of tagged probes incorporated with one-colored ddNTP for analyzing single-nucleotide polymorphisms. Anal. Chem. 75, 3345–3351 (2003)
- 78. Bird, A.: DNA methylation patterns and epigenetic memory. Genes Dev. 16, 6–21 (2002)
- Reik, W., Walter, J.: Genomic imprinting: parental influence on the genome. Nat. Rev. Genet. 2, 21–32 (2001)
- 80. Robertson, K.D., Wolffe, A.P.: DNA methylation in health and disease. Nat. Rev. Genet. 1, 11–19 (2000)
- Feng, F., Wang, H., Han, L., Wang, S.: Fluorescent conjugated polyelectrolyte as an indicator for convenient detection of DNA methylation. J. Am. Chem. Soc. 130, 11338–11343 (2008)
- Sadri, R., Hornsby, P.J.: Rapid analysis of DNA methylation using new restriction enzyme sites created by bisulfite modification. Nucleic Acids Res. 24, 5058–5059 (1996)
- Feng, F., Liu, L., Wang, S.: Fluorescent conjugated polymer-based FRET technique for detection of DNA methylation of cancer cells. Nat. Protocols 5, 1255–1264 (2010)

Part III Detection Strategies – Sensors

Chapter 6 Optical Detection of Non-amplified Genomic DNA

Di Li and Chunhai Fan

Abstract Nucleic acid sequences are unique to every living organisms including animals, plants and even bacteria and virus, which provide a practical molecular target for the identification and diagnosis of various diseases. DNA contains heterocyclic rings that has inherent optical absorbance at 260 nm, which is widely used to quantify single and double stranded DNA in biology. However, this simple quantification method could not differentiate sequences; therefore it is not suitable for sequence-specific analyte detection. In addition to a few exceptions such as chiralrelated circular dichroism spectra, DNA hybridization does not produce significant changes in optical signals, thus an optical label is generally needed for sequencespecific DNA detection with optical means. During the last two decades, we have witnessed explosive progress in the area of optical DNA detection, especially with the help of simultaneously rapidly developed nanomaterials. In this chapter, we will summarize recent advances in optical DNA detection including colorimetric, fluorescent, luminescent, surface plasmon resonance (SPR) and Raman scattering assays. Challenges and problems remained to be addressed are also discussed.

6.1 Introduction

Optical detection has long been a powerful tool for biomedical applications. Optical sensors based on bio-recognition events possess some unique advantages over other analytical methods. To name a few, light beam produces much less interference to biological events compared with electronic, electrochemical or magnetic signal sources. Optical signals are immune to electromagnetic interference, capable of

D. Li • C. Fan (⊠)

Laboratory of Physical Biology, Shanghai Institute of Applied Physics, Chinese Academy of Sciences, Shanghai 201800, China e-mail: lidi@sinap.ac.cn; fchh@sinap.ac.cn

G. Spoto and R. Corradini (eds.), *Detection of Non-Amplified Genomic DNA*, Soft and Biological Matter, DOI 10.1007/978-94-007-1226-3_6, © Springer Science+Business Media Dordrecht 2012

performing remote sensing and providing multiplexed detection within a single device. Optical labels are safe and ease of manipulation compared with classical radioactive labels.

The basic principle of DNA optical sensor is to generate a recordable optical signal upon sequence-specific hybridization. While a few excellent reviews have given comprehensive summarize of the history and updated process of DNA sensors and DNA microarrays [1–7], we will focus on some recent advances on optical detection of non-amplified genomic DNA in this chapter. In the end, we will present a brief outlook for questions remained to be addressed and future potential for these optical DNA sensors.

6.2 Optical Methods for DNA Analysis

6.2.1 Colorimetric Assays for Nucleic Acid Detection Using Au Nanoparticles

Colorimetric biosensors, which mean a detection method based on the color change induced by a bio-recognition event, are particularly impressive since they minimize or eliminate the necessity of using expensive and complicated instruments, sometimes the detection can even be realized by unaided naked eyes.

The milestone work was the development of colorimetric DNA sensor using Au nanoparticles (AuNPs) from Mirkin's group in 1996, which in a sense opens the era of bionanotechnology [8, 9]. Herein we first provide a short description of the interesting optical propertiesy of AuNPs in order to better understand the principle of AuNPs-based colorimetric DNA sensor. AuNPs, or Au colloid has a long history that could date back to at least to the fourth century AD when it was used to stain the Lycurgus cup owing to its brilliant wine color [7, 10]. But correlation of the color of Au colloidal with its small size was first was heralded by Michael Faraday in 1857 in a lecture entitled "Experimental Relations of Gold (and other Metals) to Light" [11]. We must pay a tribute to his insight, even 150 years ago, he noted, "The state of division of these particles must be extreme; they have not as yet been seen by any power of the microscope". The explosion of nanotechnology in the last two decades has confirmed that Au colloids prepared by Faraday were actually spherical AuNPs with a diameter of 10–100 nm. The intense red color of AuNPs is due to the interaction of incident light with a collective oscillation of free electrons in the particles known as localized surface plasmon resonance. When AuNPs aggregate, color of AuNPs changed from red to blue, which was attributed to the interaction of interacting particle surface plasmons and aggregate scattering properties. This simple red-to-blue color change has motivated scientists to design colorimetric sensing strategies by utilizing the analyte-induced AuNPs aggregation. For example, AuNPs loaded with antibodies or antigens has been used in commercialized test strips for HIV or pregnancy testing.



Fig. 6.1 Upon hybridization with the complementary target DNA, probe DNA -functionalized AuNPs will aggregate (**a**), resulting in a *red*-to-*blue* color change (**b**). The aggregation process is accompanied by a sharp increase of the absorbance in the melting profile as compared with regularly DNA hybridization in solution phase (**c**) (Reproduced with permission from Ref. [7]. Copyright (2007) American Chemical Society)

Mirkin's group pioneered the use of aggregation-induced color change of AuNPs in nucleic acid detection. Briefly, AuNPs of 13 nm in diameter was modified with thiolated-probe DNA via Au-S chemistry. Thus upon the addition of target DNA that cross-linked the probe DNA-modified AuNPs through sequence-specific hybridization, a red-to-purple color change is therefore observed. More interestingly, given the nature of DNA hybridization is reversible depending temperature, the aggregation process is reversible; denaturation of the hybridized DNA duplex at elevated temperature (above the melting temperature, Tm) causes the dissociation of aggregates into dispersed AuNPs (Fig. 6.1). Further studies indicated that the melting profiles of the nanoparticle-labeled DNA aggregates were extraordinarily sharp, occurring over a temperature range much narrower than the transition for unlabeled or conventional fluorophore-labeled DNA [12, 13]. By elaborate optimization of influence factors such as AuNPs size, densities of probe DNA on AuNPs and salt concentrations, this colorimetric assay enables the detection of target DNA with a limit of detection (LOD) down to nM.

This simple colorimetric technique offered several advantages for quantitative analysis of nucleic acids: (1) AuNPs is ease of conjugation with thiolated probe DNA. The large surface area of nanomaterials ensures a spherical AuNPs of 13 nm load hundreds of capture probe DNA, meanwhile the three dimensional (3D) assembly of probe with AuNPs lowers the steric hindrance and favors targetprobe hybridization; (2) AuNPs possesses extremely high extinction coefficients (e.g., $2.7 \times 10^8 \text{ M}^{-1} \text{ cm}^{-1}$ at ~520 nm for 13 nm spherical AuNPs, ca. 1,000 times higher than those of organic dyes) [14], thus slight aggregation may result in intense color, suggesting relatively high sensitivity; (3) The simple red-to-blue color change readout does not require expensive, sophisticated instrumentation; (4) last but not least, the sharp absorbance changes in Tm profile could be utilized to discriminate between perfectly matched target oligonucleotides and targets with single basepair mismatches, holding huge promise in single nucleotide polymorphisms (SNPs) analysis. Therefore, ever since first developed by Mirkin and co-workers, this platform has been increasingly applied for the detection of a wide spectrum of analytes ranging from nucleic acids to proteins, saccharides, small molecules, metal ions, and even cells. It is quickly becoming an important alternative detection tool, holding great potential in clinical diagnostics, drug discovery and environmental contaminant analysis [15, 16].

In addition to this target nucleic acid-induced cross-linking aggregations, Rothberg and Li have proposed an alternative colorimetric method using unmodified citrate-reduced AuNPs in 2003 [17–19]. They found that single-stranded (ss) DNA can be attached on citrate-reduced AuNPs through DNA base-gold interactions and can stabilize AuNPs electrostatically from salt-induced aggregation. In contrast, double stranded (ds-) DNA shows little binding affinity to AuNPs, and, therefore it provides little stabilization against salt. In other words, at an appropriate high salt concentration (e.g., 200 mM of NaCl), citrate-capped AuNPs are stabilized in the presence of ss-DNA, but aggregate in the presence of ds-DNA (Fig. 6.2). This simple phenomenon realized a rapid, label-free sequence-specific detection of target DNA with a LOD also of nM range. Later on, our group further investigated this system and provided a slightly modified mechanism [20–22]. The different affinity of ss- and ds-DNA to AuNPs is attributed to at least three effects. First, DNA bases attach to AuNPs with high affinity via Au-N coordination. Importantly, DNA bases are encapsulated within the negatively charged phosphate backbone of ds-DNA, while they are exposed in ss-DNA. Second, the surface of AuNPs is negatively charged because they are stabilized by citrate ions. Since the charge density of ds-DNA is higher than that of ss-DNA, the electrostatic repulsion between AuNPs and DNA is larger for ds-DNA. Third, ds-DNA is much more rigid than ss-DNA; thus ss-DNA possesses higher freedom to wrap on AuNPs than ds-DNA.

6.2.2 Fluorescent Nucleic Acid Assays

6.2.2.1 Fluorescent Assays Based on Molecular Beacons

Molecular beacons (MB), which were first described in 1996 by Tyagi and Kramer [23], are a class of newly developed probes that are widely used as fluorescent probes. Tyagi and Kramer named this fluorescent DNA probe "molecular beacon",



Fig. 6.2 ds- and ss-DNA reveal different affinity of unmodified citrate-reduced negatively-charged AuNPs. ss-DNA possesses higher affinity to AuNPs, it could be attached on AuNPs surface that stabilize AuNPs against salt-induced aggregation, while ds-DNA doesn't

because it emits a light signal upon hybridization with target sequence. Generally, a typical MB is a hairpin structured single stranded oligonucleotide that is modified with a fluorophore and a quencher group at each end of its stem end. The sequence of a MB is composed with two segments including a recognition region of about 15–30 bases flanked by two short complementary stem sequences, which force the entire sequence to form a stem–loop structure at room temperature. In the absence of target DNA, the fluorophore and quencher is forced to be close proximity, where fluorescence is quenched effectively. While in the presence of target DNA that is hybridized with the loop segment of MB, and the resulting duplex structure between target DNA and loop is thermodynamically stable compared with the weaker stem helix, thus the stem-loop structure is forced to open. As a result, the spatial separation of fluorophore and quencher blocks the energy transfer and restores the fluorescence signal (Fig. 6.3a). The application of MBs in DNA sensing was covered by a number of reviews [24–26].

The distinct advantage of MB lies in its ability to real-time monitoring the quantity of target DNA, therefore MB is widely used in RT-PCR to provide a real time measurement of PCR progress [27–29]. In addition to DNA, MB was also employed for real-time quantification of mRNA both *in vitro* and *in vivo* [30, 31]. For example, with the help of confocal fluorescence microscope, MBs could provide a real-time visualization and localization the detection of mRNA in living cells [32].

Despite the variety of applications of MBs in biosensors, the basic principle of detection remains unchanged: the generation of a fluorescent signal is based on hybridization between target oligonucleotides and MBs in a 1:1 stoichiometric ratio. That is, one target oligonucleotide could open only one MB to give fluorescent



Fig. 6.3 (a) Mechanism of a molecular beacon. The MB adopts a stem-loop structure and thus holds the fluorophore (*orange*) and quencher (*blue*) in close proximity. As a result, the fluorescence emission of the fluorophore is strongly suppressed (in the absence of a target). Hybridization with target DNA opens the hairpin structure and recovers the fluorescent signal to allow readout the hybridization event (Reproduced with permission from Ref. [24]. Copyright Wiley-VCH Verlag GmbH & Co. KGaA) (b) Working principle of a basic NESA strategy. A nicking enzyme is introduced to cleaves the MB that already be opened by target DNA, resulting dissociation of target from the cleavaged beacon and in turn be released to solution. The as-released target nucleic aid could further hybridize to another beacon and initiate the second cycle of cleavage (Reproduced with permission from Ref. [34]. Copyright (2008) Oxford University Press)

signal. Given the instrument limit of commercial fluorophotometers, MBs could only detect nucleic acids down to nanomolar range.

Recently, a strategy called nicking enzyme signal amplification (NESA) was designed to improve the sensitivity of MB-based fluorescence assay [33–35]. Nicking enzymes are a kind of restriction endonucleases that can recognize a

specific sequence along a double-strand DNA; however, they cleave only one specific strand instead of both strands, leaving a nick in the DNA (Fig. 6.3b). In brief, this strategy introduces a nicking enzyme recognition site in the loop of MB. Upon hybridization with target nuclei acid that yields the duplex recognition domain for nicking enzyme, the nicking enzyme specifically cleaves the MB and this process induces a decreased stability of the two duplexes obtained. Thus target nucleic acid is dissociated from the cleaved beacon and released into solution. The released target nucleic acid could further hybridize to another beacon and initiate the second cycle of cleavage. Eventually, one target nucleic acid could go through many cycles, resulting in cleavage of many beacons. Therefore the limit of detection (LOD) could be significantly pushed down to picomolar range [34].

Despite the three orders of magnitude improvement in sensitivity, the NESA still suffers from one major drawback. It requires the target sequence contains a nicking enzyme recognition site, thus is not appropriate for detection of target with random sequences. Several extended visions of NESA strategy were designed for random sequence detection. For example, Xie and co-workers coupled rolling circle amplification (RCA) with NESA that further improves the LOD down to 85 fM [34]. More recently, Zhou and co-workers [36] presented another modified version of NESA named cascade enzymatic signal amplification (CESA), which they integrated a invasive signal amplification with NESA. Unlike RCA that requires a e padlock probe DNA to capture target nucleic acid, invasive signal amplification does not rely on specific recognition sequences but on a specific structure formed by the specific binding of an upstream probe and a downstream probe to a target DNA; so any target sequence can be detected.

A further improved LOD of 1 fM was achieved in this strategy. In addition to nicking enzyme, sequence-independent exo-exonuclease III and DNA polymerase were also used to digest MBs that "recycles" target nucleic acid to improve the sensitivity [37]. The coupling of MBs with enzymatic digestion cycling possesses great potential for ultrasensitive detection, real-time imaging and location of nucleic acid in-vivo.

6.2.2.2 Fluorescent Assays Based on Nanomaterials

Organic Dye Fluorophores Incorporated in Nano-Scale Matrix

While organic dyes have to be powerful fluorescent labels in bio-diagnostics, they also suffer from intrinsic drawbacks, for instance, limited sensitivity and photo-stability. Moreover, most organic dyes present a certain level of toxicity that hinders their application in vivo cellular studies and imaging. To overcome these disadvantages, researchers attempted to incorporate dyes inside silica NPs [38], which provide an effective barrier keeping the dye from the surrounding environment, both photobleaching and photodegradation phenomena that often affect conventional dyes can be minimized. Using appropriate synthetic conditions, a large number of dye molecules can be incorporated inside a single silica particle

(there can be tens of thousands of dye molecules). Even though there remain some fluorescence quenching phenomena within an NP with a large amount of dye incorporated in a small volume, the goal of obtaining a particle with brighter luminescence is largely successful. Dye-doped NPs produce a highly amplified optical signal compared with a single dye molecule.

Tan's group has pioneered the application of this dye-doped Silica NPs in nucleic acid detection and imagining [39]. They first attached a capture probe DNA on a dye-doped Silica NPs via biotin-avidin interactions. The dye-doped Silica NPs modified capture probe was further used in a sandwich-type fluorescent DNA assay [40]. Hybridization of target DNA with the capture probe brings the dye-doped silica NP to the surface. This provides a large number of dye molecules on the surface for signaling. By monitoring the luminescent intensity from the surface-bound NPs, DNA target molecules can be detected with increased sensitivity (as low as 0.8 pM). The protocol could be also used in array-based multiple DNA analysis and imaging by using different dye-doped Silica NPs. Zhou et al. [41] prepared a class of dye-doped core-shell NPs by attaching thiolated DNA probes to the surface of AuNPs. The AuNPs were then coated with a 10–15 nm silica layer through the (3-mercaptopropyl)trimethoxysilane (MPTS). Two-color DNA microarray-based detection and imaging was demonstrated using Cy3- and Cy5-doped NPs in sandwich hybridization.

Fluorescent Nanomaterials

Quantum dots (QDs) are a class of II-IV or III-V semiconductor nanoparticles with a size scale approximately less than 10 nm [42, 43]. The distinct optical property of QDs is the size-dependent fluorescence properties. QDs possess narrow sizetunable fluorescence spectra with narrow full width-at-half-maxima. The absorption spectra of QDs are broad. The combination of broad absorption spectra and narrow symmetric emission spectra enables QDs attractive for multiplex targets sensing [44]. Meanwhile, QDs have higher quantum yields and are more robust against photobleaching compared with organic dyes, thus also great attention was given to its application in imaging [45].

The size-tunable fluorescence emission and simultaneous excitation feature of QDs render them as ideal fluorophores for wavelength-and-intensity multiple labeling. About a decade ago, Nie and co-workers pioneered the use of QDs embedded beads for multiple DNA detection [46]. The principle of multiplexed optical coding is based on multicolor QDs embedded in polymer beads. The polymer beads were modified with DNA probes. Target DNA was directly labeled with a fluorescent dye. Fluorescence spectra of the single-bead level yield both the coding and the target signals. The coding signals identify the DNA sequence, whereas the target signal indicates the presence and the abundance of that sequence (Fig. 6.4).

In addition to direct labeling, QDs is mainly used as electron donor in fluorescence DNA assays. FRET (or Förster resonance energy transfer) is a through space



Fig. 6.4 Schematic illustration of DNA hybridization assays using QD-tagged beads. Multicolor optical coding for DNA could be achieved by embedding different-sized QDs into polymeric microbeads at precisely controlled ratios (Reproduced with permission from Ref. [46]. Copyright (2001) Nature Publishing Group)

dipolar coupling interaction that allows electronic energy to be transferred from a donor to an acceptor [47, 48]. The rate of energy transfer depends on the distance between the donor and the acceptor, their relative orientations, and the spectral overlap. The fluorescence emission of QDs has a spectra overlap with the excitation of regular organic dyes. Therefore by careful arrangement of their distance from DNA and of their orientation, FRET may be observed between QDs and dyes.

A typical FRET sensor for nucleic acid, involves a probe DNA attached on a QDs surface. Then a dye labeled target DNA is hybridized with the immobilized probe DNA. As a result, the target DNA brings the fluorophore in close proximity of QDs enough to raise FRET. Several FRET-based sensing strategies for nucleic acids were proposed in the last decades (Fig. 6.5). For example, Ozkan and co-workers modified a CdSe/ZnS QDs with a stem-loop structured probe DNA with an organic quencher 4-(4'-dimethylaminophenylazo) benzoic acid (DABCYL) conjugated at one end of the stem (Fig. 6.5a). In the closed state, the fluorescence of QDs was quenched by DABCYL. In the presence of target that complementary to the loop segment of the probe DNA, DABCYL was brought away from QDs, resulting recovery of fluorescence from QDs [49]. Krull's group designed a strategy to simultaneously detect two target nucleic acid sequences using FRET (Fig. 6.5b) [50]. These two targets were labeled with Cy3 and Alexa Fluor 647, respectively. Probe sequences complementary to these labeled targets were paired with a green QDs and a red QDs. Upon probe-target hybridization, the induced donor-acceptor proximity resulted in FRET-sensitized acceptor fluorescence.



Wang and co-workers have raised a sandwich-typed FRET assay for DNA (Fig. 6.5c) [51]. A streptavidin-coated CdSe/ZnS QDs with emission at 605 nm was paired with Cy5 as an acceptor. The sandwich was constructed with a biotinylated probe DNA complementary to half of target sequence and a Cy5-labeled probe complementary to the other half of target. The strong biotin–streptavidin binding interaction brought the sandwich to the surface of the QD. FRET between QD donor and Cy5 acceptor was observed. Travas-Sejdic employed a cationic polymer as a linker to electrostatically link a dye-labeled probe DNA and QDs (Fig. 6.5d) [52]. The differential interaction of ss- and ds-DNA with CdTe⁺ results in differential changes of FRET efficiency, which was used to recognize the hybridization event.

Besides II-IV semiconductor nanoparticles QDs, Silicon nanoparticlesor nanowires are another class of fluorescence QDs. While bulk silicon is not fluorescent, silicon at nanoscale could generate intense room temperature photoluminescence, which is generally believed to result from a combination of quantum confinement effects. Although of quantum yield (QY) of silicon-based QDs is relatively low (<10 %), they still possesses some unique superiority over II-IV semiconductor nanoparticles QDs in bio-labeling and imaging. For example, the potential toxicity problem of the II-VI QDs associated with release of heavy metal ions (e.g. Cd ions) has not yet been fully addressed, which limits their widespread biological and medical applications [50, 53]. Nontoxic silicon-based QDs reveal the potential as promising candidates. Intense exploration has been done to improve the QY of silicon QDs to facilitate their applications in bio-labeling.

Kortshagen and co-workers recently reported successful preparation of SiQDs with remarkably high ensemble quantum yields exceeding 60 % by using plasmaassisted synthesis with strict removal of oxygen and elaborate surface passivation [54]. Especially, we and our collaborators recently developed a new class of fluorescent silicon nanospheres (SiNSs), each containing several hundreds of SiQDs [55, 56]. The as-prepared nanospheres possess excellent aqueous dispersibility, strong fluorescence (quantum yield: >15 %), robust photo-stability and favorable biocompatibility. Nevertheless the process of synthesis SiQDs with higher QYs, their application in DNA detection assay is still far behind that of QDs, and we believe that this is a promising field for future research.

In addition to QDs, up-converting rare-earth nanophosphors (UCNPs) are another class of fluorescence nanoparticles [57]. In contrast to QDs that adsorb UV light and emit visible light, UCNPs adsorb low energy light, usually near-infrared (NIR) or infrared (IR), and emit high energy light (UV or visible) through multi photon absorptions or energy transfers. UCNPs consist of rare earth atoms that are embedded in a crystalline host matrix. Among the varies of up converting materials, NaYF4 nanocrystals are reported as the most effective host for infrared-to-visible up conversion when doped with Er or Yb ions [58]. Although the quantum yield of UCNPs is not even comparable with QDs, is still possesses some unparalleled advantages in bio-labeling applications. For example, the noninvasive and deep penetration of NIR radiation could realize deep tissue imaging. Meanwhile, since the background fluorescence of living organisms could not be excited by NIR light, UCNPs labeling provides unmatched sharp contrast in imaging.

The bottle-neck of UCNPs in nucleic acid assay comes from the tough conjugation of DNA with UCNPs. Despite recent advances in synthesis of UCNPs with controllable size and shape, the as-synthesized UCNPs are usually coated with hydrophobic organic ligands and suffer from poor water-solubility. Currently, several surface modification strategies have been proposed to enable UCNPs water-soluble and conjugate with oligonucleotides. One general strategy to convert hydrophobic nanoparticles water soluble is to encapsulate with silica or amphiphilic copolymers [59]. For example, Tanke's group has coated Y2O2S:Er UCNPs with silica, which was then conjugated with biotin [60]. This biotinylated-UCNPS were successfully employed as fluorescence reported in nucleic acid arrays. Compared with Cy5 label, the assays with UCNPs reveals four times improved sensitivity. Another versatile strategy for synthesizing water-soluble and carboxylic acidfunctionalized UCNPs was performed by directly oxidizing oleic acid ligand to azelaic acid (HOOC(CH₂)₇COOH) with the Lemieux-von Rudloff reagent. The free carboxylic acid groups on their surface enable further conjugation with DNA probes [61]. By using this protocol, Huang and co-workers modified UCNPs with a DNA capture probe and constructed a sandwich-type DNA assay including a reporter dyelabeled DNA on UCNPs. Upon illumination of the UCNPs with NIR light, the visible emission light of UCNPs could be adsorbed by the dye, producing FRET from UCNPs to dye; the fluorescence of the dye could be observed as the readout of this UCNPs-based DNA assay.

Nanomaterials as Fluorescence Quenchers

Fluorescent detection methods are still the most employed in nucleic acid assay due to the commercial availability of a wide spectrum of fluorophores, the ease of fluorescent labeling, and the inherent capability for real-time detection. While fluorescent detection is usually highly sensitive, the detection sensitivity is often limited by the presence of background emission. While organic quenchers have proven their utility, they suffer from relatively low quenching efficiency, and, more importantly, quenching efficiencies often vary significantly from one dye to another.

Interestingly, some metallic or carbon nanomaterials have shown ultrahigh fluorescence quenching ability. For example, the quenching mechanism of metal clusters has been well theoretically studied and proven to be a resonant energy transfer process [62-64]. Meanwhile, AuNPs have a Stern-Volmer quenching constant (KSV) that is greater with several (>5) orders of magnitude than that of typical small molecule dye quencher pairs [65]. This "super-quenching" ability of AuNPs has found great implications in fluorescence-based nucleic acid assays that great suppress the background. Dubertret et al. designed an AuNPs-based nanobeacon that replaced organic quenchers with 1.4-nm AuNPs, and found that AuNPs served as a universal quencher that could quench a range of fluorophores with 100-fold higher quenching efficiency than organic quenchers [66]. This so called "nanobeacon" not only led to improved sensitivity but also remarkably high selectivity for single-base mismatch discrimination. Nie and coworkers later reported a stem-less probe on 2.5-nm AuNPs that still demonstrated target-induced conformational change and fluorescence variation in response to the binding of target DNA [67]. Although these small-sized AuNPs revealed decent performance as effective quenchers in nanobeacon, their crowded surface could only accommodate one or several oligonucleotides probes. Recently, we employed larger-sized AuNPs of 15 nm in diameter to construct multicolor nanobeacons [68]. Importantly, it is possible to anchor many DNA probes at one single particle due to the increased surface area. As a proof-of-concept experiment, three DNA probes designed for three tumor-suppressor genes were immobilized at the AuNPs surface, each carrying a unique fluorophore. This nanobeacon exhibited rapid hybridization kinetics (minutes), and could respond specifically to different gene targets (Fig. 6.6). The organic dye in the nanobeacon could also be replaced by fluorescent QDs [15, 69], providing new opportunities for nanobeacons incorporating multiple functional nanoparticles.

In addition to metallic nanoparticles, some carbon nanomaterials including one dimensional (1D) single-walled carbon nanotubes (SWNTs) and two dimensional (2D) graphene have been observed to interact differentially with ss- and ds-DNA, which also enables them to be used as effective quenchers in fluorescence DNA detection. To take SWNTs as an example, ss-DNA has been demonstrated to interact noncovalently with SWNTs [70, 71]. In detail, ss-DNA molecules could wrap around SWNTs by means of π -stacking interactions between the nucleotide bases and the sidewalls of individual SWNT, resulting stable complexes, while for dsDNA, the affinity is significantly weaker. This difference in the binding



Fig. 6.6 Multicolor nanobeacons for multiplexing detection of three different cancer DNA markers. The multicolor nanobeacons are hybridized with three perfectly matched targets and three single mismatched targets, showing high sequence specificity that arises from the conformational constraint (Reproduced with permission from Ref. [68]. Copyright Wiley-VCH Verlag GmbH & Co. KGaA)

interactions of SWNTs with ssDNA and dsDNA has provided the basis for their use to report DNA hybridization. Plus, SWNTs was found as an effective quencher to quench fluorophores nearby. This quenching was found to be quite universal to a variety of fluorophores, as well as QDs, and it is thought that either energy transfer or electron transfer might contribute to such high quenching efficiency [72, 73].

Tan and Yang combined the super quenching feature of SWNTs together with its' ability to differentiate ss- and ds-DNA, and designed a SWNTs-based strategy for DNA sensing [74]. They demonstrated that fluorophore-labeled DNA probes were efficiently quenched in the presence of SWNTs while they were released from SWNTs upon hybridization, leading to fluorescence recovery [74, 75]. This strategy, however, needs a fluorophore-tagged ss-DNA probe, which increases the detection cost. They also moved one-step further to introduce a DNA intercalating



Fig. 6.7 (a) Scheme for the fluorescent DNA detection based on the ss-DNA/ds-DNA discrimination ability of GO. (b) Scheme for the target hybridization-induced probe liberation from GO (Reproduced with permission from Ref. [80]. Copyright © 2011 WILEY-VCH Verlag GmbH & Co. KGaA, Weinheim)

dye ethidium bromide (EB) to replace fluorophore and developed a cost-effective label free method for fluorescence DNA detection [76].

As a planar molecule, ethidium bromide can absorb on the sidewall of SWNTs. A mixture of the dye, the capturing ssDNA, and SWNTs formed a self-assembly of three components, generating complete quenching of fluorescence. After the introduction of target DNA, ds-DNA formed between the capturing DNA and the target. The adsorbed ethidium bromide preferred to intercalate into the hybridized bases, which disturbed the interaction of the dye and SWNTs, resulting in fluorescence recovery.

Graphene is a single layer of carbon atoms in a closely packed honeycomb twodimensional lattice [77, 78]. Recently, graphene was also found to be an excellent super fluorescence quencher, however, different from SWNTs, the quenching mechanism of graphene was theoretically and experimentally attributed to be only long-range resonance energy transfer [64, 79]. We carried out molecular dynamics (MD) simulation to describe the interaction of graphene oxide (GO) with ss- and ds-DNA, which indicated that ss-DNA adsorbed strongly to graphene oxide (GO) sheets while ds-DNA did not [80]. Thus, similar to SWNTs, Lu et al. employed GO to adsorb and quench dye-labeled DNA probes, which could be liberated upon hybridization, leading to fluorescence increment [81]. He et al. designed a new approach that relied on different binding kinetic of ss-DNA and ds-DNA to GO, which led to sequence-specific detection with high sensitivity and rapid response time (<5 min) [50]. More significantly, due to the availability of large planar surface of GO, multicolor DNA detection was realized by using differentially labeled DNA probes (Fig. 6.7). Another unparalleled advantage of carbon nanomaterials, is their ability to cross cell membrane [82, 83]. Thus, in vivo analysis or imaging nucleic acid is highly anticipated [64, 84].

6.2.3 Chemiluminescent and Electro-Generated Chemiluminescent DNA Assay

6.2.3.1 Chemiluminescent DNA Assay

Luminescence is the emission of light without heat, and more specially the generation of light from chemical reactions (chemiluminescence, CL). This long history of CL has been summarized in a few review articles [85–87]. Since excitation light is not required for sample radiation, problems frequently encountered in fluorescence, such as light scattering, source instability and high backgrounds due to unselective photoexcitation are absent in CL assay. Consequently, CL has found great advantages in biodiagnostics [88].

In a typical CL reaction, an organic dye, which is not luminescent in the ground state, was oxidized by a strong oxidizing agent in the presence of a catalyst (chemical or biological). The most commonly used dye is luminol, while oxidation of luminol with hydrogen peroxide (oxidizing agent) in the presence of a catalyst results in the conversion of the ground state of luminol into an activated state (chemically induced electronic excited states). A strong blue emission (at 450 nm wavelength) can be observed as a result of the decay of the excited states back to the ground state [89]. The most intensively used catalysts that enhance the CL of luminol is horseradish peroxidase (HRP). The mechanism of HRP enhanced CL has been extensively studied [86], suggesting HRP reacts with H_2O_2 to form an oxidized HRP (HRP I) that reacts with the anion of luminol to form a half reduced enzyme (HRP II) and a radical of luminol. The enzyme returns to the reduced form (HRP) by reaction with a second molecule of luminol. The catalytic phenols form preferentially phenoxy radicals in contact with horseradish peroxidase and act as electron-transfer mediators to increase the efficiency of luminol radical formation. However, the luminol-based CL assay is mainly applied to immunoassays instead of DNA detection, which was attributed to the poor sensitivity. In the early phase of luminol-based DNA assay, luminol was directly conjugated with DNA as reporter probe in a sandwich-typed CL assay. A luminol derivative, N-(4-aminobutyl)-Netylisoluminol (ABEI) is the only tracer used in immunoassay. However, luminol is more efficient in its free state. Upon conjugation with DNA, it lost most of CL properties; thereby this ABEI-tagged CL assay did not reveal satisfied performance. An alternative protocol is to use HRP as an enzyme label in sandwich-typed DNA assay and luminol-H2O2 mixture was added to generate CL signal and report the hybridization event, however, which is also found limited success.

In addition to HRP, some other catalysts, including Co^{2+} [90], Cu^{2+} [91, 92], or AuNPs [93, 94] could also enhance the CL of luminol-H₂O₂ system, although the detailed mechanism is not completely clear yet. Among these enhancers, AuNPs has found widely implications in CL DNA assay owing to the ease conjugation of DNA with AuNPs. For example, Willner's group reported that horseradish peroxidaselike DNAzyme-functionalized AuNPs could act as catalytic labels for the amplified detection of DNA with a LOD of 0.1 nM [93]. Li and co-workers demonstrated a sandwich-typed CL assay for DNA detection using tadpole-shaped AuNPs [94]. Zhang and co-workers found that Au nanoaggregates induced by salt possesses higher enhancement towards the CL of luminol- H_2O_2 than AuNPs [95]. The Au nanoaggregates were used as catalyst in a CL DNA assay and a LOD of 1.1 fM was achieved.

Historically, the CL of luminol system has been preceded by a more sensitive dye of acridinium. Interaction of an acridinium ester (AE) with an alkaline hydrogen peroxide solution would trigger a short flash of light at 430 nm within 5 s or less. Nelson's group have synthesized a number of derivatives of this AE and characterized their chemiluminescent properties [96]. These derivatives display significant differences in the kinetics of the chemiluminescence reaction as well as optimal pH for light production. The first acridinium derivative of practical use was 4-(2-succinimidyl-oxycarbonylethyl)-phenyl-10-methyl-acridinium-9- carboxylate (AE-NHS). AE-NHS does not show the best quantum yield and is not very stable especially at room temperature although its stability is increased after coupling. More efficient compounds are found in thiol, sulphonamide, hydroxamic, oxime and chloroxime series [86]. After covalent conjugation to DNA helix, acridinium labels show increased stability toward hydrolysis. These property enables AE-derivatives to be used in important applications in Chemiluminescence DNA assay, including simultaneous detection of gag and pol regions of HIV [97] and discrimination of the single-base mismatch between wild-type and mutant sequences corresponding to the reverse transcriptase coding region of HIV-1 [98].

6.2.3.2 Electro-Generated Chemiluminescent Assay

Electro-generated chemiluminescence, or electrochemiluminescence (ECL) is an analogue of CL. ECL is chemiluminescence triggered by electrochemical techniques that undergoes a high-energy electron transfer reaction to generate an excited state. The combination of electrochemistry with chemiluminescence, has proven to be a powerful analytical tool [99, 100]. For example, in the classical ECL of tris(2,2'-bipyridyl) ruthenium (Ru(bpy)₃)²⁺ that was developed by Bard [101], its ECL mechanism has been well-illustrated. (Ru(bpy)₃)²⁺ ECL reaction occurs by oxidizing (Ru(bpy)₃)²⁺ to produce(Ru(bpy)₃)^{2+*} in the presence of a strong reducing agent. The most commonly used reducing agent for this type of ECL reaction is tripropylamine (TPrA). This mechanism is termed the oxidative-reduction ECL mechanism is as follows:

$$(\text{Ru}(\text{bpy})_3)^{2+} - e^- \to (\text{Ru}(\text{bpy})_3)^{3+}$$
 (6.1)

$$\operatorname{TprA} - e^{-} \to \operatorname{TprA}^{\bullet +}$$
 (6.2)



Fig. 6.8 Schematic diagram of DNA hybridization on a polystyrene bead as the ECL label carrier and a magnetic bead for the separation of analyte-contained ECL label/polystyrene beads (Reproduced with permission from Ref. [106]. Copyright (2004) American Chemical Society)

$$\operatorname{TprA}^{\bullet+} \to \operatorname{TprA}^{\bullet} + \mathrm{H}^{+}$$
 (6.3)

$$(\operatorname{Ru}(\operatorname{bpy})_3)^{3+} + \operatorname{TprA}^{\bullet} \to (\operatorname{Ru}(\operatorname{bpy})_3)^{2+*} + \operatorname{products}$$
(6.4)

$$(\text{Ru}(\text{bpy})_3)^{2+*} \to (\text{Ru}(\text{bpy})_3)^{2+} + h\nu$$
 (6.5)

The distinct advantage of ECL is its high sensitivity. $Ru(bpy)_3^{2+}$ can be highly sensitively detected at subpicomolar concentrations with an extremely wide dynamic range of greater than six orders of magnitude. Consequently, this super high sensitivity of ECL has found great implications in DNA sensors. The application of ECL in DNA assay was summarized in an excellent review [104].

This field was pioneered by Bard's group in 1995 [105], lately they integrated ECL with magnetic concentration to further improve the sensitivity [106]. In their protocol (Fig. 6.8), $Ru(bpy)_3^{2+}$ label was encapsulated in polystyrene (PS) beads. The advantage of PS beads is that one PS bead could carry up to 7.5×10^9 molecules of $Ru(bpy)_3^{2+}$. Then biotinylated target DNA was modified on the surface of these $Ru(bpy)_3^{2+}$ loaded PS beads via biotin-avidin interaction. The target DNA was complementary to a probe DNA that was modified on a magnetic beads surface. Upon addition of a magnetic field, the target DNA-PS beads containing $Ru(bpy)_3^{2+}$ were magnetically separated and transferred into an acetonitrile solution where the polystyrene beads dissolve the ECL label is released. This is followed by ECL detection and a LOD of 1 fM was achieved.

In addition to these solution phase ECL assays, solid state ECL DNA assays were also developed. The first model assay was also proposed by Bard's group [107]. In brief, they first modified a Au(111) electrode with a capture probe DNA. Upon hybridization with target DNA carrying a $Ru(bpy)_3^{2+}$ ECL label, $Ru(bpy)_3^{2+}$ was brought to the surface of electrodes and generated ECL signal. This simple method, however, requires the covalent modification of target DNA

with $Ru(bpy)_3^{2+}$, which is actually inappropriate for clinical diagnostics. Thereby, some sandwich-typed hybridization assays were further developed [108, 109]. An alternative type of hybridization assay was presented by Zhang and co-workers by using the conformational change of probe DNA [110]. The modified Au electrode with thiolated hairpin probe DNA was tagged with $Ru(bpy)_3^{2+}$. In the absence of target ss-DNA, the ECL probe immobilized on the surface of the electrode was in its folded configuration, in which its termini were held in close proximity to the electrode, and thus a strong ECL signal could be generated. In the presence of target ss-DNA, a stem-loop of the ECL probe on the electrode was converted into a linear double-helix configuration due to hybridization, resulting in the tag moving away from the electrode surface, which in turn decreased the ECL signal. This method avoids the introduce of a $Ru(bpy)_3^{2+}$ -tagged reporter DNA probe; however the "signal-off" detection mode is not a favorite option in sensor design.

In these works, the employed solid state ECL reporter DNA probe carries only one $Ru(bpy)_3^{2+}$ label, thus it produces only one ECL photon for one hybridization event. Motivated by the work of Bard's group in solution phase [107], researchers also employed polymer beads and nanoparticles that load multiple number of $Ru(bpy)_3^{2+}$ as ECL reporters, thereby one hybridization event could correspond to multiple ECL photons, which may improve the sensitivity. For example, AuNPs prepared by the classical citrate-reduced method is negatively charged, thereby AuNPs are "sticked" to aggregates upon interaction of positively charged $Ru(bpy)_3^{2+}$ [111]. The resulted Au nanoaggregates carry a large number of $Ru(bpy)_3^{2+}$, which might be a powerful label for nucleic acid detection. Another example was reported by Zhang and co-workers by using AuNPs [112]. The large surface area of AuNPs enables them to load multiple number, or multiple kind, of biomolecules. For example, one AuNPs of 12 nm in diameter could carry up to a few hundreds of ss-DNA [113]. Zhang et al. modified a thiolated ss-DNA with a $Ru(bpy)_3^{2+}$ at the other terminal, and the $Ru(bpy)_3^{2+}$ -functionalized ss-DNA was further covalently attached to AuNPs, thereby one as-prepared AuNP was loaded with hundreds of $Ru(bpy)_3^{2+}$. The resulted DNA- $Ru(bpy)_3^{2+}$ -AuNPs was then used as reporter probe in a sandwich-typed ECL DNA assay and a LOD of 5 pM was achieved.

Some DNA intercalators were recently reported as co-reactants for ECL DNA hybridization assays. Park and co-workers found some duplex DNA intercalators including doxorubicin, daunorubicin or 2-phenylindole shows favorable ECL with $Ru(bpy)_3^{2+}$ at +1.19 V (versus Ag/AgCl), while the non-intercalated ssDNA does not [114]. Several pathogens were detected using this ECL approach, and a good specificity of single point mutations for hepatitis disease was obtained by the 2-phenylindole-intercalated Ru(bpy)_3^{2+} ECL.

It should also be noted that a brand-new ECL system using semiconductor QDs was proposed by Ju's group [115–117]. They found that aqueous-soluble CdTe or CdSe QDs show intensive anodic ECL with a peak value at +1.17 V (vs Ag/AgCl) in basic buffer at an indium tin oxide (ITO) electrode. The ECL emission was demonstrated to involve the participation of superoxide ion produced at the ITO surface, which could inject an electron into the 1Se quantum-confined orbital of CdTe to form QDs anions. The collision between these anions and the oxidation
products of QDs led to the formation of the excited state of QDs and ECL emission. Therefore, CdTe QDs could also be employed as ECL labels in DNA assay and a LOD at fM scale was achieved [118].

6.2.4 Surface Plasmon Resonance (SPR) Nucleic Acid Assays

6.2.4.1 SPR Spectroscopy

Surface plasmon resonance (SPR) spectroscopy is a surface analysis method based on changes in the optical reflectivity of a thin metal film (typically gold) when analytes adsorb or bind to its surface or to any material coated onto its surface (see Chap. 9). Surface plasmon (SP) is defined as the plasma oscillations that can propagate at the interface of a metal and a dielectric medium. The electromagnetic field of a SP is confined at the metal–dielectric boundary and decreases exponentially into both media, thereby SP is extremely sensitive to changes in the refractive index of the dielectric. In a typical SPR spectroscopy, a polarized monochromatic light beam is passed through a prism and its attached gold-coated glass slide, and reflected off the thin gold coating, which is in contact with the liquid solution of interest. Excitation of SP at the gold/solution interface results in nearly complete attenuation of the specular reflected light intensity for incident angles very near the SPR angle. The specular reflected light intensity versus angle at fixed wavelength or versus wavelength at fixed angle reflects the refractive index change at the interface.

Ever since the first SPR instrument (BIAcore) was commercialized in 1990 [119, 120], SPR has been a powerful label-free method for real-time monitoring the biomolecular recognition events at surfaces including immunological analysis, studies of protein-protein interaction, molecular-biological studies on the mechanisms of gene expression, signal transduction and cell-cell interactions, the screening of new ligands, the quantification of protein adsorption and immobilization, the evaluation of surfaces for biocompatibility, epitope mapping, the determination of affinity constants, and the examination of the kinetics of binding [121–123]. In addition to quantitatively detection of analytes, one distinct feature of SPR sensor is that it could also monitor the conformational changes of surface-immobilized biomolecules.

The application of SPR in nucleic acid detection was started in the middle of 1990s [124, 125]. The early stage SPR sensors include one capture on the SPR substrate and the complementary target strand in solution. The hybridization of target strand with capture probe resulted changes of SPR angles. Unfortunately, the inability of conventional SPR to measure extremely small changes in refractive index hinders its application in ultrasensitive detection. To address this drawback, several approaches have been developed. Among them, substantial interest has been focused on utilizing external labels to enhance the sensitivity of the current technique. Keating and co-workers introduced sandwich-typed assay and an AuNPs labeled reporter probe [126]. Since the bulk refractive index of AuNPs is significantly higher than that of DNA and the electromagnetic interaction between metallic nanoparticles and metal surfaces may also help influence the plasmon mode

propagation, the introduced AuNPs label significantly amplifies the SPR response. With the help of AuNPs, LOD for a target was improved from 10 nM of regular assay to 10 pM.

6.2.4.2 SPR Microscopy

Recently, another SPR-related technique, SPR microscopy (SPRM), or "SPR imaging" (SPRi), has been developed to provide high-throughput real time information about the amount and distribution of adsorbed molecules at the interfaces with high spatial resolution [121]. The invention of SPR microscopy [127, 128], led the application of SPR to the imaging of biomolecular surface interactions. In principle of taking SPR images is by introducing a parallel light beam for uniform illumination of a surface at a fixed angle and recorded the reflected beam forming the image at a CCD camera. This configuration is sufficient to determine thickness with Angstrom-level resolution.

The thickness-sensitive feature of SPR microscopy enables it to be a powerful tool to study protein-protein, protein-DNA interactions that induce significant surface topography changes [129]. Corn's group has been very active in the development of SPRi-based detection of oligonucleotides [129–131]. In their early work, a single stranded capture probe DNA was first immobilized on SPR substrate. Upon hybridization with target DNA, the SPR image evolved rapidly, resulting in brighter spots, showing the potential of SPRi in DNA detection [130]. However, DNA hybridization does not induce significant changes in the layer height, thus this proof-of-concept label-free method is actually found limited practical success. Later on, the same group further amplified the SPRi response by introducing a sandwich-typed hybridization and a streptavidin-tagged reporter probe to improve the thickness of the DNA layer resulted by target-induced hybridization [132]. The combination of DNA hybridization and streptavidin-biotin binding improved the LOD for DNA hybridization using SPR imaging by a factor of \sim 4. Interestingly, the SPRi signal could be further improved by multilayers of streptavidin/DNA formed by successively assembly of biotin-labeled target DNA and streptavidin. This in situ monitoring of DNA hybridization, could be easily integrated to 1D arrays, or 2D arrays created by microfluidic techniques, to realize highthroughput nucleic acid analysis [133]. This hybridization event occurred at the SPR substrate, could be cycled by digestion with ExoIII. The enzymaticallyassisted SPR imaging process realized a 10^2-10^3 improvement of sensitivity for a 16-mer target ssDNA [134].

6.2.5 Surface-Enhanced Raman Scattering (SERS) in Nucleic Acid Detection

Raman is a vibrational spectroscopic technique to study vibrational, rotational, and other low-frequency modes of molecules. Raman spectrum is obtained from the inelastic scattering, or Raman scattering, of monochromatic light in the visible, near infrared, or near ultraviolet range. The limit of conventional Raman spectroscopy is its weak intensity, often requiring the use of powerful and costly laser sources for excitation. Interestingly, Raman scattering could be enhanced by factors of up to 10^8 when a compound is adsorbed on or near special metal surface, or 10^{14} – 10^{15} when adsorbed on rough coinage metallic (Cu, Ag, Au) nanoparticles surface [135–137]. This phenomenon is called surface-enhanced Raman scarring (SERS). After decades of debate, the mechanism of SERS could be sorted into two classes which were called electromagnetic and chemical, respectively [137]. Briefly, the former focus on the enhanced electromagnetic fields which can be supported on metal surfaces with appropriate morphologies, while the latter on changes in the electronic structure of molecules which occur upon adsorption and which can lead to resonance.

In addition to ultra-high enhancement factor, SERS also possesses other attractive advantages, such as narrow Raman band, and high resistance to interferences of environmental factors (e.g., humidity, oxygen, and foreign species), thereby has been considered as an attractive method for DNA detection [138, 139]. Mirkin and coworkers pioneered the use of SERS for DNA targets detection [140]. They designed a Raman nanoprobe that consists of 13-nm-diameter AuNPs functionalized with Raman dye-labeled oligonucleotides. After Ag staining, the Ag particles can grow around the Cy3-labeled nanoprobes, leading to large Raman scattering enhancements. Thus, they demonstrated that six different oligonucleotide targets could be simultaneously detected with AuNPs labeled with different Raman probes. Significantly, the detection limit was down to 20 fM due to the strong SERS signal intensity. Moskovits and co-workers proposed an alternative SERS-based strategy for DNA detection (Fig. 6.9) [141]. By self-assembling probe-tethered Ag nanoparticles to a smooth Ag film using the complementary target species, they created electromagnetic "hot spots" on smooth Ag surface, which strongly enhance the Raman signal of the species present in the hot spot. Kim and Lee [142] replaced the smooth metal surfaces with rough metal nanowires to further improve the interface roughness and in turn enhancement factor, and demonstrated a nanoparticle-onnanowire configuration SERS sensor for multiplex pathogen DNA. Very recently, our group proposed an alternative Ag nanoparticle-on-Silicon nanowire SERS platform and found that the in situ grew AgNPs on Silicon nanowires (SiNWs@AgNPs) revealed a SERS enhancement factor of $\sim 10^{10}$ [143]. Then a sandwich-typed DNA assay with A Raman tag was further constructed on the SiNWs@AgNPs substrate (Fig. 6.10). This Ag nanoparticle-on-Silicon nanowire platform revealized a remarkably low LOD of ~ 1 fM for DNA target. Faulds and Graham also intensively devoted to the design of novel SERS-based DNA sensor [144-146]. One of their representative work is that they prepared a kind of dye-coded, DNAfunctionalized silver nanoparticles [147]. Target DNA-induced sequence-specific hybridization produced the assembly of individual nanoparticles to nanoaggregates, which strongly enhanced the SERS signal of surfaces-coated dyes.

In addition to these efforts of bringing SERS tag-loaded nanoparticles to the close proximity of metal (bulk or nano) surfaces with DNA, another type of SERS-based



Fig. 6.9 (A) Schematic illustration for the detection of ss- DNA by SERS. (B) SERS spectra (vertically offset for clarity) indicating the target DNA hybridization-induced enhanced Raman signal. (C) Representative AFM images confirming the attachment of AgNPs to Au surfaces by target DNA (Reproduced with permission from Ref. [141]. Copyright (2007) American Chemical Society)

sensor involves the attachment of DNA-labeled SERS tag to the close surface of metal nanoparticles. For example, Vo-Dinh and co-workers [148] represented a plasmonics-based nanoprobe, referred to as a "molecular sentinel" (MS), consisting of a DNA hairpin loop having a Raman label molecule at one end and a metal nanoparticle at the other end. The nanoprobes combines the modulation of the plasmonics effect to change the SERS intensity of the label and the specificity of a DNA hairpin loop sequence to recognize and discriminate a variety of molecular target sequences. Upon hybridization with complementary target sequence that opens the hairpin capture probe, the Raman label was physically removed away from the metal nanoparticle, thus quenching the SERS signal. This "molecular sentinel" strategy was successfully employed for SERS detection o HIV-I virus segment. Johnson and co-workers [149] presented indirect nucleic acid capture strategy to draw SERS tag to the surface of AuNPs with the help of hybridization with target DNA. In their design, a 53-mer West Nile Virus (WNV) genome target DNA was



Fig. 6.10 Schematic illustration for the detection of DNA by the SiNWs@AgNPs SERS platform (Reproduced with permission from Ref. [141]. Copyright (2011) Elsevier)

first hybridized in solution with thiolated complementary capture (26-mer) probes and SERS tag-modified reporter (25-mer) oligonucleotide probes, after which the hybridized ternary complex was added to a colloidal suspension of AuNPs and SERS spectra were immediately acquired upon laser excitation.

6.3 Outlook

In this chapter, we have summarized different approaches for optical detection of non-amplified DNA targets. Nanotechnology has shown great advantage in this area. Signal amplification with biomolecular nanoprobes has led to extremely sensitive assays with attomolar detection limit, which represents sensitivity improvement up to six orders of magnitude compared to conventional fluorescent detection (1-10 pM).

Despite such tremendous progress, there remain several challenges for optical DNA detection technologies before they can be widely accepted in clinical or foodrelated applications. First, DNA hybridization occurring at the nanomaterial surface is inherently low in recognition efficiency and slow in binding kinetics. This is because the use of inorganic nanomaterials introduces interfaces that bring about the problems of heterogeneous diffusion and convection that are not present in homogenous solution. Second, while some DNA assays have realized ultrahigh sensitivity, they typically require multiple steps that significantly increase the operation complexity. Third, only a few of the mentioned assays show excellent performance in complex biological matrixes (e.g. serum). Thus, non-specific binding is still a major problem that should be studied in order to realize practical applications of these new assay methods.

Notably, the emerging DNA nanotechnology might provide unprecedented opportunities to meet these challenges. For example, the recently developed DNA origami-based chips [150, 151] provide a potentially promising solution to overcome the interface problem since DNA origami structures, in contrast to inorganic nanomaterials, are essentially in aqueous solution. While this method is rather immature in their current form, the combination with high-speed and high throughput readouts as well as automated devices [152], might eventually lead to an unparalleled platform for DNA detection, as well as for the detection of a range of biomolecules.

References

- Sassolas, A., Leca-Bouvier, B.D., Blum, L.J.: DNA biosensors and microarrays. Chem. Rev. 108, 109–139 (2008)
- 2. Cosnier, S., Mailley, P.: Recent advances in DNA sensors. Analyst 133, 984–991 (2008)
- 3. Vercoutere, W., Akeson, M.: Biosensors for DNA sequence detection. Curr. Opin. Chem. Biol. 6, 816–822 (2002)
- 4. Teles, F.R.R., Fonseca, L.R.: Trends in DNA biosensors. Talanta 77, 606-623 (2008)
- 5. Song, S.P., Qin, Y., He, Y., Huang, Q., Fan, C.H., Chen, H.Y.: Functional nanoprobes for ultrasensitive detection of biomolecules. Chem. Soc. Rev. **39**, 4234–4243 (2010)
- 6. Li, D., Song, S.P., Fan, C.H.: Target-responsive structural switching for nucleic acid-based sensors. Acc. Chem. Res. 43, 631–641 (2010)
- 7. Rosi, N.L., Mirkin, C.A.: Nanostructures in biodiagnostics. Chem. Rev. 105, 1547–1562 (2005)
- Elghanian, R., Storhoff, J.J., Mucic, R.C., Letsinger, R.L., Mirkin, C.A.: Selective colorimetric detection of polynucleotides based on the distance-dependent optical properties of gold nanoparticles. Science 277, 1078–1081 (1997)
- Mirkin, C.A., Letsinger, R.L., Mucic, R.C., Storhoff, J.J.: A DNA-based method for rationally assembling nanoparticles into macroscopic materials. Nature 382, 607–609 (1996)
- Wilson, R.: The use of gold nanoparticles in diagnostics and detection. Chem. Soc. Rev. 37, 2028–2045 (2008)
- Edwards, P.P., Thomas, J.M.: Gold in a metallic divided state from Faraday to present-day nanoscience. Angew. Chem. Int. Ed. 46, 5480–5486 (2007)
- Reynolds, R.A., Mirkin, C.A., Letsinger, R.L.: Homogeneous, nanoparticle-based quantitative colorimetric detection of oligonucleotides. J. Am. Chem. Soc. 122, 3795–3796 (2000)
- Jin, R.C., Wu, G.S., Li, Z., Mirkin, C.A., Schatz, G.C.: What controls the melting properties of DNA-linked gold nanoparticle assemblies? J. Am. Chem. Soc. 125, 1643–1654 (2003)
- Zhao, W., Brook, M.A., Li, Y.F.: Design of gold nanoparticle-based colorimetric biosensing assays. ChemBioChem 9, 2363–2371 (2008)

- Dai, Z., Zhang, J.M., Dong, Q.X., Guo, N., Xu, S.C., Sun, B., Bu, Y.H.: Adaption of Au nanoparticles and CdTe quantum dots in DNA detection. Chin. J. Chem. Eng. 15, 791–794 (2007)
- Taton, T.A., Mirkin, C.A., Letsinger, R.L.: Scanometric DNA array detection with nanoparticle probes. Science 289, 1757–1760 (2000)
- Li, H.X., Rothberg, L.: Colorimetric detection of DNA sequences based on electrostatic interactions with unmodified gold nanoparticles. Proc. Natl. Acad. Sci. USA 101, 14036– 14039 (2004)
- Li, H.X., Rothberg, L.: Detection of specific sequences in RNA using differential adsorption of single-stranded oligonucleotides on gold nanoparticles. Anal. Chem. 77, 6229–6233 (2005)
- Li, H.X., Rothberg, L.J.: Label-free colorimetric detection of specific sequences in genomic DNA amplified by the polymerase chain reaction. J. Am. Chem. Soc. 126, 10958–10961 (2004)
- Wang, L.H., Liu, X.F., Hu, X.F., Song, S.P., Fan, C.H.: Unmodified gold nanoparticles as a colorimetric probe for potassium DNA aptamers. Chem. Commun. 3780–3782 (2006)
- Wang, J., Wang, L.H., Liu, X.F., Liang, Z.Q., Song, S.P., Li, W.X., Li, G.X., Fan, C.H.: A gold nanoparticle-based aptamer target binding readout for ATP assay. Adv. Mater. 19, 3943–3946 (2007)
- Zhang, J., Wang, L.H., Pan, D., Song, S.P., Boey, F.Y.C., Zhang, H., Fan, C.H.: Visual cocaine detection with gold nanoparticles and rationally engineered aptamer structures. Small 4, 1196–1200 (2008)
- Tyagi, S., Kramer, F.R.: Molecular beacons: probes that fluoresce upon hybridization. Nat. Biotechnol. 14, 303–308 (1996)
- 24. Wang, K.M., Tang, Z.W., Yang, C.Y.J., Kim, Y.M., Fang, X.H., Li, W., Wu, Y.R., Medley, C.D., Cao, Z.H., Li, J., Colon, P., Lin, H., Tan, W.H.: Molecular engineering of DNA: molecular beacons. Angew. Chem. Int. Ed. 48, 856–870 (2009)
- Tan, W.H., Wang, K.M., Drake, T.J.: Molecular beacons. Curr. Opin. Chem. Biol. 8, 547–553 (2004)
- Marras, S.A.E., Tyagi, S., Kramer, F.R.: Real-time assays with molecular beacons and other fluorescent nucleic acid hybridization probes. Clin. Chim. Acta 363, 48–60 (2006)
- Whitcombe, D., Theaker, J., Guy, S.P., Brown, T., Little, S.: Detection of PCR products using self-probing amplicons and fluorescence. Nat. Biotechnol. 17, 804–807 (1999)
- Wittwer, C.T., Herrmann, M.G., Gundry, C.N., Elenitoba-Johnson, K.S.J.: Real-time multiplex PCR assays. Methods 25, 430–442 (2001)
- Wabuyele, M.B., Farquar, H., Stryjewski, W., Hammer, R.P., Soper, S.A., Cheng, Y.W., Barany, F.: Approaching real-time molecular diagnostics: single-pair fluorescence resonance energy transfer (spFRET) detection for the analysis of low abundant point mutations in K-ras oncogenes. J. Am. Chem. Soc. **125**, 6937–6945 (2003)
- Wong, M.L., Medrano, J.F.: Real-time PCR for mRNA quantitation. Biotechniques 39, 75–85 (2005)
- Santangelo, P.J., Nix, B., Tsourkas, A., Bao, G.: Dual FRET molecular beacons for mRNA detection in living cells. Nucleic Acids Res. 32(9) (2004)
- Bratu, D.P., Cha, B.J., Mhlanga, M.M., Kramer, F.R., Tyagi, S.: Visualizing the distribution and transport of mRNAs in living cells. Proc. Natl. Acad. Sci. USA 100, 13308–13313 (2003)
- 33. Kiesling, T., Cox, K., Davidson, E.A., Dretchen, K., Grater, G., Hibbard, S., Lasken, R.S., Leshin, J., Skowronski, E., Danielsen, M.: Sequence specific detection of DNA using nicking endonuclease signal amplification (NESA). Nucleic Acids Res. 35, 9 (2007)
- 34. Li, J.W.J., Chu, Y.Z., Lee, B.Y.H., Xie, X.L.S.: Enzymatic signal amplification of molecular beacons for sensitive DNA detection. Nucleic Acids Res. 36, 17 (2008)
- Xu, W., Xue, X.J., Li, T.H., Zeng, H.Q., Liu, X.G.: Ultrasensitive and selective colorimetric DNA detection by nicking endonuclease assisted nanoparticle amplification. Angew. Chem. Int. Ed. 48, 6849–6852 (2009)
- 36. Zhou, B.J., Ma, Y.J., Wu, H.P., Zhou, G.H.: Ultrasensitive DNA detection by cascade enzymatic signal amplification based on Afu Flap endonuclease coupled with nicking endonuclease. Angew. Chem. Int. Ed. 50, 7395–7398 (2011)

- Zuo, X.L., Xia, F., Xiao, Y., Plaxco, K.W.: Sensitive and selective amplified fluorescence DNA detection based on Exonuclease III-aided target recycling. J. Am. Chem. Soc. 132, 1816–1818 (2010)
- Zhao, X.J., Bagwe, R.P., Tan, W.H.: Development of organic-dye-doped silica nanoparticles in a reverse microemulsion. Adv. Mater. 16, 173–176 (2004)
- Yan, J.L., Estevez, M.C., Smith, J.E., Wang, K.M., He, X.X., Wang, L., Tan, W.H.: Dye-doped nanoparticles for bioanalysis. Nano Today 2, 44–50 (2007)
- 40. Zhao, X.J., Tapec-Dytioco, R., Tan, W.H.: Ultrasensitive DNA detection using highly fluorescent bioconjugated nanoparticles. J. Am. Chem. Soc. **125**, 11474–11475 (2003)
- Zhou, X.C., Zhou, J.Z.: Improving the signal sensitivity and photostability of DNA hybridizations on microarrays by using dye-doped core-shell silica nanoparticles. Anal. Chem. 76, 5302–5312 (2004)
- Yin, Y., Alivisatos, A.P.: Colloidal nanocrystal synthesis and the organic-inorganic interface. Nature 437, 664–670 (2005)
- Peng, Z.A., Peng, X.G.: Formation of high-quality CdTe, CdSe, and CdS nanocrystals using CdO as precursor. J. Am. Chem. Soc. 123, 183–184 (2001)
- Algar, W.R., Massey, M., Krull, U.J.: The application of quantum dots, gold nanoparticles and molecular switches to optical nucleic-acid diagnostics. TrAC Trends Anal. Chem. 28, 292–306 (2009)
- Sutherland, A.J.: Quantum dots as luminescent probes in biological systems. Curr. Opin. Solid State Mater. Sci. 6, 365–370 (2002)
- 46. Han, M.Y., Gao, X.H., Su, J.Z., Nie, S.: Quantum-dot-tagged microbeads for multiplexed optical coding of biomolecules. Nat. Biotechnol. **19**, 631–635 (2001)
- 47. Clapp, A.R., Medintz, I.L., Mattoussi, H.: Forster resonance energy transfer investigations using quantum-dot fluorophores. ChemPhysChem 7, 47–57 (2006)
- Clapp, A.R., Medintz, I.L., Uyeda, H.T., Fisher, B.R., Goldman, E.R., Bawendi, M.G., Mattoussi, H.: Quantum dot-based multiplexed fluorescence resonance energy transfer. J. Am. Chem. Soc. 127, 18212–18221 (2005)
- Kim, J.H., Morikis, D., Ozkan, M.: Adaptation of inorganic quantum dots for stable molecular beacons. Sens. Actuator B Chem. 102, 315–319 (2004)
- Algar, W.R., Krull, U.J.: Towards multi-colour strategies for the detection of oligonucleotide hybridization using quantum dots as energy donors in fluorescence resonance energy transfer (FRET). Anal. Chim. Acta 581, 193–201 (2007)
- Zhang, C.Y., Yeh, H.C., Kuroki, M.T., Wang, T.H.: Single-quantum-dot-based DNA nanosensor. Nat. Mater. 4, 826–831 (2005)
- Peng, H., Zhang, L.J., Kjallman, T.H.M., Soeller, C., Travas-Sejdic, J.: DNA hybridization detection with blue luminescent quantum dots and dye-labeled single-stranded DNA. J. Am. Chem. Soc. 129, 3048–3049 (2007)
- 53. Su, Y.Y., Hu, M., Fan, C.H., He, Y., Li, Q.N., Li, W.X., Wang, L.H., Shen, P.P., Huang, Q.: The cytotoxicity of CdTe quantum dots and the relative contributions from released cadmium ions and nanoparticle properties. Biomaterials **31**, 4829–4834 (2010)
- Jurbergs, D., Rogojina, E., Mangolini, L., Kortshagen, U.: Silicon nanocrystals with ensemble quantum yields exceeding 60 %. Appl. Phys. Lett. 88, 23316 (2006)
- 55. He, Y., Kang, Z.H., Li, Q.S., Tsang, C.H.A., Fan, C.H., Lee, S.T.: Ultrastable, highly fluorescent, and water-dispersed silicon-based nanospheres as cellular probes. Angew. Chem. Int. Ed. 48, 128–132 (2009)
- 56. He, Y., Su, Y.Y., Yang, X.B., Kang, Z.H., Xu, T.T., Zhang, R.Q., Fan, C.H., Lee, S.T.: Photo and pH stable, highly-luminescent silicon nanospheres and their bioconjugates for immunofluorescent cell imaging. J. Am. Chem. Soc. **131**, 4434–4438 (2009)
- Zijlmans, H., Bonnet, J., Burton, J., Kardos, K., Vail, T., Niedbala, R.S., Tanke, H.J.: Detection of cell and tissue surface antigens using up-converting phosphors: a new reporter technology. Anal. Biochem. 267, 30–36 (1999)

- Kamimura, M., Miyamoto, D., Saito, Y., Soga, K., Nagasaki, Y.: Design of poly(ethylene glycol)/streptavidin coimmobilized upconversion nanophosphors and their application to fluorescence biolabeling. Langmuir 24, 8864–8870 (2008)
- Emoto, K., Nagasaki, Y., Kataoka, K.: Coating of surfaces with stabilized reactive micelles from poly(ethylene glycol)-poly(DL-lactic acid) block copolymer. Langmuir 15, 5212–5218 (1999)
- van de Rijke, F., Zijlmans, H., Li, S., Vail, T., Raap, A.K., Niedbala, R.S., Tanke, H.J.: Upconverting phosphor reporters for nucleic acid microarrays. Nat. Biotechnol. 19, 273–276 (2001)
- 61. Chen, Z.G., Chen, H.L., Hu, H., Yu, M.X., Li, F.Y., Zhang, Q., Zhou, Z.G., Yi, T., Huang, C.H.: Versatile synthesis strategy for carboxylic acid-functionalized upconverting nanophosphors as biological labels. J. Am. Chem. Soc. **130**, 3023–3029 (2008)
- Dulkeith, E., Morteani, A.C., Niedereichholz, T., Klar, T.A., Feldmann, J., Levi, S.A., van Veggel, F., Reinhoudt, D.N., Moller, M., Gittins, D.I.: Fluorescence quenching of dye molecules near gold nanoparticles: radiative and nonradiative effects. Phys. Rev. Lett. 89, 4 (2002)
- 63. Das, P.C., Puri, A.: Energy flow and fluorescence near a small metal particle. Phys. Rev. B 65, 8 (2002)
- 64. Fan, C.H., Wang, S., Hong, J.W., Bazan, G.C., Plaxco, K.W., Heeger, A.J.: Beyond superquenching: hyper-efficient energy transfer from conjugated polymers to gold nanoparticles. Proc. Natl. Acad. Sci. USA 100, 6297–6301 (2003)
- Kamat, P.V., Barazzouk, S., Hotchandani, S.: Electrochemical modulation of fluorophore emission on a nanostructured gold film. Angew. Chem. Int. Ed. 41, 2764–2767 (2002)
- Dubertret, B., Calame, M., Libchaber, A.J.: Single-mismatch detection using gold-quenched fluorescent oligonucleotides. Nat. Biotechnol. 19, 365–370 (2001)
- Maxwell, D.J., Taylor, J.R., Nie, S.M.: Self-assembled nanoparticle probes for recognition and detection of biomolecules. J. Am. Chem. Soc. 124, 9606–9612 (2002)
- Song, S.P., Liang, Z.Q., Zhang, J., Wang, L.H., Li, G.X., Fan, C.H.: Gold-nanoparticlebased multicolor nanobeacons for sequence-specific DNA analysis. Angew. Chem. Int. Ed. 48, 8670–8674 (2009)
- Pons, T., Medintz, I.L., Sapsford, K.E., Higashiya, S., Grimes, A.F., English, D.S., Mattoussi, H.: On the quenching of semiconductor quantum dot photoluminescence by proximal gold nanoparticles. Nano Lett. 7, 3157–3164 (2007)
- Zheng, M., Jagota, A., Semke, E.D., Diner, B.A., McLean, R.S., Lustig, S.R., Richardson, R.E., Tassi, N.G.: DNA-assisted dispersion and separation of carbon nanotubes. Nat. Mater. 2, 338–342 (2003)
- 71. Zheng, M., Jagota, A., Strano, M.S., Santos, A.P., Barone, P., Chou, S.G., Diner, B.A., Dresselhaus, M.S., McLean, R.S., Onoa, G.B., Samsonidze, G.G., Semke, E.D., Usrey, M., Walls, D.J.: Structure-based carbon nanotube sorting by sequence-dependent DNA assembly. Science **302**, 1545–1548 (2003)
- Liang, F., Chen, B.: A review on biomedical applications of single-walled carbon nanotubes. Curr. Med. Chem. 17, 10–24 (2010)
- Zhu, Z., Yang, R.H., You, M.X., Zhang, X.L., Wu, Y.R., Tan, W.H.: Single-walled carbon nanotube as an effective quencher. Anal. Bioanal. Chem. 396, 73–83 (2010)
- 74. Yang, R.H., Tang, Z.W., Yan, J.L., Kang, H.Z., Kim, Y.M., Zhu, Z., Tan, W.H.: Noncovalent assembly of carbon nanotubes and single-stranded DNA: an effective sensing platform for probing biomolecular interactions. Anal. Chem. 80, 7408–7413 (2008)
- Yang, R.H., Jin, J.Y., Chen, Y., Shao, N., Kang, H.Z., Xiao, Z., Tang, Z.W., Wu, Y.R., Zhu, Z., Tan, W.H.: Carbon nanotube-quenched fluorescent oligonucleotides: probes that fluoresce upon hybridization. J. Am. Chem. Soc. **130**, 8351–8358 (2008)
- 76. Liu, Y., Wang, Y.X., Jin, J.Y., Wang, H., Yang, R.H., Tan, W.H.: Fluorescent assay of DNA hybridization with label-free molecular switch: reducing background-signal and improving specificity by using carbon nanotubes. Chem. Commun. 665–667 (2009)

- Rao, C.N.R., Sood, A.K., Subrahmanyam, K.S., Govindaraj, A.: Graphene: the new twodimensional nanomaterial. Angew. Chem. Int. Ed. 48, 7752–7777 (2009)
- Dreyer, D.R., Park, S., Bielawski, C.W., Ruoff, R.S.: The chemistry of graphene oxide. Chem. Soc. Rev. 39, 228–240 (2010)
- 79. Swathi, R.S., Sebastian, K.L.: Resonance energy transfer from a dye molecule to graphene. J. Chem. Phys. **129**, 9 (2008)
- He, S.J., Song, B., Li, D., Zhu, C.F., Qi, W.P., Wen, Y.Q., Wang, L.H., Song, S.P., Fang, H.P., Fan, C.H.: A graphene nanoprobe for rapid, sensitive, and multicolor fluorescent DNA analysis. Adv. Funct. Mater. 20, 453–459 (2010)
- Lu, C.H., Yang, H.H., Zhu, C.L., Chen, X., Chen, G.N.: A graphene platform for sensing biomolecules. Angew. Chem. Int. Ed. 48, 4785–4787 (2009)
- Wang, Y., Li, Z.H., Hu, D.H., Lin, C.T., Li, J.H., Lin, Y.H.: Aptamer/graphene oxide nanocomplex for in situ molecular probing in living cells. J. Am. Chem. Soc. 132, 9274–9276 (2010)
- Feng, L.Z., Zhang, S.A., Liu, Z.: Graphene based gene transfection. Nanoscale 3, 1252–1257 (2011)
- Peng, C., Hu, W.B., Zhou, Y.T., Fan, C.H., Huang, Q.: Intracellular imaging with a graphenebased fluorescent probe. Small 6, 1686–1692 (2010)
- Gerardi, R.D., Barnett, N.W., Lewis, S.W.: Analytical applications of tris(2,2'bipyridyl)ruthenium(III) as a chemiluminescent reagent. Anal. Chim. Acta 378, 1–41 (1999)
- Dodeigne, C., Thunus, L., Lejeune, R.: Chemiluminescence as a diagnostic tool. A review. Talanta 51, 415–439 (2000)
- Marquette, C.A., Blum, L.J.: Electro-chemiluminescent biosensing. Anal. Bioanal. Chem. 390, 155–168 (2008)
- Kricka, L.J.: Clinical applications of chemiluminescence. Anal. Chim. Acta 500, 279–286 (2003)
- Aslan, K., Geddes, C.D.: Metal-enhanced chemiluminescence: advanced chemiluminescence concepts for the 21st century. Chem. Soc. Rev. 38, 2556–2564 (2009)
- Lin, J.M., Shan, X.Q., Hanaoka, S., Yamada, M.: Luminol chemiluminescence in unbuffered solutions with a cobalt(II)-ethanolamine complex immobilized on resin as catalyst and its application to analysis. Anal. Chem. **73**, 5043–5051 (2001)
- Li, Z.P., Li, K.A., Tong, S.Y.: Study of the catalytic effect of copper(II)-protein complexes on luminol-H2O2 chemiluminescence reaction and its analytical application. Anal. Lett. 32, 901–913 (1999)
- Zhang, S.S., Zhong, H., Ding, C.F.: Ultrasensitive flow injection chemiluminescence detection of DNA hybridization using signal DNA probe modified with Au and CuS nanoparticles. Anal. Chem. 80, 7206–7212 (2008)
- Niazov, T., Pavlov, V., Xiao, Y., Gill, R., Willner, I.: DNAzyme-functionalized Au nanoparticles for the amplified detection of DNA or telomerase activity. Nano Lett. 4, 1683–1687 (2004)
- Wang, Z.P., Hu, J.Q., Jin, Y., Yao, X., Li, J.H.: In situ amplified chemiluminescent detection of DNA and immunoassay of IgG using special-shaped gold nanoparticles as label. Clin. Chem. 52, 1958–1961 (2006)
- Qi, Y.Y., Li, B.X., Zhang, Z.J.: Label-free and homogeneous DNA hybridization detection using gold nanoparticles-based chemiluiminescence system. Biosens. Bioelectron. 24, 3581–3586 (2009)
- Nelson, N.C., BenCheikh, A., Matsuda, E., Becker, M.M.: Simultaneous detection of multiple nucleic acid targets in a homogeneous format. Biochemistry 35, 8429–8438 (1996)
- Nelson, N.C., Kacian, D.L.: Chemiluminescent DNA probes-A comparison of the acridinium and dioxetane detection systems and their use in clinical diagnostic assays. Clin. Chim. Acta 194, 73–90 (1990)
- Nelson, N.C., Hammond, P.W., Matsuda, E., Goud, A.A., Becker, M.M.: Detection of all single-base mismatches in solution by chemiluminescence. Nucleic Acids Res. 24, 4998–5003 (1996)

- Yin, X.B., Dong, S.J., Wang, E.K.: Analytical applications of the electrochemiluminescence of tris (2.2'-bipyridyl) ruthenium and its derivatives. TrAC Trends Anal. Chem. 23, 432–441 (2004)
- Hu, L.Z., Xu, G.B.: Applications and trends in electrochemiluminescence. Chem. Soc. Rev. 39, 3275–3304 (2010)
- 101. Tokel, N.E., Bard, A.J.: Electrogenerated chemiluminescence. 9. electrochemistry and emission from systems containing Tris(2,2'-bipyridine)Ruthenium(II) dichloride. J. Am. Chem. Soc 94, 2862–2863 (1972)
- 102. Zu, Y.B., Bard, A.J.: Electrogenerated chemiluminescence. 66. The role of direct coreactant oxidation in the ruthenium tris(2,2')bipyridyl/tripropylamine system and the effect of halide ions on the emission intensity. Anal. Chem. **72**, 3223–3232 (2000)
- 103. Kanoufi, F., Zu, Y.B., Bard, A.J.: Homogeneous oxidation of trialkylamines by metal complexes and its impact on electrogenerated chemiluminescence in the trialkylamine/Ru(bpy)₍₃₎⁽²⁺⁾ system. J. Phys. Chem. B **105**, 210–216 (2001)
- 104. Miao, W.J.: Electrogenerated chemiluminescence and its biorelated applications. Chem. Rev. 108, 2506–2553 (2008)
- 105. Xu, X.H., Bard, A.J.: Immobilization and hybridization of DNA and Aluminum (III) alkanebisphosphonate thin-film with electrogenerated chemiluminescent detection. J. Am. Chem. Soc. 117, 2627–2631 (1995)
- 106. Miao, W.J., Bard, A.J.: Electrogenerated chemiluminescence. 77. DNA hybridization detection at high amplification with [Ru(bpy)(3)](2+)-containing microspheres. Anal. Chem. 76, 5379–5386 (2004)
- 107. Miao, W.J., Bard, A.J.: Electrogenerated chemluminescence. 72. Determination of immobilized DNA and C-reactive protein on Au(111) electrodes using Tris(2,2'bipyridyl)ruthenium(II) labels. Anal. Chem. 75, 5825–5834 (2003)
- Lo, W.Y., Baeumner, A.J.: Evaluation of internal standards in a competitive nucleic acid sequence-based amplification assay. Anal. Chem. 79, 1386–1392 (2007)
- Lo, W.Y., Baeumner, A.J.: RNA internal standard synthesis by nucleic acid sequencebased amplification for competitive quantitative amplification reactions. Anal. Chem. 79, 1548–1554 (2007)
- 110. Zhang, J., Qi, H.L., Li, Y., Yang, J., Gao, Q., Zhang, C.X.: Electrogenerated chemiluminescence DNA biosensor based on hairpin DNA probe labeled with ruthenium complex. Anal. Chem. 80, 2888–2894 (2008)
- 111. Sun, X.P., Du, Y., Dong, S.J., Wang, E.K.: Method for effective immobilization of Ru(bpy)₃²⁺ on an electrode surface for solid-state electrochemiluminescene detection. Anal. Chem. **77**, 8166–8169 (2005)
- Wang, H., Zhang, C.X., Li, Y., Qi, H.L.: Electrogenerated chemiluminescence detection for deoxyribonucleic acid hybridization based on gold nanoparticles carrying multiple probes. Anal. Chim. Acta 575, 205–211 (2006)
- 113. Zhang, J., Song, S.P., Zhang, L.Y., Wang, L.H., Wu, H.P., Pan, D., Fan, C.H.: Sequence-specific detection of femtomolar DNA via a chronocoulometric DNA sensor (CDS): effects of nanoparticle-mediated amplification and nanoscale control of DNA assembly at electrodes. J. Am. Chem. Soc. **128**, 8575–8580 (2006)
- 114. Lee, J.G., Yun, K., Lim, G.S., Lee, S.E., Kim, S., Park, J.K.: DNA biosensor based on the electrochemiluminescence of Ru(bpy)₍₃₎⁽²⁺⁾ with DNA-binding intercalators. Bioelectrochemistry **70**, 228–234 (2007)
- Jiang, H., Ju, H.X.: Enzyme-quantum dots architecture for highly sensitive electrochemiluminescence biosensing of oxidase substrates. Chem. Commun. 404–406 (2007)
- 116. Jiang, H., Ju, H.X.: Electrochemiluminescence sensors for scavengers of hydroxyl radical based on its annihilation in CdSe quantum dots film/peroxide system. Anal. Chem. 79, 6690–6696 (2007)
- 117. Liu, X., Jiang, H., Lei, J.P., Ju, H.X.: Anodic electrochemiluminescence of CdTe quantum dots and its energy transfer for detection of catechol derivatives. Anal. Chem. **79**, 8055–8060 (2007)

- 118. Hu, X.F., Wang, R.Y., Ding, Y., Zhang, X.L., Jin, W.R.: Electrochemiluminescence of CdTe quantum dots as labels at nanoporous gold leaf electrodes for ultrasensitive DNA analysis. Talanta 80, 1737–1743 (2010)
- 119. Fivash, M., Towler, E.M., Fisher, R.J.: BIAcore for macromolecular interaction. Curr. Opin. Biotechnol. 9, 97–101 (1998)
- Malmborg, A.C., Borrebaeck, C.A.K.: BIAcore as a tool in antibody engineering. J. Immunol. Methods 183, 7–13 (1995)
- 121. Campbell, C.T., Kim, G.: SPR microscopy and its applications to high-throughput analyses of biomolecular binding events and their kinetics. Biomaterials **28**, 2380–2392 (2007)
- 122. Homola, J.: Surface plasmon resonance sensors for detection of chemical and biological species. Chem. Rev. **108**, 462–493 (2008)
- 123. Homola, J., Yee, S.S., Gauglitz, G.: Surface plasmon resonance sensors: review. Sens. Actuator B Chem. **54**, 3–15 (1999)
- 124. Persson, B., Stenhag, K., Nilsson, P., Larsson, A., Uhlen, M., Nygren, P.A.: Analysis of oligonucleotide probe affinities using surface plasmon resonance: a means for mutational scanning. Anal. Biochem. 246, 34–44 (1997)
- Nilsson, P., Persson, B., Uhlen, M., Nygren, P.A.: Real-time monitoring of DNA manipulations using biosensor technology. Anal. Biochem. 224, 400–408 (1995)
- 126. He, L., Musick, M.D., Nicewarner, S.R., Salinas, F.G., Benkovic, S.J., Natan, M.J., Keating, C.D.: Colloidal Au-enhanced surface plasmon resonance for ultrasensitive detection of DNA hybridization. J. Am. Chem. Soc. **122**, 9071–9077 (2000)
- 127. Rothenhausler, B., Knoll, W.: Surface-plasmon microscopy. Nature 332, 615-617 (1988)
- 128. Shumaker-Parry, J.S., Zareie, M.H., Aebersold, R., Campbell, C.T.: Microspotting streptavidin and double-stranded DNA Arrays on gold for high-throughput studies of protein-DNA interactions by surface plasmon resonance microscopy. Anal. Chem. **76**, 918–929 (2004)
- 129. Brockman, J.M., Nelson, B.P., Corn, R.M.: Surface plasmon resonance imaging measurements of ultrathin organic films. Annu. Rev. Phys. Chem. **51**, 41–63 (2000)
- Thiel, A.J., Frutos, A.G., Jordan, C.E., Corn, R.M., Smith, L.M.: In situ surface plasmon resonance imaging detection of DNA hybridization to oligonucleotide arrays on gold surfaces. Anal. Chem. 69, 4948–4956 (1997)
- 131. Nelson, B.P., Grimsrud, T.E., Liles, M.R., Goodman, R.M., Corn, R.M.: Surface plasmon resonance imaging measurements of DNA and RNA hybridization adsorption onto DNA microarrays. Anal. Chem. **73**, 1–7 (2001)
- 132. Jordan, C.E., Frutos, A.G., Thiel, A.J., Corn, R.M.: Surface plasmon resonance imaging measurements of DNA hybridization adsorption and streptavidin/DNA multilayer formation at chemically modified gold surfaces. Anal. Chem. 69, 4939–4947 (1997)
- Lee, H.J., Goodrich, T.T., Corn, R.M.: SPR imaging measurements of 1-D and 2-D DNA microarrays created from microfluidic channels on gold thin films. Anal. Chem. 73, 5525– 5531 (2001)
- 134. Lee, H.J., Li, Y., Wark, A.W., Corn, R.M.: Enzymatically amplified surface plasmon resonance imaging detection of DNA by exonuclease III digestion of DNA microarrays. Anal. Chem. 77, 5096–5100 (2005)
- 135. Nie, S.M., Emery, S.R.: Probing single molecules and single nanoparticles by surfaceenhanced Raman scattering. Science **275**, 1102–1106 (1997)
- Vodinh, T., Houck, K., Stokes, D.L.: PROBES surface-enhanced Raman gene probes. Anal. Chem. 66, 3379–3383 (1994)
- 137. Campion, A., Kambhampati, P.: Surface-enhanced Raman scattering. Chem. Soc. Rev. 27, 241–250 (1998)
- 138. Graham, D.: The next generation of advanced spectroscopy: surface enhanced Raman scattering from metal nanoparticles. Angew. Chem. Int. Ed. **49**, 9325–9327 (2010)
- Graham, D., Faulds, K.: Quantitative SERRS for DNA sequence analysis. Chem. Soc. Rev. 37, 1042–1051 (2008)
- 140. Cao, Y.W.C., Jin, R.C., Mirkin, C.A.: Nanoparticles with Raman spectroscopic fingerprints for DNA and RNA detection. Science **297**, 1536–1540 (2002)

- 141. Braun, G., Lee, S.J., Dante, M., Nguyen, T.Q., Moskovits, M., Reich, N.: Surface-enhanced Raman spectroscopy for DNA detection by nanoparticle assembly onto smooth metal films. J. Am. Chem. Soc. **129**, 6378–6379 (2007)
- 142. Kang, T., Yoo, S.M., Yoon, I., Lee, S.Y., Kim, B.: Patterned multiplex pathogen DNA detection by Au particle-on-wire SERS sensor. Nano Lett. **10**, 1189–1193 (2010)
- 143. He, Y., Su, S., Xu, T.T., Zhong, Y.L., Antonio Zapien, J., Li, J., Fan, C.H., Lee, S.-T.: Silicon nanowires-based highly-efficient SERS-active platform for ultrasensitive DNA detection. Nano Today 6, 122–130 (2011)
- 144. Thompson, D.G., Enright, A., Faulds, K., Smith, W.E., Graham, D.: Ultrasensitive DNA detection using oligonucleotide-silver nanoparticle conjugates. Anal. Chem. 80, 2805–2810 (2008)
- 145. MacAskill, A., Crawford, D., Graham, D., Faulds, K.: DNA sequence detection using surfaceenhanced resonance Raman spectroscopy in a homogeneous multiplexed assay. Anal. Chem. 81, 8134–8140 (2009)
- 146. Graham, D., Stevenson, R., Thompson, D.G., Barrett, L., Dalton, C., Faulds, K.: Combining functionalized nanoparticles and SERS for the detection of DNA relating to disease. Faraday Discuss. 149, 291–299 (2010)
- 147. Graham, D., Thompson, D.G., Smith, W.E., Faulds, K.: Control of enhanced Raman scattering using a DNA-based assembly process of dye-coded nanoparticles. Nat. Nanotechnol. 3, 548– 551 (2008)
- 148. Wabuyele, M.B., Vo-Dinh, T.: Detection of human immunodeficiency virus type 1 DNA sequence using plasmonics nanoprobes. Anal. Chem. **77**, 7810–7815 (2005)
- 149. Harpster, M.H., Zhang, H., Sankara-Warrier, A.K., Ray, B.H., Ward, T.R., Kollmar, J.P., Carron, K.T., Mecham, J.O., Corcoran, R.C., Wilson, W.C., Johnson, P.A.: SERS detection of indirect viral DNA capture using colloidal gold and methylene blue as a Raman label. Biosens. Bioelectron. 25, 674–681 (2009)
- 150. Zhang, Z., Wang, Y., Fan, C.H., Li, C., Li, Y., Qian, L.L., Fu, Y.M., Shi, Y.Y., Hu, J., He, L.: Asymmetric DNA origami for spatially addressable and index-free solution-phase DNA chips. Adv. Mater. 22, 2672–2675 (2010)
- 151. Ke, Y.G., Lindsay, S., Chang, Y., Liu, Y., Yan, H.: Self-assembled water-soluble nucleic acid probe tiles for label-free RNA hybridization assays. Science **319**, 180–183 (2008)
- Husale, S., Persson, H.H.J., Sahin, O.: DNA nanomechanics allows direct digital detection of complementary DNA and microRNA targets. Nature 462, 1075–1078 (2009)

Chapter 7 Electrochemical Detection of DNA Using Nanomaterials Based Sensors

Sergio Marín and Arben Merkoçi

Abstract The detection of non-amplified DNA sequences plays a crucial role in the rapid diagnosis of genetic-related diseases especially for early stage treatment. Among the various biosensors that have been used for DNA detection, electrochemical sensors show great promise because they present efficient signal transduction and are capable of precise DNA recognition at a relatively low cost in addition to the broad range of designs with interest to be applied in different kinds of samples. Advancements in micro- and nanotechnologies, specifically fabrication techniques and new nanomaterials, have enabled for the development of highly sensitive, highly specific electrochemical sensors making them attractive for the detection of small sequence variations. Furthermore, the integration of sensors with sample preparation and fluidic processes enables for rapid, multiplexed electrochemical DNA detection essential for point of care clinical diagnostics.

7.1 Introduction

The recent discovery and sequencing of the human genome has provided valuable insight into understanding how genetic factors contribute to the development of diseases. One of the most important factors is the detection of DNA sequence variations that play an important role in the diagnosis of genetic-related diseases, especially for early stage treatment and monitoring [1-3].

Conventional technologies as for example the PCR requires a long time so as to obtain a final diagnostic report. The biosensors in general and especially

S. Marín • A. Merkoçi (🖂)

Nanobioelectronics and Biosensors Group, CIN2 (ICN-CSIC), Institut Català de Nanotecnologia, Barcelona, Spain e-mail: arben.merkoci@icn.cat

G. Spoto and R. Corradini (eds.), *Detection of Non-Amplified Genomic DNA*, Soft and Biological Matter, DOI 10.1007/978-94-007-1226-3_7, © Springer Science+Business Media Dordrecht 2012

electrochemical DNA sensors can be an interesting alternative. The electrochemical DNA sensors offer simplicity in operation and sample manipulation and provide highly sensitive and specific measurements for a broad spectrum of biomolecules [4–8]. The sample size required is small, ranging several microliters to hundred of nanoliters, which includes the sample pretreatment reagent. Additionally, the detection time is relatively fast, varying from a few minutes to 10 s. However, the most important feature of electrochemical DNA sensors is their potential to be easily transformed from a laboratory-based instrument to a commercializable point of care device. Because of all these advantages, electrochemical biosensing for DNA diagnostics is becoming a very promising area of research and development.

Nowadays, micro- and nanotechnologies have shown emerging potential in electrochemical DNA diagnostics. Electrochemical sensors offer perfect interface for incorporating these technologies, which includes a variety of new materials and fabrication processes. Nanomaterials can be used in various aspects of the detection system including capture probes, reporting molecules, electrode fabrication, and electrode coatings [9–16]. These materials offer improved biocompatibility, additional binding sites and higher signal intensities (via enhanced electrical properties) compared with traditional materials in electrochemical sensors [8, 12, 16–19]. Nanofabrications allow for miniaturization of the sensor, which improve the sensitivity and reduce the samples and reagent volumes, making the detection process more efficient.

A typical electrochemical DNA sensor consists of an electrode, a capture probe and a reporter probe. A capture probe is a DNA strand used to recognize and bind to the target DNA and is usually immobilized onto solid substrate, such as the electrode surface. A reporter probe is a DNA strand that either alone or through a label makes possible the generation of the electrochemical signal as response to the positive hybridization reaction. Both the capture probe and reporter probe are designed so as to show high specificity to the target DNA. Probe, target, and reporter molecules can all be modified or linked with properly integrated nanomaterials. Because of their high surface surface-to-volume ratios and biologic compatibilities, nanomaterials not only increase the signal intensity but also help to accumulate/separate specific DNA molecules during electrochemical reactions, which greatly improve the signalto-noise ratio, especially for sequence-specific recognition [19]. A wide variety of nanomaterials can be applied, where the most common include metal nanoparticles, quantum dots, carbon nanotubes and nanowires.

7.2 Nanoparticles as Quantification Tags

Nanoparticles can be used in a variety of bioanalytical formats with electrochemical detection. By analogy to fluorescence-based methods, several electrochemical-detection methods have been pursued in which DNA sequences have been labelled with active nanoparticles. The characteristic electrochemical response of nanoparticle reporter signals the hybridizations event.



Fig. 7.1 Detection strategies for gold nanoparticles. (A) Potentiometric/voltammetric stripping assay. The hybridization event occurs between DNA strand (*a*) and gold-tagged DNA (*b*). The gold-labeled duplex (*c*) formed is then detected according to each strategy: (**I**) direct detection of the nanoparticles onto the bare electrode without the need for tag dissolution (Adapted from reference [20]) (**II**) the gold nanoparticles are dissolved with HBr/Br₂ treatment and then detected by stripping techniques (Adapted from reference [21]) and, (**III**) the gold nanoparticles are first covered with Ag by a deposition treatment and then detected by stripping techniques (Adapted from reference [22]) (**B**). Conductivity assay. Probe DNA immobilized in a small gap between two electrodes (*a*) is hybridized with DNA target (*b*) and then with gold-modified DNA probes (*b'*). Gold is accumulated in the gap (*c*). Silver enhancement (*d*) is performed in the presence of hydroquinone (*HQ*). The silver precipitated onto the gold nanoparticles (*e*) improves the sensitivity of the assay by lowering the resistance across the electrode gap (Adapted from reference [24])

7.2.1 Gold Nanoparticles

Gold nanoparticles are the most popular nanoparticles that have been used as labels to hybridization signals of DNA. Three strategies for DNA detections of gold have been reported (see Fig. 7.1).

According to the first strategy, a direct detection of gold nanoparticle without the previous dissolution is reported [16, 20]. A DNA strand modified with gold nanoparticles is hybridized with DNA probe that is marked with paramagnetic beads. These complexes are concentrated onto the graphite-epoxy composite electrode that incorporates a magnet in its matrix. The differential pulse voltammetry is used for a direct voltammetric detection of resulting gold-DNA-paramagnetic bead complex [20] (see Fig. 7.1AI).

According to the second strategy (see Fig. 7.1AII), the intrinsic electrochemical signal of the nanoparticle can be observed after dissolving it with HBr/Br₂ [21]. The gold (III) ions obtained were preconcentrated by electrochemical reduction onto an electrode and subsequently determined by anodic-stripping voltammetry.

The third strategy used for detection of gold nanoparticles uses the silver deposition on the gold nanoparticles after the hybridization and enhanced electrochemical signal attributed to silver is obtained [22, 23]. The accumulated silver is dissolved and detected afterwards using a potentiometric stripping protocol (Fig. 7.1AIII).

Nevertheless, other interesting methods have been reported. Mirkin et al. [24] have exploited the silver-deposition technique to construct a sensor based on conductivity measurements. In their approach, a small array of microelectrodes with gaps ($20 \ \mu m$) between the electrodes leads is constructed, and probe sequences are immobilized on the substrate between the gaps. Using a three-component sandwich approach, hybridized target DNA is used to recruit gold nanoparticle-tagged reporter probes between the electrode leads. The nanoparticle labels are then developed in the silver-enhancer solution leading to a sharp drop in the resistance of the circuit (Fig. 7.1B).

7.2.2 Quantum Dots

In a similar way as the gold nanoparticles quantum dots (QDs) also are detected using three different strategies (see Fig. 7.2). According to the first strategy, after DNA hybridization quantum dots are dissolved using nitric acid solution and the electrochemical detection of cadmium (II) ions is performed (Fig. 7.2A) [25, 26].

The second strategy is based on nanoparticle-promoted cadmium precipitation (see Fig. 7.2B), using a fresh cadmium solution hydroquinone, used to enlarge the nanoparticle tag and amplify the stripping DNA hybridization signal [27]. In addition to measurements of the dissolved cadmium, it was demonstrated also in direct "solid-state" measurements following a "magnetic" collection of a "magnetic bead/DNA hybrid/CdS tracer" assembly onto a thick-film electrode transducer.

Fig. 7.2 Schematic representation of a DNA biosensor that uses magnetic beads. The formed CdSlabeled duplex is detected according to each strategy: (I) Multi-target electrical DNA detection protocol based on different inorganic colloid nanocrystal tracers. (*A*) Introduction of probemodified magnetic beads. (*B*) Hybridization with the DNA targets. (*C*) Second hybridization with the QD-labeled probes. (*D*) Dissolution of QDs and electrochemical detection. (Adapted from reference [25]) and,(II) the CdS QDs are first enhanced with cadmium by a deposition treatment using hydroquinone and afterwards dissolved with HNO₃ and detected by stripping techniques (adapted from reference [29]) (III) Schematic representation of sandwich protocol (not toscale): (*A*) immobilization of the biotinylated CF-A probe onto the streptavidin-coated magnetic beads; (*B*) first hybridization between CF-T and CF-A; (*C*) second hybridization between CF-T and CF-B modified thiol; (*D*) addition and capture of the CdS quantum dots (Adapted from reference [30])







According to the third strategy, electrochemical detection of a cadmium sulfide quantum dots (CdS QDs)–DNA complex connected to paramagnetic microbeads (MB) was performed without the need for chemical dissolving (Fig. 7.2C) [28].

ZnS

CdS

PbS

The labeling of probes bearing different DNA sequences with different nanoparticles enables the simultaneous detection of more than one target in a sample, as shown in Fig. 7.3. The number of targets that can be readily detected simultaneously can be controlled by the number of voltammetrically distinguishable nanoparticle markers.

Following this idea a multi-target sandwich hybridization assay involving a dual hybridization event, with probes linked to three tagged inorganic crystals and to magnetic beads has been reported [25].

The DNA-connected QDs yielded well-defined and resolved stripping peaks at -1.12 V (Zn), -0.68 V (Cd) and -0.53 V (Pb) at the mercury coated glassy carbon electrode (vs. the Ag/AgCl reference-electrode) (Fig. 7.3).

7.2.3 Other Nanoparticles

Other nanoparticles such as silver nanoparticleshave been used in electrochemical DNA sensors to amplify the obtained signal (Fig. 7.4I). As reported by Fu et al. using DNA biosensors based in situ DNA amplification with nanosilver as label and horseradish peroxidase enzyme (HRP) [29]. The thiolated oligomer single-stranded DNA (ssDNA) was initially directly immobilized onto a gold electrode. A competitive format, hybridization reaction was carried out via immersing the DNA biosensor into a stirred hybridization solution containing different concentrations of the complementary ssDNA and constant concentration of nanosilver-labeled ssDNA, followed by binding with HRP. The adsorbed HRP amount on the probe surface decreased with the increment of the target ssDNA in the sample. The hybridization events were monitored by using differential pulse voltammetry (DPV) toward the reduction of H_2O_2 generated by HRP. The reduction current from the enzyme-generated product was related to the number of target ssDNA molecules in the sample.

In another assay the hybridization of the target DNA with the silver nanoparticle– oligonucleotide DNA probe, followed by the release of the silver metal atoms anchored on the hybrids by oxidative metal dissolution and the indirect determination of the solubilized Ag^+ by anodic stripping voltammetry (ASV) at a carbon fiber ultramicroelectrode is performed (Fig. 7.4II) [30].

Besides QDs, gold-coated iron nanoparticles also have been used in DNAdetection assays [31]. After hybridization, the captured gold-iron nanoparticles used as DNA labels are dissolved and the released iron is quantified by cathodic stripping voltammetry in the presence of the 1-nitroso-2-naphthol ligand and a bromate catalyst. The DNA-labeling mode developed offers high sensitivity, well-defined dependence on concentration, and minimal contributions from non-complementary nucleic acids (Fig. 7.4III).

7.3 Nanowires and Nanotubes for Labelling and Signal Enhancement

7.3.1 Carbon Nanotubes as Substrates for DNA Attachment

Carbon nanotubes (CNTs) were first noticed and characterized in 1991 by Iijima [32] of NEC Corporation in Japan. CNTs are a new allotrope of carbon originated



from fullerene family, which can be described as a graphite sheet rolled up into a nanoscale-tube (which are single-wall carbon nanotubes, SWCNTs), or with additional graphene tubes around the core of an SWCNT (which are multi-wall CNTs, MWCNTs) [33]. MWCNTs consist of two or more concentric cylindrical shells of graphene sheets coaxially arranged around a central hollow area with spacing between the layers which is close to that of the interlayer separation as in graphite (0.34 nm). In contrast, SWCNT are made of single graphene (one layer of graphite) cylinders and have a very narrow size distribution (1–2 nm). CNTs offer unique electronic and mechanical properties combined with chemical stability. The combination of a biological compound with CNT to monitor a biochemical event seems to be a promising alternative for biosensors designs due to the catalytic effects that this nanomaterial can bring to the detection event.

DNA-end attachment via covalent linkage to the CNTs surface has been demonstrated and further extended for the hybridization of the captured strands to their complementary target sequences. CNTs as sensing surfaces for nucleic acids detection have been extensively demonstrated [34–37].

Tang et al. [34] modified the gold electrodes by carboxylic group-functionalized CNTs activated using N-ethyl-N'-(3-dimethylaminopropyl) carbodiimide (EDC) and N-hydroxysuccinimidobiotin (NHS) and covalently connected the ssDNA.

Abdullin et al. [35] modified glassy-carbon electrodes (GCEs) with preoxidized CNTs. According to the data of atomic force microscopy, the layers of CNTs on GCEs possess a homogeneous nanostructurized surface. Guanine and deoxyguanosine monophosphate could be strongly adsorbed on GCE/CNT and oxidized at +690 and +930 mV (pH 7.0), respectively. Similar attachment of DNA onto CNTs was also reported by Kerman et al. [36] and Erdem et al. [37].

7.3.2 CNTs as Signal Amplifier for Hybridization

The conductive properties of CNTs suggest that they could mediate electron transfer reactions and enhance the relative electrochemical reactivity with electroactive species in solution when used as the electrode material or modifiers. As electrode or substrates modified materials, CNTs show better electrochemical behavior than traditional glass carbon electrodes, carbon paste electrodes and other carbon electrodes [38].

Fig. 7.4 (**I**) Schematic representation of electrochemical detection of DNA using nanosilver as label and HRP as enhancer (Adapted from reference [31]). (**II**) Schematic representation of electrochemical stripping detection of DNA hybridization based on a silver nanoparticle label (Adapted from reference [32]). (**III**) Schematic representation of the DNA-detection protocol based on gold-coated iron nanoparticle tags: (*A*) introduction of the streptavidin-coated polystyrene beads; (*B*) attachment of the biotinylated target; (*C*) hybridization with the iron/gold particle-labeled probe; (*D*) dissolution of the iron/gold particle tag and stripping voltammetric detection at a hanging mercury drop electrode (Adapted from reference [33])

Therefore, CNTs used as substrates for DNA attachment in electrochemical biosensors, as explained previously [39] could also be regarded as electrochemical signal amplifier. The guanine oxidation signal of double stranded calf-thymus DNA after 3 min accumulation was 20 times higher at a CNTs modified glassy carbon electrodes cross-linked with glutaraldehyde (GTA) than at a bare GCE using differential pulse voltammetry, while the peak potential was around 45 mV less positive. The guanine oxidation signal was demonstrated to be highly reproducible, with 3.4% RSD for five different electrodes [40].

Recently, CNTs have also been utilized as a novel support material to concentrate nanoparticles or enzyme molecules on it as a more powerful DNA hybridization indicator than using a single nanoparticle or enzyme molecule (see Fig. 7.5). Wang et al. [41] described an effective method for amplifying electrical detection of DNA hybridization based on CNTs carrying a large number of CdS particle tracers. Such use of CNT amplification platforms was combined with an ultrasensitive stripping voltammetric detection of the dissolved CdS tags following dual hybridization events of a sandwich assay on a streptavidin modified 96-well microplate.

Anchoring of the monolayer-protected quantum dots to the acetone-activated CNT was accomplished via hydrophobic interactions. SEM images indicated that the nanocrystals were attached along the CNT sidewall, with a loading of around 500 particles per CNT. A substantial (500 fold) lowering of the detection limit was obtained compared to conventional single particle stripping hybridization assays, reflecting the CdS loading on the CNT carrier. A large excess (250 fold) of non-complimentary oligonucleotides had minimal effect on the response. They also demonstrated the use of CNTs for dramatically amplifying enzyme-based bioaffinity electrical sensing DNA [42]. In the new bioaffinity assays, CNTs played a dual amplification role in both the recognition and transduction events, namely as carriers for numerous enzyme tags and for accumulating the product of the enzymatic reaction. Such coupling of several CNT-derived amplification processes led to the low detection limit (Fig. 7.5II). Li et al. [43] developed an ultrasensitive electrogenerated chemiluminescence (ECL) detection method of DNA hybridization based-on SWCNT carrying a large number of ruthenium complex tags. The probe of single strand DNA (ssDNA) and ruthenium complex were loaded at SWCNT, which was taken as an ECL probe (Fig. 7.5III). When the capture ssDNA with a thiol group was self-assembled onto the surface of gold electrode, and then hybridized with target ssDNA and further hybridized with the ECL probe to form DNA sandwich conjugate, a strong ECL response was electrochemically generated.

7.3.3 Nanowires for DNA Attachment

Multisegment nanowires also have been used via a straightforward surface functionalization method for DNA detection [44]. Nanowires comprising CdTe-Au-CdTe segments are fabricated using electrochemical deposition. The electrical characterization indicates a p-type behavior for the multisegment nanostructures,



Fig. 7.5 Carbon nanotubes as carriers (I) of QDs. Schematic representation of the analytical protocol: (a) Dual hybridization event of the sandwich hybridization assay, leading to capturing of the CdS-loaded CNT tags in the microwell; (b) dissolution of the CdS tracer; (c) stripping voltammetric detection of cadmium at a mercury-coated glassy carbon electrode. P1, DNA probe 1; T, DNA target; P2, DNA probe 2 (Adapted from reference [43]). (II) of alcalino phosphatase (ALP). Schematic representation of the analytical protocol: (A) Capture of the ALP -loaded CNT tags to the streptavidin-modified magnetic beads by a sandwich DNA hybridization (a) or Ab-Ag-Ab interaction (b). (B) Enzymatic reaction. (C) Electrochemical detection of the product of the enzymatic reaction at the CNT-modified glassy carbon electrode; MB Magnetic beads, P DNA probe 1, T DNA target, P2 DNA probe 2, Ab1 first antibody, Ag antigen, Ab2 secondary antibody, S and P substrate and product, respectively, of the enzymatic reaction, GC glassy carbon electrode, CNT carbon nanotube layer (Adapted from reference [44]) (III) of Rutenium complex (Ru). Schematic diagram of the electrogenerated chemiluminescence (ECL) detection for DNA hybridization using sandwich DNA detection model and the ECL probes of carbon-nanotubes loaded with probe ss-DNA and Ru 1 tags. ECL measurement was performed at a constant potential of +1.30V in 2.0mL of 0.10M PBS (pH 7.4) containing 0.10M TPA (Adapted from reference [45])

in a back-to-back Schottky diode configuration. Such nanostructures modified with thiol-terminated probe DNA fragments could function as sensors for biomolecules at very low concentration. The gold segment is utilized for functionalization and binding of single strand DNA (ssDNA) fragments while the CdTe segments at both ends serve to modulate the equilibrium Fermi level of the heterojunction device upon hybridization of the complementary DNA fragments (cDNA) to the ssDNA over the Au segment. Employing such multisegment nanowires could lead to the fabrication of more sophisticated and high multispecificity biosensors via selective functionalization of individual segments for biowarfare sensing and medical diagnostics applications (Fig. 7.6I).

Arrayed gold nanowires represent a novel and useful platform for electrochemical DNA detection. One factor contributing to the heightened sensitivity is the high signal-to-noise ratio achieved with the large electrocatalytic signals observed at DNA-modified nanowire (Fig. 7.6II). Lapierre-Devlin et al. [45] explain the improved sensitivity of the DNA detection with the fact that the electrocatalysis of nanowire membrane at DNA-modified nanostructures generates amplified signals that are significantly larger than those observed at bulk gold surfaces. The results strongly suggest that the three-dimensional architectures of the nanowires facilitate the electrocatalytic reduction of Ru(III) because of enhanced diffusion occurring around these structures.

Zhu et al. reported a novel and sensitive electrochemical DNA biosensor based on electrochemically fabricated polyaniline nanowire and methylene blue for DNA hybridization detection [46]. Nanowires of conducting polymers were directly synthesized through a three-step electrochemical deposition procedure in an aniline-containing electrolyte solution, by using the glassy carbon electrode (GCE) as the working electrode (Fig. 7.6III). The diameters of the nanowires range from 80 to 100 nm. The polyaniline nanowires-coated electrode exhibited very good electrochemical conductivity. Oligonucleotides with phosphate groups at the 5' end were covalently linked onto the amino groups of polyaniline nanowires on the electrode. The hybridization events were monitored with differential pulse

Fig. 7.6 (**I**) (*a*) An individual CdTe-Au-CdTe multisegment nanowire, fabricated by electrochemistry, is functionalized on the Au segment with thiol-terminated molecules. (*b*) Fabricated modified nanowire FET device using lithography. (*c*) Immerse the nanowire into the solution containing different biospecies. Specific DNA targets are self-assembled to the receptors at the nanowire surface (*d*) Immerse the nanowire into the solution containing different DNA strands. (*e*) Specific DNA strand are self-assembled to the receptors at the nanowire surface. (*f*) Sensitive conductance responses was obtained after bound with targets (Adapted from reference [46]). (**II**) Schematic illustration of Ru(III)/Fe(III) electrocatalysis at a DNA-modified Au NEE (Adapted from reference [47]). (**III**) Schematic representation of the DNA-sensors. First step the electrochemical deposition of polyaniline nanowire layer; the hybridization of DNA probe with DNA target is performed. The metaline Blue was added (to hybridization complex and the DNA probe step) to carry out the electrochemical detection, obtaining the electrochemical signal when the hybridization reaction was performed (Adapted from reference [48])



voltammetry (DPV) measurement using methylene blue (MB) as an indicator. The approach described here can effectively discriminate complementary from noncomplementary DNA sequence, with a detection limit of 1.0×10^{-12} mol 1^{-1} of complementary target, suggesting that the polyaniline nanowires hold great promises for sensitive electrochemical biosensor applications.

7.4 Conclusions and Future Perspectives

The integration of nanotechnology with biology and electrochemistry is expected to produce major advances in the field of electrochemical sensors. Recent progress has led to the development of functional nanoparticles that are covalently linked to biological molecules, such as peptides, proteins and nucleic acids. To be biologically relevant, nanoparticles need to have surface functionality amenable to biological modification, and long-term stability in a range of buffered saline solutions and pH values, and limited non-specific binding.

The future direction of electrochemical DNA sensors is focused on the development of point-of-care systems, which seek to integrate sample handling, fluidic processing, and detection on a portable platform. Although electrochemical biosensors have been widely developed for laboratory-based detection within the past several years, there are very few successful point-of-care devices for clinical diagnostics that are currently commercialized (i.e. glucometers).

Biosensor technology in clinical testing offers several potential advantages over other clinical analysis, including high analytical sensitivity and specificity, increased assay speed and flexibility, automation, capability for multi-target analyses, reduced costs of diagnostic testing and the potential to bring health care delivery closer to the community. Electrochemical devices have traditionally received the major share of the attention in biosensor development. Thick film technology using a screenprinted procedure is a widely used method for the simple and fast mass-production of inexpensive disposable electrochemical sensors.

Clinical applications for electrochemical DNA sensors are still far from reality due to several important issues. Although much work has been done to improve the performance of electrochemical DNA sensors, the sensitivity/specificity is still a key issue. Specifically, the detection of clinical samples requires high sensitivity/ specificity as well as high repeatability/reliability, which is still an unresolved problem. To address these issues, new nanomaterials with effective and stable performance are required along with higher stringency control during manufacturing. Additionally, the accuracy for clinical detection can be enhanced though bio-statistic support based on multiple DNA biomarkers. To improve the application of electrochemical sensors for real clinical tests, a simple detection process is desired, which incorporates automatic sample processing or in situ detection. This can be achieved through using micro/nanotechnologies, which offers new materials and sensor fabrication processes. Furthermore, the safety of nanomaterials is becoming a significant issue, especially as applications of these materials become more widespread.

Several other electrochemical based sensing technologies reported for proteins [47–51] detection using nanoparticles and other nanomaterials can also be applied in the future for DNA sensing. Catalytic methods, applied for example for a model protein (Human IgG) detection based on hydrogen evolution [52] or even nanochannels [53, 54]could be extended in the future in DNA sensing.

Acknowledgments We acknowledge funding from MICINN (Madrid) for the projects MAT2008-03079 and MAT2011-25870.

References

- 1. Zarbin, M.A., Montemagno, C., Leary, J.F., Ritch, R.: Nanomedicine in ophthalmology: the new frontier. Am. J. Ophthalmol. **150**, 144–162 (2010)
- Wei, F., Lillehoj, P.B., Ho, C.: DNA diagnostics: nanotechnology-enhanced electrochemical detection of nucleic acids. Pediatr. Res. 67, 458–468 (2010)
- Nie, S., Xing, Y., Kim, G.J., Simons, J.W.: Nanotechnology applications in cancer. Annu. Rev. Biomed. Eng. 9, 257–288 (2007)
- Liepold, P., Wieder, H., Hillebrandt, H., Friebel, A., Hartwich, G.: DNA-arrays with electrical detection: a label-free low cost technology for routine use in life sciences and diagnostics. Bioelectrochemistry 67, 143–150 (2005)
- Lucarelli, F., Capponcelli, S., Marrazza, G., Sangiorgi, L., Mascini, M.: Split hybridisation probes for electrochemical typing of single-nucleotide polymorphisms. Analyst 134, 52–59 (2009)
- Pohlmann, C., Wang, Y.R., Humenik, M., Heidenreich, B., Gareis, M., Sprinzl, M.: Rapid, specific and sensitive electrochemical detection of foodborne bacteria. Biosens. Bioelectron. 24, 2766–2771 (2009)
- Wakai, J., Takagi, A., Nakayama, M., Miya, T., Miyahara, T., Iwanaga, T., Takenaka, S., Ikeda, Y., Amano, M.: A novel method of identifying genetic mutations using an electrochemical DNA array. Nucleic Acids Res. 32, e141 (2004)
- Zhang, J., Song, S.P., Zhang, L.Y., Wang, L.H., Wu, H.P., Pan, D., Fan, C.: Sequence specific detection of femtomolar DNA via a chronocoulometric DNA sensor (CDS): effects of nanoparticle-mediated amplification and nanoscale control of DNA assembly at electrodes. J. Am. Chem. Soc. **128**, 8575–8580 (2006)
- Lord, H., Kelley, S.O.: Nanomaterials for ultrasensitive electrochemical nucleic acids biosensing. J. Mater. Chem. 19, 3127–3134 (2009)
- Radwan, S.H., Azzazy, H.M.: Gold nanoparticles for molecular diagnostics. Expert Rev. Mol. Diagn. 9, 511–524 (2009)
- Wei, F., Liao, W., Xu, Z., Yang, Y., Wong, D.T., Ho, C.M.: Bio/abiotic interface constructed from nanoscale DNA dendrimer and conducting polymer for ultrasensitive biomolecular diagnosis. Small 5, 1784–1790 (2009)
- Pandey, P., Datta, M., Malhotra, B.D.: Prospects of nanomaterials in biosensors. Anal. Lett. 41, 159–209 (2008)
- Kerman, K., Morita, Y., Takamura, Y., Ozsoz, M., Tamiya, E.: Modification of Escherichia coli single-stranded DNA binding protein with gold nanoparticles for electrochemical detection of DNA hybridization. Anal. Chim. Acta 510, 169–174 (2004)
- Liao, W.C., Ho, J.A.: Attomole DNA electrochemical sensor for the detection of Escherichia coli O157. Anal. Chem. 81, 2470–2476 (2009)
- Ozsoz, M., Erdem, A., Kerman, K., Ozkan, D., Tugrul, B., Topcuoglu, N., Ekren, H., Taylan, M.: Electrochemical genosensor based on colloidal gold nanoparticles for the detection of Factor V Leiden mutation using disposable pencil graphite electrodes. Anal. Chem. 75, 2181–2187 (2003)
- Castañeda, M.T., Merkoçi, A., Pumera, M., Alegret, S.: Electrochemical genosensors for biomedical applications based on gold nanoparticles. Biosens. Bioelectron. 22, 1961–1967 (2007)
- Mao, X., Liu, G.D.: Nanomaterial based electrochemical DNA biosensors and bioassays. J. Biomed. Nanotechnol. 4, 419–431 (2008)
- Wang, F., Hu, S.S.: Electrochemical sensors based on metal and semiconductor nanoparticles. Microchim. Acta 165, 1–22 (2009)
- Xu, K., Huang, J.R., Ye, Z.Z., Ying, Y.B., Li, Y.B.: Recent development of nanomaterials used in DNA biosensors. Sensors 9, 5534–5557 (2009)

- Pumera, M., Castañeda, M.T., Pividori, M.I., Merkoçi, A., Alegret, S., Eritja, R.: Magnetically trigged direct electrochemical detection of DNA hybridization using Au67 quantum Dot as electrical tracer. Langmuir 21, 9625–9629 (2005)
- Wang, J., Xu, D., Kawde, A., Polsky, R.: Metal nanoparticle-based electrochemical stripping potentiometric detection of DNA hybridization. Anal. Chem. 73, 5576–5581 (2001)
- Wang, J., Polsky, R., Xu, D.: Silver-enhanced colloidal gold electrochemical stripping detection of DNA hybridization. Langmuir 17, 5739–5741 (2001)
- Wang, J., Xu, D., Polsky, R.: Magnetically-induced solid-state electrochemical detection of DNA hybridization. J. Am. Chem. Soc. 124, 4208–4209 (2002)
- Park, S., Taton, T.A., Mirkin, C.A.: Array-based electrical detection of DNA with nanoparticle probes. Science 295, 1503–1506 (2002)
- Wang, J., Liu, G., Merkoçi, A.: Electrochemical coding technology for simultaneous detection of multiple DNA targets. J. Am. Chem. Soc. 125, 3214–3215 (2003)
- Zhu, N., Zhang, A., Hea, P., Fang, Y.: Cadmium sulfide nanocluster-based electrochemical stripping detection of DNA hybridization. Analyst 128, 260–264 (2003)
- Wang, J., Liu, G., Polsky, R., Merkoçi, A.: Electrochemical stripping detection of DNA hybridization based on cadmium sulfide nanoparticle tags. Electrochem. Commun. 4, 722–726 (2002)
- Marin, S., Merkoçi, A.: Direct electrochemical stripping detection of cystic-fibrosis-related DNA linked through cadmium sulfide quantum dots. Nanotechnology 20(1–6), 055101 (2009)
- 29. Fu, X.: Electrochemical measurement of DNA hybridization using nanosilver as label and horseradish peroxidase as enhancer. Bioprocess Biosyst. Eng. **31**, 69–73 (2008)
- Cai, H., Xu, Y., Zhu, N., He, P., Fang, Y.: An electrochemical DNA hybridization detection assay based on a silver nanoparticle label. Analyst 127, 803–808 (2002)
- Wang, J., Liu, G., Merkoçi, A.: Particle-based detection of DNA hybridization using electrochemical stripping measurements of an iron tracer. Anal. Chim. Acta 482, 149–155 (2003)
- 32. Iijima, S.: Helical microtubules of graphitic carbon. Nature 354, 56–58 (1991)
- Trojanowicz, M.: Analytical applications of carbon nanotubes: a review. Trac Trends Anal. Chem. 25, 480–489 (2006)
- 34. Tang, T., Peng, T.Z., Shi, Q.C.: Sequence determination of DNA pieces using carbon nanotube modified gold electrodes. Acta Chim. Sin. 63, 2042–2046 (2005)
- Abdullin, T.I., Nikitina, I.I., Ishmukhametova, D., Budnikov, G.K., Konovalova, O.A., Salakhov, M.K.: Carbon nanotube-modified electrodes for electrochemical DNA-sensors. J. Anal. Chem. 62, 599–603 (2007)
- 36. Kerman, K., Morita, Y., Takamura, Y., Tamiya, E.: Escherichia coli single-strand binding protein-DNA interactions on carbon nanotube-modified electrodes from a label-free electrochemical hybridization sensor. Anal. Bioanal. Chem. 381, 1114–1121 (2005)
- Erdem, A., Papakonstantinou, P., Murphy, H.: Direct DNA hybridization at disposable graphite electrodes modified with carbon nanotubes. Anal. Chem. 78, 6656–6659 (2006)
- Katz, E., Willner, I.: Biomolecule-functionalized carbon nanotubes: applications in nanobioelectronics. Chem. Phys. Chem. 5, 1084–1104 (2004)
- Zhang, R.Y., Wang, X.M., Chen, C.: Electrochemical biosensing platform using carbon nanotube activated glassy carbon electrode. Electroanalysis 19, 1623–1627 (2007)
- Bollo, S., Ferreyra, N.F., Rivas, G.A.: Electrooxidation of DNA at glassy carbon electrodes modified with multiwall carbon nanotubes dispersed in chitosan. Electroanalysis 19, 833–840 (2007)
- Wang, J., Liu, G., Jan, M.R., Zhu, Q.: Electrochemical detection of DNA hybridization based on carbon-nanotubes loaded with CdS tags. Electrochem. Commun. 5, 1000–1004 (2003)
- 42. Wang, J., Liu, G.D., Jan, M.R.: Ultrasensitive electrical biosensing of proteins and DNA: carbon nanotube derived amplification of the recognition and transduction events. J. Am. Chem. Soc. **126**, 3010–3011 (2004)
- 43. Li, Y., Qi, H.L., Fang, F., Zhang, C.X.: Ultrasensitive electrogenerated chemiluminescence detection of DNA hybridization using carbon-nanotubes loaded with tris(2,2'-bipyridyl) ruthenium derivative tags. Talanta **72**, 1704–1709 (2007)

- Wang, X., Ozkan, C.S.: Multisegment nanowire sensors for the detection of DNA molecules. Nano Lett. 8, 398–404 (2008)
- Lapierre-Devlin, M.A., Asher, C.L., Taft, B.J., Gasparac, R., Roberts, M.A., Kelley, S.O.: Amplified electrocatalysis at DNA-modified nanowires. Nano Lett. 5, 1051–1055 (2005)
- Zhu, N., Chang, Z., Heb, P., Fang, Y.: Electrochemically fabricated polyaniline nanowiremodified electrode for voltammetric detection of DNA hybridization. Electrochim. Acta 51, 3758–3762 (2006)
- Escosura-Muñiz, A., Maltez, M., Merkoçi, A.: Controlling the electrochemical deposition of silver onto gold nanoparticles: reducing interferences and increasing the sensitivity of magnetoimmuno assays. Biosens. Bioelectron. 24, 2475–2482 (2009)
- Escosura-Muñiz, A., Ambrosi, A., Merkoçi, A.: Electrochemical analysis with nanoparticlebased biosystems. Trac Trends Anal. Chem. 27, 568–584 (2008)
- Ambrosi, A., Castañeda, M.T., Killard, A.J., Smyth, M.R., Alegret, S., Merkoçi, A.: Doublecodified gold nanolabels for enhanced immunoanalysis. Anal. Chem. 79, 5232–5240 (2007)
- Escosura-Muñiz, A., Parolo, C., Merkoçi, A.: Immunosensing using nanoparticles. Mater. Today 13, 17–27 (2010)
- 51. Merkoçi, A. (ed.): Biosensing Using Nanomaterials. Wiley-Interscience, Hoboken (2009)
- Maltez, M., Escosura-Muñiz, A., Merkoçi, A.: Electrochemical quantification of gold nanoparticles based on their catalytic properties toward hydrogen formation: application in magnetoimmunoassays. Electrochem. Commun. 12, 1501–1504 (2010)
- Escosura-Muñiz, A., Merkoçi, A.: Label-free voltammetric immunosensor using a nanoporous membrane based platform. Electrochem. Commun. 12, 859–863 (2010)
- 54. Escosura-Muñiz, A., Merkoçi, A.: A nanochannel/nanoparticle-based filtering and sensing platform for direct detection of a cancer biomarker in blood. Small **7**, 675–682 (2011)

Chapter 8 Piezoelectric Sensing for Sensitive Detection of DNA

Maria Minunni

Abstract Piezoelectric sensing has been widely applied for affinity sensing, and recently sensitive DNA detection has been reported in different matrices for different analytes (i.e. target sequences). In this chapter, the detection principle and the approaches used in DNA-based sensing with focus on detection of microsatellite DNA, present in high number of copy as well as target sequence detection of genes present in one or few copy number per haploid genome will be presented and discussed. Particular attention will be devoted to the pre-analytical steps which may influence the sensor response to the target analyte such as genomic DNA fragmentation and denaturation. Comparison between immobilization chemistries is also presented. In particular, finding in microsatellite detection with both biotinylated and thiolated probes is reported and discussed.

8.1 Introduction

The utilization of piezoelectric quartz crystal oscillators as microbalances (QCM) has been applied in Analytical Chemistry, both in gas and in liquid phases. In literature there is a wide number of papers based on commercially available or in house developed devices, using crystals ranging from 5, up to 10, 30 MHz, using in QCM in the fundamental frequency or in the relative overtones. Applications of QCM's as thin film monitors and controls; in surface science, plasma-assisted etching, analytical chemistry, and space system contamination and for aerosol mass measurements have been reported over the last 20 years. Details of their methodology are scattered widely throughout the literature.

M. Minunni (🖂)

Dipartimento di Chimica "Ugo Schiff", Università di Firenze, Via della Lastruccia 3, 50019 Sesto F.no (FI) Florence, Italy e-mail: maria.minunni@unifi.it

G. Spoto and R. Corradini (eds.), *Detection of Non-Amplified Genomic DNA*, Soft and Biological Matter, DOI 10.1007/978-94-007-1226-3_8, © Springer Science+Business Media Dordrecht 2012

8.1.1 Piezoelectric Biosensors: Theory and Applications of Piezoelectric Effect

Piezoelectric, similarly to Surface Plasmon Resonance (SPR) transduction methods is widely used in biosensors development, mainly because it allows the detection of label-free targets and the kinetic measurement of molecular interactions in real-time. Moreover, piezoelectric crystals as SPR chips can be regenerated, allowing a multiuse of the sensor. Piezoelectric crystals are acoustic transducers generally utilized as microbalance or microviscosimeters. In particular, piezoelectric quartz crystals are the basic elements of quartz crystal microbalance (QCM) device. Quartz resonators are the most used crystals, a crystal variant of Silicon-dioxide (SiO₂).

However, independently from the material used, the term "piezoelectric" derived from the Greek word piezen meaning "to press". The first investigation on the piezoelectricity was performed in 1880 by Jacques and Pierre Curie [1], who observed that a mechanical stress applied to the surfaces of various kinds of crystal caused a corresponding electrical potential across the crystal, whose magnitude was proportional to the applied stress. The Curies also verified the converse piezoelectric effect, in which application of a voltage across these crystals caused a corresponding mechanical strain. This causes a vibrational, or oscillatory, motion in the crystal, resulting in the generation of acoustic standing waves at a characteristic resonant frequency. The wave is called bulk acoustic wave (BAW) or surface acoustic wave (SAW) in the case of propagation through the substrate or on the surface, respectively. A few naturally abundant crystals (i.e. quartz, tourmaline and Rochelle salt) are piezoelectric, but many other materials exhibit this effect, including quartz analogue crystals, such as berlinite (AlPO₄), ceramics with perovskite or tungsten-bronze structures (BaTiO₃, KNbO₃, LiNbO₃, LiTaO₃, BiFeO₃, NaxWO₃, Ba₂NaNb₅O₅, Pb₂KNb₅O₁₅). However, recently piezoelectric crystals based on gallium orthophosphate (GaPO₄) have been reported for different application, including biosensing [2, 3]. GaPO₄ production has been, first set up by AVL List GmbH, Graz Austria, for applications as pressure monitoring in motors combustion engines, since the thermal coefficient of this material is compatible with the high temperature reached in car engines. Characteristics of this material are, in particular, high sensitivity, stability up to 970 °C, no pyroelectricity, no stress induced twinning, high electric resistivity >1,015 Ω cm. For all these interesting properties different applications have been identified and investigated also within an European project entitled "Growth of Large GaPO₄ Single Crystals and their use for Special Sensor Applications", (contract n° G5RD-CT-2002-00709, coordinated by Prof. Dr. Peter Kremple at AVL), who had foreseen in particular some Bulk Wave Applications such as high shear GaPO₄ crystal rheometer, GaPO₄ crystal particulate microbalance (GCPM) and Nanobalance Affinity Sensor for DNA-based sensing (i.e. GMO) and drug detection. Within this project sensitive DNA detection was studied using conventional quartz crystals and eventually GaPO₄.

However, only quartz provides the unique combination of mechanical, electrical, chemical and thermal properties, which allowed its commercial success. Direct

and converse piezoelectricity has a wide range of applications. Piezoelectric quartz crystals are used in quartz watches, computers and in many high-performance devices (such as Atomic Force Microscopy) to apply tiny mechanical displacements on the scale of nanometers. Furthermore, such crystals are employed as quartz crystal microbalance for thickness monitoring.

The quartz crystal microbalance is a bulk acoustic wave device based on the converse piezoelectric effect, in which a quartz crystal is sandwiched between two electrodes. The resonant frequency of the quartz crystal depends on several parameters, such as size, density and cut. The most used devices employ AT-cut quartz crystals, sliced with an angle of $35^{\circ}10'$ respect to the optical z-axis (Fig. 8.1a). AT-cut quartz crystals show a high frequency stability and a temperature coefficient close to zero between 0 and 50 °C [4]. AT-cut crystals oscillate in the thickness shear mode (TSM) [5].

The application of a voltage between the two electrodes causes a shear deformation of the crystal, which is maximized at the crystal faces, making the device sensitive to surface interactions. The resonant condition with the acoustic wave is satisfied by including the crystal into an oscillation circuit, where the frequency of the alternating potential difference applied to the electrodes matches the fundamental frequency of the crystal. The fundamental frequency depends upon the thickness of the wafer, its chemical structure, its shape and its mass [6]. Since the oscillation frequency depends on the crystal mass, deposition of thin films on the crystal surface increases the resonator thickness and decreases the frequency in proportion to the film mass. Measurements of the crystal frequency allow the detection of the film mass, therefore the device operates like a 'microbalance'. The first quantitative investigation of the piezoelectric effect was performed by Sauerbrey [7], who derived the relationship for the change in frequency ΔF (in Hz) caused by the added mass Δm (in g) in vacuum or in air:

$$\Delta F = -\frac{2F_0^2}{A\sqrt{\mu_0\rho_0}} \cdot \Delta m \tag{8.1}$$

where F_0 is the fundamental resonant frequency of unloaded quartz, μ_Q is the shear modulus of AT-cut quartz (2.947 × 1,011 g cm⁻¹ s⁻²), ρ_Q is the density of the quartz (2.648 g cm⁻³) and A is the surface area in cm². The Sauerbrey equation assumes a uniform distribution of mass on the entire electrode portion of an AT-cut quartz crystal. Mass sensitivity decreases monotonically with the radius, in a Gaussian manner becoming negligible at and beyond the electrode boundary [8]. Another assumption of this equation is that the mass added or lost at the crystal surface does not experience any deformation during the oscillation: this is true for thin, rigid layers in vacuum or in air. For thicker, less rigid layers, as it happens for quartz crystals operating in liquid, a more complex theory is necessary. Many factors such as density, viscosity, conductivity and dielectric constant of the liquid may influence the oscillating behavior. When a quartz crystal oscillates in contact with a liquid, a shear motion on the surface generates motion in the liquid near the interface.



Fig. 8.1 (a) The cut-angle with respect to crystal orientation (so-called AT cut) determines the mode of induced mechanical vibration. AT-cut quartz crystals with a cut angle of 35° 10' with respect to the optical z-axis perform shear displacements perpendicular to the resonator surface [Wegener website]; (b) instrumentation: Frequency meter with the oscillator; methacrylate cell where the crystal is housed and scheme of the sensor

The resonant frequency change of a quartz crystal having one face in contact with liquid is described by the Kanazawa and Gordon equation [9]:

$$\Delta F = -F_0^{3/2} \left(\rho_L \eta_L / \pi \mu_Q \rho_L \right)^{1/2}$$
(8.2)

where ρ_L is the density of the liquid and η_L is the viscosity of the liquid. Piezoelectric crystals have been used as microbalances and as a microviscometer owing to their small size, high sensitivity, simplicity of construction and operation, low cost, lightweight and the low power required [10]. The quartz-crystal microbalance has

traditionally been used in many applications such as thin film deposition control, etching studies, aerosol mass measurements and space system contamination studies. Recently however, the interest in the application of piezoelectric devices in the field of analysis has increased, since it was realized that many opportunities for molecular sensing can be opened up once a suitable recognition layer or molecule is coated on the crystal [11]. Therefore, immobilizing an affinity ligand on the surface of a quartz crystal covered with gold or silver, an affinity mass sensor is realized. When the affinity reaction with the target in solution takes place, the binding ligand can be determined at nanogram levels from the piezoelectric frequency shift. Such piezoelectric biosensors have found a wide range of applications in food [12, 13], environmental [14] and clinical [15, 16] analysis. A typical instrumentation used is displayed in Fig. 8.1b.

8.1.2 Nucleic Acid–Based Piezoelectric Sensing

Among different Affinity Based Biosensors (ABBs), nucleic acid sensing has appeared to be one of the most explored. When developing nucleic acid based approaches, a capturing sequence (probe), responsible for the system selectivity has to be immobilized on the sensing surface. The sequence, complementary to the probe added in solution, binds the surface with consequent frequency decrease, due to surface hybridization reaction and complex (dsDNA) formation. In Fig. 8.2 a typical hybridization signal is displayed. The recorded analytical signal is the frequency shift between the frequency value before sample injection, taken in buffer (base line) and its value after sample injection followed by washing with buffer to remove the sample excess and unbound material, leaving on the surface just hybridized sequences. Ideally, the frequency shift should be different from 0 if the analyte, i.e. the target sequence, is present in the sample and 0 if not. For checking the system selectivity negative control (non complementary sequence) should be also injected on the surface and the recorded should be 0. Negative control should be always used to control eventually present unspecific binding i.e. adsorption, phenomena at the surface. This is extremely important for real analytical system application.

Generally, the complementary sequence to the target analyte (a gene, a fragment or a short oligonucleotide) is first considered during the choice of probe. The probe length is generally set ranging from 15 to 20 bases, with one end usually linked to a functional group to be exploited for the immobilization chemistry. For example, biotinylated probes are used in surface functionalization involving streptavidin; thiolated probes [17], are required for direct probe coupling to gold surfaces via Self Assembled Monolayer (SAM) formation [18]. In Fig. 8.3 is reported a scheme of the relative immobilization protocols.

Some immobilization approaches are summarized in Tombelli et al. [19, 20]. Another criterion taken into account in probe selection is the C-G base content (three hydrogen bonds vs. two with A-T pairing); preferred composition varies from at least 40 to 60 % to stabilize the hybrid on the surface [21]. Finally, to facilitate



Fig. 8.2 Scheme of the signal recorded by piezolelectric sensing during a measurement cycle (see text). The hybridization reaction between the immobilized probe and the target sequence in solution is displayed

surface hybridization, it is important to avoid probe hybridization on regions that can assume conformations and may obscure the binding site of interest, i.e. by formation of secondary structures such as hairpins or loops.

Recently it has been eventually reported by our group, an optimized and reproducible strategy for probe design for nucleic acid-based sensing based on a free available software computational assisted approach. The *in silico* selection, was validated by experiments conducted using optical transduction for DNA-sensing development, in particular Surface Plasmon Resonance imaging (SPRi), demonstrating that "smart" probe design for DNA-sensing significantly improves the sensor's analytical performances for DNA-DNA hybridization measurements. This approach can be easily transferred to piezoelectric transduction [22].

Behind the probe selection, the immobilization chemistry is strategic for the success of the nucleic acid sensor development. Many immobilization chemistries, using different approaches, have been developed, depending from the sensor surface. Among the available methods, we mainly focused on thiols based approaches, using thiolated probes and dextran modified chip following the chemistry used in the gold chip of the SPR-based Biacore family instrumentation. During the sensor developments the system is first studied using synthetic oligonucleotides to evaluate the system's main analytical parameters such as sensitivity, selectivity, reproducibility expressed as coefficient of variation (CV%), detection limit (DL),


Fig. 8.3 Immobilization chemistry using (a) thiol/dextran/streptavidin modified surfaces for biotinylated probe binding; (b) direct coupling of thiolated probe with further surface passivation with short thiols

analysis time, etc. Once the assay conditions have been optimized one can move to real sample analysis. For real analysis thus matrix effects have to be evaluated and pre-analytical steps, i.e. sample-pre-treatment, should also be considered. In nucleic acid sensing, the sample pre-treatment is a key step. DNA can be detected in different cellular compartments (nucleic, mitochondrial, etc.) but it should be extracted from the cell. Double stranded DNA is the form in which the sample is obtained after its extraction.

8.2 Development of Piezoelectric Sensing for DNA-Based Sensitive Detection

The extracted DNA can be then purified by precipitation and different widely used protocols are available [23]. Re-suspension of DNA can be achieved in buffer or distilled water. In any case the extracted DNA is a double helix, with the target sequence hybridizing the probe immobilized on the sensor surface, hidden in the double helix. Thus it is clear that the dsDNA should be opened to allow surface hybridization of the target sequence.

8.2.1 Sample Pre-treatment for Target Sequence Analysis

8.2.1.1 DNA Fragmentation

Fragmentation of DNA is achieved using ultrasound generating random fragments or by enzymatic digestion by nucleases that breaks the helix at specific sites, different for the various nucleases, able to originate fragments eventually containing the intact target sequence. In literature related to genomic detection of sequences by DNA-based sensing using ultrasound has been successfully reported [24, 25]. Alternatively fragmentation can be achieved by restriction enzymes, proteins that recognize specific, short oligonucleotides sequences (the "restriction enzyme cutting sites") and cut DNA at those sites and nowhere else [26]. Bacteria contain over 400 such enzymes that cut over 100 different DNA sequences. The restriction enzymes cleave DNA in a very specific fashion. Type II restriction enzymes, most commonly used for DNA analysis and genetic engineering, have a unique nucleotide sequence at which they cut a DNA molecule. The recognition sequence is often a six base pair palindromic sequence (the top DNA strand from 5' to 3' is the same as the bottom DNA strand from 5' to 3'), but some recognize four or even eight base pair sequence. Some sites occur frequently in DNA (e.g. every several hundred of base pairs), other much less frequently (rare-cutter; e.g., every 10,000 base pairs).

There are two kinds of restriction enzymes: restriction "endonuclease" and restriction "exonuclease". Nuclease is the general term for enzymes that catalyse the hydrolysis of nucleic acids by clearing chains of nucleotides into smaller units. The endonuclease is a nuclease that claves nucleic acid at interior bonds and so produces fragments of various size; while exonuclease is a nuclease that releases one nucleotide at a time (serially) starting from a nucleic acid termination. Other restriction enzymes can be used to further characterize a particular DNA molecule. The location of these restriction enzyme cleavage sites on the DNA molecule can be compiled to create a "restriction enzyme map". These maps are very useful for identifying and characterizing a particular DNA plasmid or region. Restriction fragment length polymorphism map (RFLP) are widely employed in clinical diagnostic for point mutation detection as well as in food analysis (see below). The selection of the enzyme to be applied in genomic DNA digestion is performed by the analysis with free available software.

8.2.1.2 Denaturation of dsDNA

In this section we will discuss about the importance of the sample pre-treatment in terms of dissociating the two strands of the DNA double helix (dsDNA) using different strategies. We strongly believe that for sensitive DNA detection, such as direct sequence analysis in unamplified genomic DNA, the dissociation step is of key importance. For this reason we spent quite a lot of time in studying and evaluating different approaches, starting from the most popular denaturing thermal treatment up to the addition of suitable chemicals or oligonucleotides to the sample to prevent strand re-annealing. Thermal treatment is very simple and for this reason is widely employed, consisting in heating up the dsDNA to 90 °C for 5 min with further cooling and injection into the instrumentation. Many examples demonstrated the success of this procedure, which is particularly suited with amplified DNA material, i.e. by PCR. This procedure, on the contrary, ends in very low reproducible results, not applicable analytically, when target sequence sensitive detection has to be achieved, as in the case of single copy sequences per haploid genomic DNA in systems without any amplification [25]. Thermal denaturation can be eventually applied to satellite DNA, significantly (i.e. in many copies) present in haploid genome [27], as we will discuss later on in this chapter. We had this feeling already with optical detection using Biacore family instrumentation (BIAcore XTM) where we demonstrated that in this system re-annealing of the injected sequence occurs with important reflection on the system analytical performances [28]. To be clearer, in the case of this instrumentation or eventually in flow mode systems, we think re-annealing of the strands can occur, preventing hybridization from the analyte (target sequence) with the immobilized probe. As consequence reduced amount of ssDNA analyte are available for hybridization. This reduced amount may became significant and critic in case of genomic DNA, where only few copies of target are available for probe hybridization. If significant target depletion by re-annealing occurs, no significant analytical signal may be registered, unless when signal amplification systems i.e. by gold nanoparticles are used [25].On the base of our initial findings with sample thermal denaturation, we then focused on dedicated sample pre-treatments optimization for improving sensitive DNA sensing performances, eventually starting with amplified DNA to go further into genomic DNA target sequence detection. We initially studied different approaches both at the PCR and post-PCR level aiming to improve sensor signals and reproducibility for further application to sensitive DNA sensing. The developed approaches have a general validity, independently of the transduction principle used. We focused our approaches in particular on two sequences dealing with detection of transgenosis markers contained in the promoter region (P35S) of the cauliflower mosaic virus (CAMV) ribosomal RNA and TNOS terminator, indicated in the official protocols as the target analyte for genetically modified organisms (GMO) detection [29]. The rational we used in these approaches was to evaluate the possibility either to isolate single stranded target DNA or to prolong the half-life of the ssDNA in the denaturation step, by preventing re-annealing. ssDNA can be obtained using magnetic particles for capturing and precipitation or by enzymatic digestion [13]. The approaches used to obtain selective extraction of the single strand, containing the target sequence, are based on the use of proper primers, modified by biotinylation or phosphorylation. In particular streptavidin modified particles were used to bound biotinylated DNA, while one phosphorylated strand was obtained in the amplified DNA, using phosphorylated primers. The modified ssDNA was selectively recognized and digested by a lambda exonuclease. Other researchers, as well as our group, have employed asymmetric PCR with optical sensing to obtain significant amount ssDNA target in solution as with asymmetric amplification (i.e. SPR, Biacore family instrumentation), by suitable ration of reverse and forward primers, an excess of ssDNA containing the target sequence can be synthetized, preventing re-annealing with the other strand of the dsDNA, now present in much lower amount [30-32]. We will not further comment these just mentioned approaches because they can be applied only to amplified DNA samples, since primers have to be employed. However the full understanding of encountered problems and the developed solutions, stresses the importance of the denaturing step in improving DNA-based sensing performances. Thus further strategies have to be introduced for sensitive sequence detection in genomic DNA. To this aim, behind this PCR-based mentioned approaches, we evaluated other denaturation methods assisted by the use of chemical reagents added to the DNA sample. In particular denaturation treatments can be performed by the combination of a strong alkaline environment and a formamide treatment at 42 °C. Denaturation of dsDNA in strong alkaline conditions is well established. At a pH 13 the charge of the DNA bases changes, thus preventing H-bond formation [33]. On the other hand, the influence of organic compounds such as formamide, urea and formaldehyde on the thermal DNA denaturation process has been well documented [23]. Formamide is a helix destabiliser that replaces the native DNA bases for inter-strand hydrogen bonds, thus inducing the denaturation of dsDNA [34]. Results were obtained again with amplified and genomic DNA [32], using different concentration of chemical, but still the applied treatment was found to be insufficient to generate reproducible and significant SPR sensor signals. These treatments however were eventually applied to piezoelectric DNA to evaluate the ability of sensitive genomic DNA detection [35]. Finally an innovative approach was developed, based on the use of small selective oligonucleotides added to the denaturing mixture, for hybridizing, at a selected annealing temperature, the two ssDNA of the helix. The oligos attachment site can be designed and should not overlap the ssDNA region involved in the surface probe binding. The hybridization of these small oligos prevents re-annealing of the two strands, leaving the target sequences to hybridize the immobilized probe, thus prolongating the single stranded state of the target sequence. For its re-annealing blocking behaviour, the approach was named "denaturation by blocking oligos".

In Fig. 8.4 is displayed a scheme of the developed approach and the differences with the conventional thermal denaturation. Initially Biacore XTM instrumentation was used [36] in the protocol optimization as proof of principle. Once demonstrated the suitability of the "blocking oligonucleotides approach" for improving sensitive DNA detection of three different sequences in genomic DNA, the latter developed denaturing approaches based on the addition of formamide at different percentages or of blocking oligos were applied to sensitive detection of target DNA present in single or few copies per haploid DNA by piezoelectric sensing [37].

We will here report about the finding obtained initially by our group and more recently appeared in literature relative to sequence detection in unamplified genomic DNA. We will first discuss the detection of specific micro satellite DNA detection with application to food analysis and further detection of sequences present in few copies per haploid genome, identified as marker of transgenosis.



Fig. 8.4 Denaturation of dsDNA (a) thermal denaturation; (b) thermal with blocking oligonucleotides

8.3 Detection of Microsatellite Sequences in Bovine Genomic DNA

First the detection of microsatellite sequences was achieved to demonstrate the ability of piezoelectric sensing to operate directly in genomic DNA. To approach the problem of direct sequence detection in unamplified genomic DNA, we first focused on target sequences highly repeated in genomes, which are species specific and, in some cases, could represent a significant part of the genome. In particular, detection of the highly repeated sequence called satellite 13, present in *Bos taurus* genome was achieved in genomic DNA, for the first time by piezoelectric sensing. In the bovine genome, eight highly repetitive and several minor repetitive sequences have been detected comprising a total of 27% of the DNA [38]. The target sequence chosen in this case is situated in highly repeated DNA and is therefore present in the genome in a high number of copies, increasing the amount of target sequence available for the hybridization with the probe [39] immobilized on the sensing element. The developed quartz crystal microbalance-based sensor has been applied to the identification of animal species in meat samples, which represents

assays and blocking oligonucleotide u	sed in the denaturation protocol
Biotinylated and thiolated probe:	Biotin or SH-5' TCACGCAGCTCAGCAGGCCCT 3'
Complementary target:	5' AGGGCCTGCTGAGCTGCGTGA 3'
Non-complementary sequence:	5' GGCAGAGGCATCTTCAACGATGGCC 3'
Blocking oligonucleotide 1:	5' GTCTGCTCATCTGCTTGACAATTTC 3'
Blocking oligonucleotide 2:	5' TAATCAAGTAGATGAGCAGGCAG 3

Table 8.1 Oligonucleotide sequences of immobilized probe, modified with biotin or SH in 5' end, complementary and non complementary sequences (negative control) used in the optimization assays and blocking oligonucleotide used in the denaturation protocol

a relevant problem in food analysis for economical, religious or public health concerning reasons. Numerous analytical methods, currently available for species differentiation, rely on protein analysis, such as electrophoresis techniques [40, 41], liquid chromatography [42, 43], and immunoassays [44].

Modern methods for meat identification are instead based on DNA analysis. They allow species-specific DNA sequences identification, which has some advantages over protein analysis [45]. Molecular biology methods allow the determination of DNA also in heat-treated nourishment and are, therefore, suitable for the identification of species-specific DNA in meat and bone meal and concentrate mixtures [46]. Furthermore, they allow the discrimination between related species, such as sheep and goat or chicken and turkey [47].

Earlier DNA sequence analysis was performed using genomic DNA as speciesspecific probe, which was hybridized to DNA extracted from meat samples [48–51]. Later, probes derived from highly repetitive (satellite) DNA sequences were developed [52].

The analysis of restriction fragment length polymorphism (RFLP) of PCR fragments is an alternative DNA detection system and it has already been successfully applied to species differentiation [53–56]. PCR-RFLP allows the amplification of a conserved region of DNA sequence using PCR, and the detection of the genetic variation between species by digestion of the amplified fragment with restriction enzymes. This technique was used for speciation by exploiting DNA sequence variations within the mitochondrial D-loop region [57], cytochromeB (cytB) gene [52] and satellite DNA sequences [58].

More recently, real-time PCR for meat species identification has also been reported [59].

DNA-based piezoelectric sensing was tested as alternative method in the identification of species-specific DNA sequences for meat analysis, based on determination of a highly repetitive and species-specific DNA sequence present in bovine (*B. taurus*) genomic (non-amplified). The sensor was developed by immobilizing the specific probe on the surface using thiol chemistry with thiolated probes, and surface saturation with short thiols, following the protocol previously developed [27]. Behind thiolated probes, dextran modified surfaces using biotinylated probes were also used. The base sequences of the 5-thiol functionalized probe (21-mer), complementary (21-mer) and non-complementary (25-mer) oligonucleotides and blocking oligonucleotides (25-mer and 23-mer) are reported in Table 8.1. The analytical parameters of the sensor are evaluated first by synthetic oligonucleotides,



Fig. 8.5 (a) Biotinylated oligonucleotide probe. Hybridisation with synthetic oligonucleotide target. (b) Thiolated oligonucleotide probe. Hybridisation with synthetic oligonucleotide target

complementary and non-complementary to the 21-mer immobilized probe; once assessed good analytical behavior in terms of selectivity, sensitivity and reproducibility, the system is then applied to real samples.

8.3.1 Sensor Optimization: Biotinylated and Thiolated Probes

A synthetic 21-mer biotinylated and thiolated probes, complementary to a sequence, which is present inside the *Bos taurus* satellite n.13 (247 bp), were used. The selectivity of the systems was tested with the non-complementary sequence. Comparison between the two immobilization chemistries was achieved to select the best performing sensor with standard solutions for further application to sequence detection in genomic, unamplified DNA.

The hybridization between the probe and 21-mer complementary oligonucleotide solutions were tested and the results are shown in Fig. 8.5. In case of sensor modified with biotinylated probe, the CV% has been calculated for all the concentrations and the average is 10 %. For the thiolated probe, the sensor performances are similar to the biotinylated one. The average CV% calculated is 10 % for both. The selectivity of the crystals surfaces just towards the target was confirmed by the absence of frequency shift with non-complementary sequence 1.8 ppm in both modified surfaces. Both sensor were regenerable. The surface was regenerated by treatment with HCl (10 mM, 30 s), which allowed to perform up to 20 hybridization cycles on a single surface, without affecting the sensor surface.



Fig. 8.6 Electrophoretical analysis of digestion reaction. (**a**) Commercially available genomic DNA (Novagen). Lanes: *1* marker 1-DNA-HindIII, *2* genomic bovine DNA, *3–4* digested genomic bovine DNA, *5* genomic porcine DNA, *6–7* digested genomic porcine DNA. (**b**) Real sample genomic DNA. Lanes: *1* marker 1-DNA-HindIII, *2* genomic bovine DNA, *3-4-5* digested genomic bovine DNA

In this case of *Bos taurus* micro satellite detection was achieved first in Genomic bovine (*B. taurus*); Porcine DNA (*Sus scrofa*) was used as negative control. DNA commercially available and finally the sensor were applied to real samples consisting in genomic bovine DNA extracted from animal muscle.

8.3.2 Sample Pre-treatment

8.3.2.1 DNA Fragmentation

To allow detection of target sequence in genomic DNA, fragmentation of the sample is necessary, achieved by enzymatic digestion. Enzymatic digestion was applied here to keep the target sequence in one piece, although this procedure is much longer then fragmentation by ultrasounds. The genomic DNA was digested using the restriction enzyme EcoRI (overnight), to obtain DNA fragments of around 400–500 bp containing the target sequence (available on GenBank accession number no. V00122).

It was verified that the consensus sequence recognized by the enzyme was not present inside the *B. taurus* satellite no. 13 (247 bp) to maintain integer the target sequence thus allowing the maximum amount of target sequence in solution hybridizing the probe. Checking the restriction site of the enzyme respect the target sequence is important to ensure that the fragmentation does not affect the ability of the target sequence to hybridize to the immobilized probe. The enzymatic fragmentation can be then confirmed by electrophoretical analysis, on agarose gel (1 % in TAE electrophoresis buffer) (Fig. 8.6).



Fig. 8.7 Genomic bovine DNA: thermal denaturation. (a) Biotinylated oligonucleotide probe. Signals: *I* Non-complementary oligonucleotide 8 ppm (n = 3), 2 Complementary oligonucleotide 2 ppm (n = 3), 3 Bovine genomic DNA 10 ppm (n = 3), 4 Bovine genomic DNA 20 ppm (n = 3), 5 Porcine genomic DNA 10 ppm (n = 3), 6 Blank solution (n = 1). (b) Thiolated oligonucleotide probe. Signals: *I* Non-complementary oligonucleotide 8 ppm (n = 3), 2 Complementary oligonucleotide probe. Signals: *I* Non-complementary oligonucleotide 8 ppm (n = 3), 2 Complementary oligonucleotide 1 ppm (n = 3), 3 Bovine genomic DNA 25 ppm (n = 1), 4 Bovine genomic DNA 10 ppm (n = 1), 5 Porcine genomic DNA 10 ppm (n = 1)

8.3.2.2 DNA Denaturation: Thermal Versus Thermal and Blocking Oligos

The genomic fragmented DNA, after precipitation in ethanol and re-suspension, underwent denaturation. Both simple thermal and thermal with blocking oligos were applied to the samples and results compared. In the case of thermal blocking oligo, after the addition of the two oligonucleotides (1 μ M), the sample was incubated at 95 °C for 5 min and then 1 min at 50 °C. This second temperature is the appropriate temperature for the annealing of the oligonucleotides to the complementary DNA sequences. The choice of the oligonucleotides depends on the length of the digested fragment and on the position of the target inside the fragment.

8.3.3 Analysis of Genomic Bovine DNA

8.3.3.1 Thermal Denaturation: Biotinylated Versus Thiolated Probe

Hybridization signals obtained after the interaction between both biotinylated and thiolated probes, and the commercially available genomic DNA samples previously treated with the restriction enzyme (EcoRI) and thermally denatured are next discussed.

Figure 8.7 shows the signals relative to the hybridisation of the probes with: noncomplementary and complementary oligonucleotide; bovine genomic DNA; porcine genomic DNA as negative control and blank solution, obtained by applying the digestion treatment and the thermal denaturation to a solution containing all the reagents but DNA [60].

Both sensors are capable to distinguish between the complementary sequence (bovine DNA) and non-complementary solutions (porcine DNA and blank solution). Hybridisation with a double concentration of bovine DNA reports a frequency shift nearly double. In the case of biotinylated surface, the regeneration was fully accomplished with HCl 1 mM.

On the contrary, in the case of thiolated probes, although it was possible to recognize the bovine from the porcine DNA, maybe there is aspecific adsorption that causes a significant signal of the non-complementary sequence. Furthermore, because of this adsorption, the regeneration solution (HCl 1 mM) optimized for oligonucleotides was not sufficient to regenerate the sensor surface and so improved regeneration approach, has to be tested. A different regeneration treatment was also employed. This treatment consists of a first addition of alkaline solution (15 s with NaOH 100 mM), which dissociated the two strands, followed by 30 s with regeneration solution (200 mmol/l Tris·Cl, pH 7, 0.1×SSC, 0.1 % (w/v) SDS). The alkaline pH separates the double strand DNA and then the surfactant, contained in the regeneration solution, removes the target sequence. Several surface washing steps with hybridization buffer were necessary to remove all the surfactant in the regeneration solution.

The surface performances were controlled, using standard solutions of synthetic oligonucleotides, before and after the hybridisation-regeneration cycles. However, after many regeneration cycles, only a partial surface regeneration was achieved, consequently the surface activity results lower. In the case of thiolated probe results were not as satisfying as the ones obtained with the biotinylated one, based on dextran-streptavidin-biotin protocol.

8.3.3.2 Thermal with "Blocking Oligos" Denaturation: Biotinylated Versus Thiolated Probe

The frequency shifts in Fig. 8.8 were obtained after the interaction between the probe and commercially available genomic DNA samples previously treated with the restriction enzyme (EcoRI) and then thermally denatured with blocking oligonucleotides.

Hybridization signals relative to the hybridisation of the probe with: noncomplementary and complementary oligonucleotide; bovine genomic DNA; porcine genomic DNA as negative control and blank solution, are shown in Fig. 8.8.

The results are similar to the ones found with thermal denaturation with biotinylated probe (Sect. 8.3.3.1) with regard to the values of frequency shifts and the easiness of regeneration with HCl.

Considering that the results obtained with thermal denaturation (Sect. 8.3.3.1, thiolated probe) do not meet the requirements for analytical applications, the



Fig. 8.8 Genomic bovine DNA: thermal with blocking oligonucleotides denaturation. (**a**) Biotinylated oligonucleotide probe. Signals: *1* Non-complementary oligonucleotide 8 ppm (n = 3), 2 Complementary oligonucleotide 2 ppm (n = 3), 3 Bovine genomic DNA 10 ppm (n = 3), 4 Porcine genomic DNA 10 ppm (n = 3), 5 Blank solution (n = 1). (**b**) Thiolated oligonucleotide probe. Signals: *1* Bovine genomic DNA 5 ppm (n = 4), 2 Bovine genomic DNA 10 ppm (n = 6), 3 Bovine genomic DNA 20 ppm (n = 4), 4 Porcine genomic DNA 5 ppm (n = 1), 5 Porcine genomic DNA 10 ppm (n = 3), 6 Porcine genomic DNA 20 ppm (n = 1), 7 Blank solution for 5 ppm (n = 2), 8 Blank solution for 10 ppm (n = 3), *11* Complementary oligonucleotide 1 ppm (n = 3), *12* Complementary oligonucleotide 2 ppm (n = 3)

denaturation treatment based on the use of "blocking oligo" was combined to the improved regeneration procedure.

Figure 8.8b shows the frequency shifts after the hybridisation reaction between the probe and: non-complementary and complementary oligonucleotides; bovine genomic DNA, porcine genomic DNA and different volumes of blank solution.

The sensor was able to distinguish between complementary and noncomplementary sequences both in synthetic oligonucleotidic samples and in digested non-amplified genomic DNA, both with biotinylated and eventually with thiolated probes immobilized on the sensor surfaces.

8.3.3.3 Analysis of Real Genomic Bovine DNA Samples, Extracted from Muscle by "Blocking Oligos" Denaturation; Biotinylated vs Thiolated Probes

Real samples were analysed, with biotinylated probe and with thiolated probe. Both samples were bovine genomic DNA extracted from animal muscle; and they were denatured with the thermal and blocking oligonucleotide treatment since it yielded the best results with both modified surfaces.



Fig. 8.9 Real genomic bovine DNA sample, extracted from muscle (thermal denaturation + blocking oligonucleotides). (a) Biotinylated oligonucleotide probe. Signals: *1* Noncomplementary oligonucleotide 8 ppm (n = 3), 2 Complementary oligonucleotide 2 ppm (n = 3), 3 Real sample (bovine genomic DNA) 10 ppm (n = 3), 4 Porcine genomic DNA 10 ppm (n = 1), 5 Blank solution (n = 1). (b) Thiolated oligonucleotide probe. Signals: *1* Non-complementary oligonucleotide 8 ppm (n = 3), 2 Complementary oligonucleotide 8 ppm (n = 3), 2 Complementary oligonucleotide 8 ppm (n = 3), 2 Complementary oligonucleotide 1 ppm (n = 3), 3 Complementary oligonucleotide 2 ppm (n = 3), 4 Real sample (bovine genomic DNA) 5 ppm, 20 min (n = 4), 5 Real sample (bovine genomic DNA) 5 ppm, 35 min (n = 1), 6 Real sample (bovine genomic DNA) 10 ppm, 20 min (n = 5), 7 Real sample (bovine genomic DNA) 10 ppm, 35 min (n = 1), 8 Real sample (bovine genomic DNA) 20 ppm, 20 min (n = 5), 9 Real sample (bovine genomic DNA) 20 ppm, 35 min (n = 1), *10* Porcine genomic 5 ppm, 20 min (n = 2), *11* Porcine genomic DNA 10 ppm, 20 min (n = 1), *12* Porcine genomic DNA 10 ppm, 35 min (n = 1), *13* Blank solution (n = 2)

Figure 8.9 shows the signals relative to the hybridisation of the probe with: non-complementary and complementary oligonucleotide; the real sample (bovine genomic DNA); porcine genomic DNA as the negative control and blank solution.

The frequency shifts obtained with biotinylated probe (Fig. 8.10a) with the real sample are well comparable with the ones found with commercially available DNA (Sect. 8.3.3.1), with regard to the values of frequency shifts and the easiness of regeneration with HCl.

Figure 8.9b shows the signals relative to the hybridisation of the probe with: non-complementary oligonucleotide, complementary oligonucleotide, real sample (bovine genomic DNA); porcine genomic DNA as negative control and blank solution.

As in the previous case the frequency shifts are comparable with the ones found with commercially available DNA. The signals 4, 6 and 8 have been recorded employing 20 min as contact time; while the 5, 7 and 9 are relative to 35 min contact time. There are no significant differences between the two data series. This is in contrast with the results obtained with the commercially available bovine DNA (see signals 4 and 5 in Fig. 8.9b). A reason of this can rely on the degradation of the



Fig. 8.10 Thiolated oligonucleotide probe. Porcine 20 ppm. Signals: *1* Non-complementary oligonucleotide 8 ppm (n = 3), 2 Complementary oligonucleotide 1 ppm (n = 3), 3 Complementary oligonucleotide 2 ppm (n = 3), 4 Bovine genomic DNA 5 ppm, 20 min (n = 4), 5 Bovine genomic DNA 10 ppm, 20 min (n = 6), 6 Porcine genomic DNA 20 ppm, 20 min (n = 1), 7 Blank solution for 5 ppm, 20 min (n = 2), 8 Blank solution for 10 ppm, 20 min (n = 3), 9 Porcine 20 ppm + Bovine 5 ppm genomic DNA, 20 min (n = 1), *10* Porcine 20 ppm + Bovine 10 ppm genomic DNA, 20 min (n = 3)

real sample, observed in the electrophoresis analysis, that causes a smaller number of non damaged copies. The aspecific adsorbtion was absent, as confirmed by the signals 10, 11, 12 and 13 relative to porcine genomic DNA samples.

8.3.3.4 Analysis of Mixture of Bovine and Porcine Genomic DNA Samples, by "Blocking Oligos" Denaturation

In order to investigate the applicability of this sensor to mixtures composed by DNA of different species, some mixtures with bovine and porcine DNA in different proportion have been prepared and analysed. The bovine and porcine DNA have been first separately digested by restriction enzymes, then mixed together and denatured with the new thermal and blocking oligonucleotides procedure. Considering that both modified surfaces showed similar behaviour, the experiments were performed employing the thiolated probe because of its easiest and quickest preparation. In Fig. 8.10 results obtained with genomic DNA mixtures, previously digested with the restriction enzyme (EcoRI), and denatured are shown. The samples were prepared adding various bovine DNA concentrations to a fixed porcine DNA concentration.

Comparing the signals of the bovine DNA samples and the ones of the mixtures of bovine and porcine DNA it is evident that the signals obtained with mixtures are specific and comparable to the ones observed with the solution containing just the bovine DNA. No matrix effect has been found, confirming the applicability of this system to sequence detection even in mixed samples, containing DNA from different species. More in detail, the comparison must be made between the signals 13 and 6, 11 and 4, 12 and 5. The contact time for the hybridization was increased from 20 to 35 min, in order to obtain a higher signal from the samples because of their complexity. In fact, the presence, in solution, of many non-complementary sequences hinders the target sequence that can't reach easily the surface. The employment of a longer contact time allowed an increase in the frequency shift while the aspecific adsorbtion did not rise. It is confirmed by the signal 8, obtained with 10 ppm of porcine DNA and a contact time of 35 min. From the comparison of the two denaturing methods we can conclude that although with simple thermal denaturation specific signal is recorded, unspecific bindings occurs and regeneration of the surface is not achieved by simple 1 mM HCl treatment. The sample pretreatment, in term of denaturation step, was here strategic for the sensor re-use and reproducibility. This already in case of highly frequent sequences. We will see how the same step is crucial in case of single of few sequences per haploid genome directly in genomic DNA. In conclusion, the sensor is able to distinguish between complementary and non-complementary sequences both in synthetic oligonucleotide samples and in digested genomic DNA. It is also able to recognize the presence of bovine DNA extracted from muscle (real sample) and also in mixed samples. The only required sample pre-treatment is the enzymatic digestion to fragment the DNA. The system is specific, sensitive, and reusable, with fast analysis time. On the base of our findings, the sensor could represent an alternative method, to the traditional biomolecular techniques, for the identification of species-specific DNA sequences, directly in enzymatically digested DNA, bypassing the PCR step.

8.4 Detection of Target Sequences Present in Single Copy Per Haploid Genome in Plant Genomic DNA

Once assessed the ability of the sensor to detect highly present sequences per haploid DNA we moved further to the detection of target DNA present in single or in few copies in haploid genomes. For this reason we focused on markers of transgenosis sequences, in particular we developed a system for detecting the P35 promoter in Genetically Modified Organisms (GMOs). Here we used a genetically

modified (GM) tobacco plant, Nicotiana glauca, an ornamental plant, transformed with pBI121, which contained a gene cassette carrying the P35 form cauliflower virus. A gene cassette consists of a promoter (P), a coding region and a terminator (T). Officially established method on European scale for GMOs control are based on the search of Promoter region (P35S) of the CAMV (cauliflower mosaic virus) ribosomal RNA and on NOS terminator (TNOS) of the nopalin synthase gene from the soil bacterium Agrobacterium Tumefaciens [29]. This promoter is used in approved GM soya and maize. Analytical methods for GMOs are based on realtime PCR (required for labelling) or end-point PCR [61]. Our group has been quite active in the development of GMOs DNA-based sensor using different transduction principles with application to GMO Reference Certified Material or real-samples i.e. optical (SPR), piezoelectric and electrochemical. We thus selected the P35S as target analyte for the study of the piezolectric based detection of genomic DNA. The GM plant Nicotiana glauca (GR4), was not commercially available, but was transformed at Laboratorio di Genetica, Università di Firenze, Italy. The not transformed plant (wild type) was taken as negative control in all the experiments.

Similarly to the bovine DNA, extraction of the plant DNA (from leaves) was achieved using available protocols [23].

8.4.1 Sample Pre-treatment

8.4.1.1 DNA Fragmentation

Fragmentation was conducted by enzymatic digestion with DNA digestion enzymes Bam HI e Hind III, which ended into a 872 bp fragment containing the P35S target sequence.

The DNA immobilized probe mapped within this fragment. The enzyme selection was achieved by free available software (DNA club). Purification of fragmented DNA by precipitation in ethanol and re-suspension is necessary prior testing on the sensor.

8.4.1.2 DNA Denaturation

Here three different denaturation protocols were tested and compared. In particular we compared chemical denaturation with formamide used in our previous work with optical sensing (SPR) [32] and blocking oligos" denaturation.

The sensor surface was modified by thiol/dextran chemistry and biotinylated probes used for the immobilization via streptavidin anchoring [35].

The oligonucleotides used are reported in Table 8.2.

The P35S sensor has been already optimized and tested on PCR amplified DNA in different real matrices such as soft drinks, crackers, as well as in PCR amplified DNA from *Nicotiana glauca*, thus we will directly report the results on

assays and blocking ongoinucleotide used in t	ne denaturation protocor
Biotinylated probe (35S, 25 mer):	5' Biotin-GGCCATCGTTGAAGA TGC CTC TGC C-3'
Complementary target (35S, 25 mer):	5'-GGC AGA GGC ATC TTC AAC GAT GGC C-3'
Non complementary (Tnos, 25 mer):	5'-GAT TAG AGT CCC GCA ATT AAT CAT T-3'
Blocking oligonucleotide 1 (35S1, 21 mer):	5'-GCT CCT ACA AAT GCC ATC ATT-3'
Blocking oligonucleotide 2 (35S2, 18 mer):	5'-CTC CAA ATG AAA TGA ACT-3'

Table 8.2 Olignucleotide sequences of immobilized probe, modified with biotin in 5' end,complementary and non complementary sequences (negative control) used in the optimizationassays and blocking oligonucleotide used in the denaturation protocol

the optimized system, using this time genomic DNA samples extracted from GM plant, fragmented by enzymatic reaction, and denatured differently to compare, also in this case, the different sample pre-treatments with the aim to stress the key importance of denaturation steps. In parallel, as control oligonucleotides and PCR sample undergoing to same treatments were used as controls, to study any possible interference with the sensor.

8.4.1.3 Analysis of PCR DNA Samples, by Chemical and "Blocking Oligos" Denaturation

Three approaches were tested: thermal (95 °C for 5 min, cooling in ice for 1 min), chemical (20 % formamide, 0.3 M NaOH at 42 °C for 30 min), and thermal combined with blocking oligonucleotides.

The complementary 35S oligonucleotide (25-mer, 1 ppm), a non complementary oligonucleotide (25-mer, 8 ppm), the 35S PCR fragment (243 bp), a PCR negative control (180 bp), and PCR blanks were analyzed after the three denaturation procedures (Fig. 8.11a, b).

The effect of the chemical denaturation on oligonucleotides resulted in an increase (46 %) of the signal if compared with the one obtained with the thermal treatment. This increase could be due to a non specific effect since it is also observed with the non complementary oligonucleotide. On the contrary, the thermal and blocking oligonucleotides' treatment did not affect the hybridization signal. The same findings were obtained with the same treatments applied to PCR fragments and PCR negative controls.

The same hybridization signal was obtained after treating the 35S samples with the three denaturation methods, but the reproducibility (expressed as coefficient of variation, CV%) (n = 3) was quite different (9 % for the thermal one, 12 % for the thermal and blocking oligonucleotides one, and 25 % for the chemical one). The same denaturation treatments applied to the negative controls gave less homogeneous results since a high signal was observed in the case of the chemical treatment, while negligible results were found in the cases of thermal plus blocking and thermal treatments. The sensor behaviour is shown in Fig. 8.11a.



Fig. 8.11 (a) Effect of denaturation treatment on oligo and on PCR samples. Samples analyzed: oligonucleotide complementary p35S, oligonucleotide non complementary, fragment PCR 35S (54 ppm), PCR negative control, (35 ppm) and blank (n = 3); (b) Analysis of target P35S sequence in fragmented genomic transgenic DNA (10 ppm) GR4 (n = 6), non transgenic WT DNA (n = 6) and blank (n = 3). DNA denatured by three approaches (thermal, thermal with blocking oligonucleotides, and chemical). (Reproduced with permission from Minunni et al. [35]. Copyright (2005) American Chemical Society)

After the initial evaluation of the sample pre-treatments on the sensor behavior, these were finally applied to more complex, unamplified genomic DNA.

8.4.1.4 Analysis of Plant Genomic DNA Samples, by Chemical and "Blocking Oligos" Denaturation

Genomic DNA, previously fragmented by enzymatic digestion by restriction enzymes (BamH), previously checked to leave intact the target sequence. The length of the fragments containing the target sequence, complementary to the immobilized probe, was 872 bp. All of the denaturation procedures were applied to 10 ppm of sample. The results are reported in Fig. 8.11b.

The thermal treatment did not results in a measurable signal. On the contrary the thermal plus blocking oligonucleotides and chemical denaturations allow significant hybridization of the samples. Furthermore, thermal plus oligo oligonucleotide denaturation, results in a better reproducibility then the one obtained by the chemical one. Also the treatment may influence the sensor lifetime, since little regeneration was achieved with chemical denaturation, while with thermal with blocking the surface could be regenerated more then ten times, before losing sensitivity and specificity, thus more then doubling the sensor life-time.

This results, stresses again the importance of selecting the best sampledenaturation.

The piezoelectric sensor detection was very selective, the target sequence was detected in the transgenic sample (GR4) and the signal were significantly different from the negative control (wild type, WT), except when the thermal treatment alone was applied. Moreover, the best results were found when the thermal plus blocking oligonucleotides' treatment was applied, as demonstrated by an evaluation of the reproducibility and the lower nonspecific effect when testing the negative controls, as confirmed by the findings observed with ssDNA and PCR-amplified DNA. This is in line with our previous finding with microsatellite DNA detection in bovine genomic DNA.

This work demonstrated that it is possible to detect the target sequence directly in unamplified genomic DNA, even considering the low concentration of the target in the sample $(4 \times 10^5 \text{ copies in 10 ppm of sample})$. PCR-amplified DNA (243 bp) represents an enriched sample where the target sequence is present in a very high number of copies $(4 \times 10^{11} \text{ copies})$. To explain the detection of such a low number of copies of target DNA, additional contributions (i.e., viscoelastic effects) to the biosensor signal other than mass loading may be taken into account. Moreover, it must be considered that the signals due to oligonucleotides, to the PCR samples, and to the genomic digested DNA on the surface cannot be compared due to the very different matrix complexity of all these samples and to the different secondary structure once the DNA is hybridized to the probe.

The real-time and label-free DNA sequence detection in non amplified DNA, as reported here, represents an important improvement in DNA analysis. Since the specificity of the system relies on the probe immobilized on the surface, the

applicability of direct genomic sensing is wide, from environmental to food and clinical analysis. Recently some work on sensitive DNA in genomic DNA has been reported by some authors and will be here summarized and discussed.

8.5 State of Art

Recently some groups have reported about sensitive DNA detection by piezoelectric sensing. In Table 8.3, are summarized the approaches. Crystals with different fundamental frequencies, from 5 up to 12 MHz, are used. Different frequency means also different sensitivities as can be deduced by Sauerbrey model, (Eq. 8.1) [7]. Also different operating temperatures (from room temperature up to 55 °C) and flow modes are used, instead, except the case of "Dip and Dry" approach, were the crystal is exposed to the target solution, followed by rinsing to remove the unbound material and then dried. The frequency shift is taken before and after target exposure, when the crystal is dried.

Gene detection directly in genomic DNA is achieved in microorganisms, i.e., *Mycobacterium tuberculosis*, is detected in human specimens (sputum), and the approach used followed our first report, in other words the sample is digested by enzymatic reaction and sample denaturation is achieved by thermal with blocking oligonucleotides [62]. Signal requires amplification and the detection approach is by "dip and dry" method, not directly in liquid. This procedure may be cumbersome and may have important reflections on system reproducibility, at least to our experience The QCM, for its sensitivity to humidity is also used as hygrometer, and in the DNA measurement this effect may be present and difficult to be controlled. However the application reported encourages further development for real clinical application of these devices.

Hepatitis B virus (HBV) genomic DNA is genotyped by hybridization of target sequence to biotinylated Peptide nucleic acids (PNA) probes, which have been proved to be suitable probes especially in mismatch detection [63]. The analysis of the one and two base mismatch is performed without any fragmentation of the DNA molecule and simply by thermal sample denaturation. The hybridization temperature is set at 55 °C. To the dsDNA complex is further bond Rec protein to improve system sensitivity reported to be 8 pg/L in standard solution. Detection of the target sequences with DNA samples extracted from animal feed containing 30 % RR soybean amplified by the PCR and unamplified DNA. The detection limit for genomic DNA was in the range of 4.7·105 numbers of genomic copies contained EPSPS gene in the QCM cell. Mismatch analysis was achieved in DNA extracted from human blood cells. Good correlation (r = 0.9613) was found with RT-PCR method, indicating potential real application of the device to detect HBV in clinical diagnostics.

Two papers deal with genetically modified material. In particular detection in Soybean Roundup Ready 5-enolpyruvylshikimate-3-phosphate synthase (EPSPS) gene, which is an active component of an insert integrated into RR soybean genome,

Table 8.3 Literature report	rting abou	tt DNA genomic (detection by piezoelectric sensin	а С			
Target analyte	T°C	Probe mer	Digestion	Denaturation	Sample	Signal amplification	Ref.
Mycobacterium tuberculosis Gene IS6110	Room	25	Enzymatic BstDSI	Thermal with blocking	Sputum	Streptavidin	[62]
Hepatitis, point mutation in L50 (2562–2613)	55	29 PNA with linkers	No	Thermal	Blood	Rec -protein	[63]
OGM EPSPS gene	Room	21	No	Thermal	R-Ready soybean 30 %	No	[64]
OGM P35S GM tobacco plants	Room	25	EnzymaticBamHI/HindIII vs ultrasounds	Thermal	GM tobacco plants	No	[65]

٠	_
	U.
	Ċ
	57
	d,
	Ū.
	-
	0
•	ñ
	۰.
1	+
	C
	กั
_	9
2	•
	₩
	C
	Ñ
	53
	đ,
٠	-
	9
	5
	۶
-	<u> </u>
	_
	F
	1
	\leq
۰,	Ξ
1	7
	4
	¢,
	1
	n)
-	~
	4
	C
	-
	<u>_</u>
	⊏
	7
	-
	-
	57
	d,
	a
	-
	-
1	<1
	_
۲	-
5	Ż
-	Ž
2	\tilde{z}
2	Ž
2	Ż
1	
1	
1	
54	
100.1	NOT 1100
1.0.1	about DN
144	N about DN
144	o about DN
144	Jo about DN
144.	Ing about DN
1.1.1	ting about DN
1.00.	rting about DN
1 · D · 1	orting about DN
144.	Orting about DN
144.	norting about DN
TAG . 1	Phorting about DN
	enorting about DN
14CL	reporting about I)N
14CL	e reporting about 1)N
144.	e reporting about DN
TAG . 1	the reporting about DN
	lire reporting about DN
	tilre reporting about DN
	athre reporting about DN
	rathre reporting about DN
	Stature reporting about I)N
	erative reporting about I)N
	terature reporting about I)N.
	iteratine renorting about DN
	I iterative reporting about DN
	Literatine reporting about DN
	Literatine reporting about DN
	I literative reporting about 1N.
	A Literature reporting about 10N
	.5 Literature renorting about DN
	X.5 Literature reporting about DN.
	X.5 Literative reporting about 1)N.
	e X.5 iterative reporting about 1)N.
	e X.5 Literature reporting about 1)N.
	DIE X.5 Literature renorting about DN.
	ble X.5 Literature reporting about DN.
	able X.3 Literature reporting about DN
	able X.3 Literature reporting about DN
	Iable X.3 Literature reporting about DN

is reported [64]. EPSPS gene renders plants' resistance against herbicide glyphosate (Roundup Ready). Glyphosate is toxic for plants because it prevents the production of aromatic aminoacids (tryptophan, tyrosine, phenylalanine). A 21-mer single stranded biotinylated oligonucleotide (probes) is immobilized on sensor (8 MHz quartz crystal) surface via avidin binding, previously covalently attached to gold. No sample digestion has been performed and thermal denaturation applied. Very low reproducibility is reported: CV% 20 %, may be due to pre-analytical steps i.e. sample pre-treatment. The sensor proposed in this study was tested for the detection of EPSPS sequence in PCR non-amplified DNA samples extracted from animal feed containing 30 % of the genetically modified soybean Roundup Ready. The sensor was able to distinguish transgene sequence between modified and unmodified soybean DNA (extracted from animal feed) at the following level: 3.6, 4.6 and $5.4 \mu g$ of genomic DNA in 200 μ l of QCM cell.

The influence of different used probe on the sensor behavior is also reported, confirming the importance of probe optimization in the development of nucleic acid-based sensing as underlined before [22]. The negative control, genomic DNA extracted from unmodified soybean, generated small frequency shift, which might be attributed to the direct adsorption of DNA on gold electrode surface without hybridization to the probe or weak, non-specific interactions between the probe and partially complementary sequences present in very long genomic DNA. The detection limit was in the range of 4.7×10^5 numbers of genome copies with EPSPS gene in the QCM cell with 200 µl of investigated samples.

Finally, detection of P35S has also been reported in pflp (ferredoxin like protein)gene inserted in *Nicotiana tabacum* plants [65]. Wild type tobacco DNA was used as a control. The extraction protocol applied was the same applied in our laboratory, i.e. CTAB, and BamH/Hind III restriction enzymes were used in the fragmentation by enzymatic digestion, while we used only BamH. Sonication in parallel was used to compare the fragmentation efficacy of the two approaches. It is interesting the comparison between the two fragmentation approaches performed in this work. By the comparison can be observed the same frequency decreases were observed at the same concentrations of both digested and sonicated samples but with different standard deviations, having the sonicated samples the highest. This result was assumed since fragmentation of the DNAs was non-selective during sonication process. However, the sonication even if nonselective is much faster to performed (minutes compared to overnight digestion used in enzymatic fragmentation).

Very sensitive target sequence detection directly in unamplified DNA can be achieved by label free and real-time detection using piezoelectric sensing. The sensor development as well as the preanalytical steps are strategic for the success of the detection.

First the probe immobilization chemistry should allow selecting binding by preventing any unspecific adsorption from the bulk to the surface. For this reason we have selected and compared in the presented application both biotinylated and thiolated probes. Biotinylated probes are bound to the previously modified surface by thiol/dextran/strepatvidin, while thiolated probes are directly bound to the surface. At the same time the probe selection is also a key step. In fact the presence

of loops or hairpins in the probe structure could affect the hybridization with the target sequence, thus careful selection, eventually assisted by *in silico* selection, could be employed. The presence of secondary structures can be also prevented by tuning the denaturation protocols to prevent non only re-annealing but also intra strand bonding.

Fragmentation of genomic DNA is also important, but enzymatic digestion may be too long for use in biosensing. In this sense, once assessed the sensitive DNA detection one could move on to ultrasound fast fragmentation.

The application of these systems to a wide panorama of clinical, food and environmental analysis is the next challenge. In this sense, multi-array approaches based on piezoelectric sensing are welcome, allowing parallel analysis of target sequences. For that different sensor designs can be foreseen. Over years interesting papers dealing with the development of automated instrumentation have been presented.

References

- 1. Curie, J., Curie, P.: An oscillating quartz crystal mass detector. Rendu 91, 294–297 (1880)
- 2. Krempl, P., Schleinzer, G., Wallnöfer, W.: Gallium phosphate, GaPO₄: a new piezoelectric crystal material for high-temperature sensorics. Sens. Actuat. A **61**, 361–363 (1997)
- Vasilescu, A., Ballantyne, S.M., Cheran, L.E., Thompson, M.: Surface properties and electromagnetic excitation of a piezoelectric gallium phosphate biosensor. Analyst 130, 213–220 (2005)
- Janshoff, A., Steinem, C.: Quartz crystal microbalance for bioanalytical applications. Sensor Update 9, 313–354 (2001)
- Bruckenstein, S., Shay, M.: Experimental aspects of the use of quartz crystal microbalance solution. Electrochim. Acta 30, 1295–1300 (1985)
- 6. O'Sullivan, C.K., Guilbault, G.G.: Commercial quartz crystal microbalances theory and applications. Biosens. Bioelectron. **14**, 663–670 (1999)
- Sauerbrey, G.: The use of quartz oscillators for weighing thin layers and for microweighing. Z. Physik 155, 206–222 (1959)
- Hiller, A.C., Ward, M.D.: Scanning electrochemical mass sensitivity mapping of the quartz crystal. Anal. Chem. 64, 2539–2554 (1992)
- 9. Kanazawa, K.K., Gordon, J.G.: Frequency of a quartz microbalance in contact with liquid. Anal. Chem. **57**, 1770–1771 (1985)
- Chang, S., Muramatsu, H., Nakamura, C., Miyake, J.: The principle and applications of piezoelectric crystal sensors. Mater. Sci. Eng. C 12, 111–123 (2000)
- Minunni, M., Mascini, M., Guilbault, G.G., Hock, B.: The quartz crystal microbalance as biosensor. A status report on its future. Anal. Lett. 28, 749–764 (1995)
- Kim, N., Park, I.-S., Kim, D.-K.: Characteristics of a label-free piezoelectric immunosensor detecting Pseudomonas aeruginosa. Sens. Actuat. B 100, 432–438 (2004)
- Mannelli, I., Minunni, M., Tombelli, S., Mascini, M.: Quartz Crystal Microbalance (QCM) affinity biosensor for Genetically Modified Organisms (GMOs) detection. Biosens. Bioelectron. 18, 129–140 (2003)
- Tombelli, S., Mascini, M., Sacco, C., Turner, A.P.F.: A DNA piezoelectric biosensor assay coupled with a polymerase chain reaction for bacterial toxicity determination in environmental samples. Anal. Chim. Acta 418, 1–9 (2000)

- Sklàdal, P., dos Santos Riccardi, C., Yamanaka, H., Inàcio da Costa, P.: Piezoelectric biosensor for real time monitoring of hybridization and detection of hepatitis C virus. J Virol Methods 117, 145–151 (2004)
- Dell'Atti, D., Tombelli, S., Minunni, M., Mascini, M.: Detection of clinically relevant point mutations by a novel piezoelectric biosensor. Biosens. Bioelectron. 21, 1876–1879 (2006)
- Kukanskis, K., Elkind, J., Melendez, J., Murphy, T., Miller, G., Garner, H.: Detection of DNA hybridization using the TISPR-1 surface plasmon resonance biosensor. Anal. Biochem. 274, 7–17 (1999)
- Allara, D.L., Nuzzo, R.G.: Adsorption of bifunctional organic disulfides on gold surfaces. J. Am. Chem. Soc. 105, 4481–4483 (1983)
- Tombelli, S., Mascini, M., Turner, A.P.F.: Improved procedures for immobilisation of oligonucleotides on gold-coated piezoelectric quartz crystals. Biosens. Bioelectron. 17, 929–936 (2002)
- Tombelli, S., Minunni, M., Mascini, M.: Piezoelectric biosensors: strategies for coupling nucleic acid to piezoelectric devices. Methods 37, 48–56 (2005)
- Powdrill, T.F.: Publication number: WO03057858 (A2), European patent: C12Q1/68B10A; Y01N6/00, Application number: WO2003US00069 20030102 (2003)
- Ermini, M.L., Scarano, S., Bini, R., Banchelli, M., Berti, D., Mascini, M., Minunni, M.: A rational approach in probe design for nucleic acid-based biosensing. Biosens. Biolectron. 26, 4785–4790 (2011)
- Sambrook, J., Fritsch, E.F., Maniatis, T.: Molecular Cloning: A Laboratory Manual. Laboratory Press, New York (1989)
- Almadidy, A., Watterson, J., Piunno, P.A.E., Raha, S., Foulds, I.V., Horgen, P.A., Castle, A., Krull, U.: Direct selective detection of genomic DNA from coliform using a fiber optic biosensor. Anal. Chim. Acta 461, 37–47 (2002)
- 25. D'Agata, R., Corradini, R., Ferretti, C., Zanoli, L., Gatti, M., Marchelli, R., Spoto, G.: Ultrasensitive detection of non-amplified genomic DNA by nanoparticle-enhanced surface plasmon resonance imaging. Biosens. Bioelectron. 25, 2095–2100 (2010)
- 26. Lewin, B.: Genes VI. Oxford University Press, Oxford (1997)
- Minunni, M., Mannelli, I., Spiriti, M.M., Tombelli, S., Mascini, M.: A biosensor for the detection of highly repeated sequences in non-amplified genomic DNA. Anal. Chim. Acta 526, 19–25 (2004)
- Mariotti, E., Minunni, M., Mascini, M.: Surface Plasmon Resonance (SPR) biosensor for Genetically Modified Organism (GMOs) detection. Anal. Chim. Acta 453, 165–172 (2002)
- Minunni, M., Mascini, M., Mascini, M., Cozzani, I.: Screening methodologies for genetically modified organisms (GMOs). Anal. Lett. 215(33), 3093–3126 (2000)
- Bianchi, N., Rutigliano, C., Tomassetti, M., Feriotto, G., Zorzato, F., Gambari, R.: Biosensor technology and surface plasmon resonance for real time detection of HVI1 genomic sequence amplified by polymerase chain reaction. Clin. Diagn. Virol. 8, 199–208 (1997)
- Feriotto, G., Borgatti, M., Mischiati, C., Bianchi, N., Gambari, R.: Biosensor technology and surface plasmon resonance for real-time detection of genetically modified Roundup Ready soybean gene sequences. J. Agric. Food Chem. 50, 955–962 (2002)
- Giakoumaki, E., Minunni, M., Tombelli, S., Tothill, I.E., Mascini, M., Bogani, P., Buiatti, M.: Combination of amplification and post-amplification strategies to improve optical DNA sensing. Biosens. Bioelectron. 19, 337–344 (2003)
- Alberts, B., Bray, D., Lewis, J., Raff, M., Roberts, K., Watson, J.D.: Molecular Biology of the Cell. Garland Publishing Inc., New York (1994)
- Bhattacharyya, A.J., Feingold, M.: Single molecule study of reaction between DNA and formamide. Talanta 55, 943–949 (2001)
- Minunni, M., Tombelli, S., Fonti, J., Spiriti, M.M., Mascini, M., Bogani, P., Buiatti, M.: Detection of genomic DNA by PCR-free piezoelectric sensing. J. Am. Chem. Soc. 127, 7966–7967 (2005)

- Wang, R., Minunni, M., Tombelli, S., Mascini, M.: A new approach for the detection of specific DNA sequences in amplified nucleic acids by surface plasmon resonance biosensor. Biosens. Bioelectron. 20, 598–605 (2004)
- Minunni, M., Tombelli, S., Mascini, M.: Biosensor approach for DNA sequences detection in non-amplified genomic DNA. Anal. Lett. 40(7), 1360–1367 (2007)
- Pech, M., Streeck, R.E., Zachau, H.G.: Patchwork structure of a bovine satellite DNA. Cell 18, 883–893 (1979)
- Hunt, D.J., Parkes, H.C., Lumley, I.D.: Identification of the species of origin of raw and cooked meat products using oligonucleotide probes. Food Chem. 60, 437–442 (1997)
- Kim, H., Shelef, L.A.: Characterization and identification of raw beef, pork, chicken and turkey meats by electrophoretic patterns of their sarcoplasmic proteins. J. Food Sci. 51, 731–741 (1986)
- Skarpeid, H.J., Kvaal, K., Hildrum, K.I.: Identification of animal species in ground meat mixtures by multivariate analysis of isoelectric focusing protein profiles. Electrophoresis 19, 3103–3109 (1998)
- 42. Toorop, R.M., Murch, S.J., Ball, R.O.: Methodology and development of prediction equations for the determination of pork substitution in veal. Food Res. Int. **30**, 629–636 (1997)
- Ashoor, S.H., Monte, W.C., Stiles, P.G.: Liquid-chromatographic identification of meats. J. Ass. Off. Anal. Chem. 71, 397–403 (1998)
- 44. Hsien, Y.H., Sheu, S.C., Bridgman, R.C.: Development of a monoclonal antibody specific to cooked mammalian meats. J Food Prot **61**, 476–481 (1998)
- Lenstra, J.A., Buntjer, J.B., Janssen, F.W.: On the origin of meat DNA techniques for species identification in meat products. Vet. Sci. Tomorrow 2, 1–15 (2001)
- Calvo, J.H., Rodellar, C., Zaragoza, P., Osta, R.: Beef- and bovine-derived material identification in processed and unprocessed food and feed by PCR amplification. J. Agric. Food Chem. 50, 5262–5264 (2002)
- 47. Partis, L., Croan, D., Guo, Z., Clark, R., Coldham, T., Murby, J.: Evaluation of a DNA fingerprinting method for determining the species origin of meats. Meat Sci. **54**, 369–376 (2000)
- Baur, V.C., Teifel-Greding, J., Liebhardt, E.: Identification of heat-processed meat by DNA analysis. Arch. Lebensmittel Hyg. 38, 149–176 (1987)
- 49. Chikuny, K., Ozutsumi, K., Koishikawa, T., Kato, S.: Species identification of cooked meats by DNA hybridization assay. Meat Sci. 27, 119–128 (1990)
- 50. Ebbehøj, K.F., Thomsen, P.D.: Species differentiation of heated meat-products by DNA hybridization. Meat Sci. **30**, 221–234 (1991)
- Winterø, A.K., Thomsen, P.D., Davies, W.: A comparison of DNA-hybridization, immunodiffusion, countercurrent immunoelectrophoresis and isoelectric-focusing for detecting the admixture of pork to beef. Meat Sci. 27, 75–85 (1990)
- 52. Verkaar, E.L.C., Nijman, I.J., Boutaga, K., Lenstra, J.A.: Differentiation of cattle species in beef by PCR-RFLP of mitochondrial and satellite DNA. Meat Sci. **60**, 365–369 (2002)
- Meyer, R., Candrian, U., Lüthy, J.: Detection of pork in heated meat products by the polymerase chain-reaction. J. AOAC Int. 77, 617–622 (1995)
- Meyer, R., Höfelein, C., Lüthy, J., Candrian, U.: Polymerase chain reaction-restriction fragment length polymorphism analysis: a simple method for species identification in food. J. AOAC Int. 78, 1542–1551 (1995)
- Meyer, R.: Nachweis gentechnologisch veränderter Pflanzen mittels der Polymerase Kettenreaktion (PCR) am Beispiel der FLAVRSAVRTM-Tomate, Z. Lebensm, Unters. Forsch. 201, 583–586 (1995)
- Meyer, R., Candrian, U.: PCR-based DNA analysis for the identification and characterization of food components. Lebensm. Wiss. Technol. 29, 1–9 (1996)
- 57. Murray, B.W., McClymont, R.A., Strobeck, C.: Forensic identification of ungulate species using restriction digests of PCR-amplified mitochondrial DNA. J. Forensic Sci. **40**, 943–951 (1995)

- Saez, R., Sanz, Y., Toldrá, F.: PCR-based fingerprinting techniques for rapid detection of animal species in meat. Meat Sci. 66, 659–665 (2004)
- Jonker, K., Tilburg, J., Hagele, G., De Boer, E.: Species identification in meat products using real-time PCR. Food Addi. Contam. 25, 527–533 (2008)
- 60. Mannelli, I.: PhD thesis, University of Florence (2006)
- 61. EU Labelling And Traceability Regulation, Regulation (Ec) No 1830/2003 of The European Parliament and of the Council, 22 September (2003)
- 62. Kaewphinit, T., Santiwatanakul, S., Promptmas, C., Chansiri, K.: Detection of mycobacterium tuberculosis in clinical specimens. Sens. Transd. J. **113**, 115–126 (2010)
- 63. Yao, C., Zhu, T., Tang, J., Wu, R., Chen, Q., Chen, M., Zhang, B., Huang, J., Fu, W.: Hybridization assay of hepatitis B virus by QCM peptide nucleic acid biosensor. Biosens. Bioelectron. 23, 879–885 (2008)
- 64. Stobiecka, M., Ciesla, J.M., Janowska, B., Tudek, B., Radecka, H.: Piezoelectric sensor for determination of genetically modified soybean roundup ready in samples not amplified by PCR. Sensors 7, 1462–1479 (2007)
- Karamollaoglu, I., Öktem, H.A., Mutlu, M.: QCM-based DNA biosensor for detection of genetically modified organisms (GMOs). Biochem. Eng. J. 44, 142–150 (2009)

Chapter 9 Surface Plasmon Resonance-Based Methods

Roberta D'Agata and Giuseppe Spoto

Abstract Surface plasmon resonance represents a new tool for the simple, fast and cheap nucleic acid detection. Large efforts have been paid during the last decade with the aim to develop even more sensitive and specific SPR-based methods to be used for the direct detection of genomic DNA. This Chapter, after a description of fundamentals of surface plasmon resonance and allied techniques, will review the state-of-the-art of recent platform and assay developments and will provide an overview of recent uses of SPR techniques for the direct detection of non-amplified nucleic acids.

9.1 Introduction

Nucleic acid microarray technologies [1] have revolutionized bioanalytical capabilities in different areas including genomics [2], clinical diagnostics [3], food analysis [4], and environmental control [5].

The most widely adopted nucleic acid microarray platforms rely on nucleic acid amplification procedures and require the introduction of labels for the identification of the target sequence. These requirements are costly, labour-intensive and timeconsuming. Moreover, the target amplification and labelling procedures may cause sample contaminations that could interfere with the original interaction event.

The detection and the sequence-identification of nucleic acids are commonly achieved after the hybridization of the target strand to a surface immobilized complementary probe strand. The efficiency of the hybridization reaction depends

R. D'Agata • G. Spoto (🖂)

Dipartimento di Scienze Chimiche, Università di Catania, Viale A. Doria 6, Catania, Italy e-mail: gspoto@unict.it

G. Spoto and R. Corradini (eds.), *Detection of Non-Amplified Genomic DNA*, Soft and Biological Matter, DOI 10.1007/978-94-007-1226-3_9, © Springer Science+Business Media Dordrecht 2012

on factors such as ionic strength, temperature, length of the nucleotide sequence, and the degree of mismatch. The surface confinement of the reaction introduces additional factors like surface immobilized probe steric accessibility and orientation, electrostatic repulsion between strands, mass transfer of the target from the bulk of the solution to the solid surface, and probe surface density. Each of them should be optimized in order to control the stability and the kinetic of the probe-target interactions [6].

Surface Plasmon Resonance (SPR) represents the most advanced label-free biosensor technology for detection of chemical and biological analytes [7]. In recent years, SPR biosensors have been applied for the analysis of numerous analytes related to medical diagnostics. In particular, it has been widely used to detect nucleic acids in real-time and by using multiplexed formats. Issues related to the sensitivity of SPR-based methods in detecting nucleic acids have been fully addressed in recent years and will be discussed in following sections.

In this chapter, fundamentals of SPR methods will be introduced and recent advances in the development of SPR-based platforms will be discussed. Particular attention will be paid in discussing new materials and nanostructures that aim at advancing multiplexed and ultrasensitive nucleic acid detection. Capabilities offered by nanostructure-enhanced SPR in detecting non-amplified genomic DNA will be discussed at the end.

9.2 Theory of Surface Plasmon Resonance

After the first observation of the underlying physical phenomenon given by Wood in 1902 [8], the SPR technique has become one of the fastest-growing analytical tool over the last twenty years. Advantages associated with the SPR detection of biomolecular systems (including good sensitivity, label-free and real-time detection), together with the commercial availability of equipments and sensing surfaces, have made SPR the technique-of-choice for interaction analyses and biosensing [9].

Surface plasmons [10] are the quanta associated with periodical oscillations of the electron charge density occurring at the interface between a metal and a dielectric medium.

The concept of plasmon originates in the plasma approach of Maxwell's theory: the free electrons of a metal are treated as an electron liquid (density of about 10^{23} cm⁻³) called "plasma", ignoring the lattice in a first approximation. Electron density fluctuations propagate through the volume of a metal with a characteristic frequency given by:

$$\hbar\omega_p = h\sqrt{\frac{4\pi ne^2}{m_e}} \tag{9.1}$$



Fig. 9.1 Schematic of surface plasmon waves at the interface between a metal (dielectric constant ε_1) and a dielectric (ε_2). The electromagnetic field of SPs propagating on a surface in the x direction and decaying exponentially into z- direction are also shown schematically. The exponential dependence of the field E_z is seen on the *right*

where ω_p is the frequency associated to plasmon oscillations, *n* is the free electron density of the material, *e* is the electron charge, and m_e is the effective mass of an electron. When the plasma is excited by an external source with a frequency equal to ω_p , electrons collectively and coherently oscillate in the metal [11].

Electrons at the boundary between a metallic surface and a medium having a dielectric constant ε_2 can perform coherent fluctuations called surface plasmons (SPs). These fluctuations are confined at the boundary and vanishes both sides of the metal surface. Surface plasmon waves are *p*-polarized and are described by a wave vector \mathbf{k}_x parallel to the *x* direction (Fig. 9.1)

$$k_x = \frac{\omega}{c} \sqrt{\frac{\varepsilon_1 \varepsilon_2}{\varepsilon_1 + \varepsilon_2}}$$
(9.2)

where $\varepsilon_1 = \varepsilon'_1 + i\varepsilon''_2$ is the complex dielectric constant of the metal and ε_2 is the dielectric constant of the medium.

The above mentioned surface charge fluctuations, which can be localized in the z-direction within the length of about of 1 Å around the metal surface, are accompanied by a mixed transversal and longitudinal electromagnetic field (E) which disappears at $|z| \rightarrow \infty$ and has its maximum in the surface z = 0 [12]. The frequency ω_p of the surface plasmon longitudinal oscillations is tied to its wave vector \mathbf{k}_x by a dispersion relation $\omega_p(\mathbf{k}_x)$ (Fig. 9.2) [13].

The wave vector (\mathbf{k}_x) associated with SPs always remains larger than wave vector of light $(\mathbf{k}_{\text{light}})$ having the same energy $\hbar \omega_p$ and traveling through the medium ε_2 :

$$k_{light} = \frac{\omega}{c} \sqrt{\varepsilon_2} \tag{9.3}$$

For plasmon excitation by a photon to take place, the energy and the momentum must both be conserved during the photon-plasmon coupling. This requirement is met when the wave vector for the photon $\mathbf{k}_{\text{light}}$ and the plasmon \mathbf{k}_x are equal in magnitude and direction for the frequency of both waves. The direction of the wave



Fig. 9.3 (a) Diffraction grating; (b) Otto geometry

vector is the direction of the wave propagation, while its magnitude depends on the dielectric constant of media at the interface.

Since the SP's dispersion relation never intersects the dispersion relation of electromagnetic radiation (Fig. 9.2), SPs cannot couple with a freely propagating light beam incident upon the metal surface and for this reason SPs are "non-radiative" fluctuations of the surface electron density. In order to couple photon at a given energy $h\omega$ with SPs the wave vector has to be increased by a $\Delta \mathbf{k}_x$ (Fig. 9.2). A similar increase can be obtained by using specific experimental methods.

Photons can excite plasmons via a grating coupler (Fig. 9.3a) [14]. When light hits a grating with a grating constant *a*, at a θ angle, the component of the wave vector in the surface is increased by $2\pi/a$:

$$k_x = \frac{\omega}{c}\sin\theta + \frac{2\pi}{a} \tag{9.4}$$



Fig. 9.4 (a) Configuration of the Kretschmann geometry; (b) the dispersion relation of photons in a coupling prism; (c) SPR curve for SF10 (n = 1.723)|gold(50 nm, n1 = 0.1726 + i3.4218)|air(n = 1.0) for collimated white light source (830 nm)

The matching condition for resonance is fulfilled when:

$$\frac{\omega}{c}\sin\theta + \frac{2\pi}{a} = \frac{\omega}{c}\sqrt{\frac{\varepsilon_1\varepsilon_2}{\varepsilon_1 + \varepsilon_2}}$$
(9.5)

The most commonly used methods for surface plasmon excitation exploit the attenuated total reflection (ATR) effect. The main experimental configurations are based on the Otto (Fig. 9.3b) [15] and the Kretschmann geometries (Fig. 9.4a) [16].

ATR configurations increase the wave vector of the radiation travelling through an optically denser medium (*e.g.* glass prism, dielectric constant ε_{pr}) by $\sqrt{\varepsilon_{pr}}$:

$$k_x = \frac{\omega}{c} \sqrt{\varepsilon_{pr}} \sin \theta \tag{9.6}$$

The matching condition is obtained when:

$$\frac{\omega}{c}\sqrt{\varepsilon_{pr}}\sin\theta = \frac{\omega}{c}\sqrt{\frac{\varepsilon_1\varepsilon_2}{\varepsilon_1 + \varepsilon_2}}$$
(9.7)

In the Otto configuration (Fig. 9.3b) the metal surface (ε_2) is separated from the prism (ε_0) by an additional dielectric layer having a dielectric constant value $\varepsilon_1 < \varepsilon_0$ and the SPs resonance occurs at the metal-dielectric interface.

The Kretschmann geometry (Fig. 9.4a) uses a prism (ε_0) with a thin (about 50 nm for gold) metallic layer (ε_1) on one side.

If the match condition (Eq. 9.7) is satisfied when a p-polarized light beam hits the prism on a side opposite to the thin metal layer a drop in intensity of reflected light is observed (Fig. 9.4c) [17]. Any change of the ε_2 value caused by chemical or physical effects modifies the matching condition (Eq. 9.7) and a shift in the energy position of the minimum of the reflected light (reflectance dip) is observed [18]. The SP wave is therefore highly sensitive to changes of the dielectric constant (simply related to the refractive index) of the medium at the metallic interface. Such a distinguishing

property is the basic principle which makes the surface plasmon resonance useful as biosensor [19].

Conditions for surface plasmon resonance are obtained in the infrared and visible wavelength region for air/metal and water/metal interfaces. Absorption in the metal layer influences significantly the width of the reflectance dip. Since the dissipation in metals is determined mainly by the imaginary part of the refractive index, one may choose a metal type with a low imaginary part to reduce the dissipation and, consequently, to obtain a narrower dip. Silver and gold are noble metals that satisfy this condition, while aluminum has a large imaginary part and exhibits wider dips compared to silver and gold.

Other points to be considered for the choice of a metal for SPR are: (i) the chemical purity of the surface: oxides and sulfides formed after the metal interaction with the atmosphere can interfere with the SPR phenomenon; (ii) the compatibility of the metal reactivity with the surface chemistry needed for the SPR experiment. Gold are the substrate of choice for SPR measurements. It is relatively stable in aqueous environments needed for the study of bimolecular interactions and a versatile chemistry-mainly based on the formation of functional self-assembled monolayers (SAMs)-has been developed for this metal.

A surface plasmon resonance biosensor is able to measure the binding between target analyte molecules and receptor molecules immobilized on the gold surface. During the receptor/analyte binding event, the shift of the dip in the spectrum of reflected light is monitored over time and kinetics information regarding biomolecules interactions are gathered.

SPR has the potential to investigate interactions between antigens and antibodies, nucleic acid sequences and their complementary strands, and substrates and enzymes [20] with no need for labeling of the interacting components. The time required for the analysis is typically in the range of a few minutes and the sample consumption is in the nanomole-picomole range.

The kinetic events at the metal surface, displayed as a sensorgram, can be investigated by monitoring the SPR signal change as a function of time. The time interval during which the analyte interacts with the surface bound receptor defines the "association phase" whereas the time interval following the analyte interaction is termed "dissociation phase". In the association phase there is a simultaneous association and dissociation equilibrium. A steady-state condition is reached when the association rate equals the dissociation rate. Under ideal experimental conditions, only dissociation should take place during the dissociation phase.

In order to perform replicate SPR experiments any bound analyte molecule should be removed with no disruption of the receptor structure and activity. So, a typical SPR experiment involves several steps displayed in Fig. 9.5.

The change in the refractive index produced by the capture of biomolecules depends on the concentration of the analyte molecules at the sensor surface. If the binding event occurs within a layer of thickness h, the SPR response is proportional to the binding induced refractive index change, which can be expressed as:



Fig. 9.5 Pictorial description for the receptor-analyte interaction: the analyte is captured by receptors immobilized on the sensor surface. A sensorgram with the steps of an analysis cycle is also shown

$$\Delta n = \left(\frac{dn}{dc}\right)\frac{\Gamma}{h} \tag{9.8}$$

where (dn/dc) denotes the refractive index increment of the analyte molecules and Γ denotes the surface concentration in mass/area. Consequently, quantitative measurements of biomolecules adsorption can be obtained with the SPR technique [21].

Different instrumental formats that can be used to carry out SPR experiments: angle modulation [22], wavelength modulation [23], and intensity modulation also known as SPR imaging [24].

The popularity of the scanning angle SPR, in which the reflectivity of the incident light upon a metal film is monitored as a function of the incident angle, can be attributed to the existence of Pharmacia Biosensor AB company, that in 1990 launched the first commercial SPR platform (Biacore). Both angle and wavelength modulated SPR sensors typically provide few data points at a time. In contrast, SPR imaging uses changes in reflectivity to generate simultaneously binding difference images in a parallel manner. In addition, SPR imaging offers advantages in freedom of design of the liquid handling system.

New aspects of SPR have been explored in recent years concerning the development of combined techniques (SPR-MS [25, 26], SPFS [27]), the use of new materials (glassified substrates [28], carbon-coated metal substrates [29]), and the development of new SPR-based assays (nanoparticle enhancement [30], enzymatic amplification [31]).



Fig. 9.6 Schematic of a SPFS set-up in the Kretschmann-configuration combined with elements for fluorescence detection (Reprinted with permission from Elsevier: Ref. [38. Copyright (2000))

9.3 Surface Plasmon Fluorescence Spectroscopy (SPFS)

Although SPR provides suitable sensitivity for many label-free applications, it suffers from reduced sensitivity for specific assays when compared to other labelled methods such as electrochemistry or fluorescence [32].

In 1991, Attridge et al. [33] described how the field enhancement from surfaceplasmon modes could be used for fluorescence spectroscopy, providing surfacesensitive analysis. However, only a limited number of applications were described before the Schmidt et al. [34] and Neumann et al. [35]. After 2005, the number of papers using SPFS significantly increased [36].

SPFS experiments can be performed simultaneously with SPR measurements. In a basic SPFS setup (Fig. 9.6) a photomultiplier tube (PMT) is placed perpendicular to the plane of the substrate with a suitable filter to reject stray light. A chopper is placed in front of a laser beam in order to prevent photobleaching of fluorophores while measurements are not being taken.

In normal SPR operation, the polarized light is reflected to the photodiode at angles greater than the critical angle for total internal reflection. If a chromophore is in the evanescent field near the metal–dielectric interface, large enhancement of fluorescence excitation can occur at the angle where plasmon resonance reaches its maximum. The resulting fluorescence emission can be measured with high sensitivity in the plane perpendicular to the substrate. Although only fluorophores within the evanescent field can be excited, their fluorescence signal can be significantly quenched by the close metal surface that acts as a very efficient quencher that absorbs the excitation energy of the fluorophore [37]. Therefore, the separation distance from the metal substrate, at which biomolecular interactions will occur, is a compromise between staying close enough to the peak evanescent field value and getting far enough from the gold surface to prevent de-excitation. An ideal distance has been demonstrated to exist at about 15–25 nm separation from the metallic surface. That distance reduces the optical field by only 10 %, while the fluorescence intensity is more than 80 % [38].

SPFS represents a powerful tool for DNA detection and for this reason extensive DNA hybridization studies have been carried out [39] with the aim to demonstrate the enhanced sensitivity in detecting PCR amplicons (detection limit 100 fmol) [40] and SNPs [41].

The combined use of SPFS and the electrochemical detection offers further possibilities in DNA detection that are discussed in Chap. 10.

9.4 Surface Plasmon Resonance Imaging (SPRI)

SPR imaging, also known as SPR microscopy, was first introduced by Rothenhäusler and Knoll in the 1988 [42]. Since then, SPRI has been proposed as a powerful technique to investigate interactions with biomolecules arrayed onto chemically modified gold surfaces [43]. SPRI combines typical advantages of the traditional SPR in terms of real time analysis, no label requirements and high sensitivity, with those associated to an imaging system such as the easy coupling with microfluidic devices, the high spatial resolution and the possibility to adopt an array-compatible detection approach.

Typical SPRI applications fall into the genomics and proteomics areas. Examples of genomic applications include DNA and RNA detection through the hybridization to RNA and DNA microarrays [44–47], and DNA-protein binding on DNA array [48, 49]. Proteomic applications include the determination of protein film thickness [50], the detection of conformational changes [51], and expression profiling [52]. SPRI has been also used to study kinetics of enzymatic reactions [53, 54] and to evaluate protein interactions with protein [55–57] and carbohydrate microarrays [58, 59].

SPRI measures spatial differences in the intensity of the reflected light (expressed as percent reflectivity %R) at a fixed angle and wavelength [60]. In particular, SPRI platforms based on a Kretschmann geometry use a p-polarized light beam to illuminate a prism/thin gold film assembly at a fixed incident angle. The intensity of the reflected light is detected with a 2D detector (CCD) to obtain an SPR image. The spatially resolved surface functionalization of the metallic SPRI surface allows to detect in real time parallel interaction events.

The spatial contrast in an SPR image comes from the heterogeneity in the dielectric constant due to differences in refractive index or film thickness near the surface at different positions across the surface. If an adsorbate has a refractive index (that is related to its dielectric constant) different from that of the solvent,



Fig. 9.7 Schematic diagram of SPR imaging

its presence on the surface can be detected with spatially resolution by monitoring changes in the reflected light intensity (Fig. 9.7) [61]. Difference images obtained by the subtraction of a reference image from a post-binding image provide visual confirmation of the binding events.

For small changes in the SPR angle, changes in reflected intensity are proportional to changes in the effective refractive index and effective thickness or surface coverage of an adlayer on the SPR-active surface [62]. The proportionality factor that relates the reflected intensity change to the change in refractive index can be determined by measuring SPRI intensity versus angle curves from different regions of an SPRI surface. Broad, nearly linear regions on either sides of the minimum are observed. These regions where the slope is large and remains constant are referred to as "linear regions" or "high-contrast angles" (Fig. 9.7).

The measured reflected intensity decreases or increases as the minimum shifts to higher angles when the incident angle is set to an angle in the linear region on the left or right side of the curve, respectively. The SPRI response is proportional to the change in bulk refractive index near the sensing surface. The relationship is linear for intensity changes of less than 20 % while the SPRI response becomes nonlinear for larger refractive index changes and the intensity eventually saturates.

The SPRI lateral resolution is limited by the propagation length l_d , and varies for different metals and different source wavelengths. For a 44-nm-thick gold layer, l_d decreases as the excitation wavelength decreases ranging from $l_d = 14 \ \mu m$ at

676 nm to $l_d = 0.5 \ \mu m$ at 530 nm [63]. The reduction in the lateral resolution is counterbalanced by the possibility to obtain images with an higher contrast when using near-IR sources [64]. Hence, the source wavelength for SPRI experiment should be optimized as to obtain the highest level of contrast by maintaining a sufficient lateral resolution. Typically, excitation wavelengths ranging from 800 to 1,000 nm provides lateral resolution of about 25 μm and are well suited for imaging an array with micrometric dimensions.

The SPRI system allows simultaneous and independent measurements to be made on different locations of the same sensor surface. In addition, the fact that the sensor surface can be visually observed in real time establishes the strength of SPRI, along with other advantages which are the inherent optical sensitivity and the instrumentation simplicity.

As stated above, there are a number of different reasons why a spatially resolved immobilization of biosystems is desirable:

- 1. kinetic curves can be monitored simultaneously;
- 2. when only a low analyte sample volume is available, its solution flows in contact with a number of sensor surface areas at the same time;
- 3. multi-analyte mixtures can be measured and pattern recognition becomes feasible;
- 4. reference and duplicate measurements can easily be included and employed to correct for non specific adsorptions and refractive index changes.

The SPRI experimental format increases the analysis throughput. However, to take fully advantage of this approach, a precise control of both the patterning of biomolecules onto the SPRI sensor surface as well as the fluidic of the analyte solution is imposed. In this perspective, the use of microfluidic devices provides an SPRI-compatible convenient mean for both manipulating small amounts of sample solutions as well as for controlling the patterning of a variety of different biomolecules.

Microscale conditions induce the laminar flow of fluids [65] and optimize the way in which liquids are put in contact to the SPRI sensor surface [66]. The use of microfluidic devices significantly reduces the experiment duration and the sample consumption. In addition, diffusion distances are reduced, thus allowing a better control over reaction conditions [67]. Finally, microfluidic platforms allows to manage the interaction experiments in order to decrease the influence of nonspecific interactions and instrument drifts by internal signal referencing [67–69].

9.5 Platforms and Experimental Designs for SPRI Assays

SPRI technology has greatly advanced over the last five years as a consequence of the development of new surface functionalization strategies and new instrumental designs.
Different probe immobilization procedures have been described [70] most of which are based on the use of SAM [71].

DNA arrays can be fabricated by spotting single-stranded DNA (ssDNA) probe solutions on functionally active SPRI sensor surfaces. The spotting procedure suffers from a not uniform probe immobilization and can be performed manually or by using automatic spotters. The manual spotting results in a not uniform spot size distribution and immobilized probe density. The quality of ssDNA spots has been improved by creating mixed ssDNA/oligo(ethylene glycol) (OEG) SAMs [72]. Alternatively, thiolated oligonucleotides can be spotted onto the gold substrate, and thiolated OEG can be used to functionalize the un-modified gold surface [73, 74]. OEG minimizes surface nonspecific adsorption and helps in controlling the ssDNA probe orientation on the surface. DNA arrays offer the possibility to carry out a parallel detection of nucleic acids and proteins by immobilizing antibodies conjugated to ssDNA sequences on DNA arrays carrying a complementary sequence.

The spatially controlled covalent immobilization of amine-terminated ssDNAs has been obtained through the amide bond formation with the NHS/EDC activated poly-L-glutamic acid (pGlu) previously electrostatically adsorbed onto an 11-amino-1-undecanethiol (MUAM) monolayer [75]. The surface architecture assured a high surface hybridization efficiency and minimized non-specific adsorption effects.

The surface chemistry developed for gold exploits the high stability of the goldsulfur bond. However, the susceptibility of the gold-sulfur bond to oxidation and photodecomposition has hampered SPR sensing uses in areas such as on-surface combinatorial chemistry and photolithography due to the adverse effects of the required chemical condition or ultraviolet radiation on the gold-sulfur bond [76].

Silicon oxide-based materials like glass are standard materials for biosensing. They are inexpensive and benefit the well-developed silane-coupling chemistry for surface functionalization. A number of functionalization strategies developed on glass have already been adapted to gold. However, resulting procedures are complex and require time-consuming syntheses and extensive manipulation of the existing protocols to be effective on the new surface. A simple way to facilitate procedures is obtained by coating the gold SPRI substrate with thin silicon oxide-based materials. Since the SPRI signal decays exponentially within about 200 nm from the metallic surface [77], the coating should be some nanometers thick in order to retain a high detection sensitivity. The potentials of a similar approach have been demonstrated by using glassified gold surfaces to demonstrate the detection of 10–100 attomoles of mesophilic DNA polymerase products by nanoparticle-enhanced SPRI [28]. The approach offers stable and reusable surfaces but is limited by the not fully controlled silane chemistry and by the easy hydrolytic cleavage of Si–O–Si bonds when exposed to harsh environments or to increased temperatures.

Carbon-based surfaces could overcome most of the above mentioned limitations. In this case the ligand immobilization to the surface is obtained after the deposition of carbon [78] or amorphous silicon-carbon alloys [79] on the metallic surface. The biomolecule immobilization is obtained by exploiting the well-developed and robust chemistry based on the attachment of alkene-containing molecules to the substrate through either UV radiation mediated formation of C-C bonds or Si-C bonds.

Thin layers of amorphous carbon (7.5 nm) can be also deposited by sputtering it onto the SPR gold film [29]. In this case, an alcohol-terminated surface is obtained after a plasma treatment and an UV photo-functionalization with 9-decene-1-ol of the hydrogen terminated carbon surface. A similar interface is useful for the development of SPRI-compatible procedures for the in situ fabrication of oligonucleotide arrays through the use of photochemically protected oligonucleotide building blocks [80]. Photochemical procedures are not compatible with the traditional thiol-based gold surface chemistry because it is affected by the extended exposure to ultraviolet light and oxidizing chemical conditions. Unfortunately, the above mentioned advantages for the in situ oligonucleotide synthesis are partially counterbalanced by a reduced SPRI sensitivity caused by both a decreased photon-plasmon coupling efficiency as well as a broadening of the SPR curve generated by the complex dielectric function of amorphous carbon.

Biointerfaces based on nanomaterials with large surface-to-volume ratio, facile surface modification and overall structural robustness are particularly suitable for the development of improved the DNA detection assays. In particular, nanostructure-based SPR sensors have drawn tremendous interest in recent years as a consequence of the associated enhanced detection sensitivity [81].

The use of nanostructured tags results in a dramatic enhancement of sensitivity due to three main factors: (i) an increase of the absolute mass in each binding event, (ii) an increase in the bulk refractive index of the analyte, and (iii) coupling between the localized surface plasmon (LSP) of metallic nanoparticles and SPR of the sensing film. The role of plasmonic coupling in sensitivity enhancement is still an open question.

LSPs differ from the above discussed extended SP, that are described as a longitudinal electromagnetic wave in a 2D electron gas that exists on the interface between metals and dielectrics, because they are excited in metallic structures with dimensions less than half the wavelength of the exciting electromagnetic wave [82].

Different nanostructures such as nanoparticles, nanowires and nanotubes deposited on the SPR metallic layer have been used in combination with the conventional Kretschmann configuration of SPRI platforms [83–85]. The use of surface relief nanostructures directly built on a metal film has been the topic of recent theoretical and experimental studies. These studies have contributed to demonstrate the enhanced detection of multiple targets achieved by using LSP resonance excited by metallic nanogratings [86]. The experimental results have also shown that the LSPR-based assay may be used for the detection of a variety of different interactions, envisaging the extended feasibility and applicability of these platforms for rapid and efficient screenings.

The widespread application of surface relief nanostructures for SPR experiments is still limited by the inherent uses of nanofabrication procedures such as electron beam lithography, focused ion-beam milling, or photolithography. Only recently, some soft-lithography-based techniques have been used in this context [87]. The future development of nanostructure-based SPR chips is closely related to the development of rapid and cost-effective nanofabrication methods. In this context, low-cost nanostructured plastic substrates can be used to enhance the sensitivity of conventional SPR imaging [88]. The field enhancement provided by the nanostructures is function of both the composition, dimension and geometry of the nanostructure and of the dielectric environment. It has been shown that the addition of a monolayer at the surface of the nanostructured substrates created a maximum electromagnetic field enhancement in the vicinity of the monolayer, which was larger than that experienced on a bare gold surface.

The integration of SPRI with array systems is quickly advancing. Whether it is made in situ under the microfluidic control or offline by using spotting technologies, there is a vast collection of investigated patterning technique. These procedures are aimed at obtaining surface micropatterning, multiplexed analyte detection and fluids control.

Besides the biointerface architecture, the SPRI biosensor performances are highly dependant on optical, electrical and structural features of the platform. An accurate control of those features have been shown to produce a refractive index resolution as low as 1×10^{-7} RIU (refractive index unit) [89].

Enhanced detection capabilities result also from the exploitation of long range SPR (LRSPR). LRSPR biosensors based on the Kretschmann configuration comprise four basic layers, namely the prism, the dielectric buffer layer, the metal layer, and the analyte. The dielectric buffer layer separates the thin metal layer from the prism. If the dielectric constant of the dielectric buffer layer is similar to that of the analyte, and is lower than that of the prism, a symmetric environment is achieved on either side of the metal thin film. If the thickness of the thin metal layer (d_m) is such that $k_{zm} d_m \ll 1$ (where k_{zm} is the z component of the SP wavevector in the metal layer) the resulting symmetric configuration causes the same surface plasmon wave to exist on both sides of the thin metal layer, and interaction then takes place between the two SP waves at both surfaces.

Compared to the conventional SPR biosensor, the propagation length of LRSPR is increased by a factor of approximately ten times and specific combinations of metal thickness and buffer layer thickness can be used to achieved an enhanced sensitivity compared to standard SPR [90].

LRSPR imaging has been obtained by using an inert and optically transparent fluoropolymer with a refractive index very close to that of water (Cytop) as the dielectric buffer layer [91]. In this case, compared with conventional SPR, LRSPR has longer propagation length and longer field penetration. The longer propagation length causes a reduced lateral resolution of the produced image. The LRSPR propagation length is of 200 μ m compared to 25 μ m measured for SPR modes at 814 nm [63]. Additionally, the penetration depth of the LRSPR mode into the bulk water phase is longer than the 200 nm of conventional SPR at the same wavelength. Potentials offered by LRSPR imaging have been demonstrated by detecting DNA hybridization with enhanced sensitivity compared to SPR imaging [91].

A different SPR imaging platform has been obtained by placing between two crossed polarizers a prism with an SPR chip containing a spatially patterned multilayer structure [92]. The output polarizer blocked the light reflected from the

SPR chip areas outside the sensing areas, generating high-contrast images. Two types of SPR multilayers with opposite sensitivities to changes of refractive index at the surface were employed. The output signal was defined as a ratio of the intensities generated from the two neighboring multilayers. The SPR imaging platform was shown to be able to detect refractive index changes down to 2×10^{-6} RIU and to detect short oligonucleotides at concentrations as low as 100 pM [93].

Grating couplers offer some attractive features for SPRI in comparison to prism couplers [7]. In fact, diffraction-based sensors discriminate against nonspecific adsorption processes to the entire substrate because only patterned bioaffinity adsorption will lead to changes in the diffraction signal. Moreover, in this case the operating wavelength can be readily tuned to a spectral region where the plasmon resonance enhancement generates a maximum optical effect.

Enhanced SPRI sensitivity has been obtained by using a diffraction grating made of Au stripes [94]. In this case, the detection of femtomolar levels of unmodified DNA in a sandwich assay using Au nanoparticles has been obtained. The method termed nanoparticle-enhanced diffraction grating (NEDG) is based on the enhanced diffraction obtained by the optical coupling of planar surface plasmons in the grating to LSPR in the gold nanoparticles. The nanoparticle-enhanced DNA detection has been also investigated by coupling the diffraction grating platform with an *ex situ* nanoparticle assay [95].

An alternative approach for the SPR detection is based on the measurement of the shift in phase of the light wave interacting with SPs at a single angle of incidence and wavelength of the light wave. In fact, p-polarized light shows a very sharp phase shift of 360° in the region of the SPR angle, relative to the phase shift at the critical angle. Changes in this phase shift upon adsorption of biomolecules on the metallic surface have been used in various optical configurations as a highly sensitive method of monitoring surface adsorption processes, with detection limits up to two orders of magnitude lower than amplitude detection approaches. SPR images based on the phase shift measurement have been acquired by using a variable retarder that creates a spatially periodic retardation pattern on the metal interface that is converted into a fringe pattern by a subsequent linear polarizer. The position of the interference fringes defines is related to the p-polarized light phase shift and is used to monitor any phase shift changes due to a molecular adsorption [96]. The SPR phase shift imaging method have been demonstrated to be able to detect ssDNA hybridization with a dynamic range that is comparable to conventional intensity based SPR techniques and a detection limit roughly 100 times lower than traditional intensity-based SPR imaging measurements.

9.6 Detection of Non Amplified Genomic DNA

The successful sequencing of the human genome [97] has improved our understanding of the complex network of gene expression [98] and has prompted towards the development of new methodologies for an even more simple, fast and cheap detection of nucleic acids to be used for both applicative and research needs [1, 99]. Most of the currently available nucleic acid detection methods require the amplification of the target species and detect the duplex formation by using labels and transducers able to generate signal as a consequence of the specific hybridization event. Both amplification and labelling processes are laborious, can introduce artefacts and may interfere with the hybridization between the complementary strands.

PCR is the most widely used method for nucleic acid amplification. It enhances both the specificity and the sensitivity of the assay by increasing the target concentration. A similar enhancement is balanced by the need to use complex reaction mixtures, by the potential contamination of the genetic sample [100] and by the difficulty in amplifying long target sequences (more than about 40,000 bases) [101]. In addition, sequences different from the target can be erroneously amplified thus producing non-specific amplifications [102].

The identification of single nucleotide polymorphisms (SNPs) [103–105] introduces further complexity in the detection process. In fact, the large number of known SNPs [106] easily allows to evaluate the resulting complexity of the PCR-based detection process since the amplification of each SNP carrying sequence imposes the identification and the synthesis of two different primers to be used to define the genomic DNA portion to be amplified.

The DNA detection is expected to be significantly improved by using simple and economic detection methods which require minimal DNA modifications and provide enhanced signal amplification. In this perspective, the direct detection of non amplified genomic DNA appears an excellent cost-effective alternative the PCR-based approach since extra labour and costs from the amplification procedure are reduced.

Unfortunately, the direct detection brings out constraints caused by the complexity of the genomic DNA that are normally minimized when using PCR amplified targets. The complexity mainly results from the presence of sequences nearly identical to the targeted one and the high melting temperature of the genomic DNA. The melting temperature of the short duplex formed after the probe targeting of a selected DNA sequence could be too low to provide for the stringency needed for a base-level discrimination in the presence of total genomic DNA by resulting in a lack of specificity. An efficient probe hybridization of the target sequence is possible when the latter is available as ssDNA. Simple thermal treatments are sufficient to give a significant amount of ssDNA. However, additional treatments are required when unamplified genomic DNA is going to be analyzed. In fact, a proper genomic DNA fragmentation is required in order to facilitate its detection. Sonication is the preferred method instead of using restriction enzymes in view of the greater randomness of the resulting fragmentation. Sonication is expected to cleaves DNA also according to its composition and sequence instead of only its size [107]. In contrast, the random action of restriction enzymes is compromised by their sequence dependence since their cleaving action is performed only in the presence of a specific base sequence.

9.7 SPR Methods for the Direct Detection of Genomic DNA

The SPR sensitivity for the direct detection of unamplified genomic DNA has been first evaluated by using plant, bovine and human genomic DNAs [108]. Before the SPR detection, unamplified genomic DNAs were fragmented with EcoRI or the BaMH 1 restriction enzymes which were used for bovine and human or plant DNAs respectively. The fragmentation was followed by a thermal denaturation process. In this case, the re-association of the denatured DNA was prevented by means of two blocking oligonucleotides that hybridized to the genomic DNA close to target-probe sequences thus preventing the re-association by steric hindrance. The SPR detection was obtained by immobilizing thiol-capped oligonucleotide probes on the sensor surface and by using only 25 μ l of 10 ppm solutions, containing about 10⁵ copies of maize or human targets.

The limitations suffered by SPR for the parallel detection of different probe/target interactions are overcome by SPR imaging [109]. The possibility to detect unamplified genomic DNA by using an SPRI-based multiplexed approach has been first demonstrated by detecting the testis-specific protein, Y-encoded, (TSPY) gene located in the Y chromosome of the human genomic DNA [110]. The detection approach was based on an RNA array and a surface enzymatic amplification procedure based on the use of the RNase H enzyme. RNase H specifically hydrolyzes RNA strands in RNA-DNA heteroduplexes. The heteroduplexes were formed on the SPRI sensor surface after the exposure of RNA probes to DNA targets. As a consequence of the enzymatic reaction ssDNA molecules released into the solution can bind to other RNA probes on the surface. New heteroduplexes formed on the SPRI surface are in turn subject to the hydrolytic RNase H action. The final process caused an SPRI signal amplification due to the fact that one target DNA molecule removed approximately 10⁴ RNA probe molecules from the surface. The approach produced a 7 fM sensitivity in detecting the TSPY gene from unamplified human genomic DNA.

The enhanced sensitivity required when unamplified genomic DNA is going to be detected has prompted efforts aimed at identifying new strategies for the amplification of signals generated by the transducer after the DNA detection. With this aim the use of nanoparticles has been widely investigated in combination with optical or electrochemical transducers (see Chaps. 3, 6 and 7). The nanoscale dimension of particles gives rise to a high reactivity and to chemically tailorable physical properties. Among the other, gold nanoparticles (AuNPs) have been so far the most versatile and extensively studied for enhanced DNA detection and have been the focus of research for many decades as a result of their intriguing electronic and optical properties [30].

The use of AuNPs for the enhanced SPRI detection of oligonucleotides was first proposed in 2000 by using oligonucleotide probes and a static assay [111]. The method showed a 10 pM sensitivity. A similar strategy, was more recently adopted in combination with the use of peptide nucleic acid (PNA) probes for the detection of DNA sequences and continuous-flow microfluidics [112]. The detection

of oligonucleotide targets was obtained by adopting a sandwich hybridization strategy using AuNPs conjugated to an oligonucleotide complementary to the final tract of the target, which was not involved in the hybridization with the SPRI sensor surface-immobilized PNA probe. The sandwich hybridization strategy allowed discrimination between fully matched and single-base-mismatched sequences, even at a concentration of 1 fM. In addition, the microfluidic management of the fluids allowed multiplexed determination of SPRI responses. The discrimination was obtained using 150 zeptomoles of the DNA target.

An optimized probe design and a proper surface architecture represent fundamental requirements for a selective and specific DNA detection. Probe design is usually optimized on the basis of sequence text analysis and nearest-neighbour and thermodynamic calculations [113] which are aimed at minimizing crosshybridization with homologous sequences and at achieving higher thermodynamic stability of the complex formed with the target sequence [114].

An SPRI study of a multispot DNA chip fabricated with mixed ssDNA/oligo (ethylene glycol) self-assembled monolayers have demonstrated the influence of hairpins and homodimers on the hybridization process. The presence of secondary structures generate interferences with the DNA hybridization reaction and decrease the efficiency of the detection process [115]. ΔG values are good indicators to predict the effect of secondary structures and reduce the extent of the hybridization between complementary ssDNA. Interestingly, an increase of the temperature to an appropriate value minimizes the influence of secondary structures and enhances the efficiency of the hybridization reaction.

The replacement of ssDNA probes with suitable synthetic probes such as PNAs has been proved to successfully increase the detection specificity and sensitivity. Unique structural and hybridization features of PNAs make them superior to DNA as sequence-specific hybridization probes. PNAs are DNA mimics in which the negatively phosphate deoxyribose backbone is replaced by a neutral N-(2-aminoethyl) glycine linkage. They interact with their complementary DNA sequence more quickly, with a stronger binding and more specifically than the analogous DNA (see also Sect. 4.2.2) [116]. These properties are largely caused by the absence of the coulombic repulsion established between the two negatively charged strands in DNA. PNA complexes are more thermally stable than DNA complexes and, by virtue of the nature of their backbone, less susceptible to biological degradation by nucleases, proteases, and peptidases [117]. Additionally, their interaction with the complementary strand is almost unaffected by ionic strength and pH changes.

The successful detection of oligonucleotides by using PNA probes has prompted their use for the direct detection of genomic DNA [112, 118–120]. In particular, the nanoparticle-enhanced SPRI approach has been also used to demonstrate the detection of unamplified genomic DNA samples carrying different amounts of genetically modified (GM) sequences (Roundup Ready soybean, RR) [119]. The described approach was able to selectively detect the GM target sequence down to zM concentrations in solutions containing GM and GM-free genomic DNA at aM concentrations, even in the presence of a large excess of non-complementary DNAs.



Fig. 9.8 Pictorial description of the nanoparticle-enhanced SPRI strategy used to detect the normal, heterozygous, and homozygous genomic DNAs. In order to simplify the pictorial representation only specifically adsorbed DNA is shown. Nonspecifically adsorbed DNA is also present on the surface and contributes to generate the SPRI-detected signal. (a) PNA probes complementary to the normal (PNA-N) and the mutated (PNA-M) DNA target sequence; (b–c) Capture of the unamplified normal DNA by the PNA-N probe; (d) Absorption of oligonucleotide-conjugated AuNPs; (e) Aggregation of AuNPs on the surface; (f) Time-dependent SPRI curves obtained after the adsorption of conjugated AuNPs (Reprinted with permission from American Chemical Society: Ref. [120. Copyright (2011))

A similar SPRI-based strategy using PNA probes and microfluidic control was used to detect a point mutation in non-amplified genomic DNA isolated from patients affected by an hereditary genetic disease (Fig. 9.8) [120]. Human genomic DNAs, not subject to costly, time-consuming, and prone to contamination PCR-based amplification procedures, obtained from both healthy individuals and homozygous or heterozygous patients affected by β -thalassemia were used for the experiments. The targeting of genomic DNA was made easier by sonication, generating fragments of approximately 400 bp that are of sufficient size to localize the fragment within the human genome with a good degree of statistical certainty [121], and by denaturation of fragments at high temperature followed by cooling on ice before the introduction of the denatured fragments into the SPRI microfluidics apparatus.

After the above described treatment each sample was fluxed into each of six microchannels of the SPRI fluidic system in order to allow the direct interaction of normal, homozygous and heterozygous genomic DNAs with the PNA probe complementary to the normal DNA sequence (PNA-N) and the PNA probe complementary to DNA sequence carrying the point mutation (PNA-M), respectively



Fig. 9.9 Time-dependent SPRI curves obtained after the adsorption of conjugated AuNPs on normal (**a**), heterozygous (**b**), and homozygous (**c**) DNAs previously adsorbed to the surface immobilized PNA-N and PNA-M probes. Solutions of 5 pg μ L⁻¹ genomic DNAs were used for the experiments. A representative SPR difference image (**d**) demonstrating the DNA parallel detection is also shown (Reprinted with permission from American Chemical Society: Ref. [120]. Copyright (2011))

(Fig. 9.8a, b). The two different PNA probes were useful both to discriminate between normal, homozygous, and heterozygous DNAs as well as to avoid the use of external controls which were difficult to be obtained for this specific application.

The specific SPRI response patterns obtained when normal, homozygous, or heterozygous DNAs were each allowed to interact with the two different PNA probes provided a robust experiment control (Fig. 9.9). In fact, whereas normal DNA was expected to interact with only the PNA-N probe, different interactions were expected from homozygous and heterozygous DNAs, i.e., interaction between homozygous DNA and only the PNA-M probe and interactions between heterozygous DNA and both PNA-N and PNA-M probes.

The ultrasensitive detection of the genomic DNA was achieved by using AuNPs conjugated to an 11-mer oligonucleotide complementary to an exposed tract of the target DNA not involved in the hybridization with the PNA probe (Fig. 9.8d). After the capture of unamplified genomic DNAs carrying the normal or the mutated gene sequences by the complementary PNA probe(s), the conjugated AuNPs were fluxed into the microchannels in order to enhance detectability. By using this approach, a significant increase of the detected SPRI signals was obtained (Fig. 9.8e, f).

Attomolar concentrations of target genomic DNA were detected, DNAs from healthy individuals and homozygous or heterozygous patients affected by β -thalassemia were discriminated, and only simple manipulations of the genetic samples were required before the analysis (Fig. 9.9).

The simple procedures requiring only sonication and denaturation, minimize the possibility to contaminate the DNA samples to be analyzed. The reduced number of genomic DNA equivalents required for the analysis (about 470) also allows to propose the method as a potential new tool for the analysis of rare cell populations. As an example, the proposed strategy can be used to obtain a non-invasive prenatal genetic diagnosis based on the analysis of circulating free fetal DNA in the blood of the pregnant woman. In particular, the ongoing lowering of costs for massively parallel analyses in a SPRI multiplex format might lead to replacement of most of the other currently used approaches which traditionally require invasive procedures such as amniocentesis, chorionic villous sampling or fetal blood sampling, each associated to a small but finite risk of fetal loss.

SPRI has been also successfully used for the ultrasensitive detection of microRNAs (miRNAs) sequences [122]. miRNAs are involved in the regulation of the gene expression in cell proliferation, fat metabolism, and cell differentiation [123]. A combination of a surface polyadenylation reaction carried out by using poly(A) polymerase enzymes and a gold nanoparticle-based amplification of the SPR signal was employed to detect miRNAs to a concentration of 10 fM. The multiplexed detection of miRNAs with the described methodology is particularly appealing since it can be used for miRNA profiling at attomole levels.

9.8 Conclusion

Enhanced nucleic acid detection imposes the development of even more simple, fast and cheap methods. Those methods are expected to avoid the nucleic acid amplification and labeling. Surface plasmon resonance methods have been shown to offer promising opportunities for the ultrasensitive and direct detection of non amplified DNA. In this perspective, further efforts are expected to be paid with the principal aim to demonstrate the fully applicability of SPR-based methods for the ultrasensitive and specific detection of nucleic acids in complex matrices.

References

- Sassolas, A., Leca-Bouvier, B.D., Blum, L.J.: DNA biosensors and microarrays. Chem. Rev. 108, 109–139 (2008)
- 2. Alkan, C., Coe, B.P., Eichler, E.E.: Applications of next-generation sequencing genome structural variation discovery and genotyping. Nat. Rev. Gen. **12**, 363–375 (2011)
- Schaaf, C.P., Wiszniewska, J., Beaudet, A.L.: Copy number and SNP arrays in clinical diagnostics. Annu. Rev. Gen Hum. Genet. 12, 25–51 (2011)

- 4. Germini, A., Rossi, S., Zanetti, A., Corradini, R., Fogher, C., Marchelli, R.: Development of a PNA array platform for the detection of GMOs in food. J. Agric. Food Chem. **53**, 3958–3962 (2005)
- Roh, S.W., Abell, G.C.J., Kim, K.H., Nam, Y.D., Bae, J.W.: Comparing microarrays and next-generation sequencing technologies for microbial ecology research. Trends Biotechnol. 28, 291–299 (2011)
- Gong, P., Levicky, R.: DNA surface hybridization regimes. Proc. Natl. Acad. Sci. USA 105, 5301–5306 (2008)
- Homola, J.: Surface Plasmon Resonance Based Sensors. Springer Series on Chemical Sensors and Biosensors. Springer, Berlin/Heidelberg (2006)
- Wood, R.W.: Diffraction gratings with controlled groove form and abnormal distribution of intensity. Philosophical Magazine Series 6 23, 310–317 (1912)
- 9. Schasfoort, R.B.M., Tudos, A.J.: Handbook of Surface Plasmon Resonance. The Royal Society of Chemistry, Cambridge (2008)
- 10. Born, M., Wolf, E.: Principle of Optics. Pergamon Press, Oxford (1980)
- 11. Raether, H.: Surface Plasmons on Smooth and Rough Surfaces and on Gratings. Springer, Berlin (1988)
- 12. Forstamann, F., Gerhardts, R.R: Metal Optics Near the Plasma Frequency, vol. 109 of Springer Tracts in Modern Physics. Springer, Berlin (1986)
- 13. Powell, C.J., Swan, J.B.: Effect of oxidation on the characteristic loss spectra of aluminum and magnesium. Phys. Rev. **118**, 640–643 (1960)
- 14. Agranovich, V.M., Mills, D.L.: Surface Polaritons. North Holland Co., Amsterdam (1982)
- Otto, A.: A new method for exciting nonradioactive surface plasma oscillations. Phys. Stat. Sol. 26, K99–K101 (1968)
- Kretschmann, E., Reather, H.Z.: Radiative decay of non-radiative surface plasmons excited by light. Naturf. 23A, 2135–2136 (1968)
- 17. Matsubara, K., Rawata, S., Minami, S.: Multilayer system for a high-precision surface plasmon resonance sensor. Opt. Lett. 15, 75–77 (1990)
- Akimoto, T., Sasaki, S., Ikebukuro, K., Karube, I.: Effect of incident angle of light on sensitivity and detection limit for layers of antibody with surface plasmon resonance spectroscopy. Biosens. Bioelectron. 15, 355–362 (2000)
- Rich, R.L., Myszka, D.G.: Advances in surface plasmon resonance biosensor analysis. Curr. Opin. Biotechnol. 11, 54–61 (2000)
- Piliarik, M., Vaisocherová, H., Homola, J.: Surface plasmon resonance biosensing. Meth. Mol. Biol. 503, 65–88 (2008)
- Tumolo, T., Angnes, L., Baptista, M.S.: Determination of the refractive index increment (dn/dc) of molecule and macromolecule solutions by surface plasmon resonance. Anal. Biochem. 333, 273–279 (2004)
- Lyon, L.A., Musick, M.D., Natan, M.J.: Colloidal Au-enhanced surface plasmon resonance immunosensing. Anal. Chem. 70, 5177–5183 (1998)
- Frutos, A.G., Weibel, S.C., Corn, R.M.: Near-infrared surface plasmon resonance measurements of ultrathin films. 2. Fourier transform SPR spectroscopy. Anal. Chem. 71, 3935–3940 (1999)
- 24. Hickel, W., Kamp, D., Knoll, W.: Surface-plasmon microscopy. Nature 339, 186 (1989)
- Grasso, G., D'Agata, R., Rizzarelli, E., Spoto, G., D'Andrea, L., Pedone, C., Picardi, A., Romanelli, A., Fragai, M., Yeo, K.J.: Activity of anchored human matrix metalloproteinase-12 catalytic domain on Au (111) surfaces monitored by ESI-MS. J. Mass Spectrom. 40, 1565–1571 (2005)
- Grasso, G., Bush, A.I., D'Agata, R., Rizzarelli, E., Spoto, G.: Enzyme solid-state support assays: a surface plasmon resonance and mass spectrometry coupled study of immobilized insulin degrading enzyme. Eur. Biophys. J. 38, 407–414 (2009)
- Chu, L.Q., Foerch, R., Knoll, W.: Surface-plasmon-enhanced fluorescence spectroscopy for DNA detection using fluorescently labeled PNA as "DNA indicator". Angew. Chem. 119, 5032–5035 (2007)

- Gifford, L.K., Sendroiu, I.E., Corn, R.M., Luptaka, A.: Attomole detection of mesophilic DNA polymerase products by nanoparticle-enhanced surface plasmon resonance imaging on glassified gold surfaces. J. Am. Chem. Soc. 132, 9265–9267 (2010)
- Lockett, M.R., Weibel, S.C., Phillips, M.F., Shortreed, M.R., Sun, B., Corn, R.M., Hamers, R.J., Cerrina, F., Smith, L.M.: Carbon-on-metal films for surface plasmon resonance detection of DNA arrays. J. Am. Chem. Soc. 130, 8611–8613 (2008)
- Zanoli, L.M., D'Agata, R., Spoto, G.: Functionalized gold nanoparticles for ultrasensitive DNA detection. Anal. Bioanal. Chem. 402, 1759–1771 (2012)
- Lee, H.J., Wark, A.W., Corn, R.M.: Creating advanced multifunctional biosensors with surface enzymatic transformations. Langmuir 22, 5241–5250 (2006)
- Phillips, K.S., Cheng, Q.: Recent advances in surface plasmon resonance based techniques for bioanalysis. Anal. Bioanal. Chem. 387, 1831–1840 (2007)
- Attridge, J.W., Daniels, P.B., Deacon, J.K., Robinson, G.A., Davidson, G.P.: Sensitivity enhancement of optical immunosensors by the use of a surface plasmon resonance fluoroimmunoassay. Biosens. Bioelectron. 6, 201–214 (1991)
- 34. Schmidt, E.K., Liebermann, T., Kreiter, M., Jonczyk, A., Naumann, R., Offenhausser, A., Neumann, E., Kukol, A., Maelicke, A., Knoll, W.: Incorporation of the acetylcholine receptor dimer from Torpedo Californica in a peptide supported lipid membrane investigated by surface plasmon and fluorescence spectroscopy. Biosens. Bioelectron. 13, 585–591 (1998)
- Neumann, T., Johansson, M., Kambhampati, D., Knoll, W.: Surface–plasmon fluorescence spectroscopy. Adv. Funct. Mater. 12, 575–586 (2002)
- Dostálek, J., Knoll, W.: Biosensors based on surface plasmon-enhanced fluorescence spectroscopy. Biointerphases 3, FD12–FD22 (2008)
- Ekgasit, S., Yu, F., Knoll, W.: Fluorescence intensity in surface-plasmon field-enhanced fluorescence spectroscopy. Sens. Actuators B 104, 294–301 (2005)
- Liebermann, T., Knoll, W.: Surface-plasmon field-enhanced fluorescence spectroscopy. Colloids Surf. A 171, 115–130 (2000)
- Kambhampati, D., Nielsen, P.E., Knoll, W.: Investigating the kinetics of DNA-DNA and PNA-DNA interactions using surface plasmon resonance-enhanced fluorescence spectroscopy. Biosens. Bioelectron. 16, 1109–1118 (2001)
- Yao, D.F., Yu, F., Kim, J.Y., Scholz, J., Nielsen, P.E., Sinner, E.K., Knoll, W.: Surface plasmon field-enhanced fluorescence spectroscopy in PCR product analysis by peptide nucleic acid probes. Nucl. Acids Res. 32, 177–192 (2004)
- Park, H., Germini, A., Sforza, S., Corradini, R., Marchelli, R., Knoll, W.: Kinetic and affinity analyses of hybridization reactions between peptide nucleic acid probes and DNA targets using surface plasmon field-enhanced fluorescence spectroscopy. Biointerphases 1, 113–122 (2006)
- 42. Rothenhäusler, B., Knoll, W.: Surface-plasmon microscopy. Nature 332, 615-617 (1988)
- Smith, E.A., Corn, R.M.: Surface plasmon resonance imaging as a tool to monitor biomolecular interactions in an array based format. Appl. Spect. 57, 320–332 (2003)
- 44. Lee, H.J., Wark, A.W., Li, Y., Corn, R.M.: Fabricating RNA microarrays with RNA-DNA surface ligation chemistry. Anal. Chem. **77**, 7832–7837 (2005)
- 45. Lee, H.J., Li, Y., Wark, A.W., Corn, R.M.: Enzymatically amplified surface plasmon resonance imaging detection of DNA by exonuclease III digestion of DNA microarrays. Anal. Chem. 77, 5096–5100 (2005)
- 46. Zhou, W.J., Chen, Y., Corn, R.M.: Ultrasensitive microarray detection of short RNA sequences with enzymatically modified nanoparticles and surface plasmon resonance imaging measurements. Anal. Chem. 83, 3897–3902 (2011)
- Seefeld, T.H., Zhou, W.J., Corn, R.M.: Rapid microarray detection of DNA and proteins in microliter volumes with surface plasmon resonance imaging measurements. Langmuir 27, 6534–6540 (2011)
- 48. Kyo, M., Yamamoto, T., Motohashi, H., Kamiya, T., Kuroita, T., Tanaka, T., Engel, J.D., Kawakami, B., Yamamoto, M.: Evaluation of MafG interaction with Maf recognition element arrays by surface plasmon resonance imaging technique. Genes Cells 9, 153–164 (2004)

- Li, Y., Lee, H.J., Corn, R.M.: Detection of protein biomarkers using RNA aptamer microarrays and enzymatically amplified SPR imaging. Anal. Chem. 79, 1082–1088 (2007)
- Otsuki, S., Tamada, K., Wakida, S.: Wavelength-scanning surface plasmon resonance imaging. Appl. Opt. 44, 3468–3472 (2005)
- 51. Kim, M., Jung, S.O., Park, K., Jeong, E.J., Joung, H.A., Kim, T.H., Seol, D.W., Chung, B.H.: Detection of Bax protein conformational change using a surface plasmon resonance imagingbased antibody chip. Biochem. Biophys. Res. Commun. **338**, 1834–1838 (2005)
- Usui-Aoki, K., Shimada, K., Nagano, M., Kawai, M., Koga, H.: A novel approach to protein expression profiling using antibody microarrays combined with surface plasmon resonance technology. Proteomics 5, 2396–2401 (2005)
- Lee, H.J., Wark, A.W., Goodrich, T.T., Fang, S., Corn, R.M.: Surface enzyme kinetics for biopolymer microarrays: a combination of Langmuir and Michaelis-Menten concepts. Langmuir 21, 4050–4057 (2005)
- Wegner, G.J., Wark, A.W., Lee, H.J., Codner, E., Saeki, T., Fang, S., Corn, R.M.: Real-time surface plasmon resonance imaging measurements for the multiplexed determination of protein adsorption/desorption kinetics and surface enzymatic reactions on peptide microarrays. Anal. Chem. **76**, 5677–5684 (2004)
- 55. (a) Yu, X., Xu, D., Cheng, Q.: Label-free detection methods for protein microarrays. Proteomics 6, 5493–5503 (2006); (b) Huang, H., Chen, Y.: Label-free reading of microarraybased proteins with high throughput surface plasmon resonance imaging. Biosens. Bioelect. 22, 644–648 (2006)
- 56. Huang, H., Chen, Y.: Label-free reading of microarray-based proteins with high throughput surface plasmon resonance imaging. Biosens. Bioelect. **22**, 644–648 (2006)
- Kanda, V., Kariuki, J.K., Harrison, D.J., McDermott, M.T.: Label-free reading of microarraybased immunoassays with surface plasmon resonance imaging. Anal. Chem. 76, 7257–7262 (2004)
- Karamanska, R., Clarke, J., Blixt, O., MacRae, J.I., Zhang, J.Q., Crocker, P.R., Laurent, N., Wright, A., Flitsch, S.L., Russell, D.A., Field, R.A.: Surface plasmon resonance imaging for real-time, label-free analysis of protein interactions with carbohydrate microarrays. Glycoconj. J. 25, 69–74 (2008)
- D'Agata, R., Grasso, G., Iacono, G., Spoto, G., Vecchio, G.: Lectin recognition of a new SOD mimic bioconjugate studied with surface plasmon resonance imaging. Org. Biomol. Chem. 4, 610–612 (2006)
- Brockman, J.M., Nelson, B.P., Corn, R.M.: Surface plasmon resonance imaging measurements of ultrathin organic films. Annu. Rev. Phys. Chem. 51, 41–63 (2000)
- 61. Steiner, G.: Surface plasmon resonance imaging. Anal. Bioanal. Chem. 379, 328-331 (2004)
- Shumaker-Parry, J.S., Campbell, C.T.: Quantitative methods for spatially-resolved adsorption/desorption measurements in real time by SPR microscopy. Anal. Chem. 76, 907–917 (2004)
- Berger, C.E.H., Kooyman, R.P.H., Greve, J.: Resolution in surface plasmon microscopy. Rev. Sci. Instrum. 65, 2829–2836 (1994)
- 64. Yeatman, E.M.: Resolution and sensitivity in surface plasmon microscopy and sensing. Biosens. Bioelectron. 11, 635–649 (1996)
- 65. Squires, T.M., Quake, S.R.: Microfluidics: fluid physics at the nanoliter scale. Rev. Mod. Phys. **77**, 977–1026 (2005)
- 66. D'Agata, R., Grasso, G., Spoto, G.: Real-time binding kinetics monitored with surface plasmon resonance imaging in a diffusion-free environment. Open Spectrosc. J. 2, 1–9 (2008)
- 67. Ouellet, E., Lausted, C., Lin, T., Wei, C., Yang, T., Hood, L., Lagally, E.T.: Parallel microfluidic surface plasmon resonance imaging arrays. Lab. Chip. **10**, 581–588 (2010)
- Grasso, G., D'Agata, R., Zanoli, L., Spoto, G.: Microfluidic networks for surface plasmon resonance imaging real-time kinetics experiments. Microchem. J. 93, 82–86 (2009)
- Amarie, D., Alileche, A., Dragnea, B., Glazier, J.A.: Microfluidic devices integrating microcavity surface-plasmon-resonance sensors: glucose oxidase binding-activity detection. Anal. Chem. 82, 343–352 (2010)

- Mannelli, I., Minunni, M., Tombelli, S., Wang, R., Spiriti, M.M., Mascini, M.: Direct immobilisation of DNA probes for the development of affinity biosensors. Bioelectrochemistry 66, 129–138 (2005)
- Love, J.C., Estroff, L.A., Kriebel, J.K., Nuzzo, R.G., Whitesides, G.M.: Self-assembled monolayers of thiolates on metals as a form of nanotechnology. Chem. Rev. 105, 1103–1169 (2005)
- Ladd, J., Taylor, A.D., Piliarik, M., Homola, J., Jiang, S.: Hybrid surface platform for the simultaneous detection of proteins and DNAs using a surface plasmon resonance imaging sensor. Anal. Chem. 80, 4231–4236 (2008)
- Opdahl, A., Petrovykh, D.Y., Suda, H.K., Tarlov, M.J., Whitman, L.J.: Independent control of grafting density and conformation of single-stranded DNA brushes. Proc. Natl. Acad. Sci. USA 104, 9–14 (2007)
- 74. Gong, P., Lee, C.Y., Gamble, L.J., Castner, D.G., Grainger, D.W.: Hybridization behavior of mixed DNA/alkylthiol monolayers on gold: characterization by surface plasmon resonance and 32P radiometric assay. Anal. Chem. **78**, 3326–3334 (2006)
- Chen, Y., Nguyen, A., Niu, L., Corn, R.M.: Fabrication of DNA microarrays with poly (L-glutamic acid) monolayers on gold substrates for SPR imaging measurements. Langmuir 25, 5054–5060 (2009)
- Ulman, A.: An Introduction to Ultrathin Organic Films: From Langmuir- Blodgett to Selfassembly. Academic, Boston (1991)
- Knoll, W.: Interfaces and thin films as seen by bound electromagnetic waves. Annu. Rev. Phys. Chem. 49, 569–638 (1998)
- 78. Lockett, M.R., Smith, E.A.: Fabrication and characterization of DNA arrays prepared on carbon-on-metal substrates. Anal. Chem. **81**, 6429–6437 (2009)
- Touahir, L., Niedziolka-Jonsson, J., Galopin, E., Boukherroub, R., Gouget-Laemmel, A.C., Solomon, I., Petukhov, M., Chazalviel, J.N., Ozanam, F., Szunerits, S.: Surface plasmon resonance on gold and silver films coated with thin layers of amorphous silicon-carbon alloys. Langmuir 26, 6058–6065 (2010)
- Singh-Gasson, S., Green, R.D., Yue, Y.J., Nelson, C., Blattner, F., Sussman, M.R., Cerrina, F.: Maskless fabrication of light-directed oligonucleotide microarrays using a digital micromirror array. Nat. Biotechnol. 17, 974–978 (1999)
- Stewart, M.E., Anderton, C.R., Thompson, L.B., Maria, J., Gray, S.K., Rogers, J.A., Nuzzo, R.G.: Nanostructured plasmonic sensors. Chem. Rev. 108, 494–521 (2008)
- Shalabney, A., Abdulhalim, I.: Sensitivity-enhancement methods for surface plasmon sensors. Laser Photonics Rev. 5, 571–606 (2011)
- Malic, L., Cui, B., Veres, T., Tabrizian, M.: Enhanced surface plasmon resonance imaging detection of DNA hybridization on periodic gold nanoposts. Opt. Lett. 32, 3092–3094 (2007)
- Byun, K.M., Shuler, M.L., Kim, S.J., Yoon, S.J., Kim, D.: Sensitivity enhancement of surface plasmon resonance imaging using periodic metallic nanowires. J. Lightwave Technol. 26, 1472–1478 (2008)
- Hoa, X.D., Kirk, A.G., Tabrizian, M.: Enhanced SPR response from patterned immobilization of surface bioreceptors on nano-gratings. Biosens. Bioelectron. 24, 3043–3048 (2009)
- Kim, K., Kim, D.J., Moon, S., Kim, D., Byun, K.M.: Localized surface plasmon resonance detection of layered biointeractions on metallic subwavelength nanogratings. Nanotech (2009). doi:10.1088/0957-4484/20/31/315501
- Henzie, J., Lee, J., Lee, M.H., Hasan, W., Odom, T.W.: Nanofabrication of plasmonic structures. Annu. Rev. Phys. Chem. 60, 147–165 (2009)
- Malic, L., Cui, B., Tabrizian, M., Veres, T.: Nanoimprinted plastic substrates for enhanced surface plasmon resonance imaging detection. Opt. Express 17, 20386–20392 (2009)
- Piliarik, M., Homola, J.: Surface plasmon resonance (SPR) sensors: approaching their limits? Opt. Express 17, 16505–16517 (2009)
- Nenninger, G.G., Tobiška, P., Homola, J., Yee, S.S.: Long-range surface plasmons for highresolution surface plasmon resonance sensors. Sens. Actuators B 74, 145–151 (2001)

- Wark, A.W., Lee, H.J., Corn, R.M.: Long-range surface plasmon resonance imaging for bioaffinity sensors. Anal. Chem. 77, 3904–3907 (2005)
- Piliarik, M., Vaisocherova, H., Homola, J.: A new surface plasmon resonance sensor for highthroughput screening applications. Biosens. Bioelectron. 20, 2104–2110 (2005)
- Piliarik, M., Vaisocherova, H., Homola, J.: Towards parallelized surface plasmon resonance sensor platform for sensitive detection of oligonucleotides. Sens. Actuat. B 121, 187–193 (2007)
- Wark, A.W., Lee, H.J., Qavi, A.J., Corn, R.M.: Nanoparticle-enhanced diffraction gratings for ultrasensitive surface plasmon biosensing. Anal. Chem. 79, 6697–6701 (2007)
- Sendroiu, I.E., Corn, R.M.: Nanoparticle diffraction gratings for DNA detection on photopatterned glass substrates. Biointerphases 3, FD23–FD29 (2008)
- Halpern, A.R., Chen, Y., Corn, R.M., Kim, D.: Surface plasmon resonance phase imaging measurements of patterned monolayers and DNA adsorption onto microarrays. Anal. Chem. 83, 2801–2806 (2011)
- 97. Venter, J.C., Adams, M.D., Myers, E.W., Li, P.W., Mural, R.J., Sutton, G.G., Smith, H.O., Yandell, M., Evans, C.A., Holt, R.A., et al.: The sequence of the human genome. Science 291, 1304–1351 (2001)
- Pennisi, E.: Human genome. Reaching their goal early, sequencing labs celebrate. Science 300, 409 (2003)
- Gibson, G.: The environmental contribution to gene expression profiles. Nat. Rev. Genet. 9, 575–581 (2008)
- 100. Shi, L., Perkins, R.G., Fang, H., Tong, W.: Reproducible and reliable microarray results through quality control: good laboratory proficiency and appropriate data analysis practices are essential. Curr. Opin. Biotechnol. 19, 10–18 (2008)
- 101. Barnes, W.M.: PCR amplification of up to 35-kb DNA with high fidelity and high yield from lambda bacteriophage templates. Proc. Natl. Acad. Sci. USA **91**, 2216–2220 (1994)
- 102. D'Aquila, R.T., Bechtel, L.J., Videler, J.A., Eron, J.J., Gorczyca, P., Kaplan, J.C.: Maximizing sensitivity and specificity of PCR by preamplification heating. Nucleic Acids Res. 19, 3749 (1991)
- Sachidanandam, R.: A map of human genome sequence variation containing 1.42 million single nucleotide polymorphisms. Nature 409, 928–933 (2001)
- 104. Kruglyak, L.: The use of a genetic map of biallelic markers in linkage studies. Nat. Genet. 17, 21–24 (1997)
- White, K.P.: Functional genomics and the study of development, variation and evolution. Nat. Rev. Genet. 2, 528–537 (2001)
- 106. Murray, S.S., Oliphant, A., Shen, R., McBride, C., Steeke, R.J., Shannon, S.G., Rubano, T., Kermani, B.G., Fan, J.B., Chee, M.S., Hansen, M.S.T.: A highly informative SNP linkage panel for human genetic studies. Nat. Methods 1, 113–117 (2004)
- 107. Lerman, L.S., Fisher, S.G., Hurley, I., Silverstein, K., Lumelsky, N.: Sequence-determined DNA separations. Annu. Rev. Biophys. **13**, 399–433 (1984)
- Minunni, M., Tombelli, S., Mascini, M.: A biosensor approach for DNA sequences detection in non-amplified genomic DNA. Anal. Lett. 40, 1360–1370 (2007)
- Scarano, S., Mascini, M., Turner, A.P.F., Minunni, M.: Surface plasmon resonance imaging for affinity-based biosensors. Biosens. Bioelect. 25, 957–966 (2010)
- Goodrich, T.T., Lee, H.J., Corn, R.M.: Direct detection of genomic DNA by enzymatically amplified SPR imaging measurements of RNA microarrays. J. Am. Chem. Soc. 126, 4086– 4087 (2004)
- 111. He, L., Musick, M.D., Nicewarner, S.R., Salinas, F.G., Benkovic, S.J., Natan, M.J., Keating, C.D.: Colloidal Au-enhanced surface plasmon resonance for ultrasensitive detection of DNA hybridization. J. Am. Chem. Soc. **122**, 9071–9077 (2000)
- 112. D'Agata, R., Corradini, R., Grasso, G., Marchelli, R., Spoto, G.: Ultrasensitive detection of DNA by PNA and nanoparticle-enhanced surface plasmon resonance imaging. Chembiochem 9, 2067–2070 (2008)

- 113. Mei, R., Hubbell, E., Bekiranov, S., Mittmann, M., Christians, R.C., Shen, M., Lu, G., Fang, J., Liu, W., Ryder, T., Kaplan, P., Kulp, D., Webster, T.A.: Probe selection for high-density oligonucleotide arrays. Proc. Natl. Acad. Sci. USA 100, 11237–11242 (2003)
- 114. Ermini, M.L., Scarano, S., Bini, R., Banchelli, M., Berti, D., Mascini, M., Minunni, M.: A rational approach in probe design for nucleic acid-based biosensing. Biosens. Bioelectron. 26, 4785–4790 (2011)
- 115. Chen, W.Y., Hu, W.P., Su, Y.D., Taylor, A., Jiang, S., Chang, G.L.: A multispot DNA chip fabricated with mixed ssDNA/oligo (ethylene glycol) self-assembled monolayers for detecting the effect of secondary structures on hybridization by SPR imaging. Sens. Actuat. B 125, 607–614 (2007)
- 116. Egholm, M., Buchardt, O., Christensen, L., Behrens, C., Freier, S.M., Driver, D.A., Berg, R.H., Kim, S.K., Norden, B., Nielsen, P.E.: PNA hybridizes to complementary oligonucleotides obeying the Watson–Crick hydrogen-bonding rules. Nature **365**, 566–568 (1993)
- Nielsen, P.E.: A brief review of some possible applications of PNAs. Curr. Opin. Biotechnol. 10, 71–75 (1999)
- 118. Zanoli, L., D'Agata, R., Spoto, G.: Surface plasmon-based optical detection of DNA by peptide nucleic acids. Minerva Biotec. 20, 165–174 (2008)
- 119. D'Agata, R., Corradini, R., Ferretti, C., Zanoli, L., Gatti, M., Marchelli, R., Spoto, G.: Ultrasensitive detection of non-amplified genomic DNA by nanoparticle-enhanced surface plasmon resonance imaging. Biosens. Bioelectron. 25, 2095–2100 (2010)
- 120. D'Agata, R., Breveglieri, G., Zanoli, L.M., Borgatti, M., Spoto, G., Gambari, R.: Direct detection of point mutations in nonamplified human genomic DNA. Anal. Chem. 83, 8711–8717 (2011)
- 121. Mann, T.L., Krull, U.J.: The application of ultrasound as a rapid method to provide DNA fragments suitable for detection by DNA biosensor. Biosens. Bioelectron. **20**, 945–955 (2004)
- 122. Fang, S., Lee, H.J., Wark, A.W., Corn, R.M.: Attomole microarray detection of MicroRNAs by nanoparticle-amplified SPR imaging measurements of surface polyadenylation reactions. J. Am. Chem. Soc. **128**, 14044–14046 (2006)
- 123. Wark, A.W., Lee, H.J., Corn, R.M.: Multiplexed detection methods for profiling MicroRNA expression in biological samples. Angew. Chem. Int. Ed. 47, 644–652 (2008)

Chapter 10 Parallel Optical and Electrochemical DNA Detection

Wolfgang Knoll, Jianyun Liu, Lifang Niu, Peter Eigil Nielsen, and Louis Tiefenauer

Abstract This contribution introduces strategies for the sensitive detection of oligonucleotides as bio-analytes binding from solution to a variety of probe architectures assembled at the (Au-) sensor surface. Detection principles based on surface plasmon optics and electrochemical techniques are compared. In particular, cyclicand square wave voltammetry (SWV) are applied for the read-out of ferrocene redox labels conjugated to streptavidin that binds to the (biotinylated) DNA targets after hybridizing to the interfacial probe matrix of either DNA or peptide nucleic acid (PNA) strands. By employing streptavidin modified with fluorophores the identical sensor architecture can be used for the recording of hybridization reactions by surface plasmon fluorescence spectroscopy (SPFS). The Langmuir isotherms determined by both techniques, i.e., by SWV and SPFS, give virtually identical affinity constants K_A, confirming that the mode of detection has no influence on the hybridization reaction. By using semiconducting nanoparticles as luminescence labels that can be tuned in their bandgap energies over a wide range of emission wavelengths surface plasmon fluorescence microscopy allows for the parallel readout of multiple analyte binding events simultaneously.

W. Knoll (🖂)

J. Liu • L. Niu

P.E. Nielsen

L. Tiefenauer

AIT Austrian Institute of Technology, Donau City Strasse 1, 1220 Vienna, Austria e-mail: Wolfgang.Knoll@ait.ac.at

Max Planck Institute for Polymer Research, Ackermannweg 10, D-55128 Mainz, Germany

Department of Biochemistry B, The Panum Institute, IMBG, Blegdamsvej 3c, DK-2200 Copenhagen, Denmark

Department of Chemistry, Life Sciences Department, Paul Scherrer Institute, CH-5232 Villingen, PSI, Switzerland

G. Spoto and R. Corradini (eds.), *Detection of Non-Amplified Genomic DNA*, Soft and Biological Matter, DOI 10.1007/978-94-007-1226-3_10, © Springer Science+Business Media Dordrecht 2012

10.1 Introduction

The question as to which detection method – electrical/electrochemical or optical schemes – eventually will win the race for the most powerful transducer principle in bio-sensing is still completely open. The option for higher integration density on chips or in any other array format and the ease of coupling the output signals directly to data analysis electronics certainly points to electrical detection schemes, however, optics is still leading in terms of sensitivity: fluorescence spectroscopy and, in particular, single photon detection are routine methods with unmatched detection limits while the well established electrochemical techniques do not allow for the monitoring of such extremely low currents. And single electron (transistor) recordings are yet to be combined with the aqueous environment needed for bio-affinity studies [1].

A very reasonable experimental approach to further tackle this question is the parallel use of an optical scheme and an electrochemical technique for the simultaneous recording of binding events of bio-analytes from solution to interaction partners covalently attached to the sensor surface via functional architectures assembled at the transducer/analyte solution interface. Surface plasmon resonance spectroscopy [2] and cyclic or differential pulse voltammetry [3] are very well suited for such a combination approach: the (noble) metal substrate needed for the resonant excitation of a propagating surface plasmon mode at the metal/buffer interface can be simultaneously used as the working electrode of a regular three electrode electrochemical set-up. This way, bio-affinity studies can be performed with both techniques in parallel.

A certain limitation to this approach, however, is given by the need to design and assemble the interfacial layer deposited at the sensor surface for an optimized performance of the respective transduction principle. We will demonstrate this for a series of hybridization studies between surface-attached oligonucleotide catcher probes and their (fully complementary or mismatched) target sequences from solution. These target analytes were labeled with streptavidin that was either derivatized with ferrocene for the electrochemical detection or with a chromophore suitable for surface plasmon fluorescence spectroscopic investigations [4].

The use of semiconducting nanoparticles, quantum dots, offers interesting aspects for both schemes: The tunability of the emission wavelength via bandgap engineering of the quantum dots allows for multiple parallel recording of simultaneous binding events on an array sensor by color multiplexing [5]. On the other hand, the manipulation of the energy level of the substrate (by applying an electrical potential) and, hence, the electronic interaction between substrate and nanoparticles – covalently 'wired' to the (Au-) electrode – promises interesting coupling schemes [6] and the use of the observed new photo-electrochemical phenomena for completely unconventional biosensor recording principles1.

10.2 Experimental Apparatus

A schematic of the employed combination set-up is given in Fig. 10.1a. The basic instrument is a classical surface plasmon resonance (SPR) spectrometer in the Kretschmann configuration. The thin Au layer evaporated onto the base of the coupling prism is used simultaneously as the active metal/dielectric interface that guides the evanescent surface plasmon mode and can be employed as the working electrode in a typical electrochemical set-up. The two common modes of operation in SPR – the angular scans and the time-dependent kinetic recordings at a fixed angle of incidence – are schematically given in Fig. 10.1b, c, respectively. By attaching, in addition to the working electrode, also a counter and reference electrode to the flow



Fig. 10.1 (a) Schematic of a surface plasmon spectrometer in the prism-coupled Kretschmann configuration in combination with electrochemical instrumentation; (b) angular scan SPR measurements before (*left curve*) and after a thin film deposition, e.g., a layer of capture probes assembled (*right curve*); (c) kinetic SPR measurement taken as a function of time at a fixed angle of incidence (and observation), cf. (b); (d) schematic of a cyclic voltammogram taken by the electrochemical set-up coupled to the SPR instrument

cell, classical electrochemical techniques, e.g., cyclic voltammetry (cf. Fig. 10.1d), square wave voltammetry (SWV), or impedance spectroscopies [3] can be employed for the functional characterization of interfacial architectures.

A further extension of this combination set-up will be presented in Fig. 10.8: a fluorescence detection unit is attached in a way that the photons emitted by chromophores that are sufficiently close to the interface to be excited by the surface plasmon modes can be detected either as a function of the angle of incidence (tuning the resonance) or as a function of time for kinetic recordings [7].

Depending on whether electrochemical or surface plasmon optical measurements are to be performed the interfacial architectures assembled at the sensor surface need to fulfil different requirements: in order to optimize the electron transfer from a redox label to the electrode that is - in one way or another - the basis for electrochemical biosensing the redox-active unit has to be wired to the electrode interface in a most efficient way. The assembly of the sensor coating that were used for surface hybridization studies is schematically given in Fig. 10.2: onto the bare Au substrate a binary mixture of thiolated PNA oligonucleotides (i.e., the synthetic mimics of natural oligonucleotides with a neutral pseudo-peptide backbone [8]) of a particular sequence (HS-PNA) are co-adsorbed with a diluent molecule, mercaptohexanol (MCH), that optimizes the lateral packing of the probe oligonucleotides for the hybridization reactions (Fig. 10.2a). After hybridization of biotinvlated target strands from solution to the probe matrix at the sensor surface (Fig. 10.2b), the biotin groups are 'decorated' in a 1:1 stoichiometric ratio by streptavidin molecules that carry a number of ferrocene molecules covalently attached to the protein through a spacer (Fig. 10.2c). A schematic of the protein structure and one ferrocene spacer unit are given in Fig. 10.2d.

Table 10.1 summarizes details of the employed molecular structures and the specific sequences of probe and target PNAs and DNAs [9]. In addition to the probe HS-PNA-P1 with a 15-mer recognition sequence coupled via the spacer groups to the thiol unit for the linking to the Au substrate the equivalent DNA analogue with the same recognition sequence P1, however, with a spacer unit of 15 thymines and a biotin anchor that allows the molecule to be assembled onto a streptavidin monolayer employed in the optical experiments as a generic binding matrix (cf. also Fig. 10.10) was used. For studies on mismatch discrimination another probe DNA, i.e., the biotin-DNA-P2 was used. This recognition sequence differs by a single base (underlined and in *italics* in Table 10.1).

The target strands follow two different molecular construction principles: for the electrochemical detection the DNA targets used are all modified with a biotin group at the 5' end. This will allow for the coupling of the ferrocene-labeled streptavidin as redox-markers after hybridization (cf. the scheme given in Fig. 10.2b, c). For the surface plasmon fluorescence spectroscopic experiments the targets in one case were also modified with a biotin which after hybridization, however, were detected via the binding of a fluorophore-labeled streptavidin molecule. For the color multiplexing experiments QDs emitting at different wavelengths (cf. Table 10.1) were directly coupled to the various targets.



Fig. 10.2 Interfacial architecture self-assembled at the sensor surface for electrochemical readout of hybridization reactions: (a) onto the Au substrate used for SPR- and electrochemical measurements a binary mixture of thiolated PNA strands and mercaptohexanol (MCH) as diluent molecules are forming a functional monolayer; (b) after hybridization with biotinylated target strands resulting in DNA duplex formation, ferrocene-labeled streptavidin can be bound to the surface layer (c). Some structural details of the ferrocene-labeled streptavidin Fc-Stv are shown in (d). Typically, 9 ferrocenes are attached to one streptavidin resulting in a corresponding amplification of the binding signal

Name	Nucleotide sequence
Probes	
HS-PNA - P1	Cys-[eg1] ₆ - TGT ACA TCA CAA CTA - NH2
	$Cys-[eg1]_{6}: HS \bigvee_{NH_{2}}^{O} NH \xrightarrow{O} O O O O O O O O O O O O O O O O O O $
Biotin - DNA - P1	5'-biotin - (TTT) ₅ - TGT ACA TCA CAA CTA - 3'
Biotin - DNA - P2	5'-biotin - (TTT) ₅ - TGT ACG TCA CAA CTA - 3'
Targets	
T1 - biotin	3' - ACA TGT AGT GTT GAT - biotin - 5'
T1 - QD ₅₆₅	3' - ACA TGT AGT GTT GAT - QD₅₆₅ - 5'
T2 - QD ₆₅₅	3' - ACA TG <u>C</u> AGT GTT GAT - QD₆₅₅ - 5'

 Table 10.1
 Probe and target sequences used in the experiments

10.3 Electrochemical Sensing of Hybridization Reactions

The combination of the electrochemical cell with a surface plasmon spectrometer allows for the structural characterization of the sequential assembly of the interfacial architecture by surface plasmon optical (angular and/or time-dependent) scans followed by the functional assessment of the resulting sensor surface coating by, e.g., cyclic voltammetry. This is documented in Figs. 10.3 and 10.4, respectively, for surface architectures based on the assembly of thiolated molecules directly onto the Au substrate. This results in the formation of a functional monolaver exposing a specific nucleotide sequence to the target strands injected subsequently into the sample cell. The minute shift of the SPR resonance curve upon binding of the targets to the surface probe layer (cf. Fig. 10.3, dots to solid curve) indicates the limited sensitivity of SPR in this case of a small analyte molecule (small molecular mass compared to a typical protein) bound to the surface at a moderate surface density. Only upon decoration of the hybrid monolayer with a layer of streptavidin bound via the target-attached biotin-groups results in a significant shift of the SPR angular scan curve that could be used for a more quantitative analysis of the affinity of this hybridization reaction. Effectively, the streptavidin acts as a mass label in SPR detection. From the angular shift of $\Delta \theta = 0.25^{\circ}$ one calculates a thickness of the streptavidin layer of c. $\Delta d = 2.1$ nm (assuming a refractive index of n = 1.45), corresponding to a protein density of about $1.3 \cdot 10^{12}$ cm⁻².



Fig. 10.3 SPR curves recorded in 20 mM PB buffer with 0.005% Tween 20 after PNA/MCH immobilization (*dots*), after hybridization with 500 nM target (*solid curve*) and after Fc-Stv adsorption (200 nM, *dashed curve*). The *inset* is an expanded view of the SPR minima part



Fig. 10.4 SPR kinetic curve measured at a fixed incident angle ($\theta = 55.7^{\circ}$): (*a*) injection of 200 nM Fc-Stv after rinsing a 1 μ M non-complementary biotinylated target solution through the cell; (*b*) rinsing with buffer; (*c*) injection of 500 nM fully complementary biotinylated target; (*d*) after rinsing and then injection of 200 nM Fc-Stv; (*e*) rinsing with buffer; (*f*) rinsing with 50 mM NaOH in order to regenerate the probe matrix; (*g*)–(*k*) repeating the steps (*b*)–(*e*)

The equivalent kinetic experiment and a few more reference scans are shown in Fig. 10.4. After assembling the probe matrix and injecting an analyte solution with fully mismatched target strands (a), rinsing with neat buffer results in the complete return of the reflected intensity at this fixed angle of observation to the reference level (b). Injection of the complementary, biotinylated target results in a small increase of the reflectivity (c) which – upon injection and binding of the ferrocene-labeled streptavidin (Fc-Stv) – shows the strongly amplified reflectivity increase expected from the angular scans displayed in Fig. 10.3 (d). Regeneration of the probe matrix by injecting a 50 mM NaOH solution through the flow cell (f) results in the return of the reflected intensity to the level measured prior to any hybridization. This observation and the displayed fully repeatable hybridization, i.e., association/dissociation (regeneration) cycle added in Fig. 10.4 indicates the excellent control over the supramolecular functional architecture and the stability of the attached components upon repeated hybridization cycles (Fig. 10.4g–k).

Figure 10.5 shows a series of cyclic voltammograms taken with a sample that was prepared by hybridizing the fully matched target T1-biotin from a 500 nM solution to the surface-bound probe matrix and decorating the hybrids by a monolayer of Fc-Stv [10]. The flexibility of the double strand and the six glycine residues of the linker used to couple the ferrocene to the streptavidin allow for a rather effective electron transfer from the redox site to the electrode and *vice-versa*. The CV curves recorded at different scan rates are perfectly symmetrical, indicating a fully reversible electron transfer reaction. The peak currents are proportional to



Fig. 10.5 Cyclic voltammograms of Au/PNA/fully complementary target/Fc-Stv measured in buffer with different scan rates, i.e., 0.4, 0.5, 0.6, 0.7, 0.8, 1.0 and 1.2 V/s from inside to outside, respectively. The *inset* shows the linear relationship between peak currents and scan rates

the scan rates, confirming a surface-controlled redox reaction only (cf. the insert in Fig. 10.5). The slightly broadened peak width (i.e. 120 mV compared to the ideal value of 90.6 mV for a reversible one electron surface transfer reaction) possibly indicates some ferrocene interactions and a random distribution of the redox labels within the film. Assuming that one biotinylated target can bind one streptavidin and that all Fc sites are electroactive the amount of bound DNA target can be calculated by integrating the current under the anodic (or cathodic) charge transfer wave in the CV. The obtained value of $\Gamma = 1.1 \cdot 10^{12} \text{ cm}^{-2}$ corresponds well to the value obtained by SPR (1.3 \cdot 10^{12} cm^{-2}, see above), indicating that, indeed, most of the ferrocenes are electroactive.

The number of bound DNA molecules, hybridized to the PNA matrix depends on the target concentration in the bulk solution, c_0 . In particular, at low concentration only a fraction of the maximum PNA/DNA duplexes at the interface can be formed. With increasing concentration of c_0 more and more hybrids are formed until eventually a complete monolayer is asymptotically reached. Within the Langmuir model which requires certain conditions to be met, e.g., that all binding sites are equivalent and that the association and dissociation rate constants do not depend on the coverage, a simple relation for the coverage, Γ , at the sensor surface and the bulk concentration c_0 holds:

$$\theta = K_A c_0 / (1 + K_A c_0) \tag{10.1}$$



where θ is the fractional coverage defined as $\theta = \Gamma/\Gamma_{\text{max}}$ and Γ_{max} is the maximum coverage given by the number density of PNA probe strands at the sensor surface.

With the sensor architecture presented in Fig. 10.2 we can assume that one biotinylated target strand is able to bind to one Fc-labeled streptavidin molecule. Hence, the recorded current should be proportional to the DNA target coverage and will allow for the determination of the affinity constant K_A :

$$I(c_0) = I_m K_A c_0 / (1 + K_A c_0)$$
(10.2)

Figure 10.6 displays a series of SWV curves taken after Fc-Stv was bound to the biotinylated target strands after they hybridized to the sensor matrix from bulk solutions of different concentrations, as indicated. Clearly, with increasing target concentration the corresponding redox peak currents increase. If these peak currents are plotted as a function of the bulk target concentration c_0 , the saturation behavior is found (cf. Fig. 10.7, (- \diamond -)) as expected from the Langmuir model that has been demonstrated by optical techniques to describe this process at the sensor interface very well. If fitted to the Langmuir model according to Eq. 10.2, an affinity constant of $K_A = 1.5 \cdot 10^8 \text{ M}^{-1}$ is found (Fig. 10.7, blue curve). This value agrees well with other reports based, e.g., on the use fluorescently labeled targets, in particular, if differences in the interfacial architecture, i.e., the coupling chemistry, the spacer length, the probe density, etc., are taken into account.

10.4 Fluorescence Spectroscopic Sensing of Hybridization Reactions

As it has been introduced recently, fluorescence detection schemes in combination with the optical field enhancement mechanisms operating at resonant surface plasmon excitation result in significant sensitivity gains for biosensing applications: for proteins binding to a quasi-3D matrix a limit of detection (LOD) of



Fig. 10.7 Langmuir isotherms of PNA/DNA hybridization constructed from the electrochemical data presented in Fig. 10.6 (- \diamond -) and from the surface plasmon fluorescence data shown in Fig. 10.9 (- \blacksquare -) and fitting curves (*red* for SPFS data, *blue* for electrochemical data)

 $c_0 = 500$ aM [11], and for PCR hybridization to a 2D probe layer a LOD of $c_0 = 100$ fM [8] have been reported. As mentioned before the attachment to a regular SPR set-up required for surface plasmon fluorescence spectroscopy (SPFS) is rather trivial: a lens in front of the metal-coated prism base that guides the surface plasmon wave collects the fluorescence photons emitted from chromophores that are within the range of the evanescent surface mode. After passing an attenuator (in order to restrict the count rates to the linear response range of the detector) and a bandpass filter (for discrimination of the fluorescence emission against merely scattered light) this photoluminescence is then recorded by a photomultiplier tube or – in case of the microscopic mode of operation (cf. below) – by a color CCD camera. The whole arrangement is mounted to the SPR instrument in such a way that it rotates in the angular mode with the prism thus recording fluorescence photons emitted normal to the interface. The scheme of this instrumental setup is given in Fig. 10.8.

The possibility to employ fluorescently-labeled streptavidin (Fluor-Stv) (instead of the ferrocene-labeled analogue) for the decoration of surface-hybridized biotinylated target strands allows for a direct comparison of SPFC data with the electrochemical results presented in Figs. 10.6 and 10.7, respectively [10]. To this end, a series of angular SPFS curves were recorded after hybridization of targets from solutions of different concentrations and after the binding of Fluor-Stv. The results are summarized in Fig. 10.9. Similar to the current in the case of the SWV scans, here, the fluorescence intensity increases as one increases the bulk concentration of the target solutions, c₀. Note the slight shift of the angular peak position of the reflectivity minima) upon the increase in surface coverage of the



Fig. 10.8 Schematics of a surface plasmon fluorescence spectrometer, based on a normal SPR instrument with an attached fluorescence detection module. A photomultiplier tube (PMT) is used as the detector in the spectroscopic mode; however, in the imaging mode (cf. Fig. 10.11) a color CCD camera can be attached

bound streptavidin. Similar to the peak currents of the SWV scans the plot of the peak fluorescence intensities as a function of c_0 (cf. Fig. 10.7, $-\blacksquare$ -) shows the typical Langmuir isotherm behavior with the saturation at higher concentration (Fig.10.7, blue curve).

As one can see by the almost perfect superposition of the SWV and the SPFS data both methods give nearly identical K_A -values and, hence, confirm that both methods detecting their respective, yet very similar labels (Fc-Stv vs. Fluor-Stv) and the quantitative analysis of the data is not affected methodologically. As far as the sensitivity of the two techniques is concerned one should also expect similar limits of detection because the S/N levels in both data sets are rather equivalent (cf. Figs. 10.6 and 10.9).

The general interest in sensor platforms that allow for the parallel detection of multiple binding events typically requires a chip format in which the individually sensing elements functionalized for a particular analyte are arranged in an array matrix [12]. Read-out can then be achieved either in a serial manner (e.g., by a fluorescence scanner) or in a parallel way by a (microscopic) imaging mode [13].

The use of semiconducting nanoparticles, quantum dots (QDs), offers an alternative approach based on the possibility to engineer their emission wavelength by size (quantum confinement) or compositional control of their bandgap energy [14]. This way, very similar labels that all can be excited by the same laser wavelength



Fig. 10.9 A series of angular reflectivity (*dotted* and *dashed*) curves and SPFS scans (*curves* with *symbols*) taken after the hybridization of biotinylated, fully complementary target strands from solutions of different concentrations and the decoration of the duplex layer by fluorophore-derivatized streptavidin (*Fluor-Stv*). Note the slight shift of the angular peak position due to the increasing surface density of bound streptavidin; *dotted curve*: before hybridization, *dashed curve*: after hybridization with 500 nM target followed by binding of Fluor-Stv

but which emit in different luminescence colors can be synthesized and attached to the respective bio-analytes. Color multiplexing thus allows for the simultaneous recording of the binding signals of a whole set of analyte molecules in a mixture [5].

This concept and its implementation for surface hybridization studies is schematically given in Fig. 10.10. Here, the initial concept for the functional assembly of the sensor coating is modified by first assembling a monolayer of a binary mixture of 10% of a biotinylated thiol derivative laterally diluted by 90% of mercaptohexanol. Onto this coating a streptavidin monolayer as a generic binding matrix is assembled to which other biotinylated units can be attached, e.g., different oligonucleotide probe strands. Their base sequences are complementary to different DNA targets in the analyte solution which, in turn, are coded on different colors by carrying QDs of different emission wavelengths. Their simultaneous recording after hybridization allows for a multiple detection of a variety of analytes in parallel.

An example along this concept is demonstrated in Fig. 10.11. The flow cell was equipped with four Au electrodes that could be used as SPR/SPFS substrates [15]. After the general assembly of a protective coating to all electrodes (i.e., a poly (ethylene oxide) thiol SAM (PEO-SAM)) the flow cell was mounted to the spectrometer. The individual potential control of the four electrodes could then be used for the sequential functionalization of each Au area separately. Firstly, one of the electrodes was subject to a cathodic scan thus reductively desorbing its PEO-SAM, generating a bare, clean Au surface. Injecting sequentially (i) the binary



Fig. 10.10 Interfacial architecture optimized for multiple parallel surface hybridization studies in the color multiplexing mode with semiconducting nanoparticles emitting fluorescence light at different wavelengths: Onto the Au substrate a binary monolayer of biotinylated thiols mixed with OH-terminated thiols as diluent molecules is self-assembled, followed by the binding of a monolayer of streptavidin. To this generic binding matrix a mixture of different biotinylated probe oligonucleotides are assembled. They can specifically hybridize to their respective complementary target strands which are sequence-coded by quantum dots of different colors

(biotin-/OH-) thiol mixture, (ii) streptavidin and then (iii) a specific biotinylated PNA oligonucleotide solution (e.g., the Biotin-DNA-T1, cf. Table 10.1) results in the selective functionalization of this electrode only, while the other ones still remain protected by their PEO-SAM. By a step-and-repeat procedure eventually all electrodes are functionalized with no cross-contamination from one area to the neighboring electrode.

This way, the electrode array in Fig. 10.11 was functionalized with Biotin-DNA-P1 (P1), Biotin-DNA-P2 (P2), a mixture of both probes for the color multiplexing experiment, and a PEO-SAM used as an inert reference (cf. Fig.10.11a).

After injection of a 200 nM solution of the T2 target, coded with QDs emitting at $\lambda = 655$ nm, the surface plasmon fluorescence microscopic image displays the expected red color on the second and forth electrode (from top to bottom, cf. Fig. 10.11b) that both expose to the analyte solution the probe strand P2 which is fully complementary to T2 (cf. Table 10.1). The third electrode shows only a very faint reddish color corresponding to only a minute surface binding according to the significantly reduced affinity for T2 hybridizing to P1 which represents a mismatch 1 (MM1) situation with an affinity constant about two orders of magnitude lower than MM0. The reference electrode covered with the PEO-SAM remains completely dark.



Fig. 10.11 Surface plasmon fluorescence microscopy: (a) Schematic arrangement (*top view*) of the probe oligonucleotides P1 and P2 and a mixture of both, as indicated, on a four-electrode chip for hybridization studies with target oligonucleotide strands labeled with quantum dots; (b) SPFM image showing the *red color* emitted after the hybridization of the target T2 which is complementary to P2 and is carrying a QD emitting at $\lambda = 655$ nm; (c) SPFM taken after the hybridization of a second injected target sequence T1, fully complementary to P1 and coded with a QD emitting at $\lambda = 565$ nm. Note the (artificial) mixture of *red* and *green* in the CCD camera to result in a *yellow color*

After rinsing and injecting the T1 solution the result shown in Fig. 10.11c is obtained after the hybridization reaction was completed. The third electrode exposing the MM0 probe P1 turns green corresponding to the binding of the targets T1 with their QDs emitting at $\lambda = 565$ nm. The forth electrode remains red (the emission appears to be more intense only because the integration time was increased in order to account for the lower quantum yield of the green emitting QDs), while the PEO-SAM also in this case appears to be completely inert.

The second electrode area now looks yellow. However, this is only a (wellknown) artefact of the color CCD camera that mixes red and green to yellow. The same result was obtained upon the injection of a solution mixture of the target analytes rather than their sequential injection. Analysing the emitted light via a spectrometer clearly demonstrates that only light originating from green and red emitting QDs is recorded. This experiment demonstrates that color multiplexing is an option for multiple analyte sensing; however, it requires the spectral analysis by a spectrometer for the unambiguous identification of the emitted colors.

10.5 Conclusions

The presented results document that both, fluorescence-optical and electrochemical principles, are well suited to monitor bio-analyte binding from the aqueous phase to specific functional sites assembled at the sensor surface. Although the readout of redox labels doesn't quite reach the sensitivity of fluorescence recordings electrochemical techniques offer the advantage of giving direct electrical signals that can be used for data analysis.

Very attractive for both methods are semiconducting nanoparticles. The possibility of tuning their emission wavelength makes them very attractive labels in photoluminescence studies, e.g., for color multiplexing in parallel schemes for multiple analyte detection. One must bear in mind, however, that complications arise from the still unsolved problem of the luminescence blinking of QDs [16].

Particularly promising are schemes that couple the electronic orbitals of QDs to the Fermi-level of the metallic substrate. This way a completely new principle for the interplay between QDs as light emitting fluorophores and their electron donating or accepting properties are conceivable and should give rise to novel ways for the sensitive detection of bio-analyte binding [17].

Acknowledgments Partial funding for this project from the Deutsche Forschungsgemeinschaft (DFG, KN 224/13-1 and 13-2, and SFB 625) is gratefully acknowledged. We thank Thomas Basché, Ming-Yong Han, and Shengjun Tian for many helpful discussions.

References

- 1. Kastner, M.A.: The single electron transistor and artificial atoms. Ann. Phys. 9, 885–894 (2000)
- Knoll, W.: Interfaces and thin films as seen by bound electromagnetic waves. Annu. Rev. Phys. Chem. 49, 569–638 (1998)
- 3. Kaifer, A.E., Gómez-Kaifer, M.: Supramolecular Electrochemistry. Wiley- VCH, Weinheim (2001)
- 4. Liebermann, T., Knoll, W.: Colloids and Surfaces A 171, 115-130 (2000)
- Robelek, R., Niu, L., Schmid, E., Knoll, W.: Multiplexed hybridization detection of quantum Dot conjugated DNA sequences using surface Plasmon enhanced fluorescence microscopy and spectrometry. Anal. Chem. 15, 6160–6165 (2004)
- Cameron, P.J., Zhong, X., Knoll, W.: Electrochemically controlled surface Plasmon enhanced fluorescence response of surface immobilized CdZnSe quantum dots. J. Phys. Chem. C 113, 6003–6008 (2009)
- Knoll, W., Yu, F., Neumann, T., Niu, L., Schmid, E.: Principles and applications of surfaceplasmon field- enhanced fluorescence techniques. In: Lakowicz, J.R., Geddes, C.D. (eds.) Topics in Fluorescence Spectroscopy. Vol 8: Radiative Decay Engineering, pp. 305–332. Springer, Berlin/Heidelberg/New York (2005)

- Yao, D., Yu, F., Kim, J., Scholz, J., Nielsen, P., Sinner, E.K., Knoll, W.: Surface plasmon fieldenhanced fluorescence spectroscopy in PCR product analysis by peptide nucleic acid probes. Nucleic Acids Res. 32, 177–192 (2004)
- Liu, J., Tian, S., Tiefenauer, L., Nielsen, P., Knoll, W.: Simultaneously amplified electrochemical and surface plasmon resonance optical detection of DNA hybridization based on ferrocene-streptavidin conjugates. Anal. Chem. 77, 2756–2761 (2005)
- Liu, J., Tian, S., Tiefenauer, L., Nielsen, P., Knoll, W.: PNA-DNA hybridization study using labeled-streptavidin by voltammetry and surface plasmon fluorescence spectroscopy. Anal. Chem. 78, 470–476 (2006)
- Yu, F., Persson, B., Lofas, S., Knoll, W.: Attomolar sensitivity in bioassays based on surface plasmon fluorescence spectroscopy. J. Am. Chem. Soc. 126, 8902–8903 (2004)
- Zizlsperger, M., Knoll, W.: Multispot parallel on-line monitoring of interfacial binding reactions by surface plasmon microscopy. Prog. Colloid Polym. Sci. 109, 244–253 (1998)
- Liebermann, T., Knoll, W.: Parallel multispot detection of target hybridization to surfacebound probe oligonucleotides of different base mismatch by surface-plasmon field-enhanced fluorescence microscopy. Langmuir 19, 1567–1572 (2003)
- Zhong, X., Xie, R., Zhang, Y., Basché, T., Knoll, W.: High-quality violet- to red-emitting ZnSe/CdSe core/shell nanocrystals. Chem. Mater. 17, 4038–4042 (2005)
- Niu, L., Knoll, W.: Electrochemically addressable functionalization and parallel readout of a DNA biosensor array. Anal. Chem. 79, 2695–2702 (2007)
- Robelek, R., Stefani, F.D., Knoll, W.: Oligonucleotide hybridization monitored by surface plasmon enhanced fluorescence spectroscopy with bio-conjugated core/shell quantum dots. Influence of luminescence blinking. Phys. Status Solid A 203, 3468–3475 (2006)
- Sinner, E.-K., Ritz, S., Wang, Y., Dostalek, J., Jonas, U., Knoll, W.: Molecularly controlled functional architectures at biointerfaces. Mater. Today 13, 46–55 (2010)

Part IV Detection Strategies – Sequencing

Chapter 11 Recent Advances in Sequencing Technology

John F. Thompson, Fatih Ozsolak, and Patrice M. Milos

Abstract As we celebrate the tenth anniversary of the sequencing of the first human genome, we recognize the remarkable technological innovation that now provides the ability to resequence thousands of human genomes a year. While the current methods of choice utilize amplification-based methods and the corresponding challenges of sample preparation that accompany these methods, new technologies that do not require amplification have emerged. Single-molecule sequencing methods have the potential to dramatically shape the next 10 years of technological progress driven by the continuing interest of driving the cost of whole genome sequencing below the \$1000 cost threshold. Yet while whole genome sequencing remains of interest, sequencing technologies also enable new approaches for genome exploration and experimentation including direct RNA sequencing, complete transcript sequencing and real time methods for both nucleic acid and enzyme kinetics.

11.1 Introduction

The development of Maxam and Gilbert, and separately Sanger, sequencing methodologies in the mid-to late 1970s opened a new era for cataloging the genome of a multitude of organisms [1, 2]. While much of the early sequencing included bacteriophage, cDNA's, ribosomal RNAs and viral genomes using these newly developed methods, the Maxam and Gilbert technology was quickly replaced with

P.M. Milos (⊠) Pfizer Center for Therapeutic Innovation, Boston, MA, USA e-mail: Patrice.M.Milos@pfizer.com

J.F. Thompson

NABsys Inc., Providence, RI, USA

F. Ozsolak Helicos BioSciences Corporation, Cambridge, MA, USA

G. Spoto and R. Corradini (eds.), *Detection of Non-Amplified Genomic DNA*, Soft and Biological Matter, DOI 10.1007/978-94-007-1226-3_11, © Springer Science+Business Media Dordrecht 2012

the safer and more expeditious method of Sanger chain termination chemistries. In Sanger sequencing, a primer is used to hybridize to the template DNA and synthesis of the complementary DNA strand is initiated by the DNA polymerase. Initially, four separate reactions were used to distinguish each dNTP incorporation as well as the chain terminating ddNTP with termination occurring due to the lack of a hydroxyl moiety on the ddNTP. Altering the ratio of dNTP's and ddNTP's results in chain termination along the length of the molecule and the use of radioactive nucleotides allowed the electrophoretic and autoradiographic analysis of the newly formed DNA strands providing the sequence of the synthesized strand in four separate electrophoretic gel lanes. The technical advance of using fluorescently labelled ddNTP's, each A, C, G or T ddNTP containing a unique fluorescent dye with distinguishing spectral properties facilitated single tube chemistry as well as automated sequencing instrumentation using the differential fluorescence of the individual nucleotides to record the incorporation event along the DNA strand through a laser detection system [3].

The commercial opportunity for DNA sequencing was quickly realized when in 1981 two engineers from Hewlett-Packard started Applied Biosystems to manufacture and sell sequencing instrumentation to enable large scale sequencing in genome research laboratories across the globe. Continued improvements in instrumentation allowing ever increasing speed and high-throughput capacity using the same basic Sanger chain terminator chemistry culminated in the release of the ABI Prism[®] 3700 Genetic Analyzer which became the work horse for the Human Genome Project delivering sequence data at a volume never imagined possible. A consistent theme throughout the revolution of DNA sequencing we are experiencing involves scientists envisioning the need and technologists finding new, innovative solutions to meet those needs.

Initial uses of the sequencing technology focused on cDNAs and the genomes of small organisms, including the first complete 4.6-Mb Escherichia coli K-12 genome [4] representing the work of more than 250 people that required a 6 year effort. The draft human genome sequence was published concurrently by the publicly funded Human Genome Project [5] and the commercial venture Celera [6]. The government sponsored project took some 15 years and three Billion dollars yet prompted scientists and technologists alike to begin considering the potential for expanding the number of genomes beyond the initial draft genome sequences. In 2004, following the completion of the human genome sequence [7], the National Human Genome Research Institute under the guidance of Dr. Francis Collins, leader of the HGP, announced a major 70 million dollar funding initiative to fuel development of new DNA sequencing technologies with a goal of achieving the \$1000 genome by 2014. In now looking back, the initial round of funding shaped the field of both amplified and unamplified DNA next-generation sequencing as it exists today (Table 11.1). Indeed the goal of achieving the \$1000 genome by the year 2014 once again reiterates the notion that clarity of vision allows the developments necessary to achieve the desired objectives, often more quickly than projected.

Today, we have taken the basic principles of DNA sequencing chemistry described above and moved far beyond the throughput and yield envisioned by the early pioneers in sequencing. Further we have extended the bulk DNA methodology

Table 11.1 The initial genor	ne technology grants fund	led by the National Human Genome Research Ir	nstitute in 2004
Grantee	Funding amount	Company/Institution	Title
\$100,000 Genome Grants: 1	NHGRI's "Near-Term D	evelopment for Genome Sequencing"	
Stevan B. Jovanovich, Ph.D.	\$6.1 million (3 years)	Microchip Biotechnologies Inc.	Microbead Integrated DNA Sequencer (MINDS) System
Gina L. Costa, Ph.D.	\$5.4 million (3 years)	Agencourt Bioscience Corp.	Bead-based Polony Sequencing
Kenton Lohman, Ph.D.	\$2 million (2 years)	454 Life Sciences Corp.	Massively Parallel High Throughput, Low Cost Sequencing
Marcel Margulies, Ph.D.	5 million (3 years)	454 Life Sciences Corp.	454 Life Sciences Massively Parallel System DNA Sequencing
John Williams, Ph.D.	\$2.5 million (3 years)	Li-COR Inc.	Single-Molecule DNA Sequencing Using Charge-Switch dNTPs
Michael L. Metzker, Ph.D.	\$2 million (3 years)	Human Genome Sequencing Center, Baylor College of Medicine	Ultrafast SBS (Sequencing by Synthesis) Method for Large-Scale Human Resequencing
Stephen R. Quake, Ph.D.	\$1.8 million (3 years)	Stanford University	High-Throughput, Single-Molecule DNA Sequencing
Mostafa Ronaghi, Ph.D.	\$1.8 million (3 years)	Stanford Genome Technology Center	Pyrosequencing Array for DNA Sequencing
Jingyue Ju, Ph.D.	\$1.8 million (3 years)	Columbia University	An Integrated System for DNA Sequencing by Synthesis
Peter Williams, Ph.D.	\$1.7 million (3 years)	Arizona State University	Multiplexed Reactive Sequencing of DNA
Steven A. Benner, Ph.D.	\$800,000 (3 years)	University of Florida	Polymerases for Sequencing by Synthesis
Amit Meller, Ph.D.	\$600,000 (2 years)	Rowland Institute at Harvard, Harvard University	Ultra-fast Nanopore Readout Platform for Designed DNA's
"\$1000 Genome" Grants N.	HGRI's "Revolutionary	Genome Sequencing Technologies"	
J. Michael Ramsey, Ph.D.	\$2 million (2 years)	University of North Carolina, Chapel Hill	Nanotechnology for the Structural Interrogation of DNA
James Weifu Lee, Ph.D.	\$700,000 (3 years)	Oak Ridge National Laboratory	"Computational Research & Development for Rapid Sequencing Nanotechnology"
James Weifu Lee, Ph.D.	\$750,000 (3 years)	Oak Ridge National Laboratory	"Experimental Research & Development for Rapid Sequencing Nanotechnology"
Scott D. Collins, Ph.D.	\$850,000 (2 years)	University of Maine	High-speed Nanopore Gene Sequencing
Steven A. Benner, Ph.D.	\$800,000 (3 years)	University of Florida	DNA Sequencing Using Nanopores
Andre Marziali, Ph.D.	\$650,000 (3 years)	University of British Columbia, Vancouver	Nanopores for Trans-Membrane Bio-Molecule Detection
Stuart Lindsay, Ph.D.	\$550,000 (3 years)	Arizona State University	Molecular Reading Head for Single-Molecule DNA Sequencing
Ronald W. Davis, Ph.D.	\$450,000 (2 years)	Stanford University	Single Molecule Nucleic Acid Detection with Nanopipettes
Data Compiled from the Nati	onal Human Genome Res	earch Institute release announcing these awards	(http://www.genome.gov/12513210)
utilized in these early technologies to now allow the detection and sequencing of single molecules of DNA and RNA. This chapter describes the rapid pace of amplified sequencing technology developments and the emergence of nonamplified, single molecule DNA and RNA sequencing, all of which have resulted in dramatic increases in the generation of genomic information including thousands of complete human genome sequences that will help to unravel the complexity of numerous diseases, as well as the use of sequencing for basic biological measurements including cDNA sequencing and quantitation, direct RNA sequencing and quantitation, and new insight into ancient genomes which are also highlighted in this chapter. The field of non-amplified sequencing now offers the opportunity for direct measurements of both DNA and RNA, providing a true measurement of cellular biology.

11.2 Emergence of Amplification Based Short Read Sequencing

Through the recognition of the important role "shot-gun" sequencing played in the commercial venture led by Craig Venter, the important concepts emerged for the "second generation" technologies – if one could simply sequence short fragments of DNA, the problem of assembling the genomes of small organisms with an eye to the whole human genome, seemed entirely possible. 454 Life Sciences emerged as the first entrant into the field of new non-Sanger based sequencing technologies publishing data demonstrating an increase in sequencing throughput enabling some 25 million bases of sequence data or some 100 fold greater than traditional Sanger technology in a period of 4 h [8]. This advance was enabled by the massively parallel synthesis of DNA templates by polymerase chain amplification to provide sufficient substrate for sequencing. This technology relies on the basic principles of sequencing by synthesis applied to pyrosequencing in which the emission of light upon incorporation of the labeled nucleotide for subsequent detection and visualization of the incorporation event, the landscape of DNA sequencing was changed forever. Commercial introduction of the Genome Sequencer 20/FLX in 2005 enabled researchers to achieve the complete sequence of the first human genome subsequent to that published from the Human Genome Project. This genome however was completed in just two months time, using the 454 technology with average read lengths approaching 250 nucleotides, and provided a genome coverage of \sim 7.5× allowing redundancy of the reads to ensure both near complete coverage and accuracy of sequence data [9].

Solexa, a company founded in Cambridge, England in the early 2000s was one of the first commercial companies interested in the pursuit of single-molecule sequencing along with Helicos BioSciences founded at about the same time in Cambridge, MA. While both companies started as single-molecule sequencing by synthesis companies, Solexa abandoned the single-molecule approach, was acquired by Illumina in 2006 and their initial commercial platform, the GA1 was introduced in the 2007 timeframe. The GA1, an amplification based second generation system of short reads, provided the customer with a new level of sequence throughput with 36 nucleotide reads. Today, the sequencing by synthesis approach practiced by Illumina has been continually improved through continued investment in the technology surrounding the chemistry, image analysis, engineering hardware and image analysis software to the point that the current HiSeq instrument allows researchers to sequence two complete human genomes per run at $30 \times$ coverage for approximately \$10,000, closely approaching the goal of the \$1000 genome.

11.2.1 Emerging Low Cost, High Throughput Technologies

While the increasing capacity of instruments, such as the Illumina HiSeq and Life Tech Solid, provide genome centers across the globe with the capacity to sequence hundreds to thousands of genomes per year, access to lower cost platforms to allow genomic scientists at smaller research institutions and translational research centers not well served by the ultra-high-throughput capacity has been limited. Former 454 founder, Jonathan Rothberg meanwhile recognized the limitations of the sequencing-by-synthesis chemistries dependence on ultra-fast imaging requirements. Ion Torrent was founded on the principles of image-independent chemistry in which the nucleotide incorporation event could simply be monitored by local changes in pH evident when a hydrogen atom is released upon nucleotide incorporation, promising a future where scale is only dependent on the ability to create a semiconductor surface which gets smaller and smaller to allow ever increasing numbers of molecular events to be monitored. Ion Torrent, purchased by Life Technologies in 2010, has its eyes set on the ultimate prize of the \$1000 genome, however, along that path, the company has introduced the first low cost Personal Genome Analyzer (PGMTM) to address the need of scientific researchers requiring low cost and moderate throughout to enable genomic experimentation. With a throughput today of 100 megabases, and promised improvements of tenfold every 6 months the Ion Torrent technology appears ready for placing DNA sequencing within the realm of every biological researcher [10]. To exemplify, while this platform today is not able to sequence a complete genome in a single run, this technology enables comprehensive exon sequencing for important genes of biological relevance with the potential for diagnostic applications and when combined with bar-coding for each individual sample, hundreds to thousands of samples could be analyzed at the same time. Like the other sequencing technologies, a broad array of applications are available with the PGM machine.

11.3 Sequencing Applications

While all the technologies mentioned above rely on DNA or cDNA amplification to obtain sequence information, numerous applications are enabled by these methodologies and have revolutionized the manner by which biological measurements are possible. These applications are briefly highlighted below and will be discussed in more depth as we describe the applications for non-amplified DNA and RNA sequencing later in the Chapter.

11.3.1 Whole Genome Sequencing

Whole genome sequencing costs have dropped dramatically over the last 10 years to the current cost estimate of 12 cents per megabase of DNA sequence as demonstrated in Table 11.2 (www.genome.gov/sequencingcosts). As the technologies have improved read length and paired-read capabilities, the major application driven by the Genome Centers has been genomic sequencing. From viral species and bacterial genome sequencing for purposes of strain identification to the human microbiome to the baboon, chimpanzee and 1,000s of normal human genomes and tumor genomes remains, these technologies have become the workhorse for genome sequencing. While the scope of this chapter is focused on sequencing technology, a brief mention of the informatics developments accompanying the technological advances described herein is appropriate as these developments have been integral in continued sequencing improvements.

Whole genome sequencing in particular has been dramatically improved by new methods and algorithms to enable *de novo* assembly of genomes and are nicely highlighted in recent publications [11, 12].

11.3.2 cDNA Sequencing

For more than a decade much of our exploration of the transcriptome has been conducted using microarrays or hybridization based methods that allow one to reliably detect the relative abundance of the known transcripts which hybridize to known probes on the array surface. This field, much like the field of DNA sequencing, has exploded with the use of next-generation sequencing. Termed RNA-Sequencing or RNA-Seq, scientists have pushed the technology to conduct biological experimentation at a scale not previously imagined and further, in a hypothesis free manner. By generating cDNA molecules through traditional reverse transcriptase methods and including the ligation of adapters that allow for amplification of the corresponding molecules, one has the ability to interrogate the transcriptome to ask important biological questions including information on the relative quantitation of RNA transcripts [13]. Since the initial publications describing the RNA-Seq methods appeared in 2008, important biological insight on stems cells [14], the complexity of the cancer transcriptome [13] and more recently the beginnings of the analysis of single cells [15] has allowed scientists to gain new insight into the RNA which plays such an integral role in cellular biology and disease states.

Date	Cost per Mb of DNA sequence	Cost per genome
September-2001	\$5,292.39	\$95,263,072
March-2002	\$3,898.64	\$70,175,437
September-2002	\$3,413.80	\$61,448,422
March-2003	\$2,986.20	\$53,751,684
October-2003	\$2,230.98	\$40,157,554
January-2004	\$1,598.91	\$28,780,376
April-2004	\$1,135.70	\$20,442,576
July-2004	\$1,107.46	\$19,934,346
October-2004	\$1,028.85	\$18,519,312
January-2005	\$974.16	\$17,534,970
April-2005	\$897.76	\$16,159,699
July-2005	\$898.90	\$16,180,224
October-2005	\$766.73	\$13,801,124
January-2006	\$699.20	\$12,585,659
April-2006	\$651.81	\$11,732,535
July-2006	\$636.41	\$11,455,315
October-2006	\$581.92	\$10,474,556
January-2007	\$522.71	\$9,408,739
April-2007	\$502.61	\$9,047,003
July-2007	\$495.96	\$8,927,342
October-2007	\$397.09	\$7,147,571
January-2008	\$102.13	\$3,063,820
April-2008	\$15.03	\$1,352,982
July-2008	\$8.36	\$752,080
October-2008	\$3.81	\$342,502
January-2009	\$2.59	\$232,735
April-2009	\$1.72	\$154,714
July-2009	\$1.20	\$108,065
October-2009	\$0.78	\$70,333
January-2010	\$0.52	\$46,774
April-2010	\$0.35	\$31,512
July-2010	\$0.35	\$31,125
October-2010	\$0.32	\$29,092
January-2011	\$0.23	\$20,963
April-2011	\$0.19	\$16,712
July-2011	\$0.12	\$10,497

 Table 11.2
 The National Human Genome Research Institutes calculated costs for whole genome sequencing

Reprinted from Wetterstrand KA. DNA Sequencing Costs: Data from the NHGRI Large-Scale Genome Sequencing Program. Available at: www.genome.gov/ sequencingcosts. Accessed 12 Nov 2011

11.3.3 Chromatin Immunoprecipitation Studies

While DNA sequence is clearly the foundation for much of our understanding of the human genome, the additional insight provided by studying the structure of the DNA in its native chromatin state has been significantly aided by next-generation sequencing. Chromatin immunoprecipitation studies (ChIP) allow one to study the interaction of the DNA sequence with any DNA-binding protein including histone proteins and their various modified forms in the nucleosome structure of the DNA found in the nucleus. Additional proteins including transcription factors which regulate the dynamic state of gene transcription form key interactions with the chromatin. The state of these protein:DNA interactions are studied genomewide through a snapshot of antibody immunoprecipitation specific to the various proteins followed by isolation and sequencing of the specific DNA region to which the protein-antibody complex had bound. New insight into development [16], hematopoiesis [17] clinical outcome in ovarian cancer [18] has been possible through use of ChIP-Sequencing at the whole genome level.

11.3.4 Limitations of Amplification Technology

While amplification-based sequencing has led to tremendous advances across a range of biological questions, it remains limited or subject to artifacts in a variety of specific areas. Challenges still remain with respect to genomic rearrangements of large scale, accurate quantitation of both DNA and RNA as well as severe bias in regions of the genome with extreme G+C content. The new methods involving non-amplified DNA and RNA sequencing offer significant opportunity in all these areas as the technologies continue to mature [19]. Additionally, the various single-molecule approaches offer other technology-specific advantages including the ability to generate very long reads, to directly detect modified bases, and to directly sequence RNA.

11.4 Emergence of Non-amplified DNA Technologies

Researchers had long been interested in single molecule, non-amplified measurements of DNA. Non-amplified methods offered the potential to eliminate many of the challenges associated with complex sample preparation, PCR amplification, and the ability to interrogate single-cell nucleic acid as well as the potential for real-time interrogation to allow faster and cheaper detection of the sequence information. A landmark publication demonstrating single molecule fluorescence imaging to monitor the turnover of ATP molecules by single muscle myosin molecules using total internal reflection microscope [20] provided insight into the potential for such single-molecule measurements. This was rapidly followed by key developments in this field of non-amplified DNA sequencing which can be divided into three different areas which are sufficiently advanced for the purpose of this review: (1) direct imaging for single DNA molecules to allow both mapping and sequencing of large DNA molecules important for describing higher order structure of the DNA sequence; (2) Optical sequencing by synthesis technologies and (3) nanopore sequencing technologies. What follows is a summary of these various technologies as well as detailed examples of sequencing applications available today.

11.4.1 Optical Mapping Technologies

As the field of DNA sequencing was maturing and the Human Genome Project was in full swing, many questions remained about the ability of shot-gun sequencing to recapitulate the accurate sequence of a genome the size of humans. Optical mapping techniques emerged in the 1990s as a potential solution to create an ordered structural map of the human genome and was one of the first single molecule methods for visualization of the higher order structure of the genome. Following the isolation of high molecular weight DNA upwards to some 10–20 megabases, DNA molecules are fixed on a glass surface, liquid flow elongates the single DNA molecules and restriction digestion of the DNA is performed directly on the surface. Figure 11.1 illustrates an optical image obtained from human genomic DNA to allow visualization of the single DNA fragments and illustrates a schematic representation of the potential ability to assemble long stretches of DNA along a contiguous stretch thus recapitulating the higher order structure of the human genome [21].

More recently scientists at companies such as Halcyon Molecular (http:// halcyonmolecular.com/) and ZS Genetics (http://www.zsgenetics.com/index.html) have turned to the transmission electron microscopy as a new tool to investigate individual DNA molecules at the atomic level to allow visualization of the DNA sequence along the length of the molecule offering the potential to directly image and visualize at the DNA sequence level [22]. These methods while still under development offer the potential to provide sequence data on long stretches of DNA to overcome the limitations of current sequencing technologies which at present are limited by read length and throughput.

11.4.2 Optical Non-amplified Sequencing Technologies and Their Applications

11.4.2.1 Optical Sequencing Technologies

An initial demonstration of single-molecule sequencing-by-synthesis using DNA polymerase and fluorescent nucleotides to monitor the complementary nucleotide incorporation enabled DNA sequence data to be obtained from individual DNA molecules [23]. These initial studies led to the founding of Helicos BioSciences which further developed and commercialized the world's first single-molecule sequencing-by-synthesis technology.



Fig. 11.1 Single molecule optical mapping depiction of the human genome. (**a**) An image of a DNA molecule 2 mm long covering 6 megabases of DNA, digested with PacI represented by overlapping microscope images. *White bars* represent the cutting sites on the DNA molecule. (**b**) Schematic representation depicting the linking of a whole genome optical map with that of physical maps (Reprinted with permission from Trends in Biotechnology, Ref. [21] Copyright 1999, Elsevier Sciences)

The Helicos technology utilizes non-amplified fragments of DNA or RNA for direct capture of the nucleic acid on the glass flow cell surface to which either universal capture primers or gene specific capture primers are covalently affixed. The depiction of the DNA sequencing process, which is also utilized for direct RNA sequencing, as well as the actual images captured during the sequencing process are shown in Fig. 11.2 [24, 25].



Fig. 11.2 Helicos BioSciences single molecule sequencing sample preparation and imaging of single molecules of DNA (Left) Single-molecule sequencing by synthesis process. (1) Genomic DNA is fragmented and a 3' poly(A) tail added, labeled, and blocked using terminal transferase. (2) The DNA templates are captured on the surface with covalently bound oligo- dT(50). (3) Imaging of the captured templates to template sites on the surface. (4) One labeled nucleotide and polymerase mixture are added, followed by rinsing of the synthesis mixture and direct imaging. (5) Chemical cleavage of the dve-nucleotide linker to release the dve label. (6) Addition of the next nucleotide and polymerase mixture. (Right) Image series illustrating template-specific base addition, successful rinsing, and successful linker cleavage. A mix of three templates is used to allow visual sequence assignment. Template complementary sequences are shown in the table (bottom). One example of each template is outlined in the figure. Each frame is a $6.6 \mu m$ square image of the same sample position, and shows ~ 35 of the 1.8×10^6 imaged templates in this experiment. Frame 1 is the image of the template labels. Template activity in three positions is shown in the columns to the right. Frame 2 is after the first synthesis and rinse cycle. Frames 3 to 8 show the effect of six more consecutive cleave, synthesis, and image cycles, using the base identity shown in the lower right corner of the frame (Reprinted with permission from Science, Ref. [24] Copyright 2008, American Association for the Advancement of Sciences)

The universal surface consists of an Oligo-dT50 surface to allow the researcher to add a polyA tail to the individual DNA molecules with terminal transferase for subsequent hybridization on the flow cell surface. When sequencing RNA, the natural polyA tail of the RNA molecules are captured on the surface or alternatively a polyA tail can be added to RNA molecules via polyA polymerase. Following hybridization to the flow cell, fluorescently labeled nucleotides, termed "Virtual Terminators" and polymerase are added sequentially to the flow cell to allow the incorporation of complementary nucleotides into the growing strands of DNA or RNA. Once incorporated, laser excitation of the fluorophore present on the individual molecules leads to fluorescence that is captured with a charge coupled device (CCD) camera and converted to sequence reads using specialized imaging software. Following 120 cycles of this sequencing-by-synthesis and optical capture of the sequence data, more than one billion individual DNA sequence reads are generated from the DNAs held on the flow cell surface area and available for the variety of research applications of interest to the scientist. The Helicos sequencing technology provides an average read length of 35 nucleotides for each DNA molecule with a raw error profile between 3 and 5% with the predominant error form being a 'dark base' due to the incorporation of a nucleotide which is not visualized during the sequencing by synthesis process. Key is the alignment algorithm, IndexDP Genomic which allows for accurate alignment of DNA sequence data taking into account potential single nucleotide gaps in the DNA sequence [26]. Various applications of the sequencing technology are described in later sections of this Chapter.

Meanwhile, scientists at Cornell University were pursuing the real-time incorporation of fluorescent nucleotides via DNA polymerase into growing strands of DNA using a zero-mode waveguide (ZMW) technology [27]. Commericalized recently by Pacific Biosciences, the sequencing technology utilizes a SMRTTM cell or chip consisting of thousands of ZMW guides which are tiny microwells, nanometers in diameter, created in a metal film on a glass surface (shown in a schematic form in Fig. 11.3). Here the DNA polymerase is affixed to the bottom glass surface of the well. Laser illumination of the bottom 30 nm of the ZMW guide allows detection only of molecules which are near the bottom of the well. Because the DNA polymerase is attached to the bottom of the ZMWs, only labeled molecules bound to the polymerase remain in the illumination region long enough to be detected. DNA molecules are flowed across the surface to allow single molecules of DNA to bind to the polymerase and reside in the ZMW followed by addition of fluorescently-labeled nucleotides. Through diffusion, the nucleotides find their way into the ZMW where incorporation via the DNA polymerase occurs on the growing strands of DNA. Prior to each incorporation event, the fluorescent nucleotide must remain bound to the polymerase prior to incorporation and this results in fluorescence and corresponding detection of the color indicative of the nucleotide fluor. Following incorporation, the signal returns to low background level until the next nucleotide finds its way to the growing DNA strand. With the initial SMRT cell configuration consisting of some 75,000 ZMW, the incorporation events are monitored in real-time allowing the researcher to interrogate thousands of strands of individual DNA molecules in real time. In addition, the natural processivity of the DNA polymerase enzyme as well as the cleavage of the dye molecule attached to the phosphate chain of the nucleotide leaving a natural nucleotide in the growing DNA strand, has the potential to deliver read lengths well exceeding all other amplification and non-amplification based methods. Currently the average read lengths can approach 1,000 bases but continued efforts on both chemistry and detection offer the opportunity to surpass the current read length.



Fig. 11.3 Depiction of Pacific Biosciences Zero Mode Waveguide. The ZMW microwell is depicted here with a polymerase molecule to which DNA is bound. Synthesis of the DNA strand occurs through incorporation of the appropriate labeled nucleotide which will be detected upon laser excitation and emission to the recorded below the ZMW surface (Adapted with Permission from Pacific Biosciences)

Helicos and Pacific Biosciences non-amplified DNA sequencing technology have been used for genome sequencing. The publication describing the first single molecule sequencing of a human genome using the Helicos sequencing technology was indeed a remarkable achievement in the field of non-amplified DNA [28]. Requiring minimal sample preparation and three weeks time for sequencing followed by several weeks of analysis time, the field of non-amplified sequencing was established. Additionally, new insights into ancient genomes have been provided by recent studies of DNA from a Pleistocene-era horse [29] and thus the unique attributes of this ability are highlighted in a later section. Pacific Biosciences and colleagues at Harvard Medical School followed with another single-molecule sequencing study demonstrating the ability of SMRT sequencing to decipher the origin of the cholera strain responsible for the outbreak in Haiti in 2010 [30]. Yet, importantly, these two technologies each provide unique features that differentiate them from the existing amplification-based sequencing technologies. The current ability and future potential of the SMRT sequencing technology to obtain read lengths that far surpass the current technologies combined with the sheer speed of the DNA read-out offer new opportunity for more complete characterization of the genome, allowing us to address the many rearrangements and repetitive regions not possible with current technologies. In addition the ability to obtain full length cDNA transcripts to fully elucidate the complex structure of the transcriptome is entirely within the realm of the technology as throughput improvements occur.

The accuracy of quantitation with the Helicos single molecule sequencing technology is unparalleled as the ability to examine upwards of one billion molecules which have not been amplified or ligated. Thus the inherent bias caused by these molecular manipulations are avoided and the unamplified methodology provides the scientist with the purest quantitative measurement for nucleic acid from a biological source of interest [31] and enables new insight into important new biology revealing new RNA species not previously detected with amplification based technologies [32]. Additionally, the Helicos technology allows the direct sequencing of RNA. Molecules which have not been manipulated via copying with reverse transcriptase can thus be examined in a massively parallel fashion, something never before possible [33]. A more in-depth characterization of these unique applications is detailed below.

11.4.2.2 Gene Expression Measurements: Tag-Based and RNA Seq

Gene expression measurements provide a powerful window into how cells and organisms behave normally as well as in response to various stimuli. Both the number of genes that can be analyzed and the sensitivity with which those genes can be detected has increased substantially as the technology of choice moved from Northern blots and S1 assays to microarrays and qPCR and most recently to RNA Seq. Each of the methodologies has its own strengths and weaknesses with respect to identification of exon structure, crosstalk with similar genes, sensitivity, quantitative accuracy, ability to detect poorly expressed or uncharacterized transcripts and susceptibility to artifacts and errors. No technology is able to provide a complete, quantitative picture of which transcripts are expressed combined with a detailed view of their exon structure and 5'/3' ends. As such, experimenters need to decide which aspects of the true expression profile are most important for their purposes so the technology can be chosen which is best suited to the needed information.

Most early technologies were able to monitor only a small number of genes for their expression levels so are limited if one wishes to do a complete transcriptome characterization. A more thorough view on transcription became possible with the advent of various types of microarrays and the ability to analyze thousands of different genes simultaneously. However, microarrays suffer from poor sensitivity for genes that are expressed at low levels, can suffer from significant crosstalk with genes that are closely related, has difficulty determining splicing patterns, and is not generally useful for uncharacterized transcripts. To varying extents, next generation sequencing technologies have overcome all of these difficulties. For NGS technologies with a very high read count, very precise expression patterns can be generated. However, all sequencing methods also have limitations and these must be recognized so that experimental results can be properly interpreted.

Sequencing-based gene expression methods can be divided into two types: tagbased methods and methods that interrogate the entire RNA molecule. Tag-based methods include CAGE (Cap Analysis of Gene Expression) [34], SAGE (Serial Analysis of Gene Expression) [35, 36], PET-Seq (Paired End Tag Seq) [37], DGE (Digital Gene Expression) [38, 39], and DRS (Direct RNA Sequencing) [33]. These techniques capture a specific sequence at either the 5' end (CAGE and DGE) or 3' end (SAGE and DRS) or both (PET Seq) and count those molecules for determining gene expression profiles. Little or no information is captured on splicing or the opposite end of the molecule. Thus, these techniques are ideal for assessing and comparing gene expression levels. Additionally, the tag-based systems typically incorporate a selection step in which a specific feature of mRNA (5' cap or 3' polyA tail) that allows preferential sequencing of the RNAs of interest with less sequencing of ribosomal RNA that is generally of less importance for expression.

RNA Seq, in contrast to tag-based methods, captures reads from throughout each RNA molecule. This provides information about the entire RNA but, unlike microarrays and tag-based sequencing, this introduces an artifact into RNA Seq data in that the results depend on the length of the RNA being interrogated. If the RNA is long, more reads will arise from that RNA even if expressed at the same number of molecules as a short RNA. Frequently, the raw expression levels are corrected for length but these corrections are imperfect for many reasons [40] and can lead to analysis issues [41]. RNAs with extremes of GC content are less likely to be amplified and thus will appear less often than they should based on actual expression levels [42]. Efforts to eliminate GC bias in library construction have been partially successful [43] but some of these biases remain and are exacerbated by the amplification that is required during sequencing. Even after removing much of the library-induced amplification bias, genomic coverage patterns are still far from what is predicted based on known sequence content [43]. Figure 11.4 demonstrates this principle with data derived from RNA Seq experimentation in which amplification based methods are compared directly to single molecule methods, both involving cDNA analyses. Single molecule sequence data demonstrates more uniform coverage across the gene transcripts depicted [19].

The paired-end protocols used to generate long-range data for splicing analysis can introduce artifacts as well. If the raw reads from RNA Seq libraries are examined, 5–10% of all reads have paired ends that do not match the same gene [44]. Thus, one could expect that some artifacts could also be present in matched pairs, suggesting that rare splice variants should be verified using an independent technology. Additionally, the random hexamers typically used for synthesis of cDNA introduce another level of bias into the process [45], independent of sequencing platform.



Fig. 11.4 Comparison of amplification and non-amplification RNA Seq data sets derived from cancer cell line. Amplification-based sequencing leads to a bias in high-concentration transcripts. Coverage maps from amplification-based and single molecule sequencing demonstrate significantly greater coverage of (**a**) RPLP0, (**b**) RPL31, and (**c**) SPINT2. Removal of reads with the same start positions significantly reduces the "spikiness" seen in these cases. (**d**) Duplicate reads are relatively evenly distributed along the length of observed transcripts across all samples and artificially inflate the apparent expression of those genes in amplification-based sequencing but not in single-molecule sequencing (Reproduced with Permission from PLoS ONE: Ref. [19])

While amplification-based sequencing has been used extensively for high quality expression profiles [46], there are situations in which single-molecule RNA Seq methods can provide superior results for some aspects of gene expression studies. For example, the Pacific Biosciences system is capable of long reads [47] so offers the potential of a direct view of exon content and start/stop sites for different

transcript isoforms. Unfortunately, the read count with this system is far too low to be generally useful for accurate expression profiles. A reasonable splicing picture of very highly expressed genes probably could be generated but the frequency of reads from most genes will leave them with no coverage or insufficient coverage to assess splicing and start/stop sites. For that reason, no RNA Seq profiles have yet been published with this system.

In contrast to Pacific Biosciences, the Helicos system generates a very high read count and thus can provide a very accurate quantitative assessment of gene expression [38]. Because amplification is not necessary, quantitative accuracy is maintained to a much higher extent than with amplification-based systems.

Any artifacts that arise from reverse transcriptase and cDNA synthesis may still skew the true gene expression measurements unless Direct RNA Sequencing (DRS) is carried out (see below) but biased amplification during library construction and sequencing is not an issue. The Helicos read lengths are generally, though not always, sufficient for unambiguous assignment to a particular gene. With highly homologous genes, some crosstalk may occur so there will be uncertainty in the expression level of such genes. Additionally, though some information can be gleaned about splicing, complex splice patterns are not detectable. Nonetheless, very precise RNA Seq profiles have been obtained [48] and technical replication far superior to amplification-based systems has been described [40].

To a large extent, the differential expression patterns obtained with singlemolecule versus amplified sequencing are very similar but there are key differences. In particular, genes that are poorly expressed are found much less often with amplification-based sequencing [19], likely due to issues of limiting library diversity but other issues with PCR cannot be ruled out. In addition to limited visibility of poorly expressed RNAs and those with extreme GC contents, very short RNAs can also be problematic [32]. Many short RNAs are not well amplified during library construction and are too short to be amplified using bridge PCR required in the Illumina platform.

11.4.2.3 Gene Expression Measurements: Direct RNA Sequencing Technology

In addition to the frequent use of amplification, all of the gene expression methods described above rely on conversion of RNA to cDNA prior to sequencing. However, cDNA synthesis is known to be plagued by many artifacts including template switching, primer-independent first and second strand synthesis, and biased cDNA synthesis [49–51]. DRS technology can alleviate many limitations inherent in the transcriptomics methods in use today, and provide new avenues of research and applications in diagnostics. DRS not only eliminates the reverse transcription step, but also the other sample manipulation steps such as ligation and amplification, thus resulting in minimal distortion in the representation of RNA templates. The natural strand-specificity of DRS and its requirement for only femtomole quantities of RNA are advantageous for all aspects of RNA research.



Fig. 11.5 Mapping of liver-derived direct RNA sequencing reads to human Chromosome 4.Total liver RNA was sequencing using the DRS technology. Sequence reads were mapped with high stringency to the human genome reference. Reads mapping to Chromosome 4 are shown (Unpublished data used with Permission from Helicos BioSciences Corporation)

DRS offers a simple route for polyadenylation site mapping [52]. Given its nature of capturing polyadenylated RNAs on poly(dT)-coated surfaces and sequencing after a "fill & lock" step, DRS reads emerge immediately upstream of the polyA-tail. Thus, the 5' end of DRS reads signify polyadenylation addition locations. The DRS procedure is capable of capturing the polyA + mRNA population from total RNAs or cell lysates directly after blocking only the 3' hydroxyl. The data generated is quantitative, thus for the first time allows genome-wide study of alternative polyadenylation states in both quantitative and qualitative manner across biological settings of interest. Figure 11.5 demonstrates direct RNA sequence reads obtained from liver total RNA mapped to human chromosome 4. Peaks across the chromosome demonstrate the diversity of RNA reads at low resolution. This data can also be used to generate a tag-based gene expression profile of polyA + mRNAs within cells.

DRS can also be adapted for all RNA analyses being performed today. Whole transcriptome profiling can be done with RNA fragmentation with standard methods, followed by polyadenylation of the RNAs. One advantage of DRS is the universality of sample preparation steps for different applications. In other words, unlike cDNA methods which require different cDNA synthesis and sample manipulation steps for short and long RNAs, DRS requires only 3' polyadenylated templates. Thus, both short and long RNAs can be sequenced together in a single experiment.

11.4.2.4 Ancient and Degraded DNA

In most situations, long sequence reads are an advantage. However, when the DNA is degraded due to age, chemical fixation, or damaged by other deleterious conditions, it may actually be advantageous to use short read technologies. For example, ancient DNA is frequently contaminated with DNA from other species which is sometimes of more recent origin and hence potentially longer and of higher quality [53]. When such mixtures are amplified, the longer and less damaged modern DNA is preferentially replicated and hence increases its fractional composition and could potentially swamp out the desired signal from the ancient or damaged DNA. This is evident from work with DNA from a Pleistocene-era horse bone in which the same sample sequenced with an amplified versus non-amplified system yielded very different results for the per cent horse versus non-horse sequence reads [29]. On average, $>3\times$ more reads were from horse using the non-amplified sequencing. This difference was substantially increased when the sample preparation for the ancient DNA was modified slightly to remove blocking 3' phosphates from the ancient DNA [54]. The extent to which single-molecule sequencing is superior is highly dependent on the quality of the DNA sample of interest. The more degraded the sample, the higher degree of improvement can be obtained. However, it is not necessary for DNA to be Pleistocene era in order to be too degraded for analysis. For example, some remains buried at the National Memorial Cemetery of the Pacific that had been exposed to highly damaging conditions during embalming and which, even after extensive amplification, had not previously provided more than a few dozen base pairs of usable human sequence using amplification-based sequencing was able to be effectively sequenced using the Helicos system [55]. Similarly, many clinical samples are formaldehyde-treated and preserved in paraffin and thus can be significantly damaged or degraded. More recently preserved samples tend to be higher quality as experimenters have realized the importance of mild conditions to allow subsequent sequencing but many samples have not been as carefully handled and thus are problematic with sequencing systems that require longer templates. Again, single-molecule sequencing has been able to provide good sequence data for even RNA samples extracted from FFPE clinical specimens [56].

11.4.3 Nanopore Sequencing Methods

Classical next-generation sequencing techniques have dramatically dropped the price of sequencing while opening up numerous new avenues of scientific investigations. These tremendous advances have served to whet the appetite for even greater capacity at lower costs. However, the current commercial technologies are unlikely to yield orders of magnitude cheaper sequence or provide markedly different capabilities relative to what is already available. Consistent with the initial goals laid out by NHGRI for the \$1000 genome and in contrast to the existing technologies, there are a variety of nanopore technologies, though not yet



Fig. 11.6 Schematic representation of nanopore fabrication. artificial nanopores and nanochannels can be made by heavy ion and chemical ecthing (**a** and **b**) or by drilling holes through a silicon wafer with a focused electron beam (**c** and **d**). A more detailed description is provided in Ref. [60] (Reprinted with permission from Trends in Biotechnology, Ref. [60] Copyright 2011, Elsevier Sciences)

technically and/or commercially viable, that could provide dramatically different or cheaper output. Nanopore technology is attractive because, in theory, extremely long reads could be generated on single molecules with little or no reagent costs, simple sample preparation in a highly parallel fashion and with a very short time to results. However, realizing all these goals or even some of them, with a single technology will be challenging. To be commercially successful, the output from a technology will either have to provide a qualitatively different result or substantial cost/time savings relative to the already high bar created by existing technologies. Reviews dedicated to nanopores and nanopore sequencing are available and describe the wide variety of approaches being taken in this area [57–61]. Healey [62] has reviewed nanopore sequencing from a more historical perspective.

Nanopore approaches can be categorized in a number of ways based on the nature of the nanopore with the most frequent classification being biological or solid state. Biological nanopores use a protein channel with a pore of the necessary dimensions to allow DNA to pass through [63]. While early biological nanopores were in lipid membranes, some more recent versions are hybrid, using a solid-state scaffold with a protein pore. Solid-state pores (Fig. 11.6) were originally derived from silicon and

its derivatives though recent efforts have also explored graphene as an alternative. In addition to the nature of the pore, another useful categorization is the mode of sequence detection. Most nanopores detect variations in electrical signals (voltage or current) induced by blocking an ionic current between chambers but other methods, including optical detection and differences in transverse electrical currents, can also be used. The theoretical background and modeling of blocking ionic currents have been described in detail [64, 65]. Similarly, the theory and signals expected from transverse currents detected across a tunneling gap have also been described [66, 67] and confirmed experimentally [68–70].

Most efforts directed at biological nanopores have employed either α -hemolysin or MspA as the protein of choice [63, 71, 72] though other proteins may also be used [73]. All of these proteins have a pore with a diameter greater than 1 nm needed to allow single-stranded DNA (ssDNA) to pass through. Some proteins have an even larger pore allowing double-stranded DNA to pass. However, the variation in the lengths, widths and charge distributions within the pores of these proteins is not ideal for sequence detection so efforts have been directed at improving their properties. α -hemolysin and MspA have been successfully mutated to improve a variety of sequencing properties [72, 74, 75]. These mutated pore proteins have been shown to be capable of distinguishing the electrical signals from all four natural nucleosides/nucleotides [76-78]. However, distinguishing free nucleotides is not the same as sequencing extended lengths of DNA. To generate usable sequence data, the DNA must be translocated through the nanopore with sufficient force that movement is unidirectional but not so quickly that the signal from the individual bases cannot be distinguished. The length and width of the pore must be such that the base being sequenced is the primary contributor to the signal or a very complex set of signals will result. These conditions are not easily met so a variety of methods have been used in an attempt to generate sequence data. The speed of DNA translocation through protein pores has been slowed by altering viscosity [79] and by varying salt concentrations [80]. These manipulations have not yet proved sufficient for reading sequence so various proteins have been used to assist in the process.

For example, DNA polymerase has been used to detect sequence incorporations while attached to α -hemolysin [81]. Similarly, the activity of exonuclease I bound to α -hemolysin has been monitored by nanopores [82]. The kinetics of these reactions in pilot studies is not sufficient for effective sequencing but demonstrates the future possibilities if they could be optimized. In addition to detecting the electronic signal directly from nucleotides/nucleobases, there have also been efforts to detect signals optically after converting the naturally occurring sequence to fluorescent emitters [83]. However, methods described thus far introduce far more complexity into sample preparation than is required by other approaches so are not as appealing as the simpler nanopore readouts.

While biological nanopores offer advantages in terms of being easily manipulated with respect to changing the charge and inclusion of complex functional groups in or near the pore, there are only a small number of suitable pore proteins and they provide a limited set of scaffolds with which to work. Also, construction of highly parallel arrays does not have the economies of scale that solid-state nanopores offer. Solid-state nanopores have the advantage of being much thinner and thus can generate signals arising from a single base more easily. Indeed, graphene offers the thinnest possible nanopore, the thickness of a single carbon atom. Furthermore, nanopores can be constructed with a wide variety of widths and this can be readily changed using well-established techniques [84]. For the most part, solid-state nanopores are made from silicon derivatives. Very long DNAs (97 kb) have been reported to be translocated through silicon at a speed greater than 10 kb/ms [85] and it is likely that much longer DNAs could be used. Even at 97 kb, these DNAs far surpass the lengths that can be interrogated with current sequencing technologies and thus would immediately provide a substantial benefit for genomic analysis if information beyond simply the length could be obtained.

As with biological nanopores, the speed of translocation with current solidstate nanopores is too fast for sufficient signal to be generated for each base to be effectively read. As a result, modifications of the pores and translocation conditions have been carried out to slow the rate of translocation [86]. Capture rate can be adjusted by varying salt concentrations [87] or by altering the manner in which the pores are made [88, 89]. Pores can be chemically modified so that the charge slows translocation speed. Because silicon nanopores can be made wider, ssDNA and dsDNA as well as protein-bound DNA can be successfully translocated through pores. When bound to recA, dsDNA moves through the pores much more slowly and generates a much higher blockade current [90].

While varying translocation conditions can slow the rate of transit through the nanopore that creates new issues with the positioning of the DNA with respect to the nanopore constriction and assuring that the DNA is read sequentially [91]. However, if DNA is translocated through a nanopore at a speed necessary to ensure predictable positioning, it is going too fast to generate sufficient signal. As a result, a variety of methods have been tested in order to provide the optimal mix of speed and sensitivity [92, 93]. Additionally, hybrid pores modified with DNA [94] or protein [95] can also be used. Another approach employs oligonucleotide probes of known sequence to tag regions of interest in DNA [96] and generate position-specific changes in blocking current. By using pools of probes, the entire sequence can be generated *de novo*. This method has the advantage of using groups of nucleotides with each signal and spreading that signal out over a longer physical distance for enhanced signal to noise and ease of detection.

Graphene nanopores also have the subject of much recent interest due to the absolute minimum thickness of their constrictions. Thus far, translocation of DNA has been observed through pores experimentally [97–99] along with theoretical predictions of how sequence signals might be possible [100]. While offering tremendous potential, graphene nanopores are now in their infancy in terms of characterization and their very simplicity will make them difficult to modify as can be accomplished with protein and silicon-based nanopores. Thus, the varying flavors of nanopore sequencing provide the best hope for the next quantum leap in sequencing capacity and capabilities but many issues need to be overcome before they are a commercial reality.

11.5 The Future

The power of amplification-based next generation sequencing has enabled countless new approaches to a host of previously inaccessible biological questions. This treasure trove of new applications and knowledge has obscured the fact that there are severe limitations and artifacts present in this powerful but not omniscient technology. The tremendous variety of single-molecule approaches holds the potential for filling many of those gaps and expanding the reach of massively parallel sequencing ever further. An important example, sequence reads of unprecedented length achieved with orders of magnitude faster speed to results could be generated, all at lower cost than current technologies. Different single-molecule approaches existing today, or in development, offer the potential for simplifying and lowering sample requirements, improved quantitation for gene expression, protein binding, and epigenetics. Thus, we can look to the next 10 year when the promise of the \$1000 genome will be realized – and with this new milestone new opportunity for unraveling the complexity of human disease.

References

- 1. Maxam, A.M., Gilbert, W.: A new method for sequencing DNA. Proc. Natl. Acad. Sci. U.S.A. **74**, 560–564 (1977)
- Sanger, F., Nicklen, S., Coulson, A.R.: DNA sequencing with chain-terminating inhibitors. Proc. Natl. Acad. Sci. U.S.A. 74, 463–467 (1977)
- Prober, J.M., Trainor, G.L., Dam, R.I., Hobbs, F.W., Robertson, C.W., Zagursky, R.I., Cocuzza, A.J., Jensen, M.A., Baumeister, K.: A system for rapid DNA sequencing with fluorescent chain-terminating dideoxynucleotides. Science 238, 336–341 (1987)
- 4. Blattner, F.R., et al.: Escherichia coli K-12. Science 277, 1453–1462 (1997)
- International Human Genome Sequencing Consortium, Lander, E.S., Linton, L.M., Birren, B., Nusbaum, C., Zody, M.C., et al.: Initial sequencing and analysis of the human genome. Nature 409, 860–921 (2001)
- Venter, C.J., Adams, M.D., Myers, E.W., et al.: The sequencing of the human genome. Science 291, 1304–1351 (1991)
- International Human Genome Sequencing Consortium: Finishing the euchromatic sequence of the human genome. Nature 431, 931–945 (2004)
- Margulies, M., Egholm, M., Altman, W.E., et al.: Genome sequencing in microfabricated high-density picolitre reactors. Science 437, 376–380 (2005)
- Wheeler, A., Srinivasan, M., Egholm, M., Shen, Y., Chen, L., McGuire, A., He, W., Chen, Y.-J., Makhijani, V., et al.: The complete genome of an individual by massively parallel DNA sequencing. Nature 452, 872–876 (2008)
- Rothberg, J.M., Hinz, W., Rearick, T.M., et al.: An integrated semiconductor device enabling non-optical genome sequencing. Nature 475, 348–352 (2011)
- 11. Li, R., Zhu, H., Ruan, J., Qian, W., Fang, X., et al.: de novo assembly of human genomes with massively parallel short read sequencing. Genome Res. **20**, 265–272 (2010)
- Nagarajan, N., Pop, M.: Sequencing and genome assembly using next-generation technologies. Methods Mol. Biol. 673, 1–17 (2010)
- Morin, R., Bainbridge, M., Fejes, A., Hirst, M., Krzywinski, M., Pugh, T., McDonald, H., Varhol, R., Jones, S., Marra, M.: Profiling the HeLa S3 transcriptome using randomly primed cDNA and massively parallel short-read sequencing. Biotechniques 45, 81–94 (2008)

- Cloonan, N., Forrest, A.R., Kolle, G., Gardiner, B.B., Faulkner, G.J., Brown, M.K., et al.: Stem cell transcriptome profiling via massive-mRNA sequencing. Nat. Methods 5, 613–617 (2008)
- Tang, F., Barbacioru, C., Wang, Y., Nordman, E., Lee, C., Xu, N., Wang, X., Bodeau, J., Tuch, B.B., Siddiqui, A., Lao, K., Surani, M.A.: mRNA-Seq whole-transcriptome analysis of a single cell. Nat. Methods 6(5), 377–82 (2009)
- 16. Zhao, X.D., Han, X., Chew, J.L., Liu, J., Chiu, K.P., Choo, A., Orlov, Y.L., Sung, W.K., Shahab, A., Kuznetsov, V.A., Bourque, G., Oh, S., Ruan, Y., Ng, H.H., Wei, C.L.: Wholegenome mapping of histone H3 Lys4 and 27 trimethylations reveals distinct genomic compartments in human embryonic stem cells. Cell Stem Cell 1, 286–98 (2007)
- Adli, M., Zhu, J., Bernstein, B.E.: Genome-wide chromatin maps derived from limited numbers of hematopoietic progenitors. Nat. Methods 7, 615–8 (2010), Epub Jul 11 2010
- 18. Kennedy, B.A., Deatherage, D.E., Gu, F., Tang, B., Chan, M.W., et al.: ChIP-seq defined genome-wide map of TGFβ/SMAD4 targets: implications with clinical outcome of ovarian cancer. PLoS One 6, e22806 (2011). E pub 2011 Jul 25 (2011)
- Sam, L.T., Lipson, D., Raz, T., Cao, X., Thompson, J.F., Milos, P.M., Robinson, D., Chinnaiyan, A.M., Kumar-Sinha, C., Maher, C.A.: A comparison of single molecule and amplification based sequencing of cancer transcriptomes. Plos One 6, e17305 (2011)
- Funatsu, T., Harada, Y., Tokunaga, M., Saito, K., Yanagida, T.: Imaging of single fluorescent molecules and individual ATP turnovers by single myosin molecules in aqueous solution. Nature 374, 555–559 (1994)
- Aston, C., Mishra, B., Schwartz, D.C.: Optical mapping and its potential for large-scale sequencing projects. Trend. Biotech. 17, 297–302 (1999)
- Krivanek, O.L., Chisholm, M.F., Nicolosi, V., Pennycook, T.J., Corbin, G.J., et al.: Atom-byatom structural and chemical analysis by annular dark-field electron microscopy. Nature 464, 571–574 (2010)
- Braslavsky, I., Herbert, B., Kartalov, E., Quake, S.R.: Sequence information can be obtained from single DNA molecules. Proc. Natl. Acad. Sci. U.S.A. 100, 3960–4 (2003)
- Harris, T.D., Buzby, P.R., Babcock, H., Beer, E., Bowers, J., Braslavsky, I., et al.: Singlemolecule DNA sequencing of a viral genome. Science 320, 106–109 (2008)
- 25. Thompson, J.F., Reifenberger, J.G., Giladi, E., Kerouac, K., Gill, J., et al.: Single-step capture and sequencing of natural DNA for detection of *BRCA1* mutations. Genome Res.. doi:10.1101/gr.122192.111. Published in Advance July 15, 2011
- Giladi, E., Healy, J., Myers, G., Hart, C., Kapranov, P., Lipson, D., et al.: Error tolerant indexing and alignment of short reads with covering template families. J. Comput. Biol. 17, 1397–1411 (2010)
- Levene, M.J., Korlach, J., Turner, S.W., et al.: Zero-mode waveguides for single-molecule analysis at high concentrations. Science 299, 682–6 (2003)
- Pushkarev, D., Neff, N.F., Quake, S.R.: Single-molecule sequencing of an individual human genome. Nat. Biotechnol. 27, 847–852 (2009)
- Orlando, L., Ginolhac, A., Raghavan, M., Vilstrup, J., Rasmussen, M., Magnussen, K., Steinmann, K., Kapranov, P., Thompson, J.F., Zazula, G., Froese, D., Shapiro, B., Hofreiter, M., AL-Rasheid, K.A.S., Mundy, J., Gilbert, M.T.P., Willerslev, E.: True single-molecule DNA sequencing of a pleistocene horse bone. Genome Res. 21, 1705–1719 (2011)
- Chin, C.S., et al.: The origin of the Haitian cholera outbreak strain. N. Engl. J. Med. 364, 33–42 (2011)
- 31. Kanamori-Katayama, M., Itoh, M., Kawaji, H., Lassmann, T., Katayama, S., Kojima, M., Bertin, N., Kaiho, A., Ninomiya, N., Daub, C.O., Carninci, P., Forrest, A.R., Hayashizaki, Y.: Unamplified cap analysis of gene expression on a single-molecule sequencer. Genome Res. 21, 1150–9 (2011)
- 32. Kapranov, P., Ozsolak, F., Kim, S.W., Foissac, S., Lipson, D., Hart, C., Roels, S., Borel, C., Antonarakis, S.E., Monaghan, A.P., John, B., Milos, P.M.: Novel class of gene-terminiassociated human RNAs suggests a novel RNA copying mechanism. Nature 466, 642–646 (2010)

- Ozsolak, F., Platt, A., Jones, D., Reifenberger, J., Sass, L.E., McInerney, P., Thompson, J.F., Bowers, J., Jarosz, M., Milos, P.: Direct RNA sequencing. Nature 461, 814–818 (2009)
- Plessy, C., Bertin, N., Takahashi, H., Simone, R., Salimullah, M., et al.: Linking promoters to functional transcripts in small samples with nanoCAGE and CAGEscan. Nat. Methods 7, 528–534 (2010)
- 35. Asmann, Y.W., Klee, E.W., Thompson, E.A., Perez, E.A., Middha, S., Oberg, A.L., Therneau, T.M., Smith, D.I., Poland, G.A., Wieben, E.D., Kocher, J.P.: 3' tag digital gene expression profiling of human brain and universal reference RNA using Illumina Genome Analyzer. BMC Genomics 10, 531 (2009)
- Wu, Z.J., Meyer, C.A., Choudhury, S., Shipitsin, M., Maruyama, R., et al.: Gene expression profiling of human breast tissue samples using SAGE-Seq. Genome Res. 20, 1730–1739 (2010)
- Fullwood, M.J., Wei, C.L., Liu, E.T., Ruan, Y.: Next-generation DNA sequencing of pairedend tags (PET) for transcriptome and genome analyses. Genome Res. 19, 521–532 (2009)
- Lipson, D., Raz, T., Kieu, A., Jones, D.R., Giladi, E., et al.: Quantification of the yeast transcriptome by single-molecule sequencing. Nat. Biotechnol. 27, 652–658 (2009)
- Ozsolak, F., Ting, D.T., Wittner, B.S., Brannigan, B.W., Paul, S., et al.: Amplification-free digital gene expression profiling from minute cell quantities. Nat. Methods 7, 619–621 (2010)
- Raz, T., Kapranov, P., Lipson, D., Letovsky, S., Milos, P.M., Thompson, J.F.: Protocol dependence of sequencing-based gene expression measurements. PLoS One 6, e19287 (2011)
- Oshlack, A., Wakefield, M.J.: Transcript length bias in RNA-seq data confounds systems biology. Biol. Direct 4, 14 (2009)
- 42. Dohm, J.C., Lottaz, C., Borodina, T., Himmelbauer, H.: Substantial biases in ultra- short read data sets from high-throughput DNA sequencing. Nucleic Acids Res. **36**, e105 (2008)
- 43. Aird, D., Ross, M.G., Chen, W.S., Danielsson, M., Fennell, T., Russ, C., Jaffe, D.B., Nusbaum, C., Gnirke, A.: Analyzing and minimizing PCR amplification bias in Illumina sequencing libraries. Genome Biol. 12, R18 (2011)
- Mamanova, L., Andrews, R.M., James, K.D., Sheridan, E.M., Ellis, P.D., et al.: FRTseq: amplification-free, strand-specific transcriptome sequencing. Nat. Methods 7, 130–132 (2010)
- 45. Hansen, K.D., Brenner, S.E., Dudoit, S.: Biases in Illumina transcriptome sequencing caused by random hexamer priming. Nucleic Acids Res. **38**, e131 (2010)
- Oshlack, A., Robinson, M.D., Young, M.D.: From RNA-seq reads to differential expression results. Genome Biol. 11, 220 (2010)
- 47. Eid, J., Fehr, A., Gray, J., Luong, K., Lyle, J., Otto, G., Peluso, P., Rank, D., et al.: Real-time DNA sequencing from single polymerase molecules. Science **323**, 133–138 (2009)
- Kapranov, P., St Laurent, G., Raz, T., Ozsolak, F., Reynolds, C.P., Sorensen, P.H., Reaman, G., Milos, P., Arceci, R.J., Thompson, J.F., Triche, T.J.: The majority of total nuclear- encoded non-ribosomal RNA in a human cell is 'dark matter' unannotated RNA. BMC Biol. 8, 149 (2010)
- Mader, R.M., et al.: Reverse transcriptase template switching during reverse transcriptasepolymerase chain reaction: artificial generation of deletions in ribonucleotide reductase mRNA. J. Lab. Clin. Med. 137, 422–8 (2001)
- 50. Cocquet, J., Chong, A., Zhang, G., Veitia, R.A.: Reverse transcriptase template switching and false alternative transcripts. Genomics **88**, 127–31 (2006)
- Haddad, F., Qin, A.X., Giger, J.M., Guo, H., Baldwin, K.M.: Potential pitfalls in the accuracy of analysis of natural sense-antisense RNA pairs by reverse transcription-PCR. BMC Biotechnol. 7, 21 (2007)
- Ozsolak, F., Kapranov, P., Foissac, S., Kim, S.W., Fishilevich, E., Monaghan, A.P., John, B., Milos, P.M.: Comprehensive polyadenylation site maps in yeast and human reveal pervasive alternative polyadenylation. Cell 143, 1018–29 (2010)
- Malmström, H., Svensson, E.M., Gilbert, M.T., Willerslev, E., Götherström, A., Holmlund, G.: More on contamination: the use of asymmetric molecular behavior to identify authentic ancient human DNA. Mol. Biol. Evol. 24, 998–1004 (2007)

- 54. Ginolhac, A., Vilstrup, J., Stenderup, J., Raghavan, M., Rasmussen, M., Stiller, M., Shapiro, B., Zazula, G., Froese, D., Steinmann, K.E., Thompson, J.F., AL-Rasheid, K.A.S., Gilbert, T., Willerslev, E., Orlando, L.: Improving the performance of true-single molecule sequencing for ancient DNA. (2011) (Submitted)
- 55. Thompson, J., Lipson, D., Hart, C., Kapranov, P., Letovsky, S., Milos, P., Ozsolak, F., Raz, T., Reifenberger, J., Steinmann, K., Loreille, O., Coble, M.: Sequencing the unsequenceable: applying massively parallel, single-molecule sequencing to badly degraded DNAs. In: Abstracts of the 59th Annual Meeting of The American Society of Human Genetics, Honolulu, 20–20 Oct 2009. http://www.ashg.org/2009meeting/abstracts/fulltext/f21866.htm
- 56. Yee, A.J., Raz, T., Amzallag, A., Lipson, D., Giladi, E., Lopez, H., Borger, D.R., Mino-Kenudson, M., Thompson, J.F., Iafrate, A.J., Milos, P., Haber, D.A., Ramaswamy, S.: Single molecule RNA sequencing of formalin-fixed paraffin-embedded tissue derived from patients with lung cancer. J. Clin. Oncol. 29(15_suppl), 10550 (2011), http://www.asco.org/ASCOv2/ Meetings/Abstracts?&vmview=abst_detail_view&confID=102&abstractID=78488
- Branton, D., Deamer, D.W., Marziali, A., Bayley, H., Benner, S.A., et al.: The potential and challenges of nanopore sequencing. Nat. Biotechnol. 26, 1146–53 (2008)
- 58. Deamer, D.: Nanopore analysis of nucleic acids bound to exonucleases and polymerases. Annu. Rev. Biophys. **39**, 79–90 (2010)
- Timp, W., Mirsaidov, U.M., Wang, D., Comer, J., Aksimentiev, A., Timp, G.: Nanopore sequencing: electrical measurements of the code of life. IEEE Trans Nanotechnol. 9, 281– 294 (2010)
- Kowalczyk, S.W., Blosser, T.R., Dekker, C.: Biomimetic nanopores: learning from and about nature. Trends Biotechnol. 29(12), 607–614 (2011)
- Venkatesan, B.M., Bashir, R.: Nanopore sensors for nucleic acid analysis. Nat. Nanotechnol. 6, 615–24 (2011)
- 62. Healy, K.: Nanopore-based single-molecule DNA analysis. Nanomedicine 2, 459–481 (2007)
- Kasianowicz, J.J., Brandin, E., Branton, D., Deamer, D.W.: Characterization of individual polynucleotide molecules using a membrane channel. Proc. Natl. Acad. Sci. U.S.A. 93, 13770–3 (1996)
- Aksimentiev, A.: Deciphering ionic current signatures of DNA transport through a nanopore. Nanoscale 2, 468–483 (2010)
- Kowalczyk, S.W., Grosberg, A.Y., Rabin, Y., Dekker, C.: Modeling the conductance and DNA blockade of solid-state nanopores. Nanotechnology 22, 315101 (2011)
- Lagerqvist, J., Zwolak, M., Di Ventra, M.: Fast DNA sequencing via transverse electronic transport. Nano Lett. 6, 779–82 (2006)
- 67. Krems, M., Zwolak, M., Pershin, Y.V., Di Ventra, M.: Effect of noise on DNA sequencing via transverse electronic transport. Biophys. J. **97**, 1990–6 (2009)
- Chang, S., Huang, S., He, J., Liang, F., Zhang, P., Li, S., Chen, X., Sankey, O., Lindsay, S.: Electronic signatures of all four DNA nucleosides in a tunneling gap. Nano Lett. 10, 1070–5 (2010)
- Tsutsui, M., Taniguchi, M., Yokota, K., Kawai, T.: Identifying single nucleotides by tunnelling current. Nat. Nanotechnol. 5, 286–90 (2010)
- Ivanov, A.P., Instuli, E., McGilvery, C.M., Baldwin, G., McComb, D.W., Albrecht, T., Edel, J.B.: DNA tunneling detector embedded in a nanopore. Nano Lett. 11, 279–85 (2011)
- 71. Bayley, H., Cremer, P.S.: Stochastic sensors inspired by biology. Nature. 413, 226-30 (2001)
- Butler, T.Z., Pavlenok, M., Derrington, I.M., Niederweis, M., Gundlach, J.H.: Singlemolecule DNA detection with an engineered MspA protein nanopore. Proc. Natl. Acad. Sci. U.S.A. 105, 20647–52 (2008)
- Wendell, D., Jing, P., Geng, J., Subramaniam, V., Lee, T.J., Montemagno, C., Guo, P.: Translocation of double-stranded DNA through membrane-adapted phi29 motor protein nanopores. Nat. Nanotechnol. 4, 765–772 (2009)
- Maglia, G., Restrepo, M.R., Mikhailova, E., Bayley, H.: Enhanced translocation of single DNA molecules through alpha-hemolysin nanopores by manipulation of internal charge. Proc. Natl. Acad. Sci. U.S.A. 105, 19720–5 (2008)

- Stoddart, D., Heron, A.J., Mikhailova, E., Maglia, G., Bayley, H.: Single-nucleotide discrimination in immobilized DNA oligonucleotides with a biological nanopore. Proc. Natl. Acad. Sci. U.S.A. 106, 7702–7 (2009)
- 76. Clarke, J., Wu, H.C., Jayasinghe, L., Patel, A., Reid, S., Bayley, H.: Continuous base identification for single-molecule nanopore DNA sequencing. Nat. Nanotechnol. 4, 265–70 (2009)
- 77. Derrington, I.M., Butler, T.Z., Collins, M.D., Manrao, E., Pavlenok, M., Niederweis, M., Gundlach, J.H.: Nanopore DNA sequencing with MspA. Proc. Natl. Acad. Sci. U.S.A. 107, 16060–5 (2010)
- Manrao, E.A., Derrington, I.M., Pavlenok, M., Niederweis, M., Gundlach, J.H.: Nucleotide discrimination with DNA immobilized in the MspA nanopore. PLoS One 6, e25723 (2011)
- Kawano, R., Schibel, A.E., Cauley, C., White, H.S.: Controlling the translocation of singlestranded DNA through alpha-hemolysin ion channels using viscosity. Langmuir 25, 1233–7 (2009)
- de Zoysa, R.S., Jayawardhana, D.A., Zhao, Q., Wang, D., Armstrong, D.W., Guan, X.: Slowing DNA translocation through nanopores using a solution containing organic salts. J. Phys. Chem. B 113, 13332–6 (2009)
- Cockroft, S.L., Chu, J., Amorin, M., Ghadiri, M.R.: A single-molecule nanopore device detects DNA polymerase activity with single-nucleotide resolution. J. Am. Chem. Soc. 130, 818–20 (2008)
- Hornblower, B., Coombs, A., Whitaker, R.D., Kolomeisky, A., Picone, S.J., et al.: Singlemolecule analysis of DNA-protein complexes using nanopores. Nat. Methods 4, 315–17 (2007)
- McNally, B., Singer, A., Yu, Z., Sun, Y., Weng, Z., Meller, A.: Optical recognition of converted DNA nucleotides for single-molecule DNA sequencing using nanopore arrays. Nano Lett. 10, 2237–44 (2010)
- Healy, K., Schiedt, B., Morrison, A.P.: Solid-state nanopore technologies for nanopore-based DNA analysis. Nanomedicine (Lond.) 2, 875–97 (2007)
- Storm, A.J., Storm, C., Chen, J., Zandbergen, H., Joanny, J.-F., Fast, D.C.: Fast DNA translocation through a solid-state nanopore. Nano Lett. 5, 1193–1197 (2005)
- He, Y., Tsutsui, M., Fan, C., Taniguchi, M., Kawai, T.: Controlling DNA translocation through gate modulation of nanopore wall surface charges. ACS Nano 5, 5509–18 (2011)
- Wanunu, M., Morrison, W., Rabin, Y., Grosberg, A.Y., Meller, A.: Electrostatic focusing of unlabelled DNA into nanoscale pores using a salt gradient. Nat. Nanotechnol. 5, 160–5 (2009)
- van den Hout, M., Krudde, V., Janssen, X.J., Dekker, N.H.: Distinguishable populations report on the interactions of single DNA molecules with solid-state nanopores. Biophys. J. 99, 3840–8 (2010)
- Luan, B., Aksimentiev, A.: Control and reversal of the electrophoretic force on DNA in a charged nanopore. J. Phys. Condens. Matter 22, 454123 (2010)
- Smeets, R.M., Kowalczyk, S.W., Hall, A.R., Dekker, N.H., Dekker, C.: Translocation of RecA-coated double-stranded DNA through solid-state nanopores. Nano Lett. 9, 3089–96 (2009)
- Lu, B., Albertorio, F., Hoogerheide, D.P., Golovchenko, J.A.: Origins and consequences of velocity fluctuations during DNA passage through a nanopore. Biophys. J. 101, 70–9 (2011)
- Trepagnier, E.H., Radenovic, A., Sivak, D., Geissler, P., Liphardt, J.: Controlling DNA capture and propagation through artificial nanopores. Nano Lett. 7, 2824–30 (2007)
- Peng, H., Ling, X.S.: Reverse DNA translocation through a solid-state nanopore by magnetic tweezers. Nanotechnology 20, 185101 (2009)
- Iqbal, S.M., Akin, D., Bashir, R.: Solid-state nanopore channels with DNA selectivity. Nat. Nanotechnol. 2, 243–8 (2007)
- Hall, A.R., Scott, A., Rotem, D., Mehta, K.K., Bayley, H., Dekker, C.: Hybrid pore formation by directed insertion of α-haemolysin into solid-state nanopores. Nat. Nanotechnol. 5, 874–7 (2010)

- Oliver, J., Bready, B., Goldstein, P., Preparata, F.: Biopolymer sequencing by hybridization of probes to form ternary complexes and variable range alignment. US patent application 20090099786 (2008)
- 97. Garaj, S., Hubbard, W., Reina, A., Kong, J., Branton, D., Golovchenko, J.A.: Graphene as a subnanometre trans-electrode membrane. Nature 467, 190–3 (2010)
- Merchant, C.A., Healy, K., Wanunu, M., Ray, V., Peterman, N., Bartel, J., Fischbein, M.D., Venta, K., Luo, Z., Johnson, A.T., Drndić, M.: DNA translocation through graphene nanopores. Nano Lett. 10, 2915–21 (2010)
- Schneider, G.F., Kowalczyk, S.W., Calado, V.E., Pandraud, G., Zandbergen, H.W., Vandersypen, L.M., Dekker, C.: DNA translocation through graphene nanopores. Nano Lett. 10, 3163–7 (2010)
- 100. Sathe, C., Zou, X., Leburton, J.P., Schulten, K.: Computational investigation of DNA detection using graphene nanopores. ACS Nano. 11, 8842–51 (2011)

Index

A

ABEI. See N-(4-aminobutyl)-N-etylisoluminol (ABEI) Acridinium ester (AE), 168 AFLP. See Amplified fragment length polymorphism (AFLP) Aggregate-induced interchain energy transfer, 141 Alexa Fluor, 108, 134, 135, 161 Alignment algorithm, 292 Altritol nucleic acids (ANA), 103 N-(4-Aminobutyl)-N-etylisoluminol (ABEI), 167 11-Amino-1-undecanethiol (MUAM), 246 Amplified fragment length polymorphism (AFLP), 49 ANA. See Altritol nucleic acids (ANA) Approaches in diagnosis/prognosis, 5 AT-cut quartz crystals, 205, 206

B

β-D-LNA. See β-D-ribofuranosyl nucleotide (β-D-LNA)
β-D-ribofuranosyl nucleotide (β-D-LNA), 95
BHQ. See Black hole quencher (BHQ)
Bio-barcode, 71, 72
Bioconjugation, 74, 104–106
Biotinylated probes, 106, 112, 207, 209, 214, 215, 218–220, 224, 229, 275
Black hole quencher (BHQ), 110

С

CAGE. See Cap analysis of gene expression (CAGE) Campylobacter coli, 39, 40 Campylobacter spp., 36 Cap analysis of gene expression (CAGE), 295 Carbon nanotubes (CNTs), 68, 69, 74, 112, 113, 164, 186, 191-195 Cascade enzymatic signal amplification (CESA), 159 Cationic conjugated polymers (CCP), 126, 127, 130, 140, 144, 146 Cauliflower mosaic virus (CAMV), 29, 30, 211.223 35S promoter, 29, 30 CCP. See Cationic conjugated polymers (CCP) Centre for Disease Prevention and Control, 36 Certified standard materials, 33 CESA. See Cascade enzymatic signal amplification (CESA) CFS. See Charge flow separation (CFS) Charged conjugated polymers (CCP), 125-147 optical properties, 127-132 Charge flow separation (CFS), 11, 12 Cheese, 40, 41, 47 Chemiluminescence assays, 167-171 Chemiluminescent DNA assay, 167-171 Chemoresponsive diffraction gratings, 76 ChIP. See Chromatin immunoprecipitation studies (ChIP) Chitosan, 112 Cholera strain, 293 Cholesterol concentration in blood, 50 Chromatin immunoprecipitation studies (ChIP), 9-10, 83, 98, 126, 176, 204, 208, 247-249, 252, 264, 273, 276, 287-288, 292 Chromosomal FISH, 12 Circulating fluorescently-labeled DNA (CNA), 82 Circulating tumor cells (CTC), 12

G. Spoto and R. Corradini (eds.), *Detection of Non-Amplified Genomic DNA*, Soft and Biological Matter, DOI 10.1007/978-94-007-1226-3,
© Springer Science+Business Media Dordrecht 2012 Click chemistry, 105, 106 *Clostridium perfringens*, 36, 39 CNTs. *See* Carbon nanotubes (CNTs) Coffee DNA, 48 Concentrations of DNA targets, 71, 93, 136 Conjugated polyelectrolytes (CPEs), 70, 126–132, 140–143, 145, 146 DNA sensing, 131 Conjugated polymers (CPs), 125–147 CTC. *See* Circulating tumor cells (CTC) Cyclic voltammetry, 266, 268

D

DABCYL, 110, 138, 161 Dairy products, 41, 46-47 DEP. See Dielectrophoresis (DEP) Dextran, 112, 208, 209, 214, 218, 223, 229 DGE. See Digital gene expression (DGE) Dielectrophoresis (DEP), 8-10 Differential pulse voltammetry, 187, 191, 194, 264 Digital gene expression (DGE), 295 Direct RNA sequencing (DRS), 284, 295, 297-298 DNA ancient, 299 autocatalytic hybridization, 140, 141 bovine, 216-222 conductivity-based sensors, 77 degradation, 45, 48, 49, 114, 252 denaturation blocking oligos, 212, 213, 217-227, 251 thermal, 211-213, 217-218, 220, 222, 251 extraction, 26, 27, 44, 48, 49, 209, 223 food safety and quality, 25-53 fragmentation, 45, 210, 216, 223, 227, 229, 230, 250, 251 methylation detection, 143-146 sequencing, 13, 67, 68, 80, 82, 84, 282-284, 287-290, 293 DRS. See Direct RNA sequencing (DRS) Dual-color fluorescence coincidence, 81

Е

- ECL. See Electro-generated chemiluminescent (ECL)
- EFSA. See European Food Safety Authority (EFSA)

EGFET. See Extended-gate field effecttransistor (EGFET) Electrical signal transduction mode, 75 Electrochemical detection carbon nanotubes, 68, 112 nanoparticles, 186-191, 193, 194, 198, 199 quantum dots, 186, 188-191 silver nanoparticles, 191, 193 voltammetric detection, 187, 193-195 Electro-generated chemiluminescent (ECL), 167-171, 194, 195 ELISA. See Enzyme-linked immunosorbent assay (ELISA) Embryo selection, 15 Engineered nanostructures, 67-84 Enzyme-linked immunosorbent assay (ELISA), 33-35, 38, 84 Escherichia coli, 39-41, 108, 110, 282 Ethidium bromide (EB) use with charged conjugated polymers, 140 SWNT assays, 166 EU Directives, 32-35 European Food Safety Authority (EFSA), 36 European Regulation, 35 Expanded genetic information system (AEGIS), 104 Extended-gate field effect-transistor (EGFET), 78

F

- FETs. See Field effect transistor (FETs)
- Fiber-FISH, 108
- Field effect transistor (FETs), 68, 78, 81, 126, 196
- FINS. See Forensically Informative Nucleotide Sequencing (FINS)
- FISH. See Fluorescence in situ hybridization (FISH)
- Flow-FISH, 107, 108, 115
- Fluorescence
 - activated cell sorting, 7, 10
 - burst size analysis, 82
 - chain reaction (FCR), 134, 135
 - quenchers, 109, 110, 131, 136, 138, 164, 166
- Fluorescence in situ hybridization (FISH), 5, 11, 12, 16, 38, 100, 107–111, 114, 115
- Fluorescence resonance energy transfer (FRET), 31, 73, 109–111, 128, 131, 132, 135, 136, 138–140, 142–146, 160–163
- Fluorescent switching probes, 30

Fluorometric detection, 131, 134, 137 by charged conjugated polymers, 131, 134, 137 Food allergens, 28, 32-36 authenticity, 26, 28, 43-49 borne illness, 36, 37 borne pathogens, 37, 38, 43, 53 diagnostics, 37-39, 42, 43, 52, 53 labelling, 28 Foodomics, 25, 26 Forced intercalation, 104, 111 Forensically Informative Nucleotide Sequencing (FINS), 45 Free nucleic acids, 12 Free tumor DNA, 13-14 **RNA**, 14 FRET. See Fluorescence resonance energy transfer (FRET) FTDP-17 disease, 139 Fusarium graminearum, 39

G

gag region of HIV, 168 GCEs. See Glassy-carbon electrodes (GCEs) G-clamp, 103, 104 Gene expression measurements, 294-298 Gene-food interactions, 51 Genetically modified organisms (GMOs) EPSPS gene, 228, 229 Nicotiana glauca, 223 Roundup Ready soybean, 31, 227, 229, 252 Genetic analyses, 3-17 in health laboratories, 3-17 Glassy-carbon electrodes (GCEs), 191, 193-196 β-Globin allele, 16 Glutaraldehyde (GTA), 194 Gold nanoparticles aggregation, 70 colorimetric assays, 70, 71 colorimetric change, 71 extinction coefficient, 70 Gold surfaces, 98, 105, 106, 110, 112, 196, 207, 238, 240, 243, 246-248 Graphene nanopores, 302 oxide (GO), 113, 116 Graphite-epoxy composite electrode, 187 GTA. See Glutaraldehyde (GTA)

H

Hairpin probes, 141, 170 HDA. See Helicase-dependent isothermal DNA amplification (HDA) Helicase-dependent isothermal DNA amplification (HDA), 39 α-Hemolysin, 301 Hepatitis B virus (HBV), 227 Herbicide, 29, 30 resistance, 29, 30 Hexitol nucleic acids (HNA), 103 Hidden allergenes, 34 HLA-G, 15 HNA. See Hexitol nucleic acids (HNA) Horseradish peroxidase (HRP), 97, 167, 191, 193 Human Genome Project, 282, 284, 289 Hydrogels, 112

I

IACs. See Internal amplification controls (IACs) Impedance spectroscopies, 266 IndexDP Genomic, 292 Indium-tin oxide (ITO), 98, 170 Insect resistance, 29, 30 Internal amplification controls (IACs), 38 Internal Transcribed Spacer (ITS), 47 Inter-particle gap engineering, 77 Invader LNA probes, 111 In vitro fertilization and embryo transfer (IVF-ET), 15, 16 Isothermal amplification, 38, 39 Isothermal technologies, 39 ITO. See Indium-tin oxide (ITO) ITS. See Internal Transcribed Spacer (ITS) IVF-ET. See In vitro fertilization and embryo transfer (IVF-ET)

K

Kretschmann geometry, 239, 243

L

Lab-on-a-Chip, 9–10, 39, 46, 48, 73 technology, 27 Langmuir model, 92, 270, 271 LDR. *See* Ligase detection reaction (LDR) *Legionella* spp., 39 Ligase detection reaction (LDR), 30, 31, 49 Light-up probes, 111 Listeria monocytogenes, 36, 39, 41 LNA. See Locked nucleic acids (LNA) LNA-FISH, 108 Localized surface plasmon (LSP), 75, 154, 247 Locked nucleic acids (LNA), 27, 91, 95–97, 105, 108–111, 115 Loop-mediated isothermal amplification (LAMP), 30, 39, 42 real-time LAMP, 30 Low-tech technologies, 16–17 LSP. See Localized surface plasmon (LSP) Luminol, 167, 168

М

Magnetic activated cell sorting (MACS), 11 concentration, 169 relaxation switch, 73 MAQC. See MicroArray Quality Control Consortium (MAQC) MBs. See Molecular beacons (MBs) MCH. See Mercaptohexanol (MCH) Meat DNA, 44-45, 213, 214 Melting temperature (T_m), 69, 92, 94, 102, 155, 250 Mercaptohexanol (MCH), 112, 266, 267, 274 Metal enhanced fluorescence, 73 Microarray, 30, 31, 34, 49, 51, 75, 100, 105, 112, 126, 135, 138, 142, 146, 154, 160, 235, 243, 286, 294, 295 MicroArray Quality Control Consortium (MAQC), 51 Microbial contamination of food, 36–43 Microchannels, 82, 253, 254 Microfluidics, 39, 73, 80-82, 108, 172, 243, 245, 248, 251-253 microRNA, 14-15, 47, 109 SPR detection, 109 Microtubule associated protein Tau, 139 Milk, 32, 33, 35, 40, 41, 46-47 Mitochondrial DNA (mtDNA), 44, 45 MO. See Morpholino oligonucleotides (MO) Molecular beacons (MBs), 41, 97, 109, 110, 138, 156–159, 162 Molecular diagnosis, 15-16 on human embryos, 15 Molecular probes, 89–115 Morpholino oligonucleotides (MO), 27, 91, 95, 97-98, 115 MPS-PPV. See poly(2,5-methoxypropyloxysulfonate phenylene vinylene) (MPS-PPV)

MspA, 301 Multiple labeling, 160 Mycobacterium tuberculosis, 227, 228

Ν

Nanobeacon, 164, 165 Nanochannels, 80-84, 199, 300 Nanocrystal-DNA conjugates, 73 Nanodisks, 75, 76 on-wire, 75 Nanodumbbells, 76, 77 Nanogaps, 76, 77, 80-82, 84 Nanoparticle-enhanced diffraction grating (NEDG), 249 Nanoparticles color-coded, 81 hydrophobic, 163 iron oxide, 73 magnetic, 73 metallic, 164, 171, 247 promoted cadmium precipitation, 188 semiconductors, 160, 162 silica incorporated dyes, 159, 160 silicon, 162 silver, 69, 70, 72, 78, 173, 191, 193 super-paramagnetic, 73 Nanopores artificial, 300 biological, 300-302 graphene, 302 sequencing, 299-302 silicon-based, 302 Nanoprobes, 68-74, 173-175 Nano-shell, 68, 76 Nanowells, 80, 82 Nanowires gold, 196 polyaniline, 196, 198 silicon, 68, 78, 81, 98, 173 NASBA. See Nucleic acid sequence-based amplification (NASBA) NaYF4 nanocrystals, 163 NEDG. See Nanoparticle-enhanced diffraction grating (NEDG) NESA. See Nicking enzyme signal amplification (NESA) Newborn screening, 4 N-hydroxysuccinimidyl ester, 106 Nicking enzyme signal amplification (NESA), 158, 159 Non-amplified genomic DNA denaturation, 30, 31, 155, 210-214, 217-230, 251, 253, 255

detection cationic polyelectrolytes (CPE), 127, 140, 141 PNA, 27, 34, 91, 96, 99-103, 105, 109, 227.252 probes, 5, 27, 68, 89-115, 126, 155, 186, 207, 235, 264, 286 QCM, 32, 38, 43, 114, 115, 203, 204, 227, 229 SPR, 32, 38, 68, 73, 75, 100, 102, 154, 171-172, 204, 208, 235-255, 264, 265 sequencing, optical technologies, 153-176, 289-299 Non-invasive diagnosis, 9-15 Norovirus, 36, 39, 41 Nucleated red blood cells (NRBC), 11, 12 Nucleic acid sequence-based amplification (NASBA), 31, 39 implemented microarray analysis (NAIMA), 31 Nutrigenetics, 50-52 Nutrigenomics, 28, 50-52

0

Olea europaea, 48, 49 Oligonucleotides, 5, 27, 69, 90, 131, 156, 191, 207, 246, 264, 302 Olive, 48–49 Optical detection, 74, 82, 110, 153–176, 211, 301 of non-amplified genomic DNA, 153–176 Optical trap, 82 Ovarian cancer, 288 8-Oxo-guanine, 104

P

Paired end Tag seq (PET-Seq), 295
PCR-free, 17, 32, 35, 36, 38, 94, 97, 114–115
PCR-RFLP, 46, 48, 214
PDO. See Protected designation of origin (PDO)
Peptide nucleic acids (PNAs) analogues, 100–103, 107 beacons, 110 chemically modified, 100
FISH, 107, 109, 110 lysine-based chiral, 102 microarrays, 30, 34, 49, 112, 142 openers, 96, 99

probe, 29, 34, 99, 100, 102, 107-109, 111, 138, 139, 142, 143, 247, 251-254, 271 SPR, 100, 115 Peroxidase-like DNAzyme, 167 Personalized therapies, 4-8 PET-Seq. See Paired end Tag seq (PET-Seq) PFBT, 130, 142, 143 PFPB, 141, 142 Piezoelectric sensing, 114, 203-230 Polar body (PB), 16, 107 pol region of HIV, 168 poly(2,5-methoxy-propyloxysulfonate phenylene vinylene) (MPS-PPV), 128 - 130poly(3-alkoxy-4-methylthiophene), 132-134 poly(fluorene), 126, 130 poly(thiophene) (PT), 126, 128, 131-137, 142 Polyadenylation site mapping, 298 poly[9,9-bis(6'-[N,N,Ntrimethylammonium]hexyl)fluoreneco-phenylene dibromide] (PFP), 128.142 poly-(fluorene-alt-phenylene) (PFP), 138 Polymerase Chain Reaction (PCR) clamping, 29 end point PCR, 27, 33, 34, 38, 223 free, 17, 32, 35, 36, 38, 94, 97, 114-115 inhibitors, 44 ITS-PCR, 47 multiplex, 28-31, 34 nested, 27, 29 qPCR, 29, 33, 38, 294 real time PCR, 11, 27, 28, 30, 31, 33, 34, 40, 41, 52, 214 restriction fragment length difference (RFLP), 46, 48, 214 Polymorphisms fetal hemoglobin production, 6-7 SNPs, 83, 90, 94, 156, 250 poly-(p-phenylenevinylene)s (PPV), 126 poly(p-phenyleneethynylene)s (PPE), 126 PPE molecular-beacon, 138 pre-fertilized oocytes, 4, 5, 16 pre-miRNA, 14 Prenatal diagnosis genetic diseases, 15 screening, 4 pri-miRNA, 14 Probe affinity, 90-94 biotinylated, 106, 112, 207, 209, 214, 215, 218-220, 224, 229, 275

Probe (*cont.*) density, 92, 93, 271 folding, 94 immobilization chemistry, 208, 209, 229 thiolated, 112, 155, 207–209, 214–221, 229 Progenitor enrichment approach, 11 prolyl-(ACPC)-PNA, 101, 102 Protected Designation of Origin (PDO), 47–49 Protein-based allergen detection, 33 Pyrrolidinyl-PNA, 102

Q

QCM. See Quartz crystal microbalance (QCM)
Quantum dots (QDs)
II-IV semiconductors, 160, 162
multicolor, 160, 161
Quartz crystal microbalance (QCM), 32, 38, 43, 114, 115, 203–205, 227, 229
Quencher-labeled target strand (DNA-QTL), 131, 136

R

RCA. See Rolling circle amplification (RCA)
RecA, 115, 302
Redox-active functional groups, 105
Restriction enzymes, 210, 216–218, 221, 222, 229, 250, 251
Restriction fragment length polymorphism (RFLP), 48, 210, 214
RNA-Seq methods, 286
Rolling circle amplification (RCA), 39, 159
Roundup Ready soybean, 29–31, 36, 227, 229, 252
RT-PCR, 40, 47, 52, 157, 227
Ruthenium tris(2,2'-bipyridyl), 168

S

SAGE. See Serial analysis of gene expression (SAGE)
Salmonella spp., 36, 39, 40
Sample enrichment, 37, 40, 43, 134
Santa Lucia model, 92 for DNA hybridization, 92
Sauerbrey equation, 205
SAW. See Surface acoustic wave (SAW)
SCA. See Sickle-cell anemia (SCA)
SCAR. See Sequenced characterised amplified region markers (SCAR)
Sensorgram, 240, 241 Sequenced characterised amplified region markers (SCAR), 49 Sequence selectivity, 27, 94-95, 102 Sequencing amplification technology limitation, 288 cDNA, 281, 284, 286-287, 297 de novo assembly of genomes, 286 the \$1000 genome, 282, 283, 299 genome grants, 283 Maxam and Gilbert, 281 nanopores, 68, 283, 289, 299-302 non-amplified DNA, 131, 226, 286, 288, 293 optical mapping, 289 optical non-amplified technologies, 289-299 pyrosequencing, 283, 284 Sanger, 281, 282, 284 shot-gun, 284, 289 SMRT technology, 292-294 by synthesis, 283-285, 289, 291, 292 whole genome, cost, 286, 287 Serial analysis of gene expression (SAGE), 295 SERS. See Surface enhanced Raman scattering (SERS) S-FISH. See Suspension FISH Sickle-cell anemia (SCA), 4, 6 Silicon nanoparticles, 162 nanospheres, 163 nanowire morpholino-functionalized, 98 nitride, 83 Silver-enhanced scanometry, 75 Silver staining, 72, 75 Simple sequence repeats (SSR), 49 Single cell analysis, 8-10 Single-molecule detection, 76, 108, 115 Single molecule fluorescence spectroscopy (SMFS), 108 Single nucleotide polymorphism (SNP), 6, 27, 44, 49, 69, 81, 83, 90, 94, 103, 127, 139, 143-147, 156, 250 Single-walled carbon nanotubes (SWCNTs), 68, 164, 193, 194 Smart probes, 110, 208 SMFS. See Single molecule fluorescence spectroscopy (SMFS) SNP. See Single nucleotide polymorphism (SNP) S1 nuclease, 139 Sonagashira method, 128 for the synthesis of PPE, 128

Index

Square wave voltammetry (SWV), 266, 271-273 SSR. See Simple sequence repeats (SSR) Staphylococcus aureus, 142 Streptavidin, 75, 97, 105, 106, 112, 136, 162, 172, 188, 193-195, 207, 209, 211, 218, 223, 228, 264, 266-275 Superquenching, 13 Surface acoustic wave (SAW), 115, 204 Surface and nanostructure architectures, 112 - 113Surface architectures, 112-113, 246, 252, 268 Surface enhanced Raman scattering (SERS), 43, 72-77, 172-175 Surface plasmon fluorescence microscopy, 276 spectroscopy (SPFS), 241-243, 272-274 Surface plasmon resonance imaging, 32, 100, 115, 172, 208, 241, 243-245, 248, 249, 251 long range (LRSPR), 248 theory, 236-241 Suspension FISH (S-FISH), 108 SWCNTs. See Single-walled carbon nanotubes (SWCNTs) SWV. See Square wave voltammetry (SWV) SYBR[®] Green, 30, 35, 38, 41

Т

TaqMan[®], 30, 38, 40, 41, 47
Target-induced DNA strand displacement, 140
TCCD. See Two-colour coincidence detection (TCCD)
Telomere repeats, 107
β-Thalassemia, 6–8, 15, 16, 253, 255
Threose nucleic acids (TNA), 103
TNA. See Threose nucleic acids (TNA)
tNOS terminator, 211, 223

Total internal reflection ellipsometry, 114 fluorescence microscopy (TIRFM), 73, 109 Toxoplasma gondii, 36 Traceability, 28, 43-49 Transgenic, 26, 30-32, 225, 226 maize and soy, 32, 223, 229, 252 Trialkoxysilanes, 113 Tricyclo-DNA, 103 Triplex formation, 96, 133 by charged conjugated polymers, 133 Tripropylamine (TPrA), 168, 169 in electrochemiluminescence, 168 Tunneling junction, 82, 83 Two-colour coincidence detection (TCCD), 108

U

Up-converting rare-earth nanophosphors (UCNPs), 163

V

Vibrio vulnificus, 39

w

Wessling polymerization, 128

Y

Y-chromosome-specific DYS1 sequences, 11

Z

Zeptomolar DNA detection, 36, 134 by charged conjugated polymers, 134 Zero-mode waveguide (ZMW), 82, 108, 292, 293